Rigidity theorems for universal and symplectic universal lattices (普遍格子と斜交普遍格子の剛性定理)

thesis by

見村万佐人 (MIMURA, Masato)

for the Degree of Doctor of Mathematical Sciences at

the Graduate School of Mathematical Sciences,

the University of Tokyo, Tokyo, Japan

January, 2011

Acknowledgments

I am deeply indebted to my supervisor Narutaka Ozawa for introducing me to this topic, and for guidance and suggestions. This thesis could not exist without his stimulating helps. I am grateful to Yasuyuki Kawahigashi for instruction, encouragement, and advice.

I am grateful to Masahiko Kanai, Yoshikata Kida, Hiroyasu Izeki, and Shin Nayatani for their invitation to the conference "Rigidity School Kyoto, 2008." This conference was the turning point on my study.

For the first part of my work (Chapter 6), I thank Bachir Bekka and Nicolas Monod for some use of symbols. Thanks to Yves de Cornulier, Alex Furman, Nigel Higson, Martin Kassabov, Tsuyoshi Kato, Takefumi Kondo, Vincent Lafforgue, Jesse Peterson, Yehuda Shalom, Guoliang Yu, and Andrzej Żuk for comments and interests. Thanks to Uzy Hadad and Mamoru Tanaka for correcting earlier versions of the preprint. At the conference "Affine Isometric Actions of Discrete Groups" at ETH, Centro Stefano Franscini (Ascona, Switzerland) in 2009, I received an award "Award for Best Contribution" for my contributed talk on this work. I am grateful to the organizers, Alain Valette and Goulnara Arzhantseva, for accepting my talk, and to the participants there for their interests.

For the second part of my work (Chapter 10), thanks to Pierre-Emmanuel Caprace, Koji Fujiwara, Pierre de la Harpe, Nicolas Monod, Leonid Vaserstein for comments.

For the third part of my work (Chapter 8, specially Section 8.2), thanks to Takeshi Katsura and Hiroki Sako, who first raised a concerning question to me. Thanks to Nicolas Monod for comments and references.

For the fourth part of my work (Chapter 9 and Chapter 11), I thank Cornelia Druţu, Talia Fernós, Alex Furman, Ursula Hamenstädt, Alex Lubotzky, and Andrés Navas for conversations and references. Thanks to Andrei Jaikin-Zapirain and Martin Kassabov; and to Koji Fujiwara for letting me know their works in progress respectively. Also thanks to Mladen Bestvina, Martin R. Bridson, Benson Farb, David Fisher, Nigel Higson, Nicolas Monod, Pierre Pansu, Alain Valette, and Guoliang Yu for their attentions and comments.

On part of my doctoral study, I took a long-term stay (February, 2010–January, 2011) at EPFL (École Polytechnique Fédérale de Lausanne). I am truly grateful to Nicolas Monod and his secretary Marcia Gouffon for their warmhearted acceptence and hospitality. I thank Nicolas Monod and Alain Valette for many helps and fruitful conversations in Switzerland. Thanks to Aurélien Alvarez, Bachir Bekka, Marek Bożejko, Michelle Bucher-Karlsson, Marc Burger, Indira Chatterji, Laura Ciobanu, Damien Gaboriau, Matthieu Gendulphe, Cyril Houdayer, Alessandra Iozzi, Gilbert Levitt, Anders Karlsson, Roberto Longo, Tatiana Smirnova-Nagnibeda, Romain Tessera, and Andrzej Żuk during my visit.

I acknowledge the Japan Society for the Promotion of the Science (JSPS). My works are supported by JSPS Research Fellowships for Young Scientists No.20-8313, and my stay in Swizterland, which is mentioned above, is supported by the Excellent Young Researcher Overseas Visiting Program by JSPS.

I thank my friends Yoh Tanimoto and Makoto Yamashita, who and I have been encouraging each other during our doctoral (and master) courses. Finally, I wish to thank my family for their unremitting helps in my life.

Abstract

Cohomological rigidity theorems (with Banach coefficients) for some matrix groups G over general rings are obtained. Main examples of these groups are (finite index subgroups of) universal lattices $SL_m(\mathbb{Z}[x_1,\ldots,x_k])$ for m at least 3 and symplectic universal lattices $Sp_{2m}(\mathbb{Z}[x_1,\ldots,x_k])$ for m at least 2 (where k is finite). The results includes the following for certain large m:

- (1) The first group cohomology vanishing with any isometric L^p or p-Schatten coefficients, where p is any real on $(1, \infty)$. This is strictly stronger than having Kazhdan's property (T).
- (2) The injectivity of the comparison map in degree 2 from bounded to ordinary cohomology, with coefficients as in item (1) *not* containing trivial one.

As a corollary, homomorphim rigidity (, namely, the statement that every homomorphism from G has finite image) is established with the following targets: circle diffeomorhisms with low regularity; mapping class groups of surfaces; and outer automorhisms of free groups. These results can be regarded as a generalization of some previously known rigidity theorems for higher rank lattices (Bader–Furman– Gelander–Monod; Burger–Monod; Farb–Kaimanovich–Masur; Bridson–Wade) to the case of certain general matrix group cases, which are *not* realizable as lattices in algebraic groups. Note that G above does *not* usually satisfy the Margulis finiteness property.

Finally, quasi-homomorphims are studied on special linear groups over euclidean domains. This concept has relation to item (2) above for trivial coefficient case, and to the conception of the stable commutator length. In particular, a question of M. Abért and N. Monod, which was for instance stated at ICM 2006, is answered for large degree case, and a new example of groups with the following intriguing features is provided: having infinite commutator width; but the stable commutator length vanishing.

Contents

Acknowledgment iii				
Abstract				
0	Notation and convention			xi
1	Introduction and main results			1
2	Kaz	khdan's	s property (T)	11
	2.1 Original definition			11
		2.1.1	Definition	11
		2.1.2	Kazhdan constant	14
		2.1.3	Relative (T) \ldots	16
	2.2	Induct	tion and permanence properties $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	18
		2.2.1	Compact generation and compact abelianization	19
		2.2.2	Heredity to lattices	22
	2.3	Prope	rty (FH) \ldots	24
		2.3.1	Affine isometric actions and first group cohomology \ldots .	25
		2.3.2	Definition and relative (FH)	26
		2.3.3	Induction of 1-cocycles	28
	2.4	Delorr	me–Guichardet theorem	31
		2.4.1	Positive definite functions	31
		2.4.2	Conditionally negative definite functions	32
		2.4.3	Schoenberg's theorem	34
		2.4.4	Proof of Delorme–Guichardet theorem	37
	2.5	The H	Iaagerup property – as a strong negation	39
		2.5.1	Two definitions	40
		2.5.2	Amenable groups	42
		2.5.3	Other examples	44
	2.6	2.6 Examples of Kazhdan groups		
		2.6.1	$SL_2(\mathbb{R}) \ltimes \mathbb{R}^2 \supseteq \mathbb{R}^2$ has relative (T)	
		2.6.2	$\operatorname{SL}_m(\mathbb{R})$ and $\operatorname{SL}_m(\mathbb{Z})$ have (T) for $m \ge 3$	51

		2.6.3	General totally higher rank lattices	. 53
		2.6.4	Other examples	. 54
	2.7	Expan	nder graphs – as an application	. 56
3	Pro	perty	(T_B) and (F_B)	63
	3.1	Prelin	ninaries on Banach spaces	. 63
		3.1.1	Uniform convexity, uniform smoothness, and superreflexivity	. 63
		3.1.2	Changing norms and space decomposition	. 66
	3.2	Defini	tions of (\mathbf{T}_B) and (\mathbf{F}_B)	. 69
		3.2.1	Definitions and examples	. 69
		3.2.2	Property (\mathbf{F}_B) implies (\mathbf{T}_B)	. 74
		3.2.3	Property (T_B) does not implies (F_B)	. 74
	3.3	Prope	$\operatorname{rrty}(\mathbf{T}_{\mathcal{L}_p}) \text{ and } (\mathbf{F}_{\mathcal{L}_p}) \dots \dots \dots \dots \dots \dots \dots \dots \dots $. 74
		3.3.1	Facts for L^p spaces	
		3.3.2	Property (T) implies $(T_{\mathcal{L}_p})$. 76
		3.3.3	Property (T) does <i>not</i> imply $(F_{\mathcal{L}_p})$. 77
	3.4	$\mathrm{SL}_3(\mathbb{Z}$	\mathcal{L}) has $(\mathcal{F}_{\mathcal{L}_p})$	
		3.4.1	$\operatorname{SL}_3(\mathbb{R})$ has $(F_{\mathcal{L}_p})$	
		3.4.2	<i>p</i> -inductions	
	3.5	Prope	$\operatorname{erty} (T_{[\mathcal{H}]}) \text{ and } (F_{[\mathcal{H}]}) \dots \dots$. 81
4	Uni	versal	lattices	83
	4.1	Defini	tion \ldots	. 83
		4.1.1	Elementary group	. 83
		4.1.2	The Suslin stability theorem	. 85
	4.2	Bound	ded generation and (T)	. 89
		4.2.1	Relative (T) for $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$	
		4.2.2	Bounded generation of Carter–Keller	. 92
		4.2.3	Property (T) for non-algebraic matrix groups	. 93
	4.3	Bound	ded generation for universal lattices	
		4.3.1	A result of van der Kallen	. 94
		4.3.2	Vaserstein's bounded generation	. 95
5	Rec	luced o	cohomology and ultraproduct	97
	5.1	Ultrap	product of metric spaces	. 97
		5.1.1	Definition of ultralimits and properties	. 97
		5.1.2	Ultraproduct of metric spaces, and scaling limits	. 99
		5.1.3	Some facts on ultraproducts	
	5.2	Reduc	ced cohomology	
		5.2.1	Definition and uniform actions	
		5.2.2	Proposition of scaling limits	
		5.2.3	Applications	

	$5.3 \\ 5.4$	Shalom's machinery	
6	Pro	perty (\mathbf{F}_{B}) for universal lattices	113
-	6.1	Relative property $(T_{[\mathcal{H}]})$	
	0.1	6.1.1 Dixmier's unitarization of uniformly bounded representation	
		6.1.2 Proof of relative $(T_{\mathcal{H}})$	
	6.2	A key trick	
	6.3	Proof of Theorem A	
7	Pro	perty (TT) and (FF_B)	121
	7.1	Property (FF_B) and bounded cohomology $\ldots \ldots \ldots \ldots \ldots \ldots$. 121
		7.1.1 Quasi-cocycles and (FF_B)	. 121
		7.1.2 Relation to bounded cohomology	. 123
	7.2	Property (TT) and some facts	. 125
8	Pro	perty $(FF_B)/T$	127
	8.1	A shortcut without ultraproducts	
		8.1.1 Definition of property $(FF_B)/T$. 128
		8.1.2 Property $(FF_B)/T$ for universal lattices	
		8.1.3 Proof of Theorem B (i)	. 134
	8.2	The case of <i>p</i> -Schatten spaces C_p	. 136
		8.2.1 Basics on noncommutative L^p spaces $\ldots \ldots \ldots \ldots \ldots$. 136
		8.2.2 Proof of Theorem C	. 138
	8.3	Property $(TT)/T$	
		8.3.1 Ozawa's property (TTT)	. 143
		8.3.2 Universal lattices have $(TT)/T$. 144
9		plectic universal lattices	
	9.1	Definition	
		9.1.1 Elementary symplectic groups– one realization	
		9.1.2 Another realization	
		9.1.3 The Grunewald–Mennicke–Vaserstein stability theorem	
	9.2	Symplectic universal lattices have (T)	
		9.2.1 Vaserstein's bounded generation for Sp	. 151
		9.2.2 A theorem of Ershov–Jaikin-Zapirain	. 152
	9.3	Symplectic universal lattices have $(TT)/T$	154
	9.4	Property $(FF_B)/T$ for symplectic universal lattices	. 156
10	-	si-homomorphisms and scl	163
		Scl	
	10.2	Quasi-homomorphisms	. 165

		10.2.1 Definition \ldots	. 165		
		10.2.2 Examples on free groups	. 166		
		10.2.3 Higher rank versus rank one	. 168		
	10.3	Homogenization of quasi-homomorphism	. 170		
		10.3.1 Homogeneous quasi-homomorphisms	. 170		
		10.3.2 The Bavard duality theorem	. 172		
	10.4	Question of Abért and Monod	. 175		
		10.4.1 The question \ldots	. 175		
		10.4.2 A theorem of Muranov $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$. 177		
	10.5	Quasi-homomorphisms on elementary groups			
		10.5.1 Precise statement of Theorem D	. 177		
		10.5.2 Key proposition and proof \ldots \ldots \ldots \ldots \ldots \ldots	. 180		
		10.5.3 Difficulty on universal lattices	. 183		
11	11 Applications to group actions				
		Actions on the circle	. 188		
		Fundamentals on $MCG(\Sigma)$ and $Out(F_n)$			
		11.2.1 Definitions			
		11.2.2 Linear representations	. 193		
		11.2.3 Element classification	. 195		
		11.2.4 Subgroup classification			
		11.2.5 Curve graph for MCG	. 199		
		11.2.6 Homomorphism rigidity from higher rank lattices	. 200		
	11.3	Quasi-homomorphisms on $MCG(\Sigma)$ and $Out(F_n)$. 201		
		11.3.1 A theorem of Bestvina–Fujiwara	. 202		
		11.3.2 An $Out(F_n)$ complex of Bestvina–Feighn	. 203		
	11.4	Quasi-cocycles on $MCG(\Sigma)$ and $Out(F_n)$. 204		
		11.4.1 A theorem of Hamenstädt	. 204		
		11.4.2 Another approach – actions on quasi-trees			
	11.5	Homomorphism rigidity into $MCG(\Sigma)$ and $Out(F_n)$. 206		
	11.6	Proof of Theorem H–a shortcut	. 210		
		11.6.1 Distorted elements in a finitely generated group	. 210		
		11.6.2 The proof \ldots			
		11.6.3 Difficulty for symplectic universal lattices	. 212		
Aŗ	Appendix				
Ι	Pro	of of Proposition I	213		

Chapter 0

Notation and convention

Unless otherwise stating, we always assume that all topological groups in this paper are locally compact and σ -compact (our definition of the compactness contains being Hausdorff). Throughout this thesis, we also assume all rings are associative and unital, all representations and actions of a topological group on Banach spaces are strongly continuous, and all subsets and subgroups of a topological group are closed. We use the terminology representations for linear representations. We basically use the symbol Γ , G, H, N, and Λ for topological groups; S for a subset of a topological group; ρ for a representation on a Banach space; π for a unitary representation; B, E and for Banach spaces; \mathfrak{H} for a Hilbert space; \mathcal{C} for a class of Banach spaces; A and R for rings.

We use the following symbols, which are standard in mathematical literatures:

- Z[x₁,...,x_k]: the (commutative) polynomial ring of k independent generators over Z (, as a discrete ring);
 Z⟨x₁,...,x_k⟩: the noncommutative polynomial ring of k independent generators over Z (, as a discrete ring) (for k a natural number)
- \mathbb{F}_q : the finite field of order q (for q a positive power of a prime)
- S(E): the unit sphere; B(E): the unit ball;
 B(E): the Banach algebra of all bounded linear operators; O(E): the group of linear isometries on E; E*: the dual Banach space (for a Banach space E)
- $U(\mathfrak{H})$: the group of unitaries (for a Hilbert space \mathfrak{H})
- $T \mapsto T^*$: the adjoint operation $\mathbb{B}(\mathfrak{H}) \to \mathbb{B}(\mathfrak{H})$ (for a Hilbert space \mathfrak{H})

- $B^{\rho(G)}$: the subspace in B of $\rho(G)$ -invariant vectors (for a Banach representation (ρ, B) of a group G)
- |S|: a number (for a finite set S)
- l_S : a word length $G \to \mathbb{Z}_{\geq 0}$ (for a group and a symmetric generating subset S)
- diam(X): the diameter of X, namely, the maximum of distances between two vertices in X

(for a finite connected graph X)

- μ : a (nonzero) left Haar measure (for a group G)
- 1_G : trivial representation (for a group G) (therefore for a G-representation (B, ρ) , the condition of $\rho \supseteq 1_G$ means $B^{\rho(G)} \neq 0$).
- λ_G: left regular representation on L^p(G) = L^p(G, μ) (for a group G and p in (1,∞))
 Unless otherwise stating, we consider the case of p = 2.
- $A \triangle B$: symmetric deference, $:= (A \setminus B) \cup (B \setminus A)$ (for a set A and B)
- \widehat{G} : the unitary dual of G (for a group G)
- $\pi \succeq 1_G$: the weak containment of 1_G (in the Fell topology), equivalently, π having almost invariant vectors [Definition 2.1.1, Definition 2.2.11]; $\pi \succeq \sigma$: π weakly containing σ (in the Fell topology) [Definition 2.2.11] (for unitary representations π and σ of a group G)
- $\mathcal{K}(G; S)$: the Kazhdan constant [Definition 2.1.10] (for a group G and a compact subset S)
- $\mathcal{K}(G, N; S)$ the relative Kazhdan constant [Definition 2.1.15] (for a group $G, N \leq G$ and a compact subset S of G)
- $\xi \mapsto \xi^*$: the duality mapping $S(B) \to S(B^*)$ [Definition 3.1.4] (for a *uniformly smooth* Banach space B)
- ρ^{\dagger} : the contragredient representation $G \to \mathbb{B}(B^*)$ [Definition 3.1.6] (for a Banach *G*-representation (ρ, B))
- e_G (or simply, e): the group unit of G; 1_R (or simply, 1): the ring unit of R

- lim_ω: ultralimit (of bounded sequences), or ultraproduct (of sequences of metric spaces with base points) [Definition 5.1.1, Definition 5.1.2, Definition 5.1.5] (for a fixed non-principal ultrafilter ω)
- M_m(R): the ring of m × m matrices;
 I_m: the ring unit of M_m(R);
 GL_m(R): the multiplicative group of invertible matrices in M_m(R) (for m ≥ 2 and a ring R)
- $W \to {}^{t}W$: the transpose map on $M_m(A)$ (for $m \ge 2$ and a commutative ring A)
- $SL_m(A)$: the multiplicative group of matrices in $M_m(A)$ of determinant 1 (for $m \ge 2$ and A a commutative ring)
- E_m(R): the elementary group inside GL_m(R);
 E_{i,j}(r): the elementary matrix in M_m(R) with the (i, j)-th entry r [Definition 4.1.1]
 (for m ≥ 2 and a ring R; and 1 ≤ i ≤ m, 1 ≤ j ≤ m, i ≠ j, and r ∈ R)
- $U_m(R)$: the normal subroup of $GL_m(R)$ generated by all unipotent matrices in $GL_m(R)$
- $E_m(R) \ltimes R^m \succeq R^m$: these groups are respectively identified with

$$\left\{ \left(\begin{array}{c|c} W & v \\ \hline 0 & 1 \end{array} \right) : W \in \mathcal{E}_m(R), \, v \in R^m \right\} \quad \trianglerighteq \quad \left\{ \left(\begin{array}{c|c} I_m & v \\ \hline 0 & 1 \end{array} \right) : v \in R^m \right\}$$

(for $m \ge 2$ and a ring R)

• $E_{m_0}(R) \leq E_m(R)$, $SL_{m'}(A) \leq SL_m(A)$: by these we mean the inclusions are respectively realized as the subgroups sit in the left upper corner. Namely, for instance we realize $E_{m_0}(R) \leq E_m(R)$ as

$$\left\{ \left(\begin{array}{cc} W & 0 \\ 0 & I_{m-m_0} \end{array} \right) : W \in \mathcal{E}_{m_0}(R) \right\} \leqslant \mathcal{E}_m(R).$$

(For $m \ge m_0 \ge 2$, a ring R, and a commutative ring A)

- $\operatorname{St}_m(R)$: the Steinberg group [Definition 9.2.4] (for a ring R and $m \ge 3$)
- C_p : the space of *p*-Schatten class operators (for *p*) [Subsection 8.2.1] (although the space itself is defined also for $p = 1, \infty$, we always assume $p \in (1, \infty)$, as mentioned in below).

- H[•](G; ρ, B), H[•]_b(G; ρ, B): respectively, group cohomology and group bounded cohomology with Banach coefficient [Definition 7.1.5]
 (for a *discrete* group G and a Banach G-representation (ρ, B))
- H[•](G), H[•]_b(G): respectively, group cohomology, and group bounded cohomology with the trivial real coefficient (R, 1_G) [Definition 10.2.1] (for a *discrete* group G)
- H[•]_c(G; ρ, B), H[•]_{cb}(G; ρ, B): respectively, continuous group cohomology, and continuous group bounded cohomology with Banach coefficient [Definition 7.1.5] (for a topological group G and a Banach G-representation (ρ, B))
- H[•]_b(G; ρ, B) → H[•](G; ρ, B), H[•]_{cb}(G; ρ, B), H[•]_c(G; ρ, B): the comparison maps in degree [Definition 7.1.7]
 (for a dicrete/topological group G and a Banach G-representation (ρ, B));
 H[•]_b(G) → H[•](G): the comparison map in degree •, with the trivial real coefficient [Definition 7.1.7, Lemma 10.2.2]
 (for a discrete group G)
- QH(G): the (R-vector) space of quasi-homomorphisms; QH(G): the actual (R-vector) space of quasi-homomorphisms [Definition 10.2.1] (for a discrete group G)
- HQH(G): the (R-vector) space of homogeneous quasi-homomorphisms; *HQH*(G): the *actual* (R-vector) space of homogeneous quasi-homomorphisms [Definition 10.3.1] (for a discrete group G)
- [g,h]: a single commutator, $:= ghg^{-1}h^{-1}$ (for $g,h \in G$ and a group G)
- [G,G]: the commutator subgroup;
 cl (:[G,G] → Z_{≥0}), scl (:[G,G] → R_{≥0}): respectively the commutator length and the stable commutator length on [G,G]
 [Definition 10.1.1]
 (for a discrete group G)
- S¹: the unit circle on \mathbb{R}^2 , identified with $[-\pi,\pi)$
- Diff^{1+ α}(S¹): the group of orientation preserving circle homeomorphisms which are $(1 + \alpha)$ -Hölder diffrentiable [Definition 11.1.1]
- $\Sigma_g, \Sigma_{g,l}$: a compact oriented connected surface respectively of genus g (closed), and of genus g and punctures l (for $g, l \ge 0$) (therefore $\Sigma_g = \Sigma_{g,0}$);

 $MCG(\Sigma)$: the mapping class group, as a discrete group [Definition 11.2.1] (for a surface Σ)

- F_n: the free group of rank n; Aut(F_n), Out(F_n): respectively the automorphism group and the outer automorphism group of F_n, as discrete groups [Definition 11.2.1] (for n ≥ 2 finite)
- $L \star L'$: the free product (for groups L and L')
- $\overline{\mathrm{IA}}_n$: the kernel of $\mathrm{Out}(F_n) \twoheadrightarrow \mathrm{GL}_n(\mathbb{Z})$ [Definition 11.2.3] (for $n \ge 2$)

The following notation, convention and uses of symbols in this thesis may not be standard:

- For a simple algebraic group G over a local field, by rank we mean the local rank of G, namely, the dimension of maximal split torus in G (for example, for any m ≥ 1 the group Sp_{m,1} is of rank 1 in our definition).
- By a totally higher rank (algebraic) group, we mean a group of the following form: $G = \prod_{i=1}^{m} \mathbf{G}_{i}(k_{i})$, where k_{i} are local fields, $\mathbf{G}_{i}(k_{i})$ are k_{i} -points of Zariski connected simple k_{i} -algebraic groups (with finite center), and each simple factor $\mathbf{G}_{i}(k_{i})$ has rank ≥ 2 .
- By a *totally higher rank lattice*, we mean a lattice in a totally higher rank algebraic group.
- By a higher rank lattice, we mean a lattice Γ in a group the form $G = \prod_{i=1}^{m} \mathbf{G}_{i}(k_{i})$ where k_{i} are local fields and $\mathbf{G}_{i}(k_{i})$ are k_{i} -points of Zariski connected simple k_{i} -algebraic groups with finite center and with $\sum_{i} \operatorname{rank} \mathbf{G}_{i}(k_{i}) \geq 2$, which satisfies the following condition: for every i such that $\operatorname{rank} \mathbf{G}_{i}(k_{i}) = 1$, the image of Γ by the projection $G \twoheadrightarrow \mathbf{G}_{i}(k_{i})$ is dense in $\mathbf{G}_{i}(k_{i})$.

Note that essential examples of higher rank lattices which may not be of totally higher rank are *irreducible* lattices in a higher rank algebraic group. Here a higher rank algebraic group is a group of the form $G = \prod_{i=1}^{m} \mathbf{G}_{i}(k_{i})$ above with $\sum_{i} \operatorname{rank} \mathbf{G}_{i}(k_{i}) \geq 2$, and a lattice Γ in G is irreducible if each image of Γ by the projection into $\mathbf{G}_{i}(k_{i})$ is dense.

For instance, $\mathrm{SL}_2(\mathbb{Z}[\sqrt{2}])$ is a higher rank lattice (can be realized as an irreducible lattice in $\mathrm{SL}_2(\mathbb{R}) \times \mathrm{SL}_2(\mathbb{R})$), but *not* a totally higher rank lattice. And even though $\mathrm{SL}_2(\mathbb{Z}) \times \mathrm{SL}_2(\mathbb{Z})$ is a lattice in the group $\mathrm{SL}_2(\mathbb{R}) \times \mathrm{SL}_2(\mathbb{R})$ of rank 2, this is *not* a higher rank lattice in our definition.

- Let p represent any real in (1,∞) (it is important in this thesis that we exclude the case of p = 1 and p = ∞).
- Let k represent any natural number (for k = 0, we mean by $\mathbb{Z}[x_1, \ldots, x_k]$ the ring of integers \mathbb{Z}).
- Let \mathcal{H} denote the class of all Hilbert spaces.
- Let $[\mathcal{H}]$ denote the class of all Banach spaces which admit compatible norms to those of Hilbert spaces.
- The symbol K is used for a local field (we allow archimedean local fields as well).
- For a real $M \ge 1$, let \mathcal{H}_M denote the class of all Banach spaces which admit compatible norms to those of Hilbert spaces with the norm ratio $\le M$. [Theorem 5.1.10]

Here for two norms $\|\cdot\|_1$ and $\|\cdot\|_2$ on a Banach space, the norm ratio between $\|\cdot\|_1$ and $\|\cdot\|_2$ is defined by the following formula:

$$\max\left\{\sup_{x\neq 0}\frac{\|x\|_1}{\|x\|_2}, \sup_{x\neq 0}\frac{\|x\|_2}{\|x\|_1}\right\}.$$

- $\langle \cdot, \cdot \rangle$: the duality $B \times B^* \to \mathbb{C}$ (for a Banach space B)
- $\langle \cdot | \cdot \rangle$: the inner product $\mathfrak{H} \times \mathfrak{H} \to \mathbb{C}$ (for a Hilbert space \mathfrak{H})
- S^{-1} : the set of all elements of the form s^{-1} with $s \in S$ (for a subset S of a group G)
- S^n : the set of all elements of the form $g = s_1 \cdots s_n$ with $s_1, \ldots, s_n \in S$ (for a subset S of a group G, and $n \ge 1$) also we set $S^0 := \{e_G\}$.
- For p, let \mathcal{L}_p denote the class of all L^p spaces on any (σ -additive) measure.
- $\rho \succeq 1_G$: ρ having almost invariant vectors [Definition 3.2.1] (for a Banach *G*-representation (ρ, B))
- $|\rho|$: = sup_{*q*\in G} $||\rho(g)||$ (for uniformly bounded representation of a group *G*)
- $d_{\|\cdot\|}, r_{\|\cdot\|}$: respectively, the modulus of convexity $(0, 2) \to \mathbb{R}_{\geq 0}$, and the modulus of smoothness $\mathbb{R}_{>0} \to \mathbb{R}_{\geq 0}$ [Definition 3.1.1] (for a Banach space $(B, \|\cdot\|)$)

- $B'_{\rho(N)}$: the natural complement of $B^{\rho(N)}$ [Definition 3.1.11] (for a superreflexive Banach space B, an isometric G-representation (B, ρ) , and $N \leq G$)
- $\mathcal{K}(G, N; S, \rho)$: the relative Kazhdan constant for property (\mathbf{T}_B) [Definition 3.2.7] (for $N \leq G$, a compact subset S of G, and a isometric Banach G-representation (ρ, B))
- $\overline{\mathcal{K}}(G, N; S; M)$: the generalized relative Kazhdan constant for uniformly bounded representation [Definition 3.5.2] (for $N \trianglelefteq G$, a compact subset S of G, and M a real ≥ 1)
- Cay(G; S): a Cayley graph [Definition 2.5.9] (for a finitely generated group G and a finite generating set S). In this thesis, we connect edges on Cay(G; S) by *right* multiplication of s ∈ S, and consider the isometric *left* G-action on Cay(G; S).
- δ_S: displacement function [Definition 5.2.2]
 (for a finite subset S of a group G, and an isometric action of G on a metric space X)
- QH_c(G; B, ρ): the (vector) space of continuous quasi-cocycles (with Banach coefficient);
 QH_c(G; B, ρ): the actual (vector) space of continuous quasi-cocycles (with Banach coefficient)
 [Definition 7.1.5]
 (for a topological group G and a Banach G-representation (B, ρ))
- QH(G; B, ρ): the (vector) space of quasi-cocycles (with Banach coefficient); QH(G; B, ρ): the actual (vector) space of quasi-cocycles (with Banach coefficient)
 [Definition 7.1.5]

(for a discrete group G and a Banach $G\text{-representation }(\rho,B))$

• sr(R): the stable range [Definition 4.1.4] (for a ring R)

NOTE: there is an inconsistency of ± 1 in the definition of stable range in literatures.

• J_m : the alternating matrix

$$J_m := \left(\begin{array}{cc} 0 & I_m \\ -I_m & 0 \end{array}\right)$$

in M_{2m} (for $m \ge 1$)

• $\operatorname{Sp}_{2m}(A)$: the multiplicative group of symplectic matrices in $\operatorname{M}_{2m}(A)$ associated with the alternating matrix J_m , :={ $g \in \operatorname{M}_{2m}(A) : {}^tgJ_mg = J_m$ }; $\operatorname{Ep}_{2m}(A)$: the elementary symplectic group inside $\operatorname{Sp}_{2m}(A)$ associated with the alternating matrix J_m

[Definition 9.1.3]

(for $m \ge 1$ and A a commutative ring)

Note that the choice of the alternating matrix (, namely, J_m) is not a standard one in studies of symplectic groups.

• $\operatorname{Sp}_{2m_0}(A) \leq \operatorname{Sp}_{2m}(A)$ (or, $\operatorname{Sp}_{2m_0}(A) \hookrightarrow \operatorname{Sp}_{2m}(A)$): by this we mean the inclusion is realized as

$$\left\{ \begin{pmatrix} P & 0 & Q & 0 \\ 0 & I_{m-m_0} & 0 & 0 \\ R & 0 & S & 0 \\ 0 & 0 & 0 & I_{m-m_0} \end{pmatrix} : \begin{pmatrix} P & Q \\ R & S \end{pmatrix} \in \operatorname{Sp}_{2m_0}(A) \right\} \leqslant \operatorname{Sp}_{2m}(A);$$

 $\operatorname{SL}_{m_0}(A) \leq \operatorname{Sp}_{2m}(A)$ (or, $\operatorname{SL}_{m_0}(A) \hookrightarrow \operatorname{Sp}_{2m}(A)$): by this we mean the inclusion is realized as

$$\left\{ \begin{pmatrix} W & 0 & 0 & 0 \\ 0 & I_{m-m_0} & 0 & 0 \\ 0 & 0 & {}^t W^{-1} & 0 \\ 0 & 0 & 0 & I_{m-m_0} \end{pmatrix} : W \in \mathrm{SL}_{m_0}(A) \right\} \leqslant \mathrm{Sp}_{2m}(A).$$

[Definition 9.1.3]

(for $m \ge m_0 \ge 1$ and A a commutative ring)

- $S^{m*}(A^m)$: the additive group of all symmetric matrices in $M_m(A)$ (for $m \ge 2$ and A a commutative ring)
- $E_m(A) \ltimes S^{m*}(A^m) \succeq S^{m*}(A^m)$: these groups are respectively identified with

$$\left\{ \left(\begin{array}{c|c} W & v \\ \hline 0 & {}^t W^{-1} \end{array} \right) : W \in \mathcal{E}_m(A), \ v \in S^{m*}(A^m) \right\} \ \supseteq \ \left\{ \left(\begin{array}{c|c} I_m & v \\ \hline 0 & I_m \end{array} \right) : v \in S^{m*}(A^m) \right\}$$

[Definition 9.1.3]

(for $m \ge 2$ and a commutative ring A)

• $U_m R$; $L_m R$: the subgroups of $E_m(R)$ respectively consisting of all unit upper triangle matrices; all unit lower triangle matrices (for a ring R and $m \ge 2$)

Finally, we shall define the following properties in terms of B:

xviii

- property (T_B) (for a group G) [Definition 3.2.2], relative property (T_B) (for a group pair N ≤ G) [Definition 3.2.2];
- property (F_B) (for a group G) [Definition 3.2.4], relative property (F_B) (for a group pair N ≤ G) [Definition 3.2.4];
- the Shalom property for (F_B) (for a discrete group G) [Definition 5.3.1];
- property (FF_B) (for a group G) [Definition 7.1.3], relative property (FF_B) (for a group pair $N \leq G$, or a pair $Q \subseteq G$) [Definition 7.1.3];
- property (FF_B)/T (for a group G) [Definition 8.1.1], relative property (FF_B)/T (for a group pair N ≤ G, or a pair Q ⊆ G) [Definition 8.1.1, Definition 8.1.6], property (TT)/T (for a group G) [Definition 8.1.1, Definition 8.3.4], relative property (TT)/T (for a group pair N ≤ G, or a pair Q ⊆ G) [Definition 8.1.1, Definition 8.1.6, Definition 8.3.4].

When letting (P_B) represent any of these properties, we define the property (P_C) in terms of a class C of Banach spaces as follows: having (P_C) denotes having (P_B) for all $B \in C$.

Chapter 1

Introduction and main results

The special linear group $G = \operatorname{SL}_m(\mathbb{Z}[x_1, \ldots, x_k])$ over the commutative polynomial ring with k variables over \mathbb{Z} (where $m \geq 3$) is called the *universal lattice* by Y. Shalom in [Sha1]. Here in this thesis the symbol k is used for representing any finite natural number (usually $k \geq 1$. We state if we allow k = 0). It was a long standing problem to determine whether this group satisfies a property so-called Kazhdan's property (T).

Kazhdan's property (T), which was first introduced in a paper [Kaz] of D. Kazhdan in 1967, represents certain forms of rigidity of a group, and now plays an important role in wide range of mathematical fields (we will see in Chapter 2 the definition, basic properties, and some examples of applications). The original definition of property (T) is stated in terms of weak containment of the trivial representation. The celebrated Delorme–Guichardet theorem [Del], [Gui] states that for locally compact and σ -compact groups, property (T) is equivalent to a property so-called *property* (FH), which is defined as follows: a group G is said to have *property* (FH) if every (continuous) affine isometric action of G on a Hilbert space has a global fixed point. This definition of property (FH) is identical to the condition of *first* (*continuous*) group cohomology vanishing with any unitary coefficient. Therefore as we mentioned, property (T) represents extreme rigidity of groups.

We go back to the question we raised in above. In Kazhdan's original paper [Kaz] of property (T), he shown that any "totally higher rank algebraic group" (we refer to Chapter 0 for the definition: roughly speaking, rank means local field rank, and totally means each simple factor has rank at least 2) and "totally higher rank lattices" enjoy this property. For instance, the special linear group $SL_m(\mathbb{Z})$ has property (T) for $m \geq 3$ because it is a lattice in a simple algebraic group $SL_m(\mathbb{R})$ of real rank $m - 1(\geq 2)$. The special linear group $SL_m(\mathbb{Z}[\sqrt{2}])$ also enjoys property (T) for $m \geq 3$ because this group can be realized as a lattice in $SL_m(\mathbb{R}) \times SL_m(\mathbb{R})$. However, Kazhdan's proof is deeply based on representation theory of semisimple algebraic groups, and it gives no information whether the special group $SL_{m\geq 3}(\mathbb{Z}[x])$ enjoys property (T). This is one background of the question to determine whether universal lattices have property (T). Around 2006 and 2007, finally Shalom and L. Vaserstein answered this question affirmatively:

Theorem 1.0.1. (Shalom [Sha5], Vaserstein [Vas2]; Theorem 5.4.1 in this thesis) The universal lattice, namely, the group $G = \operatorname{SL}_{m \geq 3}(\mathbb{Z}[x_1, \ldots, x_k])$ has property (T). This means: for any unitary representation π of G, one has

$$H^1(G;\pi) = 0.$$

We have more words on the motivation to focus on this problem. It is easy to see by definition that property (T) passes to group quotients. Therefore once Theorem 1.0.1 has been proved, this then immediately implies that groups such as $\operatorname{SL}_{m\geq 3}(\mathbb{Z}[1/p])$ (here p is a prime number); $\operatorname{SL}_{m\geq 3}(\mathbb{Z}[\sqrt{2},\sqrt{3}])$; $\operatorname{SL}_{m\geq 3}(\mathbb{F}_q[x])$ (\mathbb{F}_q is the field of order q and q is a positive power of a prime); and $SL_{m>3}(\mathbb{Z}[t, t^{-1}])$ have property (T). Note that in four examples above, all but last one are totally higher rank (hence arithmetic (or S-arithmetic)) lattices, and Kazhdan's theorem applies. However, the last one in the examples above cannot be realized as an arithmetic lattice, and property (T) for this group had not been obtained before the Shalom-Vaserstein theorem. In general, Theorem 1.0.1 implies property (T) for *elementary* groups $E_{m>3}(A)$ over any commutative and finitely generated ring (we always assume that rings are associative and unital). Here the elementary group $E_m(A)$ is defined as the multiplicative group of $m \times m$ matrices generated by elementary matrices (for details, we will see in Chapter 4. In the examples above, the groups should be elementary groups, but they coincide in these case). Therefore, property (T) for universal lattices can be regarded as the *universal* result for elementary groups over a commutative finitely generated rings, which includes special linear groups over a ring of integers. This is the reason why groups $SL_{m>3}(\mathbb{Z}[x_1,\ldots,x_k])$ are called universal lattices: they are universal for lattices of the form $SL_m(\mathcal{O})$.

As we mentioned in above, the group $\operatorname{SL}_{m\geq 3}(\mathbb{Z}[t, t^{-1}])$ (and a universal lattice itself) can*not* be realized as a lattice in semi-simple algebraic group. This follows from the following argument: on an (irreducible) higher rank lattice, there is an extremely strong constraint, which is called the *Margulis finiteness property*:

every normal subgroup is either finite or of finite index.

This contradicts the fact that the group above contains an infinite group with infinite index (we will see in Lemma 4.1.12). Note that by the Margulis arithmeticity theorem, any (irreducible) higher rank lattice is arithmetic. In these views, Theorem 1.0.1 can be regarded as a *non-arithmetization* of extreme rigidity of totally higher rank lattices (of certain form).

We proceed to the next (but closely related) topic property (T_B) and property (F_B), where B is a (given) Banach space or a class of Banach spaces. In 2007, Bader– Furman–Gelander–Monod [BFGM] investigated similar properties to property (T)

(and property (FH)) in the broader framework of general Banach spaces B. They named the Kazhdan type property and the fixed point property respectively property (T_B) and property (F_B) (we will see the precise definitions in Section 3). Property (F_B) is a straight generalization of property (FH), and defined as first (continuous) group cohomology vanishing with any isometric coefficient on B. The classes of Banach spaces of our main interest are the class \mathcal{L}_p $(p \in (1,\infty)$ is given) and $[\mathcal{H}]$. Here the former denotes the class of all L^p spaces on any measure spaces; and the latter denotes the class of all Banach spaces which have compatible norms to ones of Hilbert spaces. The reason why we are interested in these cases is, then property (F_B) is strictly stronger than Kazhdan's property (T), and hence it represents even much more extreme rigidity of groups. Indeed, there are plenty of groups with property (T) which are known to *fail* to have property (F_B) for above class. For instance, P. Pansu [Pan] shown the group $Sp_{m,1}$ of real rank 1, which is known to have property (T) if $m \geq 2$, fails to have property ($F_{\mathcal{L}_p}$) as soon as p > 4m + 2 (therefore fore instance, property $(F_{\mathcal{L}_p})$ with $(\infty >) p > 10$ is no longer equivalent to property (FH)=property ($F_{\mathcal{L}_2}$)). Moreover G. Yu [Yu2] shown that for every (Gromov-)hyperbolic group H (we will give the definition in Subsection 2.6.4), including of wide range of groups with property (T), it has corresponding $p \gg 2$ such that H admits a (metrically) proper cocycle on an ℓ^p space. If a cocycle is a coboundary, then it is bounded (we are considering *isometric* coefficients). Therefore, existence of proper (which means, "diverging at infinity") cocycle represents strong negation of property $(F_{\mathcal{L}_p})$, hence, soft (or, well-deformed) feature of groups. On property $(F_{[\mathcal{H}]})$, Shalom has shown in his unpublished work, that every rank 1 groups, including $\text{Sp}_{m,1}$, fails to have this property.

On the other hand, in [BFGM], Bader–Furman–Gelander–Monod proved the following theorem and revealed that (F_B) is stronger than (T_B) in general, but that totally higher rank algebraic groups and lattices *remain* to have property $(F_{\mathcal{L}_p})$ and $(F_{\mathcal{H}})$ (the assertion in item *(iii)* for property $(F_{\mathcal{H}})$ is due to Shalom):

Theorem 1.0.2. ([BFGM]) Let G be a locally compact and σ -compact group.

- (i) For any Banach space B, property (F_B) implies property (T_B) .
- (ii) Property (T) is equivalent to property $(T_{\mathcal{L}_p})$, where $p \in (1, \infty)$. It is also equivalent to property $(F_{\mathcal{L}_p})$, where $p \in (1, 2]$.
- (iii) Any totally higher rank algebraic groups G and any totally higher rank lattices Γ have property, in the sense in Chapter 0 possess property $(F_{\mathcal{L}_p})$ for $1 and property <math>(F_{|\mathcal{H}|})$.

Therefore, it is now natural to ask the following questions:

Do universal lattices have property (F_{L^p}) $(1 and property <math>(F_{[\mathcal{H}]})$?

Also, specially property $(F_{\mathcal{L}_p})$ has application to group actions on the circle. We will come back to this point later in this introduction.

The first part of this thesis gives the answer to this question, with some (slight) degree condition. The answer is *affirmative*, and the following is the precise statement:

Theorem A. ([Mim1]; Theorem 6.3.1 in this thesis) Let $p \in (1, \infty)$. If $m \ge 4$, then for any p (and any k) the universal lattice $G = SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ possesses property $(F_{\mathcal{L}_p})$ and property $(F_{[\mathcal{H}]})$ in the sense in Definition 3.2.4. This means: for any ρ , an isometric G-representation on an L^p space or a uniformly bounded G-representation on a Hilbert space, one has

$$H^1(G;\rho) = 0.$$

It may be reasonable to expect that Theorem A remains true in the case of m = 3. However at the present the author has no idea how to settle this problem. As we will see in Chapter 4 and Chapter 5, in the proof of Theorem 1.0.1 Shalom employs the equivalence of property (T) and property (FH), and deduces property (FH) from a certain "relative version" of property (T). However in our setting $(B = \mathcal{L}_p, [\mathcal{H}])$, as we have seen in above, property (T_B) does *not* imply property (F_B). This means, an easy imitation of the proof of Theorem 1.0.1 does *not* provide with Theorem A. We need some idea which overcomes the gap from property (T_B) to property (F_B). We will examine this in Section 6.3.

Secondly, we consider a quasification of property (F_B): it means we consider maps which are cocycle up to bounded error. This conception has natural connection to the concept of bounded cohomology [Gro1], [Mon1]. Here bounded cohomology is defined by restricting each cochains to be bounded. We will see in Chapter 7. By naming after so-called property (TT) of N. Monod (which states any quasi-cocycle into unitary representation is bounded), we define a notion of property (FF_B) by strengthening property (F_B). We note the following: since there is a natural injection from bounded cochains to ordinary cochians, this map induces a natural map from (continuous) bounded to (continuous) ordinary group cohomology (with isometric Banach coefficient (ρ , B) of a group G),

$$\Psi^{\bullet}_{cb} \colon H^{\bullet}_{cb}(G;\rho,B) \to H^{\bullet}_{c}(G;\rho,B).$$

This map is called the *comparison map*, and in general it is *neither* injective *nor* surjective. Property (FF_B) for a group G specially implies that for any isometric G-representation ρ , the comparison map degree 2

$$\Psi^2_{\rm cb} \colon H^2_{\rm cb}(G;\rho,B) \to H^2_{\rm c}(G;\rho,B)$$

is *injective*.

We examine whether universal lattice has some *confined* property of property (FF_B) , which we name "*property* $(FF_B)/T$."

Theorem B. ([Mim1]; Theorem 8.1.10 and Theorem 8.3.5 in this thesis) Let G = $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ be universal lattice. Let $p \in (1, \infty)$.

- (i) If $m \ge 4$, then for any p, G possesses property $(FF_{\mathcal{L}_p})/T$ and property $(FF_{[\mathcal{H}]})/T$ in the sense in Definition 8.1.1. In particular the following hold: for any ρ , an isometric G-representation on an L^p space or a uniformly bounded Grepresentation on a Hilbert space, if moreover $\rho \not\supseteq 1_G$, then one has
 - (a) both $H^1(G; \rho) = 0$,
 - (b) and the comparison map in degree $2: H_b^2(G; \rho) \to H^2(G; \rho)$ is injective.
- (ii) For $m \ge 3$, G possesses property (TT)/T (, that is, property (FF_H)/T).

As the statement of above theorem suggests, " $(FF_B)/T$ " means "property (FF_B) modulo the trivial linear part."

In the third part of this thesis, we consider the case of $B = C_p$, which denotes the space of *p*-Schatten class operators acting on a separable Hilbert space. This can be seen an analogy to property $(F_{\mathcal{L}_p})$ (or $(FF_{\mathcal{L}_p})$) in noncommutative L^p -setting. Some properties on commutative L^p space are no longer true in noncommutative setting, and one of them is crucial to establish property $(F_{\mathcal{L}_p})$. Explicitly, the conditional negative definiteness of the kernel on L^p space with $p \in [1, 2]$ is not valid for C_p . However, by extending a previous work of M. Pushnigg [Pus] for totally higher rank algebraic groups and lattices, we show the following:

Theorem C. ([Mim3]; Theorem 8.2.6 in this thesis) If $m \ge 4$, then for any $p \in (1,\infty)$, any finite index subgroup Γ of $SL_m(\mathbb{Z}[x_1,\ldots,x_k])$ has property $(FF_{C_p})/T$. In particular, Γ above has property (F_{C_p}) . Here C_p denotes the space of p-Schatten class operators on a separable Hilbert space.

For any p, any totally higher rank algebraic group and any lattice therein has property (F_{C_p}) .

In the proof of this theorem, we state criteria on a class of Banach spaces for which universal lattice (with degree ≥ 4) has property (FF_B)/T and property (F_B). The class (or a single Banach space) fulfilling these criterion contains the cases of \mathcal{L}_p , [\mathcal{H}], and C_p . We hope this will provide with further study on rigidity theory for universal lattices.

In the fourth part of this thesis, we consider the symplectic version of universal lattice, namely, the group $\operatorname{Sp}_{2m}(\mathbb{Z}[x_1,\ldots,x_k])$ $(m \geq 2)$, and call it symplectic unversal lattice. As is often in various fields of mathematics, the bahavior of symplectic group is more complicated than that of special linear group. However, by employing a result of a work [EJK] in progress of Ershov, Jaikin-Zapirain, and Kassabov (of property (T) for symplectic universal lattices), we obtain the following theorem:

Theorem D. ([Mim4]; Theorem 9.4.1 and Theorem 9.3.1 in this thesis) Let $G = \operatorname{Sp}_{2m}(\mathbb{Z}[x_1,\ldots,x_k])$ be a symplectic universal lattice.

- (i) If $m \geq 3$, then for any $p \in (1, \infty)$, G possesses property $(FF_{\mathcal{L}_p})/T$ and property $(FF_{\mathcal{C}_p})/T$. In particular, G has property $(F_{\mathcal{L}_p})$ and property $(F_{\mathcal{C}_p})$.
- (ii) For $m \geq 2$, G possesses property (TT)/T.

Specially, property (TT)/T (this in particular implies that the comparison map

$$\Phi_{\rm b}^2 \colon H^2_{\rm b}(G;\pi) \to H^2(G;\pi)$$

is injective for any unitary representation which satisfies $\pi \not\supseteq 1_G$) plays a significant role in application, as we shall see in below. We also note that property (TT)/T (as well as property (T)) passes to group quotients and finite index subgroups. Thus we obtain property (TT)/T for any finite index subgroups in the *elementary symplectic* group $\operatorname{Ep}_{2m}(A)$ over a finitely generated commutative ring (we will see for details in Chapter 9).

In the fifth part of this thesis, we consider quasi-cocycles with trivial (real) coefficient, which are called *quasi-homomorphisms*. Recall that property $(FF_B)/T$ means "modulo the trivial lienar part." Hence, in studies above, we escape from dealing with quasi-homomorphisms, which are turned out to be in fact the most tough types of quasi-cocyles for universal lattices. At the moment, we have not succeeded in establishing vanishing results of quasi-homomorphisms on universal lattices. Nevertheless, we have obtained the following result (Theorem E) on elementary groups over *euclidean domains* (then in fact elementary groups coincide with special linear groups). This result covers, for instance, $SL_{m\geq 6}(K[x])$ for K being an *arbitrary* (commutative) field, and has interesting application. On quasi-homomorphisms, or equivalently in some sense, on the kernel of comparison map

$$\Psi^2_{\mathbf{b}} \colon H^2_{\mathbf{b}}(G) \to H^2(G),$$

(here coefficients are the trivial real coefficients) the following result is known as *Bavard's duality theorem*:

Theorem 1.0.3. (Bavard [Bav]) Let G be a discrete group. Then the following are equivalent:

- (i) The comparison map $H^2_{\rm h}(G) \to H^2(G)$ is injective.
- (ii) The stable commutator length scl: $[G, G] \to \mathbb{R}_{\geq 0}$, which is defined as

$$[G,G] \ni g \mapsto \operatorname{scl}(g) := \lim_{n \to \infty} \frac{\operatorname{cl}(g^n)}{n},$$

vanishes identically.

If the commutator length on [G, G] is bounded, then by the definition above, the stable commutator length vanishes identically (on [G, G]). M. Abért has asked whether there exists a counterexample of the converse, and Monod stated this question in his ICM invited lecture [Mon2]. This question is now known as a question of Abért and Monod, and our theorem provides with a natural class of counterexamples of the converse:

Theorem E. ([Mim2]; Theorem 10.5.1 in this thesis) Let A be a euclidean domain. Then for $m \ge 6$, $G = SL_m(A)$, as a discrete group (possibly uncountable), fulfills the following: the comparison map

$$H^2_b(G; 1_G, \mathbb{R}) \to H^2(G; 1_G, \mathbb{R})$$

is injective. Equivalently, the stable commutator length vanishes identically on [G,G](=G).

In particular, if K is a (commutative) field of infinite transcendence degree over its subfield (for instance, $K = \mathbb{C}$), then for $m \ge 6$ the group $G = SL_m(K[x])$ enjoys the following two properties:

(i) The commutator width of G(=[G,G]) is infinite; namely,

$$\sup_{g\in G} \operatorname{cl}(g) = \infty$$

(ii) The stable commutator length vanishes identically on [G,G] (= G); namely, for any $g \in G$,

$$\lim_{n \to \infty} \frac{\operatorname{cl}(g^n)}{n} = 0.$$

We note that A. Muranov [Mur] has shown there exists a 2-generated simple group with infinite commutator width and with the stable commutator length vanishing. He employs small cancellation theory, and it has completely different background to our one.

We also note that there exists a *countable* field K satisfying the assumption of the latter part of Theorem E. We will examine these topics in Chapter 10.

In the final part of this thesis, we apply our theorems to group actions. There are two applications. One is to group actions on the circles. By combining a theorem of A. Navas [Nav1], [Nav2] with our property $(F_{\mathcal{L}_p})$ result, we obtain the following theorem:

Theorem F. ([Mim1], [Mim4]; Theorem 11.1.2 in this thesis) Let Γ be a finite index subgroup either of $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 4)$ or of $Sp_{2m}(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 3)$. Then for any $\alpha > 0$, every homomorphism

$$\Gamma \to \operatorname{Diff}^{1+\alpha}_+(\mathrm{S}^1)$$

has finite image. Here the symbol $\text{Diff}^{1+\alpha}_+(S^1)$ means the group of orientation preserving $(1+\alpha)$ -differentiable (in the sense of Hölder continuity) diffeomorphisms on the unit circle.

Note that by a lemma of A. Selberg, in both cases there exists such Γ which is torsion free.

The next and final application is to homomorphism rigidity into mapping class groups of surfaces and into (outer) automorphism groups of free groups. Those objects have strong connection to group actions on low dimensional manifolds (here by homomorphism rigidity we mean the property that every homomorphism from the group into a target group has finite image). We note that for higher rank lattices, homomorphism rigidity into those groups are respectively obtained by Farb and Masur [FaMas] (into mapping class groups); and Bridson–Wade [BrWa] (into automorphism groups of free groups). We note that in their proof, the Margulis finiteness property for higher rank lattices plays a key role. However, as we have mentioned in above, this property is not valid for universal lattices or symplectic universal lattices. Also, we mention the following: Bridson–Wade [BrWa] have shown that if a group is \mathbb{Z} -averse, then homomorphism rigidity into the groups above holds. Here a group is said to be *not* \mathbb{Z} -averse if there exists finite index subgroup which has a normal subgroup mapping onto \mathbb{Z} . We have, however, universal lattices and symplectic universal lattices are not \mathbb{Z} -averse (see Lemma 11.2.26). Therefore, the following result is a new result:

Theorem G. ([Mim4]; Theorem 11.5.1 in this thesis) Let Γ be a finite index subgroup either of $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 3)$ or of $Sp_{2m}(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 2)$. Then for any $g \ge 0$ and $n \ge 2$, every homomorphism

$$\Phi \colon \Gamma \to \mathrm{MCG}(\Sigma_q)$$

and every homomorphism

$$\Psi \colon \Gamma \to \operatorname{Out}(F_n)$$

have finite image. Here Σ_g denotes a compact closed connected oriented surface of genus g and $MCG(\Sigma_g)$ denotes the mapping class group. The symbol F_n denotes the free group of rank n (here n is finite) and $Out(F_n)$ denotes the outer automorphism group of F_n .

For the proof of Theorem G, property (TT)/T for universal and symplectic universal lattices is one key. Counterpart are the study of quasi-cocycles on $MCG(\Sigma)$ and $Out(F_n)$; and subgroup classification for these groups. By combining our results with deep results of U. Hamenstädt [Ham], Bestvina–Bromberg–Fujiwara [BBF]; and McCarthy–Papadopoulos [McPa], and Handel–Mosher [HaMo], we establish the theorem. In Theorem G, the restriction on m is optimal. Also, the case of $\operatorname{Out}(F_n)$ targets implies homomorphism rigidity with $\operatorname{Aut}(F_n)$ targets; and with $\operatorname{MCG}(\Sigma_{g,l})$ targets. Here $\Sigma_{g,l}$ denotes a compact oriented surface with g genus and l punctures, and here we assume $l \geq 1$. This is because these groups inject into $\operatorname{Out}(F_{n'})$ for sufficiently large n'. See Subsection 11.2.1.

It is worth making a remark that Theorem G for universal lattice cases can be deduced from much easier argument. In fact, we have the following theorem along that shortcut argument:

Theorem H. ([Mim4]; Theorem 11.6.4 in this thesis) Let Γ be a finite index subgroup of noncommutaive universal lattice $E_m(\mathbb{Z}\langle x_1, \ldots, x_k \rangle)$ $(m \geq 3)$. Then for any $g \geq 0$ and $n \geq 2$, every homomorphism $\Phi \colon \Gamma \to MCG(\Sigma_g)$ and every homomorphism $\Psi \colon \Gamma \to Out(F_n)$ have finite image.

However at the moment, there seems to be a gap to extend the proof of Theorem H to symplectic universal lattice cases.

Therefore, Theorem G for symplectic universal lattices can be regarded as the high-end of this thesis. Theorem G together with Theorem H is non-arithmetization of Farb–Masur and Bridson–Wade theorems. Specially, in [FaSh], Farb and Shalen appealed to homomorphism rigidity for higher rank lattices with $Out(F_n)$ target in order to obtain rigidity results on group actions on a 3-dimensional manifold. It may be possible our theorems give some extension of their results beyond arithmetic lattice groups.

Finally, in Appendix, we make an estimation of a generalization of relative Kazhdan constant to uniformly bounded representation cases (on Hilbert spaces) as follows:

Proposition I. ([Mim4]; Proposition I.0.5 in this thesis) Let $A_k = \mathbb{Z}[x_1, \ldots, x_k]$ and set $G = \mathbb{E}_2(A_k) \ltimes A_k^2$ and $N = A_k^2 \leq G$. Set S be the set of all unit elementary matrices in $G (\subset SL_3(A_k))$ in the sense in Definition 4.1.13. Let $M \geq 1$ be a positive real. Then there is an inequality

$$\overline{\mathcal{K}}(G,N;S;M) > (15k+100)^{-1}M^{-6}.$$

In the case of k = 0, one has $\overline{\mathcal{K}}(\mathrm{SL}_2(\mathbb{Z}) \ltimes \mathbb{Z}^2, \mathbb{Z}^2; F; M) > (21M^6)^{-1}$. Here the symbol $\overline{\mathcal{K}}(G, N; S; M)$ denotes the generalized relative Kazhdan constant for uniformly bounded representations, which is defined in Definition 3.5.2.

Organization of this paper: Chapter 2 is for basic and fundamental facts on property (T) of Kazhdan. If the reader is familiar with this topic, this chapter can be omitted. Chapter 3 is on property (T_B) and property (F_B) of Bader– Furman–Gelander–Monod, and the reader also consult the original paper [BFGM]. In Chapter 4, we define elementary groups over rings, and universal lattices. Also

we see a celebrated argument by Shalom to prove property (T) from bounded generation [Sha1]. Vaserstein's bounded generation [Vas2], stated there, is a powerful tool throughout this thesis. In Chapter 5, we introduce another effective tool, called Shalom's machinery [Sha5], [Mim1]. To obtain this machinery, examine ultraproducts of metric spaces and that of isometric actions. This study has important application to reduced group cohomology. Chapter 6 is devoted to study of property (F_B) for universal lattices, in which Theorem A is proven. In Chapter 7, we introduce property (FF_B) , which is a generalization of property (TT), and see connection to bounded cohomology. In chapter 8, we introduce a notion of property $(FF_B)/T$, and prove Theorem B. Moreover we prove Theorem 8.1.7, which is a quite strong tool in deducing Property $(FF_B)/T$ (, although the proof is very elementary). We show Theorem C with the aid of this. Chapter 9 is utilized for definitions of elementary symplectic groups, and symlectic universal lattices. There we prove Theorem D, with use of Theorem 8.1.7. In Chapter 10, we consider quasi-homomorphisms and stable commutator lengths. We prove Theorem E. Chapter 11 is devoted for the proofs of Theorem F and Theorem G. The proof of Theorem G is involved and requires a number of deep facts on mapping class groups and automorphism groups of free groups. We briefly see them and complete the proof of Theorem G. We also verify Theorem H. In Appendix, we give a proof of Proposition I.

Hereafter, unless otherwise stating, we use and keep the notation and convention as in Chapter 0.

Chapter 2

Kazhdan's property (T)

In this chapter, we collect basic facts on property (T) of D. A. Kazhdan [Kaz]. One main goal is to show the Delorme–Guichardet theorem, which states Kazhdan's property (T) is equivalent to a certain cohomological property called property (FH) for locally compact and σ -compact groups. First, we give definitions of those two properties and see some permanence properties, including heredity to lattices. The other goal is to show totally higher rank algebraic groups and lattices, such as $SL_3(\mathbb{R})$ and $SL_3(\mathbb{Z})$, enjoy property (T). Main references of this chapter are Chapter 1, Chapter2, Appendix C, and Appendix E in a book of Bekka–de la Harpe–Valette [BHV].

2.1 Original definition

Kazhdan's property (T) has two famous formulations, namely, the original definition of (T), and the definition of property (FH). In this section, we state the original definition of Kazhdan, and see some properties of groups with (T).

2.1.1 Definition

We say a bounded operator U on a Hilbert space \mathfrak{H} is a *unitary* if U has a bounded inverse and if U preserves the inner product (, namely, for any $\xi, \eta \in \mathfrak{H} \langle U\xi | U\eta \rangle = \langle \xi | \eta \rangle$). Equivalently, if $U^*U = UU^* = I$, where I is the identity operator on \mathfrak{H} . The set $U(\mathfrak{H})$ of all unitary operators on \mathfrak{H} becomes a group with composition. A unitary representation of a group G is a group homomorphism $G \to U(\mathfrak{H})$ on some Hilbert space \mathfrak{H} . Recall from Chapter 0 that we always assume a group G are locally compact and σ -compact, and a representation π is strongly continuous, namely, for every $\xi \in \mathfrak{H}$, the map $G \to \mathfrak{H}$; $g \mapsto \pi(g)\xi$ is continuous.

Definition 2.1.1. Let (π, \mathfrak{H}) be a unitary representation of a group G.

(i) For a subset $S \subseteq G$ and $\kappa > 0$, a vector $\xi \in \mathfrak{H}$ is said to be (S, κ) -invariant if

$$\sup_{s\in S} \|\xi - \pi(s)\xi\| < \kappa \|\xi\|.$$

- (ii) We say π has almost invariant vectors if for any compact subset $S \subseteq G$ and $\kappa > 0$, there exists an (S, κ) -invariant vector.
- (*iii*) We write $\pi \succeq 1_G$ and say π weakly contains trivial representation if π has almost invariant vectors.
- **Remark 2.1.2.** (i) The zero vetor 0 is not (S, κ) -invariant for any $S \subseteq G$ and $\kappa > 0$ by definition.
 - (ii) Let $S \subseteq G$ and $\kappa > 0$ and $n \ge 2$. Then if ξ is (S, κ) -invariant for a unitary representation (π, \mathfrak{H}) , then ξ is $(\overline{S \cup S^{-1}}, \kappa)$ -invariant and $(S^n, \kappa/n)$ -invariant for π . (Here we refer to Chapter 0 for the definition of S^n .) Indeed, the first assertion is trivial because for any $g \in G$ and $\eta \in \mathfrak{H} \|\eta - \pi(g)\eta\| = \|\pi(g^{-1})\eta - \eta\|$ holds. The second assertion follows from the following argument, for instance for n = 2: for any $g_1, g_2 \in G$ and any $\eta \in \mathfrak{H}$,

$$\|\eta - \pi(g_1g_2)\eta\| = \|(\eta - \pi(g_1)\eta) + (\pi(g_1)\eta - \pi(g_1g_2)\eta)\|$$

$$\leq \|\eta - \pi(g_1)\eta\| + \|\pi(g_1)\eta - \pi(g_1)\pi(g_2)\eta\| = \|\eta - \pi(g_1)\eta\| + \|\eta - \pi(g_2)\eta\|.$$

Here recall $\pi(g_1)$ is a unitary.

From this point, we use this argument without mentioning. Note that this is available in a more general setting, in the case of (ρ, B) being an isometric representation on a Banach space.

We state the following basic observation, which is trivial in the case of discrete groups. We say a subset S of a group G is generating (or, S generates G) if $\bigcup_{n \in \mathbb{N}} (S \cup S^{-1})^n = G$. We say S is symmetric if $S = S^{-1}$. We say a group G is compactly generated (respectively, finitely generated) if there exists a compact (respectively finite) generating set.

Lemma 2.1.3. Let G be a compactly generated group. Then there exists a symmetric compact generating set \tilde{S} containing the group unit e. Moreover, for any compact subset $K \subseteq G$ there exists $m \in \mathbb{N}$ such that $K \subseteq \tilde{S}^m$.

Proof. By assumption, there exists a compact generating set S. Hence one can set $\tilde{S} = \overline{S \cup S^{-1} \cup e}$. For the second assertion, observe that $G = \bigcup_{n \in \mathbb{N}} \tilde{S}^n$. With recalling that we always assume G is locally compact (with the Hausdorff property), we apply the Baire category theorem. Therefore there exists $n \in \mathbb{N}$ such that \tilde{S}^n contains an open neighborhood U of e. By considering the open covering $K \subseteq \bigcup_{g \in K} gU$, we obtain a finite subcovering $K \subseteq \bigcup_{1 \leq i \leq l} g_i U$ (for some $l \in \mathbb{N}$). For each $1 \leq i \leq l$ there exists $n_i \in \mathbb{N}$ such that $g_i \in \tilde{S}^{n_i}$. Therefore, if one sets m as the maximum of n_i $(1 \leq i \leq l)$ and n, then this m works.

2.1. ORIGINAL DEFINITION

By combining Lemma 2.1.3 and item (ii) of Remark 2.1.2, we obtain the following corollary.

Corollary 2.1.4. Suppose a group G is compactly generated and let S be a compact generating set. Then for any unitary G-representation (π, \mathfrak{H}) , π has almost invariant vectors if and only if for every $\kappa > 0$ there exists (S, κ) -invariant vectors.

We shall see in Theorem 2.2.1 that every group with property (T) (see in Definition 2.1.5) is compactly generated.

Recall that we write $\pi \supseteq 1_G$ (and say π contains trivial representation) if 1_G is a subrepresentation of π , equivalently, if $\mathfrak{H}^{\pi(G)} \neq 0$ (namely, there exists a non-zero $\pi(G)$ -invariant vector). For any unitary representation π , there is an implication " $\pi \supseteq 1_G \Rightarrow \pi \succeq 1_G$." We make a remark the converse " $\pi \succeq 1_G \Rightarrow \pi \supseteq 1_G$ " is usually false for an infinite dimensional unitary representation π , as we will see in Example 2.1.6.

Definition 2.1.5. A group G is said to have Kazhdan's property (T) if for any unitary representation π of G, $\pi \succeq 1_G$ implies $\pi \supseteq 1_G$. Equivalently, if for any unitary representation π of G, whenever π has almost invariant vectors, π has a non-zero G-invariant vector.

Groups with (T) are also called *Kazhdan groups*.

Example 2.1.6. The group \mathbb{Z} is *not* Kazhdan. More precisely, we claim $\lambda_{\mathbb{Z}} \succeq 1_{\mathbb{Z}}$, here $\lambda_{\mathbb{Z}}$ denotes the left regular representation $\mathbb{Z} \to U(\ell^2(\mathbb{Z}))$ (, namely, for $m \in \mathbb{Z}$ and $f \in \ell^2(\mathbb{Z}), \lambda(m)(f(n)) := f(-m+n)$.) Indeed, take a sequence of subsets $S_n := [-n, n] \cap \mathbb{Z}$ indexed by $n \in \mathbb{N}$. Then it is easy to see that for any $m \in \mathbb{Z}$,

$$\lim_{n \to \infty} \frac{|(m \cdot S_n) \triangle S_n|}{|S_n|} = 0$$

Here $m \cdot$ means the left action of m as an element of the additive group \mathbb{Z} (, namely, "m+"), \triangle means the symmetric difference, and $|\cdot|$ means the number of sets. This means, for any finite set S and $\kappa > 0$, there exists $n \in \mathbb{N}$ such that for a unit vector $\xi := (|S_n|)^{-1/2} \cdot \chi_{S_n}$ in $\ell^2(\mathbb{Z})$ satisfies

$$\sup_{m \in S} \|\lambda_{\mathbb{Z}}(m)\xi - \xi\| < \kappa,$$

where χ is the characteristic function. This shows the claim. On the other hand, it is immediate that $\lambda_{\mathbb{Z}} \not\supseteq 1_{\mathbb{Z}}$.

This argument can be extended to the case of (locally compact) *amenable* groups, including all abelian groups, nilpotent groups, and solvable groups. For details of amenable groups, see Subsection 2.5.2.

2.1.2 Kazhdan constant

In the definition of Kazhdan's property (T), the choice of pairs (S, κ) may a priori depend on the choices of unitary representations π . The following lemma states one can take (S, κ) universal for π 's: all unitary representations.

Lemma 2.1.7. For a group G, the following are equivalent:

- (i) The group G has (T).
- (ii) There exist a compact subset $S \subseteq G$ and $\kappa > 0$ such that the following holds: for any unitary G-representation (π, \mathfrak{H}) , whenever π has an (S, κ) -invariant vector, $\pi \supseteq 1_G$ holds.

Proof. Condition (*ii*) implies condition (*i*) is trivial by definition. For the converse, we use a direct-sum argument. Suppose G does not satisfy condition (*ii*). Let I be the set of all pairs $\mu = (S, \kappa)$, where $S \subseteq G$ is compact and $\kappa > 0$, such that there exists a unitary G-representation $(\pi_{\mu}, \mathfrak{H}_{\mu})$ with $\pi_{\mu} \not\supseteq 1_{G}$ which has a unit (S, κ) -invariant vector ξ_{μ} . Set $(\pi, \mathfrak{H}) := (\bigoplus_{\mu \in I} \pi_{\mu}, \bigoplus_{\mu \in I} \mathfrak{H}_{\mu})$ be the ℓ^{2} -sum of the unitary representations. Then we claim that $\pi \succeq 1_{G}$ but that $\pi \not\supseteq 1_{G}$. Indeed, the first assertion follows from $\xi_{\mu} \in \mathfrak{H}_{\mu}$ being (S, κ) -invariant. For the second assertion, suppose $\eta = \bigoplus_{\mu \in I} \eta_{\mu}$ is $\pi(G)$ -invariant. Then each $\eta_{\mu} \in \mathfrak{H}_{\mu}$ is $\pi_{\mu}(G)$ -invariant. Hence all η_{μ} 's are zero and $\eta = 0$. Thus G cannot have (T).

We mention that in some other literature (such as [BHV]), Kazhdan's property (T) is defined as condition (*ii*) in Lemma 2.1.7.

Remark 2.1.8. The symbol $\pi \succeq 1_G$ (weak containment) comes from the concept of the *Fell topology* on sets of unitary representations. This concept is in general defined in terms of *positive definite functions* (in other words, of diagonal matrix coefficients), see Definition 2.2.11. We will shortly treat positive definite functions in Subsection 2.4.1, and here we only mention the following fact (for the proof, see [BHV, Theorem 1.2.5]). For a group G, \hat{G} denotes the *unitary dual* of G, namely, the set of all equivalence class of irreducible G-unitary representations.

Theorem 2.1.9. (Kazhdan [Kaz], Wang [Wan]) For a group G, the following are all equivalent:

- (i) The group G has property (T).
- (ii) The trivial representation 1_G is isolated in \hat{G} with respect to the Fell topology.
- (iii) Every finite dimensional irreducible unitary representation is isolated in G with respect to the Fell topology.

The very original definition of (T) in [Kaz] is condition (ii) in Theorem 2.1.9. The terminology "(T)" was named after this condition that the trivial representation , "T", is isolated, "()."

Definition 2.1.10. Let G be a group.

- (i) For $S \subseteq G$ and $\kappa > 0$, we say the pair (S, κ) is a *Kazhdan pair* for G if the following holds true: for any unitary G-representation π , whenever π has an (S, κ) -invariant vector, $\pi \supseteq 1_G$ holds.
- (ii) For a compact subset $S \subseteq G$, the Kazhdan constant $\mathcal{K}(G; S)$ is defined as the supremum of $\kappa > 0$ such that (S, κ) is a Kazhdan pair for G. If there is no such $\kappa > 0$, we set $\mathcal{K}(G; S) = 0$.

In other words, we define

$$\mathcal{K}(G;S) := \inf_{(\pi,\mathfrak{H})} \inf_{\xi \in S(\mathfrak{H})} \sup_{s \in S} \left\| \xi - \pi(s) \xi \right\|$$

where (π, \mathfrak{H}) moves among all unitary *G*-representations with $\pi \not\supseteq 1_G$, and $S(\mathfrak{H})$ denotes the unit sphere of \mathfrak{H} (recall Chapter 0).

As a corollary of Lemma 2.1.7, we have the following:

Corollary 2.1.11. A group G has (T) if there exists a compact subset S such that $\mathcal{K}(G;S) > 0$.

One importance of Kazhdan constants is due to the following lemma. This states that if G has (T) and (S, κ) is a Kazhdan pair, then for any unitary G-representation and for any vector, how close to being G-invariant is controlled uniformly by a behavior on S.

Lemma 2.1.12. Let G be a Kazhdan group and $S \subseteq G$ be a compact subset such that $\mathcal{K} = \mathcal{K}(G; S) > 0$. Let $\epsilon > 0$. Then for any unitary G-representation (π, \mathfrak{H}) , if $\xi \in \mathfrak{H}$ is (S, ϵ) -invariant, then there exists a $\pi(G)$ -invariant vector ξ_0 such that

$$\|\xi - \xi_0\| < \mathcal{K}^{-1} \epsilon \|\xi\|.$$

In particular, there is an inequality:

for any
$$g \in G$$
, $\|\xi - \pi(g)\xi\| < 2\mathcal{K}^{-1}\epsilon\|\xi\|$.

Proof. The point here is there is the following canonical decomposition of \mathfrak{H} , as *G*-representation spaces: $\mathfrak{H} = \mathfrak{H}^{\pi(G)} \oplus (\mathfrak{H}_{\pi(G)})^{\perp}$. Here $\mathfrak{H}^{\pi(G)}$ is the space of $\pi(G)$ invariant vectors (see Chapter 0), and $(\mathfrak{H}_{\pi(G)})^{\perp}$ denotes the orthogonal complement of $\mathfrak{H}^{\pi(G)}$. The *G*-invariance of $(\mathfrak{H}_{\pi(G)})^{\perp}$ follows from the unitary of representation π . Note that the following restriction of π

$$\pi' \colon G \to U((\mathfrak{H}_{\pi(G)})^{\perp})$$

satisfies $\pi' \not\supseteq 1_G$.

Decompose $\xi = \xi_0 + \xi_1$ such that $\xi_0 = (I - P)\xi \in \mathfrak{H}^{\pi(G)}$ and $\xi_1 = P\xi \in (\mathfrak{H}_{\pi(G)})^{\perp}$. Here *P* is the (orthogonal) projection $P \in \mathbb{B}(\mathfrak{H})$ onto $(\mathfrak{H}_{\pi(G)})^{\perp}$. By construction, for any $g \in G$, $\xi - \pi(g)\xi = \xi_1 - \pi'(g)\xi_1$ holds, and we have

$$\sup_{s \in S} \|\xi_1 - \pi'(s)\xi_1\| < \epsilon \|\xi\|$$

by assumption. On the other hand, by the definition of the Kazhdan constant, we have

$$\sup_{s \in S} \|\xi_1 - \pi'(s)\xi_1\| \ge \mathcal{K}\|\xi_1\|$$

Thus we conclude $\|\xi_1\| < \mathcal{K}^{-1}\epsilon \|\xi\|$, as desired. For the second part, we have for any $g \in G$,

$$\|\xi - \pi(g)\xi\| = \|\xi_1 - \pi'(g)\xi_1\| \le 2\|\xi_1\| < 2\mathcal{K}^{-1}\epsilon\|\xi\|.$$

In last part of this subsection, we see trivial examples of Kazhdan groups, more precisely, compact groups. We need the following lemma, but we shall prove this in more general setting in Chapter 3 (see Lemma 3.1.8).

Lemma 2.1.13. Let X be a bounded subset in a Hilbert space \mathfrak{H} . Then there exists a unique closed ball with the minimum radius which includes X. We define the Chebyshev center of X as the center of this ball.

Corollary 2.1.14. For any group G, (G, 1) is a Kazhdan pair. In particular, every compact group has property (T).

There is a remark: it is known that in fact $(G, \sqrt{2})$ is a Kazhdan pair for G. For the proof, see [BHV, Proposition 1.1.5].

Proof. Suppose a unitary G-representation (π, \mathfrak{H}) has (G, 1)-invariant vector ξ . Consider the orbit $X = \pi(G)\xi$, which is a bounded subset in \mathfrak{H} . Lemma 2.1.13 applies and there exists the Chebyshev center η of X. By uniqueness of η and unitary of π , $\eta \in \mathfrak{H}^{\pi(G)}$. Finally, $\eta \neq 0$ holds because $\sup_{\zeta \in X} \|\zeta - \xi\| < \sup_{\zeta \in X} \|\zeta - 0\| = \|\xi\|$ by assumption.

2.1.3 Relative (T)

The following relative version of property (T) was explicitly introduced by G. A. Margulis [Mar1] for an application to expander graphs (see Section 2.7). Now this concept is known to play an important role, as well as the original property (T).

Definition 2.1.15. Let G be a group and $N \leq G$ be a subgroup.

16

2.1. ORIGINAL DEFINITION

- (i) The pair $G \ge N$ is said to have *relative property* (T) if the following condition is satisfied: for any unitary *G*-representation (π, \mathfrak{H}) , whenever $\pi \succeq 1_G, \mathfrak{H}^{\pi(N)} \neq 0$ holds.
- (ii) A pair (S, κ) $(S \subseteq G \text{ and } \kappa > 0)$ is called a *relative Kazhdan pair* for $G \ge N$ if the following holds: for any unitary *G*-representation (π, \mathfrak{H}) , whenever π has an (S, κ) -invariant vector, $\mathfrak{H}^{\pi(N)} \neq 0$ holds.
- (*iii*) Let S be a compact subset of G. The relative Kazhdan constant for $G \ge N$, written as $\mathcal{K}(G, N; S)$, is defined as the supremum of κ such that (S, κ) is a relative Kazhdan pair for $G \ge N$. In other words,

$$\mathcal{K}(G,N;S) := \inf_{(\pi,\mathfrak{H})} \inf_{\xi \in S(\mathfrak{H})} \sup_{s \in S} \|\xi - \pi(s)\xi\|,$$

where (π, \mathfrak{H}) moves among all unitary *G*-representations with $\mathfrak{H}^{\pi(N)} = 0$.

We mention that a group G has (T) if and only if $G \ge G$ has relative (T). If $G \ge G_0 \ge N \ge N_0$, then relative (T) for $G_0 \ge N$ implies that for $G \ge N_0$ (in particular, by Corollary 2.1.14, for any group G and any compact subgroup N, $G \ge N$ has relative (T)). The next lemma can be shown in a similar way to one in Lemma 2.1.7.

Lemma 2.1.16. For a group pair $N \leq G$, the following are equivalent:

- (i) The pair $G \ge N$ has relative (T).
- (ii) There exists a compact subset $S \subseteq G$ such that $\mathcal{K}(G, N; S) > 0$.

In the following lemma, a priori the assumption of the *normality* of the subgroup N is necessary.

Lemma 2.1.17. Let G be a group and $N \leq G$ be a normal subgroup. Suppose $G \geq N$ has relative (T) and $S \subseteq G$ be a compact subset such that $\mathcal{K} = \mathcal{K}(G, N; S) > 0$. Let $\epsilon > 0$. Then for any unitary G-representation (π, \mathfrak{H}) , if $\xi \in \mathfrak{H}$ is (S, ϵ) -invariant, then there exists a $\pi(N)$ -invariant vector ξ_0 such that

$$\|\xi - \xi_0\| < \mathcal{K}^{-1}\epsilon \|\xi\|.$$

In particular, there is an inequality:

for any
$$h \in N$$
, $\|\xi - \pi(h)\xi\| < 2\mathcal{K}^{-1}\epsilon\|\xi\|$.

Proof. We imitate the proof of Lemma 2.1.12. The point here is the decomposition $\mathfrak{H} = \mathfrak{H}^{\pi(N)} \oplus (\mathfrak{H}_{\pi(N)})^{\perp}$ is a decomposition as *G*-representation spaces, provided N is normal. This follows from an equality

$$\pi(h)\pi(g)\xi = \pi(g)\pi(g^{-1}hg)\xi = \pi(g)\xi$$

for any $h \in N$, $g \in G$, and any $\xi \in \mathfrak{H}^{\pi(N)}$. Also, observe that the orthogonal projection $P \in \mathbb{B}(\mathfrak{H})$ onto $\mathfrak{H}^{\pi(N)}$ has the operator norm ≤ 1 , namely, for any $\eta = \eta_0 + \eta_1$ with $\eta_1 = P\eta$, $\|\eta_1\| \leq \|\eta\|$ holds (this is trivial). Decompose ξ as $\xi = \xi_0 + \xi_1$, where $\xi_0 = (I - P)\xi \in \mathfrak{H}^{\pi(N)}$ and $\xi_1 = P\xi \in (\mathfrak{H}_{\pi(N)})^{\perp}$. Then we have

$$\epsilon \|\xi\| > \sup_{s \in S} \|\xi - \pi(s)\xi\|$$

$$\geq \sup_{s \in S} \|P(\xi - \pi(s)\xi)\| = \sup_{s \in S} \|\xi_1 - \pi(s)\xi_1\|$$

(here P and $\pi(G)$ commute because $\mathfrak{H}^{\pi(N)}$ is $\pi(G)$ -invariant), and

$$\sup_{s \in S} \|\xi_1 - \pi(s)\xi_1\| \ge \mathcal{K}(G, N; S) \|\xi_1\|.$$

These inequalities end our proof.

Remark 2.1.18. This direct proof of Lemma 2.1.17 deeply relies on the assumption of N being normal. However with the aid of the Delorme–Guichardet theorem (see Section 2.4 and Theorem 2.4.13), it is possible to obtain a similar result for general cases (more precisely, we use the fact that relative property (T) is equivalent to relative property (FH) (Theorem 2.4.15), and that for the definition of relative (FH) there is no need to consider whether the subgroup is normal). For detailed arguments, we refer readers to a paper of P. Jolissaint [Jol].

One of the most important examples of group pairs with relative (T) is the pair $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$. More precisely, these are identified with

$$\left\{ \left(\begin{array}{c|c} W & v \\ \hline 0 & 1 \end{array} \right) : W \in \mathrm{SL}_2(\mathbb{Z}), \, v \in \mathbb{Z}^2 \right\} \quad \succeq \quad \left\{ \left(\begin{array}{c|c} I_2 & v \\ \hline 0 & 1 \end{array} \right) : v \in \mathbb{Z}^2 \right\}$$

(see Chapter 0).

We also note that $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2$ itself does *not* have property (T). For the proofs of having relative (T), and not having (T), see respectively Subsection 2.6.1 (and Subsection 4.2.1 for a quantitative proof), and Subsection 2.5.3.

2.2 Induction and permanence properties

In this section, we collect some consequences and permanence properties concerning (T). In particular, we see that having (T) implies compact generation and compact abelianization, and that (T) is inherited to lattices. For the proof of the last assertion, inductions of unitary representation play an important role.

2.2.1 Compact generation and compact abelianization

The following proposition was one of motivations of Kazhdan to introduce property (T).

Theorem 2.2.1. Property (T) implies compact generation. In particular, a discrete Kazhdan group is finitely generated.

Proof. Suppose G is a Kazhdan group. Let I be the set of all open and compactly generated subgroups of G. Observe that $G = \bigcup_{H \in I} H$ because every element in G has a compact neighborhood (we always assume local compactness) and because the subgroup generated by a set containing an open (non-empty) subset is open. For every $H \in I$, since $H \leq G$ is open, we regard G/H as a discrete set. Set $\lambda_{G/h} \colon G \to U(\ell^2(G/H))$ the quasi-regular representation, namely, $\lambda_{G/H}(g)(f(xH)) \coloneqq f(g^{-1}xH)$. Set $\delta_H \in \ell^2(G/H)$ the Dirac function at the point H. Let $\pi = \bigoplus_{H \in I} \lambda_{G/H}$ be the ℓ^2 sum of these representations.

We then claim $\pi \succeq 1_G$. Indeed, for a compact subset S of G, let $K \in I$ be the group generated by a relatively compact open neighborhood of S. Then $\sup_{s \in S} \|\delta_K - \pi(s)\delta_K\| = 0$, where we regard $\delta_K \in \ell^2(G/K)$ as an element of $\bigoplus_{H \in I} \ell^2(G/H)$.

Through property (T) for G, there exists a non-zero $\pi(G)$ -invariant vector $\xi = \bigoplus_{H \in I} \xi_H$, and in particular there exist $H \in I$ and $\xi_H \in \ell^2(G/H)$ such that $\xi_H \neq 0$ and is $\lambda_{G/H}(G)$ -invariant. This implies G/H is finite, and therefore G itself must be compactly generated.

Thanks to Theorem 2.2.1, a group G has property (T) if and only if for some compact generating set $S_0 \subseteq G$, $\mathcal{K}(G; S_0) > 0$ holds. By Lemma 2.1.3, in this case if a compact subset $S \subseteq G$ is a generating set, then $\mathcal{K}(G; S) > 0$ holds. The following proposition can be seen as a converse of this.

Proposition 2.2.2. Suppose a group G is Kazhdan. Let $S \subseteq G$ be a compact subset with a non-empty interior. Then if $\mathcal{K}(G; S) > 0$, then S generates G. (More precisely, $S \cup S^{-1}$ generates G.)

In particular, every finite subset S of a discrete Kazhdan group G satisfying $\mathcal{K}(G;S) > 0$ is a generating set.

Proof. Let H be the group generated by S. By assumption, H is an open subgroup of G. Hence one can consider the quasi-regular representation $\lambda_{G/H}: G \to U(\ell^2(G/H))$. The Dirac function δ_H at H in $\ell^2(G/H)$ is $\lambda_{G/H}(H)$ -invariant, and Lemma 2.1.12 shows this is in fact $\lambda_{G/H}(G)$ -invariant. Therefore G = H.

Remark 2.2.3. In Proposition 2.2.2, the assumption of S having a non-empty interior cannot be omitted. Indeed, Shalom shows in [Sha4], the following finite

subset S of $G = SL_3(\mathbb{R})$ satisfies $\mathcal{K}(G; S) > 0$:

$$\left\{ E_{1,2}(2) = \begin{pmatrix} 1 & 2 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix}, E_{2,1}(2) = \begin{pmatrix} 1 & 0 & 0 \\ 2 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix} \right\}$$

(recall the definition of $E_{i,j}(r)$ in Chapter 0).

Remark 2.2.4. Kazhdan [Kaz] asked whether a discrete Kazhdan group is necessarily finitely presented. The answer is negative, and the first counterexample was provided by Margulis [Mar2]. Recall from Chapter 0 we use the symbol q for a positive power of a prime number, and \mathbb{F}_q denotes the finite field of order q. It is well-known that the field $\mathbb{K} = \mathbb{F}_q((x))$ of Laurent series is a local field with respect to the absolute value $|\sum_{i=m}^{\infty} a_i x^i| := e^{-m}$ with $a_m \neq 0$. The subring $\mathbb{F}_q[x^{-1}]$, which is isomorphic to $\mathbb{F}_q[x]$ as discrete rings, is an ring of integers of the field \mathbb{K} . Therefore, $\mathrm{SL}_3(\mathbb{F}_q[x^{-1}])$, which is isomorphic to $\mathrm{SL}_3(\mathbb{F}_q[x])$ as discrete groups. is a lattice in $\mathrm{SL}_3(\mathbb{K})$. It shall be shown respectively in Subsection 2.6.2; and Subsection 2.2.2 that for any local field \mathbb{K} , $\mathrm{SL}_{m\geq 3}(\mathbb{K})$ has (T); and (T) passes to lattices. Therefore, the discrete group $\mathrm{SL}_3(\mathbb{F}_q[x])$ has (T). However, H. Behr shown in [Beh] that $\mathrm{SL}_3(\mathbb{F}_q[x])$ is *not* finitely presented.

Another example along this line is a universal lattice $SL_3(\mathbb{Z}[x])$, but the proof of property (T) for this group is much involved, as we mentioned in Chapter 1.

Concerning with this direction, we see a theorem of Shalom [Sha2] in Subsection 5.2.3.

Next we will see property (T) passes to group quotients.

Lemma 2.2.5. Let G_1 , G_2 be groups, and $\phi: G_1 \to G_2$ be a continuous homomorphism with dense range. If G_1 has (T), then so does G_2 . In particular, property (T) passes to group quotients.

Proof. This proof is very easy and uses a pull-back argument. Suppose (S_1, κ) is a Kazhdan pair for G_1 with S_1 compact. Then one can easily check that $(\phi(S_1), \kappa)$ is a Kazhdan pair for G_2 (with $\phi(S_1)$ compact), by composing ϕ and a unitary G_2 -representation.

Corollary 2.2.6. A Kazhdan group has compact abelianization. More precisely, for any Kazhdan group G, its abelianization $G/\overline{[G,G]}$ is compact. Here [G,G] denotes the commutator subgroup of G, namely, the (normal) subgroup generated by the set of single commutators $\{[g,h]: g, h \in G\}$ (see Chapter 0).

Proof. As we mentioned in Example 2.1.6, any noncompact abelian group cannot have property (T). The group quotient $G/\overline{[G,G]}$ is by definition abelian, and hence Lemma 2.2.5 ends the proof.

Finally, we see the permanence property under group extensions:

Proposition 2.2.7. For a group G and a normal subgroup $N \leq G$, the following are equivalent:

- (i) The group G has (T).
- (ii) The group G/N has (T), and the pair $G \ge N$ has relative (T).

In particular, property (T) is stable under group extensions.

Proof. Condition (i) implies condition (ii) is trivial, from Lemma 2.2.5.

To show the converse, firstly, we recall the following: If S' is a compact subset of G/N, then there exists a compact subset S of G such that $\phi(S) = S'$, where $\phi: G \to G/N$ is the canonical projection. Note that here we need not the assumption of normality of N, but the assumption of G being locally compact is necessary.

Hence by definition, there exist a relative Kazhdan pair (S_1, κ_1) for $G \geq N$, a Kazhdan pair (S', κ_2) for G/N, and a compact subset S_2 of G such that S_1, S' are compact and $\phi(S_2) = S'$. We claim $(S, \kappa) := (S_1 \cup S_2, \min_{i=1,2} \kappa_i/2)$ is a Kazhdan pair for G.

Indeed, let (π, \mathfrak{H}) be a unitary *G*-representation with a unit (S, κ) -invariant vector. Then by applying Lemma 2.1.17 with the pair being $G \geq N$ and $\epsilon = \kappa$, we have

$$||P\xi|| < \mathcal{K}(G, N; S)^{-1}\kappa \le \kappa_1^{-1}\kappa \le 1/2.$$

Here $P \in \mathbb{B}(\mathfrak{H})$ is the orthogonal projection onto $(\mathfrak{H}_{\pi(N)})^{\perp}$. (Recall that since $G \succeq N$, this projection P commutes with $\pi(G)$.) Set Q = I - P the orthogonal projection onto $\mathfrak{H}^{\pi(N)}$. Then $||Q\xi|| \ge 1/2$, and

$$\sup_{s \in S} \|Q\xi - \pi(s)Q\xi\| = \sup_{s \in S} \|Q(\xi - \pi(s)\xi)\|$$

$$\leq \sup_{s \in S} \|\xi - \pi(s)\xi\| < \kappa \le \kappa_2/2 \le \kappa_2 \|Q\xi\|.$$

Hence $Q\xi$ is (Q, κ_2) -invariant. Now note that the restriction of π to $\mathfrak{H}^{\pi(N)}$ factors through a unitary representation of G/N. From the fact that (Q', κ_2) is a Kazhdan pair for G/N, a standard pull-back argument then shows there must exists a nonzero $\pi(G)$ -invariant vector. This verifies our claim, and therefore G has (T).

The final assertion follows from the trivial observation that if N has (T), then for any $\Gamma \ge N$, the pair $\Gamma \ge N$ has relative (T).

Remark 2.2.8. As in Proposition 2.2.7, one can*not* expect there that N itself is Kazhdan. One example is $SL_m(\mathbb{R}) \ltimes \mathbb{R}^m$ with $m \ge 3$. Although this group has (T) (see Subsection 2.6.2), a normal subgroup \mathbb{R}^m is not Kazhdan.

2.2.2 Heredity to lattices

Firstly, recall the definition of lattices: a subgroup Γ in a group G (recall we always assume local compactness and σ -compactness of groups and closeness of subgroups) is called a *lattice* if it is discrete and carries a *finite invariant* regular Borel measure. If moreover G/Γ is compact, then we say the lattice Γ is *cocompact* (or, *uniform*). The main result in this subsection is that property (T) passes to lattices. To see this, we need some preliminaries on inductions of unitary representations.

Definition 2.2.9. Let G be a group, $H \leq G$ be a subgroup. Suppose G/H carries an invariant measure μ from a left Haar measure of G. Then for a unitary representation (σ, \mathfrak{K}) , we defined the *induced representation* of G by σ , written as $\operatorname{Ind}_{H}^{G}\sigma$, as follows:

- The representation space \mathfrak{H}_{μ} is the (Hilbert) space of all measurable mappings $\xi: G \to \mathfrak{K}$ such that
 - (i) For any $h \in H$ and almost all $x \in G$, $\xi(xh) = \sigma(h^{-1})\xi(x)$.
 - (*ii*) The norm $\|\xi\|$ is finite. Here the inner product on \mathfrak{H}_{μ} is defined by

$$\langle \xi_1 | \xi_2 \rangle := \int_{G/H} \langle \xi_1(x) | \xi_2(x) \rangle d\mu(xH).$$

Here $\langle \cdot | \cdot \rangle$ in the right hand side is the inner product on \mathfrak{K} .

• The unitary representation $\operatorname{Ind}_{H}^{G}\sigma$ on \mathfrak{H}_{μ} is defined by:

$$\operatorname{Ind}_{H}^{G}\sigma(g)(\xi(x)) := \xi(g^{-1}x).$$

For instance, with the assumptions in Definition 2.2.9, $\operatorname{Ind}_{H}^{G}1_{H}$ is the quasiregular representation $\lambda_{G/H}$ on $L^{2}(G/H, \mu)$. that means, $\lambda_{G/H}(g)(\xi(x)) := \xi(g^{-1}x)$

We note that here we assume G/H carries an invariant measure (and also σ compactness of groups, as mentioned above), and that in general setting the definition of induced representation is much more involved. We refer to Appendix E of
[BHV] for this topic.

We need the following proposition on containment of trivial representation:

Lemma 2.2.10. Under the assumptions of Definition 2.2.9, let σ be a unitary *H*-representation. Then the following are equivalent:

- (i) The induced representation $\operatorname{Ind}_{H}^{G}\sigma$ contains 1_{G} .
- (ii) The representation σ contains 1_H .

Proof. In this setting, the proof is very easy. First suppose condition (*ii*) and take a non-zero invariant vector $\eta \in \mathfrak{K}$. Then $\xi \colon G \to \mathfrak{K}$ defined as $\xi(x) = \eta$ (constant function) is in \mathfrak{H}_{μ} by assumption, and is a non-zero $\operatorname{Ind}_{H}^{G}\sigma(G)$ -invariant vector. Hence condition (i) holds. Conversely, suppose condition (i). Take a non-zero $\operatorname{Ind}_{H}^{G}\sigma(G)$ -invariant vector $\xi \in \mathfrak{H}_{\mu}$. Then from G-invariance, for any $x \in G \ \xi(x) = \xi(e)$ holds, and $\xi(e) \in \mathfrak{K}$ is a non-zero $\sigma(H)$ -invariant vector. Hence condition (ii) holds.

Note that in Lemma 2.2.10, without the assuption that G/H carries a finite invariant measure, it is true that condition (*i*) is equivalent to G/H carrying a finite invariant measure and having condition (*ii*). The proof of this is much harder, see Theorem E.3.1 in [BHV].

Next we need general definitions of weak containments among unitary representations. Recall for a unitary representation (π, \mathfrak{H}) of a group G, a function $f: G \to \mathbb{C}$ of the form $g \mapsto \langle \pi(g)\xi | \eta \rangle$ (for some ξ and η in \mathfrak{H}) is called a *matrix coefficient* of π . If in addition we can take $\xi = \eta$, this f is called a *diagonal matrix coefficient* of π .

Definition 2.2.11. Let G be a group and (π, \mathfrak{H}) and (σ, \mathfrak{K}) be two unitary representations of G. We say σ weakly contains π , wite as $\pi \leq \sigma$, if every diagonal matrix coefficient of π can be approximated, uniformly on compact subsets of G, by finite sums of diagonal matrix coefficients of ρ . More precisely, if the following holds true: for every $\xi \in \mathfrak{H}$, every compact subset $S \subseteq G$, and $\epsilon > 0$, there exist $n \in \mathbb{N}$ and $\eta_1, \ldots, \eta_n \in \mathfrak{K}$ such that

$$\sup_{s\in S} \left| \langle \pi(s)\xi|\xi\rangle - \sum_{i=1}^n \langle \sigma(s)\eta_i|\eta_i\rangle \right| < \epsilon.$$

- **Remark 2.2.12.** (*i*) The following three observations are easy: firstly, in Definition 2.2.11, one only has to check the condition under the assumption on ξ that $\|\xi\| = 1$. Secondly, $\pi \subseteq \sigma$ implies $\pi \preceq \sigma$. Thirdly, the weak containment relation is transitive, namely, for unitary representations π_1, π_2, π_3 of a group, if $\pi_1 \preceq \pi_2$ and $\pi_2 \preceq \pi_3$, then $\pi_1 \preceq \pi_3$.
 - (*ii*) Recall that in Definiiton 2.1.1, we introduce the symbol $1_G \leq \pi$ for the existence of almost invariant vectors for π . Here we see this definition is equivalent to the weak containment of trivial representation in the sense of Definition 2.2.11. Indeed, firstly observe for any unit vectors η_1, η_2 in a Hilbert space \mathfrak{H} ,

$$\|\eta_1 - \eta_2\|^2 = 2\operatorname{Re}(1 - \langle \eta_1 | \eta_2 \rangle).$$

Let (π, \mathfrak{H}) be a unitary representation of a group G. Suppose π has almost invariant vectors. Then for any compact subset $S \subseteq G$ and $\epsilon > 0$, there exists a unit (S, ϵ) -invariant vector η for π by assumption. Then by letting $\eta_1 = \pi(s)\eta$ $(s \in S)$ and $\eta_2 = \eta$ in the equality above, we have

$$\sup_{s \in S} \operatorname{Re}(1 - \langle \pi(s)\eta | \eta \rangle) < \epsilon^2/2.$$

By observing $\langle \pi(s)\eta|\eta\rangle$ sits on the unit disc $\{|z|\leq 1\}$ of \mathbb{C} , we conclude

$$\sup_{s \in S} |1 - \langle \pi(s)\eta |\eta \rangle| < \epsilon.$$

This means π weakly contains 1_G . The converse is also straightforward.

The following fact is the key to proof of heredity for property (T) to lattices. Note that Theorem 2.2.13 holds without the assumption of G/H carrying a finite invariant measure.

Theorem 2.2.13. (Continuity of inductions) With the assumptions in Definition 2.2.9, let $(\sigma, \mathfrak{K}_{\sigma}), (\tau, \mathfrak{K}_{\tau})$ be two unitary *H*-representations. If $\sigma \leq \tau$, then $\operatorname{Ind}_{H}^{G} \sigma \leq \operatorname{Ind}_{H}^{G} \tau$ holds.

For the proof, see Theorem F.3.5 in [BHV].

Theorem 2.2.14. Let G be a group and $\Gamma \subseteq G$ be a lattice. Then if G has (T), then so does Γ . In particular, property (T) for discrete groups is inherited to finite index subgroups.

Proof. Suppose σ is a unitary Γ -representation with $1_{\Gamma} \preceq \sigma$. Then by Theorem 2.2.13, $\operatorname{Ind}_{\Gamma}^{G} 1_{\Gamma} \preceq \operatorname{Ind}_{\Gamma}^{G} \sigma$. Since G/Γ carries a finite invariant measure, $\operatorname{Ind}_{\Gamma}^{G} 1_{\Gamma} \supseteq 1_{G}$. From item (*i*) of Remark 2.2.12, we have

$$1_G \preceq \operatorname{Ind}_{\Gamma}^G \sigma$$

Through property (T) for G, this implies

 $1_G \subseteq \operatorname{Ind}_{\Gamma}^G \sigma.$

Finally, by Lemma 2.2.10, we have $1_{\Gamma} \subseteq \sigma$. Therefore, Γ has property (T).

- **Remark 2.2.15.** (i) The relative version of the theorem is also ture. Namely, the following holds: "let G be a group, $H \leq G$ be a subgroup, and $\Gamma \leq G$ be a lattice. Then if $G \geq H$ has relative (T), then $(G \cap \Gamma) \leq (H \cap \Gamma)$ has relative (T)." For the proof, simply imitate the proof of Theorem 2.2.14.
- (ii) In fact the following hold true: "let G be a group and $H \subseteq G$ be a subgroup (we do not assume discreteness) with G/H carrying a finite invariant Borel regular measure. Then G has (T) if and only if H has (T)." For the proof of this, see Theorem 1.7.1 in [BHV].

2.3 Property (FH)

In this section, we shortly introduce *property* (FH), which happens to be equivalent to property (T), in relation to first group cohomology.

2.3.1 Affine isometric actions and first group cohomology

Definition 2.3.1. An affine isometric action α of a group G on a Hilbert space \mathfrak{H} is a group action of G of the form:

for any $g \in G$ and for any $\xi \in \mathfrak{H}$, $\alpha(g) \cdot \xi = \pi(g)\xi + c(g)$.

Here π is a unitary representation of G on H, and $c: G \to \mathfrak{H}$ be a map.

We call π the *linear part* of α , and c the *transition part* of α .

Recall from Chapter 0 we always assume (affine isometric) actions on Banach spaces are strongly continuous. In Definition 2.3.1, this corresponds to the assumptions that π is strongly continuous and c is continuous.

We note that in Definition 2.3.1, there is a constraint on the transition part $c: G \to \mathfrak{H}$ as follows. The condition of α being a group action means the following: for any $g, h \in G$ and any $\xi \in \mathfrak{H}$,

$$\alpha(gh) \cdot \xi = \alpha(g) \cdot (\alpha(h) \cdot \xi)$$

holds. This is equivalent to the following constraint of the transition part c (and π being a group representation):

for any
$$g, h \in G$$
, $c(gh) = c(g) + \pi(g)c(h)$.

This is a (1-)cocycle relation, called the *cocycle identity*, with unitary *G*-coefficient (π, \mathfrak{H}) . We shall regard an affine isometric action α (of a group *G* on a Hilbert space H) as a "trivial" affine isometric action if α has a global fixed point, namely, if there exists a vector $\eta \in \mathfrak{H}$ such that

for all
$$g \in G$$
, $\alpha(g) \cdot \eta = \eta$.

In the view of Definition 2.3.1, this is equivalent to the existence of $\eta \in \mathfrak{H}$ such that

for all
$$g \in G$$
, $c(g) = \eta - \pi(g)\eta$,

where π and c are respectively the linear part and the transition part of α . The condition above on c is a (1-) coboundary relation with unitary *G*-coefficient (π, \mathfrak{H}) (we mention that here we allow the case of $\eta = 0$. Do not confuse with linear representation cases, in which we consider *non-zero* invariant vectors).

To sum up, we have come up with the following definitions and proposition:

Definition 2.3.2. Let G be a group, and (π, \mathfrak{H}) be a unitary G-representation.

(i) A continuous map $c: G \to \mathfrak{H}$ is called a π -1-cocycle (or shortly, a π -cocycle) if the following holds:

For any $g, h \in G$, $c(gh) = c(g) + \pi(g)c(h)$.

This equality is called the *cocycle identity*.

(*ii*) A continuous map $c: G \to \mathfrak{H}$ is called a π -1-coboundary (or shortly, a π -coboundary) if the following holds:

There exists $\xi \in \mathfrak{H}$ such that for any $g \in G$, $c(g) = \xi - \pi(g)\xi$.

(*iii*) The space $Z_{c}^{1}(G; \pi, \mathfrak{H})$ (or shortly, $Z_{c}^{1}(G; \pi)$) denotes the vector space of all π -cocycles. The space $B_{c}^{1}(G; \pi, \mathfrak{H})$ (or shortly, $B_{c}^{1}(G; \pi)$) denotes the vector space of all π -coboundaries, which is a subspace of $Z_{c}^{1}(G; \pi)$. The quotient vector space

$$H^1_{\mathbf{c}}(G;\pi,\mathfrak{H}) := Z^1_{\mathbf{c}}(G;\pi)/B^1_{\mathbf{c}}(G;\pi)$$

(or shortly, $H^1_{\rm c}(G;\pi)$) is called the *first cohomology group* with π -coefficient.

Proposition 2.3.3. Let G be a group. Then for any Hilbert space \mathfrak{H} , there is a oneto-one transposepondence (as sets) between affine isometric actions α on \mathfrak{H} and pairs (π, c) of unitary G-representation π on \mathfrak{H} and π -cocycles c. The correspondence is:

$$\alpha(g) \cdot \xi = \pi(g)\xi + c(g) \quad (g \in G, \ \xi \in \mathfrak{H}).$$

For any unitary G-representation (π, \mathfrak{H}) , there is a one-to-one correspondence (as sets) between affine isometric actions, up to conjugation by a translation, with linear part π ; and elements in $H^1_c(G; \pi)$. In particular, $H^1_c(G; \pi) = 0$ if and only if every affine isometric G-action with linear part π has a global fixed point.

We mention that $H^1(G; \pi)$ is defined as the first cohomology with π -coefficient of a group G which is viewed as a discerete group. Here we allow non σ -compact (hence uncountable) discrete groups. From this point of view, we use the symbol $H^1(G; \pi)$ instead of $H^1_c(G; \pi)$ for discrete groups, such as $SL_m(\mathbb{Z})$ and universal lattices $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$.

2.3.2 Definition and relative (FH)

Definition 2.3.4. Let G be a group. We say G has property (FH) if for any unitary G-representation, $H_c^1(G; \pi) = 0$ holds. Equivalently, if every affine isometric G-action on a Hilbert space has a G-fixed point.

The terminology (FH) named after *Serre's property* (FA) (see Subsection 2.5.2), and means "Fixed point property on Hilbert spaces." We also define a relative version of (FH), similarly to the case of property (T).

Definition 2.3.5. Let G be a group and $H \leq G$ be a subgroup. We say a pair $G \geq H$ has *relative property* (FH) if for any unitary G-representation π and for every π -cocycle c, the restriction of c on H is a coboundary (with $\pi \mid_{H}$ -coefficient). Equivalently, if every affine isometric G-action on any Hilbert space has an H-fixed point.

Note that there is another formulation of relative property (FH) for $G \ge H$: for any unitary *G*-representation π , the restriction map

rest:
$$H^1_c(G;\pi) \to H^1_c(H;\pi\mid_H)$$

is a zero-map. However, this formulation is not as powerful as the cohomological formulation of property (FH).

The following lemma, which is based on Lemma 2.1.13, is of importance because this gives powerful interpretations of (FH) and relative (FH).

Lemma 2.3.6. Let G be a group and α be an affine isometric G-action on a Hilbert space. Then the following are all equivalent:

- (i) The action α has a G-fixed point.
- (ii) Any G-orbit is bounded.
- (iii) Some G-orbit is bounded.
- (iv) The cocycle c is bounded, where c is the transition part of α .

Corollary 2.3.7. Let G be a group and $H \leq G$ be a subgroup.

- (i) The group G has (FH) if and only if every affine isometric G-action on any Hilbert space has a bounded (G-)orbit. These are also equivalent to the condition that for any unitary G-representation, every π-cocycle is bounded.
- (ii) The pair $G \ge H$ has relative (FH) if and only if every affine isometric G-action on any Hilbert space has a bounded H-orbit. These are also equivalent to the condition that for any unitary G-representation, every π -cocycle is bounded on H.

We note that G has (FH) if and only if $G \ge G$ has relative (FH). Hence in Corollary 2.3.7, item (i) follows directly from item (ii).

Proof. (Lemma 2.3.6) Condition (i) implies condition (ii) is easy, and it is trivial that conditions (ii), (iii), (iv) are equivalent (note that the range of the cocycle c is the G-orbit of the origin). Hence it is enough to show condition (iii) implies condition (i). Suppose there exists a bounded $\alpha(G)$ -orbit X. By Lemma 2.1.13, there exists the Chebyshev center η of X. Because η is unique and α is affine isometric, this η is a global fixed point.

By Corollary 2.3.7, every compact group has (FH) (it is also verified more directly as follows. Let G be a compact group with μ the probability Haar measure. Then for any affine isometric G-action α , $\int_G \alpha(g) \cdot \xi d\mu$ is a global fixed point, where ξ is any vector).

2.3.3 Induction of 1-cocycles

As we have mentioned several times, property (FH) is equivalent to property (T), shall see Section 2.4. Nevertheless, we see some permanence properties of (FH). Specially, we consider inductions of affine isometric actions (or of cocycles).

The following lemma is proven in a pull-back argument, similar to the case of Lemma 2.2.5.

Lemma 2.3.8. Let G_1 , G_2 be groups, and $\phi: G_1 \to G_2$ be a continuous homomorphism with dense range. If G_1 has (FH), then so does G_2 .

In particular, property (FH) passes to group quotients.

This implies that a group with (FH) has compact abelianization (recall we always assume σ -compactness).

The next lemma is trivial in the view of Corollary 2.3.7.

Lemma 2.3.9. Property (FH) is stable under group extensions.

Our goal in this subsection is show that property (FH) is inherited to *cocompact* lattices, without the use of the Delorme–Guichardet theorem. For the proof, we need inductions of (1-)cocycles, which is introduced in [Sha3].

First, we recall the definition of Borel fundamental domains, and the following basic fact (for the proof, we refer to Proposition B.2.4 in [BHV]).

Definition 2.3.10. Let G be a group and $H \leq G$ be a subgroup. A Borel fundamental domain for H is a Borel subset $\mathcal{D} \subseteq G$ such that $G = \bigsqcup_{h \in H} \mathcal{D}h$ (this symbol means a disjoint union).

Proposition 2.3.11. Let G be a group and $\Gamma \leq G$ be a discrete subgroup.

- (i) There exists a Borel fundamental domain for Γ .
- (ii) If Γ is a lattice, then every Borel fundamental domain for Γ has finite Haar measure. If Γ is moreover cocompact, then for every Borel fundamental domain D satisfies the following: for any compact subset S ⊆ G, the set {γ ∈ Γ : Dγ ∩ S ≠ Ø} is finite.

In this subsection, henceforth, we let G be a group, $\Gamma \leq G$ be a lattice, and \mathcal{D} be a Borel fundamental domain for Γ ; we let μ be a Haar measure of G with $\mu(\mathcal{D}) = 1$; and we identify \mathcal{D} with G/Γ and regard \mathcal{D} as a (left) G-space. We define a map $\beta: G \times \mathcal{D} \to \Gamma$ by the following rule:

 $\beta(g, x) = \gamma$ if and only if $g^{-1}x\gamma \in \mathcal{D}$.

Lemma 2.3.12. With the setting above, the map β is a Borel cocycle. That means, β satisfies the following equality:

for any
$$g, h \in G$$
 and any $x \in \mathcal{D}$, $\beta(gh, x) = \beta(g, x)\beta(h, g^{-1}x)$.

2.3. PROPERTY (FH)

Proof. Recall we identify \mathcal{D} with G/Γ and endow \mathcal{D} with the associated left G-action. Therefore, in the equality in the lemma, " $g^{-1}x \in \mathcal{D}$ " means $g^{-1}x \cdot \beta(g, x) \in \mathcal{D}$. After observing this, one directly obtains the conclusion through the definition of β . \Box

With the preparation above, firstly, we shall reformulate induced representations in terms of the Borel cocycle β . Let (σ, \mathfrak{K}) be a unitary Γ -representation. Then $\operatorname{Ind}_{\Gamma}^{G}\sigma$ coincides with the following *G*-representation π :

• The representation space of π is $L^2(\mathcal{D}, \mathfrak{K})$, equipped with a natural inner product: for $\xi_i \in L^2(\mathcal{D}, \mathfrak{K})$ (i = 1, 2),

$$\langle \xi_1 | \xi_2 \rangle := \int_{\mathcal{D}} \langle \xi_1(x) | \xi_2(x) \rangle d\mu(x).$$

Here in the right hand side of the equality, $\langle \cdot | \cdot \rangle$ is the inner product of \mathfrak{K} .

• The representation is defined by the equality: for any $g \in G$, $\xi \in L^2(\mathcal{D}, \mathfrak{K})$ and $x \in \mathcal{D}$,

$$\pi(g)\xi(x) := \sigma(\beta(g, x))\xi(g^{-1} \cdot x).$$

Note that this π becomes a group representation because β is a Borel cocycle (Lemma 2.3.12). Thus we identify $\operatorname{Ind}_{\Gamma}^{G}\sigma$ with the representation π above on $\mathfrak{H}_{\sigma} := L^{2}(\mathcal{D}, \mathfrak{K}).$

Secondly, we proceed to the definition of induced cocycles. For the well-definedness of induced cocycles, we need some restriction on lattices.

Definition 2.3.13. A lattice Γ in a group G is said to be 2-*integrable* if either of the following two conditions is satified:

- (1) the lattice Γ is cocompact;
- (2) the lattice Γ is finitely generated, and for some (equivalently any) symmetric finite generating set S of Γ , there exists a Borel fundamental domain $\mathcal{D} \subseteq G$ such that

for any
$$g \in G$$
, $\int_D l_S(\beta(g, x))^2 d\mu(x) < \infty$.

Here $l_S \colon \Gamma \to \mathbb{Z}_{\geq 0}$ denotes the *word length* on Γ with respect to S. Namely, for $\gamma \in \Gamma$, $l_S(\gamma)$ is the smallest number $n \geq 0$ such that $\gamma \in S^n$.

Remark 2.3.14. We note that for a finitely generated group Γ , for every pair of (symmetric) finite generating sets S_1 and S_2 , l_{S_1} and l_{S_2} are *bi-Lipschitz equivalent*. That means, for every pair S_1 and S_2 , there exist a constant C > 1 such that for any $\gamma \in \Gamma$, $C^{-1} \cdot l_{S_2}(\gamma) \leq l_{S_1}(\gamma) \leq C \cdot l_{S_2}(\gamma)$ holds. Therefore, the inequality in condition (2) in Definition 2.3.13 does not depend on the choice of S.

The following proposition is the reason why one needs 2-integrability for lattices here:

Proposition 2.3.15. Suppose a lattice Γ in a gorup G is 2-integrable (with a Borel fundamental domain \mathcal{D}). Let (σ, \mathfrak{K}) be a unitary Γ -representation and c be a σ -(1-)cocycle. Set a map \tilde{c} defined by the following equality:

$$\tilde{c}(g)(x) := c(\beta(g, x)) \quad (g \in G, \ x \in \mathcal{D}).$$

Then this \tilde{c} ranges into $\mathfrak{H}_{\sigma} = L^2(\mathcal{D}, \mathfrak{K})$, and is an $\mathrm{Ind}_{\Gamma}^G \sigma$ -cocycle.

Proof. Firstly, we observe that this \tilde{c} satisfies the formal cocycle identity, namely,

for any $g, h \in G$, $\tilde{c}(gh) = \tilde{c}(g) + \operatorname{Ind}_{\Gamma}^{G} \sigma(g) \tilde{c}(h)$.

This is because by Lemma 2.3.12, for any $g, h \in G$ and $x \in \mathcal{D}$,

$$\begin{split} \tilde{c}(gh)(x) &= c(\beta(gh, x)) = c(\beta(g, x)\beta(h, g^{-1}x)) \\ &= c(\beta(g, x)) + \sigma(\beta(g, x))c(\beta(h, g^{-1}x)) = \tilde{c}(g)(x) + \operatorname{Ind}_{\Gamma}^{G}\sigma(g)\tilde{c}(h). \end{split}$$

Here we also use the fact that c is a σ -cocycle.

Secondly, we verify the square integrability of \tilde{c} , namely,

for any
$$g \in G$$
, $\int_{\mathcal{D}} \|\tilde{c}(g)(x)\|^2 d\mu(x) < \infty$.

Here is the main part of the proof. We firstly treat the case of that Γ is cocompact. Then by item (*ii*) of Lemma 2.3.13, for every $g \in G$, $\beta(g, x)$ takes only finitely many values, and hence we have the conclusion. We next deal with the case of that Γ satisfies condition (2) in Definition 2.3.13. Take a finite generating set S of Γ , and set $M < \infty$ as $\sup_{s \in S} ||c(s)||$. Since \tilde{c} satisfies the formal cocycle condition, for any $g \in G$ and $x \in \mathcal{D}$

$$\|c(\beta(g,x))\| \le M \cdot l_S(\beta(g,x))$$

holds. Thus the conclusion is confirmed from the inequality in condition (2) of Definition 2.3.13.

Finally, we see the continuity of \tilde{c} , but this is almost by definition (observe that \tilde{c} is a measurable map).

Theorem 2.3.16. Suppose G be a group and Γ is a 2-integrable lattice in Γ . Then if G has (FH), then so does Γ .

Proof. Take any unitary Γ -representation (σ, \mathfrak{K}) and any σ -cocycle c. By the assumption of 2-integrability of Γ , by employing Proposition 2.3.15, we obtain the induced cocycle \tilde{c} . By property (FH) for G, this \tilde{c} is a coboundary, namely, there exists $\xi \in L^2(\mathcal{D}, \mathcal{K})$ such that for any $g \in G$, $\tilde{c}(g) = \xi - \operatorname{Ind}_{\Gamma}^G \sigma(g)\xi$. Therefore, by letting $\eta = \xi(e) \in \mathfrak{K}$, we have

for any
$$\gamma \in \Gamma$$
, $c(\gamma) = \eta - \sigma(\gamma)\eta$.

This means c is a σ -coboundary. Hence Γ has (FH).

Remark 2.3.17. The relative version of the theorem is also true. Namely, the following holds: "let G be a group, $H \leq G$ be a subgroup, and $\Gamma \leq G$ be a 2-integrable lattice. Then if $G \geq H$ has relative (FH), then $(G \cap \Gamma) \geq (H \cap \Gamma)$ has relative (FH)."

2.4 Delorme–Guichardet theorem

In this section, we prove the Delorme–Guichardet theorem, which states property (T) is equivalent to property (FH). For the proof, we study on positive definite functions and conditionally negative definite functions on groups.

2.4.1 Positive definite functions

Definition 2.4.1. (i) Let X be a topological space. A positive definite kernel on X is a continuous function $\Phi: X \times X \to \mathbb{C}$ such that the following holds: for any $n \in \mathbb{N}$, any complex numbers c_1, \ldots, c_n , and any $x_1, \ldots, x_n \in X$,

$$\sum_{i=1}^{n} \sum_{j=1}^{n} c_i \overline{c_j} \Phi(x_i, x_j) \ge 0.$$

Here $\overline{c_j}$ means the complex conjugation of c_j .

(*ii*) Let G be a group. A *positive definite function* on G is a continuous function $\phi: G \to \mathbb{C}$ such that the kernel defined by

$$(g,h) \mapsto \phi(h^{-1}g)$$

is positive definite.

Note that the inequality in item (i) is equivalent to that the $n \times n$ matrix $[\Phi(x_i, x_j)]_{i,j}$ is positive definite.

Example 2.4.2. The following are typical examples:

(i) Let $f: X \to \mathfrak{H}$ be a continuous map from a topological space X to a Hilbert space \mathfrak{H} . Then a kernel

$$X \times X \to \mathbb{C}; \ (x,y) \mapsto \langle f(x) | f(y) \rangle$$

is positive definite.

(ii) Let $\pi: G \to U(\mathfrak{H})$ be a unitary representation of a group G. Then a function

$$G \to \mathbb{C}; \ g \mapsto \langle \pi(g)\xi|\xi \rangle$$

is positive definite. Here $\xi \in \mathfrak{H}$ is any vector.

Indeed, for item (*ii*), it follows from the equality that for any $n \in \mathbb{N}$, any $c_1, \ldots, c_n \in \mathbb{C}$, and any $g_1, \ldots, g_n \in G$,

$$\sum_{i=1}^{n} \sum_{j=1}^{n} c_i \overline{c_j} \langle \pi(g_i) \xi | \pi(g_j) \xi \rangle = 2 \left\| \sum_{i=1}^{n} c_i (\pi(g_i) \xi - \xi) \right\|^2.$$

The following theorem, so-called the *GNS construction* (named after Gelfand–Naimark–Segal) states that Example 2.4.2 is universal.

Theorem 2.4.3. (GNS construction for positive definite kernels)

- (i) Let X be a topological space. Then for any positive definite kernel Φ on X, there exist a Hilbert space \mathfrak{H} and a continuous map $f: X \to \mathfrak{H}$ such that for any $x, y \in X$, $\Phi(x, y) = \langle f(x) | f(y) \rangle$. Moreover, if one takes \mathfrak{H} such that the linear span of $\{f(x) : x \in X\}$ is dense in \mathfrak{H} , then the pair (\mathfrak{H}, f) is unique up to canonical isomorphism.
- (ii) Let G be a group. Then for any positive definite function ϕ on G, there exists a tripleHilbert (π, \mathfrak{H}, ξ) of a unitary G-representation (π, \mathfrak{H}) and a vector $\xi \in \mathfrak{H}$ such that for any $g \in G$, $\phi(g) = \langle \pi(g)\xi|\xi \rangle$ holds. Moreover, if one takes \mathfrak{H} such that the linear span of $\{\pi(g)\xi : g \in G\}$ is dense in \mathfrak{H} (equivalently, ξ is a cyclic vector for (π, \mathfrak{H})), then the triple (π, \mathfrak{H}, ξ) is unique up to canonical isomorphism.

Note that in item (*ii*), by uniqueness we write (π, \mathfrak{H}, ξ) as $(\pi_{\phi}, \mathfrak{H}_{\phi}, \xi_{\phi})$ (, where ξ_{ϕ} is a cyclic vector,) and call it the *GNS-triple* for ϕ . For the proof of Theorem 2.4.3, see Theorem C.1.4 and Theorem C.4.10 in [BHV].

It is clear that for a topological space X, positive definite kernels on X are closed under positive linear combinations, and under a pointwise limit (inside continuous kernels on X). The next lemma, which is a corollary of Theorem 2.4.3, states that they are also closed under (pointwise) multiplications.

Lemma 2.4.4. Let X be a topological space. If Φ_1 and Φ_2 are positive definite kernels on X, then so is $\Phi_1\Phi_2$ (here the multiplication is pointwise). In particular, if ϕ_1 and ϕ_2 are positive definite functions on a group G, then so is $\phi_1\phi_2$.

Proof. By Theorem 2.4.3, there exist corresponding (f_1, \mathfrak{H}_1) and (f_2, \mathfrak{H}_2) . Consider the tensor product Hilbert space $\mathfrak{H} = \mathfrak{H}_1 \otimes \mathfrak{H}_2$ and define $f: X \to \mathfrak{H}$ by $f(x) = f_1(x) \otimes f_2(x)$. Then (f, \mathfrak{H}) realizes $\Phi_1 \Phi_2$.

2.4.2 Conditionally negative definite functions

Definition 2.4.5. (i) Let X be a topological space. A conditionally negative definite kernel on X is a continuous function $\Psi: X \times X \to \mathbb{R}$ such that the following three conditions are satisfied:

2.4. DELORME–GUICHARDET THEOREM

- (a) For any $x \in X$, $\Psi(x, x) = 0$.
- (b) For any $x, y \in X$, $\Psi(x, y) = \Psi(y, x)$.
- (c) For any $n \in \mathbb{N}$, any real numbers c_1, \ldots, c_n with $\sum_{i=1}^n c_i = 0$, and any $x_1, \ldots, x_n \in X$,

$$\sum_{i=1}^{n} \sum_{j=1}^{n} c_i c_j \Psi(x_i, x_j) \le 0.$$

(*ii*) Let G be a group. A negative definite function on G is a continuous function $\psi: G \to \mathbb{R}$ such that the kernel defined by

$$(g,h) \mapsto \psi(h^{-1}g)$$

is conditionally negative definite.

The following are typical examples:

Example 2.4.6. (i) Let $f: X \to \mathfrak{H}$ be a continuous map from a topological space X to a Hilbert space \mathfrak{H} . Then, a kernel

$$X \times X \to \mathbb{R}; \ (x, y) \mapsto \|f(x) - f(y)\|^2$$

is conditionally negative definite.

(*ii*) Let α be a affine isometric action of a group G on a Hilbert space \mathfrak{H} . Then for any $\xi \in \mathfrak{H}$, a function

$$G \to \mathbb{R}; \ g \mapsto \|\alpha(g) \cdot \xi - \xi\|^2$$

is conditionally negative definite. In particular, for any unitary G-representation (π, \mathfrak{H}) and any π -cocycle c, a function

$$G \to \mathbb{R}; \ g \mapsto \|c(g)\|^2$$

is conditionally negative definite (note that by cocycle identity, c(e) = 0. This follows from the observation that c(g) coincides with $\alpha(g) \cdot 0$ where α is the affine isometric action associated with (π, c) , see Proposition 2.3.3).

Indeed, for item (*ii*), it follows from the equality that for any $n \in \mathbb{N}$, any $c_1, \ldots, c_n \in \mathbb{R}$ with $\sum_{i=1}^n c_i = 0$, and any $g_1, \ldots, g_n \in G$,

$$\sum_{i=1}^{n} \sum_{j=1}^{n} c_i c_j \|\alpha(g_i) \cdot \xi - \alpha(g_j) \cdot \xi\|^2 = -2 \left\| \sum_{i=1}^{n} c_i \alpha(g_i) \cdot \xi \right\|^2.$$

We note that there is a GNS construction theorem for conditionally negative definite kernels, and that states item (i) of Example 2.4.6 is universal. (See Theorem C.2.3 in [BHV].) We note that item (ii) of Example 2.4.6 is *not* universal, because what one needs are continuous *G*-action on a topological space *X*, and conditionally negative definite kernel on *X* which is *G*-invariant.

Remark 2.4.7. We also make a remark that in the view of conditionally negative kernels and functions, it is more natural to regard Hilbert spaces as *real* Hilbert paces. Also compare with the Mazur–Ulam theorem states that any (surjective) isometry on a *real* Banach space is linear. This implies that any isometric group action on a *real* Hilbert space is automatically affine.

2.4.3 Schoenberg's theorem

The following theorem, due to Schoenberg [Sch], relates positive definite functions to conditionally negative kernels, and vice versa.

Theorem 2.4.8. (Schoenberg's theorem) Let X be a topological space, and $\Psi: X \times X \to \mathbb{R}$ be a continuous kernel on X such that for any $x \in X \Psi(x, x) = 0$ and for any $x, y \in X \Psi(x, y) = \Psi(y, x)$. Then the following are equivalent:

- (i) The kernel Ψ is conditionally negative definite.
- (ii) For any $t \ge 0$, $e^{-t\Psi}$ is positive definite.

As a corollary, through Example 2.4.6 we obtain the following corollary:

Corollary 2.4.9. Let G be a group and π be a unitary G-representation. Let c be a π -cocycle. Then for any $t \ge 0$, the function

$$G \to \mathbb{R}; \quad g \mapsto \exp(-t \|c(g)\|^2)$$

is positive definite.

This corollary is a key to proving Delorme's part of the Delorme–Guichardet theorem, namely, property (T) implies property (FH). Roughly speaking, the importance of Corollary 2.4.9 is the following: property (FH) concerns affine isometric actions, and apparently this may have much information than property (T) has (because affine isometric actions have not only linear parts but also cocycle parts, and a priori property (T) does not seem to give any information for cocycle parts). However, thanks to Corollary 2.4.9, in the view of Theorem 2.4.3, one can extract some information on linear (unitary) representations from that of cocycles.

Since Corollary 2.4.9 is needed for our purpose, we only give a proof of this. For the proof of Theorem 2.4.8, see for instance Theorem C.3.2 in [BHV].

Proof. (Corollary 2.4.9) We show the case of t = 1 for simplicity. Set $\phi(g) = \exp(-\|c(g)\|^2)$. In the view of Remark 2.4.7, we may assume the Hilbert space is real. Then for $g, h \in G$,

$$\phi(h^{-1}g) = \exp(-\|c(h^{-1}g)\|^2) = \exp(-\|c(h^{-1}) + \pi(h^{-1})c(g)\|^2)$$

= $\exp(-\|\pi(h^{-1})(-c(h) + c(g))\|^2) = \exp(-\|c(g) - c(h)\|^2)$
= $\exp(-\|c(g)\|^2) \exp(-\|c(h)\|^2) \exp(2\langle c(g)|c(h)\rangle).$

Here we use $c(h^{-1}) = -\pi(h^{-1})c(h)$, which directly follows from the cocycle identity. By the equality

$$\exp(2\langle c(g)|c(h)\rangle) = \sum_{n\geq 0} \frac{(2\langle c(g)|c(h)\rangle)^n}{n!},$$

the kernel on G; $(g, h) \mapsto \exp(2\langle c(g)|c(h)\rangle)$ is a positive definite kernel. Also, the kernel on G; $(g, h) \mapsto \exp(-\|c(g)\|^2) \exp(-\|c(h)\|^2)$ is a positive definite kernel, because this can be seen as a map $(g, h) \mapsto \langle \exp(-\|c(g)\|^2)|\exp(-\|c(h)\|^2)\rangle_{\mathbb{R}}$ (here we see \mathbb{R} as a 1-dimensional Hilbert space). Therefore, the kernel on G; $(g, h) \mapsto \exp(-\|c(h^{-1}g)\|^2)$ is positive definite by Lemma 2.4.4. This means the function $G \ni g \mapsto \exp(-\|c(g)\|^2)$ is a positive definite function on G, as desired. \Box

We state the following definition and lemma, which is needed in Subsection 3.3.1.

Definition 2.4.10. A continuous function $F \colon \mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$ is called a *Bernstein* function if there exist a positive measure ν on Borel subsets of $\mathbb{R}_{>0}$ and $C \geq 0$ such that for any t > 0

$$F(t) = Ct + \int_0^{+\infty} (1 - e^{-tx}) d\nu(x)$$

holds. Here ν satisfies the following two conditions:

- (i) For any $\epsilon > 0$, $\nu(\epsilon, \infty) < \infty$.
- (*ii*) The inequality $\int_0^1 x d\nu(x) < \infty$ holds.

Lemma 2.4.11. Let Ψ be a conditionally negative definite kernel on a topological space X, and F be a Bernstein function. Then $F \circ \Psi$ is conditionally negative definite.

Note that conditionally negative definite kernels on a topological space X are closed under positive linear combinations, and under a pointwise limit (inside continuous kernels on X).

Proof. (Lemma 2.4.11) By Shoenberg's theorem (Theorem 2.4.8), for every $t > 0 e^{-t\Psi}$ is positive definite, and hence $1 - e^{-t\Psi}$ is negative definite (in particular, conditionally negative definite). Therefore conclusion follows from the observation above. Note that conditions on the measure ν in Definition 2.4.10 is needed for the well-definedness of $F \circ \Psi$, namely, $F \circ \Psi(x, y) < \infty$ for any $x, y \in X$ and $F \circ \Psi$ being continuous.

Remark 2.4.12. In Corollary 2.4.9, in the view of Theorem 2.4.3, there must exist the GNS triple for the positive definite function $g \mapsto \exp(-t\|c(g)\|^2)$. One can explicitly construct this triple by taking the (*full*) Fock space, as in below:

From the view point of Remark 2.4.7, suppose the representation (π, \mathfrak{H}) is real. Define $\operatorname{Exp}(\mathfrak{H}) := \bigoplus_{n>0} \mathfrak{H}^{\otimes^n}$, where $\mathfrak{H}^{\otimes^0} := \mathbb{C}$. Also, define a map (among sets)

$$\mathfrak{H} \hookrightarrow \operatorname{Exp}(\mathfrak{H}); \ \xi \mapsto \operatorname{Exp}(\xi) := 1 \oplus \xi \oplus \frac{\xi^{\otimes^2}}{\sqrt{2!}} \oplus \frac{\xi^{\otimes^3}}{\sqrt{3!}} \oplus \cdots$$

For any $s \ge 0$, define a map (among sets)

$$\phi^s \colon \mathfrak{H} \hookrightarrow \operatorname{Exp}(\mathfrak{H}); \ \xi \mapsto \phi^s(\xi) := \exp\left(-\frac{s^2}{2} \|\xi\|^2\right) \operatorname{Exp}(s\xi).$$

Then we have for any $s \ge 0$ and any $\xi, \eta \in \mathfrak{H}$,

$$\langle \phi^s(\xi) | \phi^s(\eta) \rangle = \exp\left(-\frac{s^2}{2} \|\xi - \eta\|^2\right).$$

In particular, this ϕ^s maps \mathfrak{H} into the unit sphere $S(\operatorname{Exp}(\mathfrak{H}))$.)

Now let α be the affine isometric *G*-action on \mathfrak{H} associated with (π, c) . For $s \geq 0$, denote by \mathfrak{K}^s the closure of linear span of $\{\phi^s(\xi) : \xi \in \mathfrak{H}\}$ in $\operatorname{Exp}(\mathfrak{H})$. We define unitary *G*-representation π^s_{α} on \mathfrak{K}^s by the following equality: for any $\lambda \in \mathbb{C}$ and any $\xi \in \mathfrak{H}$,

$$\pi^s_{\alpha}(g)(\lambda\phi^s(\xi)) := \lambda\phi^s(\alpha(g)\cdot\xi)$$

This is a priori densely defined, and can be extended to the closure \Re^s . Indeed, for a vector $\sum_i \lambda_i \phi^s(\xi_i) \in \Re^s$,

$$\begin{aligned} \left\| \pi_{\alpha}^{s}(g)(\sum_{i} \lambda_{i} \phi^{s}(\xi_{i})) \right\|^{2} &= \left\| \sum_{i} \lambda_{i} \phi^{s}(\alpha(g) \cdot \xi_{i}) \right\|^{2} = \sum_{i} \lambda_{i} \overline{\lambda_{j}} \langle \phi^{s}(\alpha(g) \cdot \xi_{i}) | \phi^{s}(\alpha(g) \cdot \xi_{j}) \rangle \\ &= \sum_{i,j} \lambda_{i} \overline{\lambda_{j}} \exp(-s^{2}/2 \cdot \|\alpha(g) \cdot \xi_{i} - \alpha(g) \cdot \xi_{j}\|^{2}) \\ &= \sum_{i,j} \lambda_{i} \overline{\lambda_{j}} \exp(-s^{2}/2 \cdot \|\xi_{i} - \xi_{j}\|^{2}) \\ &= \left\| \sum_{i} \lambda_{i} \phi^{s}(\xi_{i}) \right\|^{2}. \end{aligned}$$

Therefore, π^s_{α} is a unitary representation on \mathfrak{K}^s , with the equality

$$\langle \pi_{\alpha}^{s}(g)\phi^{s}(0)|\phi^{s}(0)\rangle = \exp\left(-\frac{s^{2}}{2}\|c(g)\|^{2}\right).$$

Thus we obtain the GNS triple $(\pi^s_{\alpha}, \mathfrak{K}^s, \phi^s(0))$ for the positive definite function $g \mapsto \exp(-t\|c(g)\|^2)$, where $s = \sqrt{2t}$.

It is worth noting that as seen in above, it is usually quite difficult (or, almost impossible) to observe the change of the corresponding GNS triples explicitly when we take an operation to positive definite (or conditionally negative definite) functions. A significant merit of considering positive definite functions (not unitary representations themselves) is that one can study properties on the whole unitary representations on a group, by means of analysis and certain operations on positive definite functions, without dealing with respective representation spaces.

2.4.4 Proof of Delorme–Guichardet theorem

Now we state the Deloreme–Guichardet theorem:

Theorem 2.4.13. (Delorme [Del], Guichardet [Gui]) For a group G, the following are equivalent:

- (i) The group has property (T).
- (*ii*) The group has property (FH).

More precisely, P. Delorme proved condition (i) implies condition (ii), and A. Guichardet proved condition (ii) implies condition (i).

Proof. Firstly, we will show that Guichardet's implication: "property (FH) implies property (T)" $((ii) \Rightarrow (i))$, which is less involved. Here recall that we always assume groups are σ -compact. Take a unitary representation (π, \mathfrak{H}) with $\pi \not\supseteq 1_G$. The point here is the space of π -cocycles $Z^1(G; \pi)$ can be endowed with a Fréchet space structure, and that the subspace of π -coboundaries $B^1(G; \pi)$ is usually not closed. Indeed, for the former part, since G is σ -compact, we can define a separating sequence of seminorms (m_S) , indexed by a countable family of compact subsets $S \subseteq G$ which covers G, by

$$m_S \colon Z^1(G;\pi) \to \mathbb{R}_{\geq 0}; \ c \mapsto \sup_{s \in S} \|c(s)\|.$$

For the latter part, we claim if $B^1(G;\pi)$ is closed in $Z^1(G;\pi)$, then $\pi \not\geq 1_G$ (, namely, π cannot have almost invariant vectors). To show the claim, note that $B^1(G;\pi)$ is the image of the bounded linear map

$$\tau \colon \mathfrak{H} \to Z^1(G;\rho); \ \xi \mapsto (\tau(\xi))(g) = \xi - \pi(g)\xi,$$

which is injective by the assumption that $\pi \not\supseteq 1_G$. Hence if $B^1(G; \pi)$ is closed, by the open mapping theorem $\tau \colon \mathfrak{H} \to B^1(G; \rho)$ must have a bounded inverse. This implies that there exist a constant M > 0 and a compact subset $S \subseteq G$ such that for any $\xi \in \mathfrak{H}$,

$$\|\xi\| \le M \|\tau(\xi)\|_S = M \cdot \sup_{s \in S} \|\xi - \pi(s)\xi\|.$$

Therefore $\pi \not\succeq 1_G$ follows.

To finish the proof of Guichardet's implication, take any unitary representation π of a group G with (FH) which satisfies $\pi \not\supseteq 1_G$. Then by (FH) the π -cohomology group vanishes, and in particular $B^1(G;\pi) \subseteq Z^1(G;\pi)$ must be closed with respect to the topology in the paragraph above. Through the argument above, this forces π to satisfy $\pi \not\succeq 1_G$. Therefore, for a unitary G representation $\pi, \pi \succeq 1_G$ implies $\pi \supseteq 1_G$. This means G has (T), as desired.

Finally, we proceed to the proof of Delorme's implication: "property (T) implies property (FH)" ((i) \Rightarrow (ii)). Take a contraposition, and suppose G does not have (FH). Then there exist a unitary G-representation σ and a σ -cocycle c which is not a coboundary. By Corolary 2.4.9 and Theorem 2.4.3, for any $t \geq 0$, there exists the GNS-triple ($\pi_t, \mathfrak{K}_t, \xi_t$) associated with the positive definite function $G \ni g \mapsto$ $\exp(-t \|c(g)\|^2)$. For explicit construction, $\pi_t = \pi^{\sqrt{2t}}$ as in Remark 2.4.12. Set

$$\pi = \bigoplus_{n=1}^{\infty} \pi_{1/n}$$

First, we cliam $\pi \not\supseteq 1_G$. This follows from the fact that for each $n \ge 1$, $\pi_{1/n} \not\supseteq 1_G$ (otherwise the affine isometric action associated with (σ, c) must have a fixed point and contradiction occurs). Next, we claim $\pi \succeq 1_G$. Indeed, $\exp(-1/n \cdot ||c(s)||^2) \rightarrow 1$ as $n \to \infty$ uniformly on every compact subset $S \subseteq G$, and this implies for each compact subset $S \subseteq G$ and every $\epsilon > 0$, there exists sufficiently large n such that $\xi_{1/n}$ is (S, ϵ) -invariant (note that for any $t \ge 0$, ξ_t is a unit vector, because $\exp(-t \cdot ||c(0)||^2) = 1$). Hence π has almost invariant vectors.

Therefore, the group G does not have (T). This argument ends our proof of Delorme's implication.

Remark 2.4.14. Here are remarks on the Delorme–Guichardet theorem:

(i) There is a relative version of the Delorme–Guichardet theorem, as follows:

Theorem 2.4.15. Let G be a group and $H \leq G$ be a subgroup. Then the following are equivalent:

- (i) The pair $G \ge H$ has relative (T).
- (ii) The pair $G \ge H$ has relative (FH).

The proof goes along a similar way to that of Theorem 2.4.13.

(ii) In the proof of Guichardet's implication (FH) \Rightarrow (T), the following is in fact hold: "for any group G and any unitary representation π with $\pi \not\supseteq 1_G$, the space $B^1(G;\pi)$ is closed in $Z^1(G;\pi)$ with the Fréchet topology (, namely, the topology of uniform convergence on compact subsets) if and only if $\pi \not\succ 1_G$." In the proof we have shown "only if" direction, and "if" direction is much easier. In general, it is much more convenient to consider the closure $\overline{B}^1(G;\pi)$ instead of treating $B^1(G;\pi)$. This yields the important concept of *reduced cohomology*. For more details and discussions, see Chapter 5.

- (*iii*) In the proof of Guichardet's implication, we do not change the representation space (π, \mathfrak{H}) . However note that in the proof of Delorme's implication $(T) \Rightarrow (FH)$, we do change the representation space. One explanation is that we use Schoenberg's theorem; and the other, more explicit explanation is we replace with the full Fock space. Therefore, for a fixed family of unitary representations $(\pi_i, \mathfrak{H}_i)_{i \in I}$ of a group G, condition (a) in below always implies condition (b); but the converse is no longer true:
 - (a) For any $i \in I$, $H^1(G; \pi_i) = 0$.
 - (b) For any $i \in I$, if $\pi_i \succeq 1_G$, then $\pi_i \supseteq 1_G$.

For instance, consider the family $\{\lambda_G\}$, namely, the single representation λ_G (the left regular representation on $L^2(G)$). Then condition (b) is satisfied for any noncompact nonamenable group (for amenability, see Subsection 2.5.2). However, F_2 , which denotes the free group of rank 2, does *not* satisfy condition (a), although it is a infinite nonamenable group and hence satisfies condition (b).

Therefore, it is worth noting Delorme's implication holds true because there is a theory of positive definite and conditionally negative definite functions. This means, firstly, this holds true thanks to a special property for *unitary* representations; and secondly, this holds true because we consider the family of *all* unitary representations on arbitrary Hilbert spaces of a group. Compare with Subsection 3.2.3.

(iv) For Guichardet's implication, we need the assumption of σ-compactness of groups. In fact, there exists a uncountable discrete group with (FH), see for instance a paper of Y. de Cornulier [dCo2]. Note that for the proof of Theorem 2.2.1, σ-compactness is not needed. Hence every (locally compact) (T) groups are compactly generated. Therefore, Guichardet's implication is no longer true for non-σ-compact groups.

2.5 The Haagerup property – as a strong negation

In this section, we briefly treat the *Haagerup property*, which is equivalent to *Gro-mov's a-T-menability*, for groups. These properties can be regarded as strong negation of Kazhdan's property (T). A main reference in this section a book of Cherix–Cowling–Jolissaint–Julg–Valette [CCJJV].

2.5.1 Two definitions

Kazhdan's property (T) (, which is equivalent to property (FH),) represents strong rigidity of groups (property (T) for groups means difficulty in finding almost invariant vectors; and property (FH) represents first cohomology vanishing with unitary coefficients). The Haagerup property [Haa] and Gromov's a-T-menability [Gro3] are repsectively strong negation of these, and represent strong non-rigidity.

Definition 2.5.1. Let G be a group and (π, \mathfrak{H}) be a unitary G-representation.

(i) The representation π is called a C_0 -representation if every matrix coefficient

$$G \ni g \mapsto \langle \pi(g)\xi | \eta \rangle \in \mathbb{C} \quad (\xi, \eta \in \mathfrak{H})$$

vanishes at infinity. Here a continuous function $f: X \to \mathbb{C}$ on a topologocal space is said to vanish at infinity if for any $\epsilon > 0$, the set $\{x \in X : |f(x)| \ge \epsilon\} \subseteq X$ is compact.

(ii) A π -cocycle $c: G \to \mathfrak{H}$ is said to be metrically proper (or shortly, proper) if for any M > 0, the set $\{g \in G : ||c(g)|| \leq M\} \subseteq G$ is compact. If this condition is satisfied, we write as " $||c(g)|| \to \infty \ (g \to \infty)$." We say an affine isometric action of G on a Hilbert space is (metrically) proper if the cocycle part is proper. In general, let a discrete group Γ acts on a metric space (X, d) $\alpha: \Gamma \curvearrowright X$ by isometries. Then we say the action α is (metrically) proper if for some (equivalently, any) $x \in X$, the following holds true: for any $M \ge 0$, the set $\{g \in G : d(g \cdot x, x) \leq M\} \subseteq G$ is finite.

Note that in item (i), we only have to check the condition for every diagonal matrix coefficients.

Example 2.5.2. An example of C_0 -representation is λ_G for any group G. Indeed, if G is compact, this is trivial. If G is non-compact, we use the fact that $C_c(G)$, the space of continuous functions with compact support, is dense in $L^2(G)$. For details, see Proposition C.4.6 in [BHV].

For the proof of the following theorem, which states the Haagerup property is equivalent to Gromov's a-T-menability, one employs Schoenberg's theorem (in a similar way to the proof of Theorem 2.4.13). For details, see Theorem 2.1.1 in [CCJJV].

Theorem 2.5.3. For a group G, the following are equivalent:

- (i) The group G has the Haagerup property. That means, there exists a C_0 -unitary representation π of G which satisfies $\pi \succeq 1_G$.
- (ii) The group G is a-T-menable in the sense of Gromov. That means, there exists a proper cocycle into a unitary representation.

The following proposition indicate precise meaning that these properties are strong negation of property (T).

Proposition 2.5.4. Let $G \ge H$, Λ be groups. Suppose $G \ge H$ has relative (T), and Λ has the Haagerup property. Then for every continuous homomorphism $\phi: G \to \Lambda$, the image of H is relatively compact in Λ .

In particular, every continuous homomorphism from a Kazhdan group into an a-T-menable group has relatively compact image.

Proof. One proof is the following: by (relative version of) Lemma 2.2.5, $\phi(G) \geq \overline{\phi(H)}$ has relative (T). Therefore there exists a C_0 -unitary representation (π, \mathfrak{H}) of Λ , which satisfies $\mathfrak{H}^{\pi(\overline{\phi(H)})} \neq 0$. By the definition of C_0 -representations, this forces $\overline{\phi(H)}$ to be compact.

Here is another proof in terms of different characterization: by (relative version of) Lemma 2.3.8, $\overline{\phi(G)} \ge \overline{\phi(H)}$ has relative (FH). Therefore there exists a proper cocycle c into a unitary representation, which is bounded on $\overline{\phi(H)}$. By the definition of properness, this forces $\overline{\phi(H)}$ to be compact.

Remark 2.5.5. Here are remarks on the Haagerup properties.

- (i) One deep implication of a group having the Haagerup property is the following theorem of Higson-Kasparov [HiKa] and J.-L. Tu [Tu]: "a group with the Haagerup property fulfills the Baum-Connes conjecture [BCH] (with coefficients)." In contrary, Kazhdan's property (T) can be seen as a main obstruction in establishing (surjectivity side of) the Baum-Connes conjecture (with coefficients). In this thesis, we will not go further into this topic.
- (ii) In the view of Proposition 2.5.4, it is natural to ask the following question: "if a group G does not have the Haagerup property, then does it follow G contains noncompact subgroup H such that $G \ge H$ has relative (T)?" This question was answered by de Cornulier [dCo1] negatively. He shown, for instance, $SO_3(\mathbb{Z}[2^{1/3}]) \ltimes \mathbb{Z}[2^{1/3}]^3$ is a counter example. In [dCo3], de Cornulier extended relative (T) and relative (FH) for a pair of a locally compact group and a closed subset.
- (iii) The Haagerup property is stable under subgroups, direct products, and increasing unions of open subgroups. This is not stable under group quotients. Indeed, free groups have the Haagerup property, as we will see in Subsection 2.5.3.
- (*iv*) By theory of induction (see Subsection 2.2.2), the following holds true: if a lattice Γ in a group G has the Haagerup property, then so does G. In particular, every virtually free group has the Haagerup property. Here a group G is said to be *virtually free* if G contains a free group with finite index. In fact, the

following more general fact is known: if a subgroup H of G which is *co-Følner* has the Haagerup property, then so does G. For the definition of co-Følner property and the proof, see Chapter 6 of [CCJJV].

2.5.2 Amenable groups

Amenablity of groups has a significant role in geometric and analytic group theory, and has thousands of formulations. For comprehensive treatment for this subject, we refer readers to a book of A. T. Paterson [Pat] and Appendix G in [BHV]. Before stating definitions of amenability, we briefly explain on means.

Definition 2.5.6. Let X be a set, \mathcal{B} be a σ -algebra of X, and μ be a measure on (X, \mathcal{B}) .

- (i) A mean m on \mathcal{B} is a finitely additive probability measure on \mathcal{B} . Namely, a nonnegative valued function $\mathcal{B} \to \mathbb{R}_{\geq 0}$ that satisfies the following:
 - (1) m(X) = 1.
 - (2) for any pair wise disjoint finitely many sets $A_1, \ldots, A_n \in \mathcal{B}, m(\bigsqcup_{i=1}^n A_i) = \sum_{i=1}^n m(A_i).$

If moreover X is endowed with an action of a group G which leaves \mathcal{B} invariant, then a mean m is said to be (*left*) G-invariant if for any $g \in G$ and $A \in \mathcal{B}$, $m(g \cdot A) = m(A)$ holds.

- (*ii*) Let *E* be a closed subspace of $L^{\infty}(X, \mathcal{B}, \mu)$ which contains $1 = \chi_X$ and is stable under complex conjugation. A *mean M* on *E* is a linear functional $E \mapsto \mathbb{C}$ that satisfies the following:
 - (1) M(1) = 1.
 - (2) M is positive, namely, for any nonnegative valued function $\phi \in E \subseteq L^{\infty}(X, \mathcal{B}, \mu), M(\phi) \geq 0.$

If in addition X is endowed with an action of a group G which leaves \mathcal{B} invariant, consider the left G-action on $L^{\infty}(X, \mathcal{B}, \mu)$ by $\phi(x) \mapsto g \cdot \phi(x) = \phi(g^{-1} \cdot x)$. If moreover E is invariant under this G-action then a mean m is said to be (*left*) G-invariant if for any $g \in G$ and $\phi \in E$, $M(g \cdot \phi) = M(\phi)$ holds.

It can be shown there is the following one-to-one correspondence:

 $\left\{\begin{array}{c} \text{means } m \text{ on } \mathcal{B} \\ \text{which are absolute continuous with respect to } \mu \end{array}\right\} \leftrightarrow \left\{\begin{array}{c} \text{means } M \\ \text{on } L^{\infty}(X, \mathcal{B}, \mu) \end{array}\right\}.$

Here *m* is said to be *absolutely continuous* with respect to μ , if m(A) = 0 whenever $A \in \mathcal{B}$ staisfies $\mu(A) = 0$. In this correspondence, the map from the right hand side to left hand side is easy: $m(A) := M(\chi_A)$. We note that for the construction of the inverse map, one needs much more care. Thanks to this correspondence, henceforth we will not distinguish these two concepts of means. Let $\ell^{\infty}(G)$ denote the ℓ^{∞} space for *G* and for this construction, we temporarily regard *G* as a (possible uncountable) discrete group. Let UCB(G) denote the closed subspace of $\ell^{\infty}(G)$ of all functions ϕ which are uniformly left continuous, equivalently, the mapping $G \ni g \mapsto g \cdot \phi \in \ell^{\infty}$ is continuous. Note that UCB(G) is invariant under the *G*-action on $\ell^{\infty}(G)$.

Here we extract some equivalent formulations of amenability. We note that in this theorem, we do not need σ -compactness assumption.

Theorem 2.5.7. For a group G, the following are all equivalent. If G satisfies either of these conditions, G is said to be amenable.

- (i) (Existence of invariant mean 1) G has a (left) invariant mean on the class of Borel subsets of G.
- (ii) (Existence of invariant mean 2) G has a (left) invariant mean on UCB(G).
- (iii) (Fixed point property on convex compact subsets) any continuous affine action of G on a non-empty convex compact subset of a locally convex topological vector space has a G-fixed point.
- (iv) (Følner's condition) there exists a Følner net. Nemely, for every compavt subset $S \subseteq G$ and every $\epsilon > 0$, there exists a Borel subset D of G with $0 < \mu(D) < \infty$ such that

$$\sup_{s \in D} \frac{\mu(sD \triangle D)}{\mu(D)}.$$

Here μ is a left Haar measure.

(v) (The Hulanicki–Reiter condition) $\lambda_G \succeq 1_G$ holds, where λ_G is the left regular representation on $L^2(G)$.

Amenability is stable under subgroups, group quotients, extensions, and inductive limits. Also amenability is extended to a group from a lattice (in general, from a *co-amenable* subgroup. See [Pat]). As we mentioned in Example 2.1.6, any abelian group is amenable. Therefore, all solvable groups are amenable. Also, any finite generated group with subexponential growth is amenable. Here we say a finitely generated group has subexponential growth if for some (equivalently, any: see Remark 2.3.14) finite generating set $S \subseteq G$, the equality

for any
$$t > 0$$
, $\lim_{n \to \infty} \frac{|\{g \in G : l_S(g) \le n\}|}{e^{tn}} = 0.$

An interesting and famous example of groups with subexponential growth is the *Grigorchuk group* (this group has *intermediate growth*). For details, see the paper of R. I. Grigorchuk [Gri].

Also, we note that the free group F_2 of rank 2 is *not* amenable (because for instance, this group admits a paradoxical decomposition). Since any subgroup of an amenable group is amenable, this implies the following: any group which contains a subgroup isomorphic to F_2 is *not* amenable. An example of nonamenable groups which does not contain F_2 is an (infinite) free Burnside group.

As a corollary of Theorem 2.5.3 and Theorem 2.5.7, we have the following:

Lemma 2.5.8. Any amenable group has the Haagerup property.

2.5.3 Other examples

In this subsection, we show that the free group F_n $(2 \le n \le \infty)$ and the special linear group $SL_2(\mathbb{Z})$ have the Haagerup property.

Definition 2.5.9. Let G be a finitely generated group, and S be a symmetric finite generated set. Then the *Cayley graph* of G associated with S, written as Cay(G; S), is defined as follows:

- The vertex set V is G, as a set.
- The edge set E is the set of all oriented pair e = (v, vs), where $v \in V(=G)$ and $s \in S$ and $vs \in V$ means the group multiplication.

Since S is symmetric, $(v, vs) \in E$ implies $(vs, v) \in E$. Hence we also identify $\operatorname{Cay}(G; S)$ with non-oriented graph. If S does not contain e_G , $\operatorname{Cay}(G; S)$, as a non-oriented graph, is a connected |S|-regular graph without self loops or multiple edges.

We endow $\operatorname{Cay}(G; S)$ with a metric space structure by the shortest path metric d (with setting each edge length= 1). Also, we equip $\operatorname{Cay}(G; S)$ with a left G-action (, namely, $G \ni g$ sends $V \ni v \mapsto gv \in V$). By the definition of the edge set E, this action on $\operatorname{Cay}(G; S)$ is isometric, and (metrically) proper in the sense of (ii) in Definition 2.5.1.

A basic example is the following: let F_2 denote the free group of rank 2, and let $a, b \in F_2$ be free generators. Then, for a generating set $S = \{a^{\pm}, b^{\pm}\}$, the Cayley graph of F_2 $T := (V, E) = \text{Cay}(F_2; S)$ is a 4-regular tree, as a non-oriented graph.

We will show that F_2 has the Haagerup property. The key here is, with the setting in the paragraph above, F_2 acts on a (4-regular) tree T properly by isometries. From this point, we regard the edge set E of T as the set of all *oriented* edges. Set a (real) Hilbert space \mathfrak{H} as the subspace of all vectors ξ in $\ell^2(E, \mathbb{R})$ with the following condiiton:

for any edge $(v, vs) \in E$, $\xi((vs, v)) = -\xi((v, vs))$.

The isometric action of F_2 on T involves a natural action on E, and this induces an orthogonal F_2 -representation on \mathfrak{H} . We write this representation as π , namely, for any $g \in F_2$, $\xi \in \mathfrak{H}$, and any $(v, vs) \in E$,

$$\pi(g)\xi((v,vs)) = \xi((g^{-1}v,g^{-1}vs))$$

Next, observe that since T is a tree, for any distinct $v, w \in V$, there exists a *unique* (shortest) path from v to w. We denote by [v, w] this path. Set $z: V \times V \to \mathfrak{H}$ by the following equality: for any $v, w \in V$ and any $e \in E$,

$$z(v,w)(e) := \begin{cases} 0 & \text{if } v = w \text{ or } e \text{ is not on } [v,w];\\ 1 & \text{if } e \text{ is on } [v,w] \text{ and points from } v \text{ to } w;\\ -1 & \text{if } e \text{ is on } [v,w] \text{ and points from } w \text{ to } v. \end{cases}$$

Note that then this z fulfills the following two conditions:

- (1) For any $g \in F_2$ and $v, w \in V$, $\pi(g)z(v, w) = z(gv, gw)$.
- (2) For any $u, v, w \in V$, z(u, v) + z(v, w) = z(u, w).

Indeed, for item (2), observe the fact that for distinct $u, v, w \in V$, there exists a unique point which lies on [u, v], [v, w], and [u, w].

Finally, fix a vertex $v \in V$, and set

$$c \colon F_2 \to \mathfrak{H}; \ c(g) = z(gv, v).$$

Then from item (1) and (2), it is easy to show that c is a π -cocycle. Moreover, this c is a proper cocycle because the action $F_2 \curvearrowright T$ is proper. More precisely, for any $g \in F_2$,

$$||c(g)|| = d(gv, v)^{1/2} = (l_S(g))^{1/2}$$

Therefore, F_2 is a-T-menable, and equivalently has the Haagerup property.

In fact, in this case, one can prove in more direct way. However, in above argument, we also obtain the following. Here we say a graph is *locally finite* if for every vertex, the valence is finite.

Theorem 2.5.10. Let Γ be a discrete group, and T be a locally finite tree. Suppose Γ acts on T properly by isometries. Then G has the Haagerup property.

Corollary 2.5.11. For any $2 \leq n \leq \infty$, the free group F_n of rank n has the Haagerup property. Here F_{∞} denotes the free group of countably many generators, viewed as a discrete group. The special linear group $SL_2(\mathbb{Z})$ has the Haagerup property.

Note that $F_1 \cong \mathbb{Z}$ is amenable.

Proof. (Corollary 2.5.11) We have seen F_n has the Haagerup property for $1 \le n < \infty$ in the argument above. Also, F_2 contains a subgroup isomorphic to F_{∞} and hence F_{∞} has the Haagerup property.

In the case of $G = SL_2(\mathbb{Z})$, note a famous fact that G contains F_2 with finite index. Indeed, consider the following subgroup of G

$$\left\langle E_{1,2}(2) = \begin{pmatrix} 1 & 2 \\ 0 & 1 \end{pmatrix}, E_{2,1}(2) = \begin{pmatrix} 1 & 0 \\ 2 & 1 \end{pmatrix} \right\rangle.$$

Then it is well-known that this subgroup is a free group of rank 2, and that it has index 12 (for the proof of the former fact, we use so-called a ping-pong lemma. Compare with the proof of Proposition 2.6.3 below). Therefore through item (iv) of Remark 2.5.5, we obtain the conclusion.

Note that in fact $SL_2(\mathbb{Z})$ itself can act on a locally finite tree properly by isometries. This follows from the decomposition of $SL_2(\mathbb{Z})$ as an amalgamated free product of finite groups:

$$\operatorname{SL}_2(\mathbb{Z}) \cong (\mathbb{Z}/4\mathbb{Z}) \star_{\mathbb{Z}/2\mathbb{Z}} (\mathbb{Z}/6\mathbb{Z}),$$

and the Bass–Serre theory. For details on trees, we refer to a book [Ser] by J.-P. Serre.

Other examples of groups with the Haagerup properties are $SO_{m,1}$, $SU_{m,1}$ ($m \ge 1$); Coxeter groups; Baumslag–Solitar groups; Thompson groups (such as Thompson's F); and groups acting property on CAT(0) cubical complexes (by isometries).

Remark 2.5.12. Here are remarks on groups acting on trees, and extensions of a-T-menable groups. For the former topic, see [Ser] for comprehensive treatment.

- (i) For a connected graph V with countable vertices (viewed as a metric space with the shortest path distance), one can endow $\operatorname{Aut}(V)$ with the topology of pointwise convergence and view $\operatorname{Aut}(V)$ as a topological group (if V is locally finite, then $\operatorname{Aut}(V)$ is locally compact with respect to this topology). Thus one can consider continuous action of a topological group on trees.
- (*ii*) In the opposite of a group acting properly on trees, Serre defined property (FA) as follows:

Definition 2.5.13. A group G is said to have *property* (FA) of Serre if any continuous G-action on a tree (by isometries) has either a globally fixed vertex or a globally fixed nonoriented edge.

This definition is equivalent to the condition that for any continuous G-action on a tree, some (equivalently any) G-orbit is bounded. We use the symbol (FA) instead of the original one FA, in order to emphasize this property represents some *rigidity* aspect. Along a similar line to a proof of Theorem 2.5.10, one can show the following theorem of Y. Watatani [Wat].

Theorem 2.5.14. If a group has (FH) (, equivalently, (T)), then it has (FA).

Note that the converse of this theorem is *not* true. Indeed, it is known that every Coxeter group has (FA). However as we stated in above, it has the Haagerup property and hence it cannot have (T) unless it is finite.

Also, it is worth noting that property (FA) does *not* pass to subgroups of finite index. A famous and surprising example is the following: $\operatorname{Out}(F_3)$, the outer automorphism group of the free group of rank 3, is known to have (FA) [Bog], [CuVo2] (this is true for any $n \geq 4$ finite). Nevertheless, J. McCool [McC] found a finite index subgroup of $\operatorname{Out}(F_3)$ which maps onto \mathbb{Z} , hence not having (FA). Note that if a (discrete) group G has (T), then any finite index subgroup has (T) (Theorem 2.2.14) and hence has (FA) by Theorem 2.5.14. This shows, for instance, any finite index subgroup of $\operatorname{SL}_{m\geq 3}(\mathbb{Z})$ has (FA), because $\operatorname{SL}_{m\geq 3}(\mathbb{Z})$ has (T) (we will see this in Subsection 2.6.2).

(*iii*) We note that the Haagerup property is *not* stable under group extensions. Indeed, we see $SL_2(\mathbb{Z})$ and \mathbb{Z}^2 have the Haagerup property. However, as we stated in Subsection 2.1.3, the semidirect product $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2$ has an infinite subgroup \mathbb{Z}^2 with relative (T) (we will show this in Subsection 2.6.1). In the view of item (*iv*) of Remark 2.5.5, it is true that the Haagerup property is stable under group extensions by *amenable* group quotients. A recent notable result of de Cornulier–Stalder–Valette [CSV] states the Haagerup property is stable under *wreath products*.

2.6 Examples of Kazhdan groups

In this section, we see some typical examples of Kazhdan groups. The main goal in this section is to show *totally higher rank lattices*, defined in Chapter 0, has property (T). We shall examine the basic case of this, more precisely, the case of $\mathrm{SL}_{m\geq 3}(\mathbb{Z})$, which can be seen as a lattice in $\mathrm{SL}_{m\geq 3}(\mathbb{R})$.

Recall from Chapter 0 that we use the symbol \mathbb{K} for local fields.

2.6.1 $\operatorname{SL}_2(\mathbb{R}) \ltimes \mathbb{R}^2 \trianglerighteq \mathbb{R}^2$ has relative (T)

The following theorem is the essential in proving property (T) for totally higher rank lattices:

Theorem 2.6.1. For any (local field) \mathbb{K} , the pair $SL_2(\mathbb{K}) \ltimes \mathbb{K}^2 \supseteq \mathbb{K}^2$ has relative (T).

Recall as in Chapter 0, the symbol above means these groups are identified with

$$\left\{ \left(\frac{W \mid v}{0 \mid 1} \right) : W \in \mathrm{SL}_2(\mathbb{K}), v \in \mathbb{K}^2 \right\} \ \succeq \ \left\{ \left(\frac{I_2 \mid v}{0 \mid 1} \right) : v \in \mathbb{K}^2 \right\}.$$

Since $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2$ is a lattice in $SL_2(\mathbb{R}) \ltimes \mathbb{R}^2$, by item (i) of Remark 2.2.15 we obtain the following corollary:

Corollary 2.6.2. The pair $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$ has relative (T).

For the proof of Theorem 2.6.1, we need spectral theory for the abelian (additive) group \mathbb{K}^2 . We briely recall the unitary dual $\hat{\mathbb{K}}^2$ of \mathbb{K}^2 is (non-canonically) identified with \mathbb{K}^2 as follows: fix a unitary character χ of \mathbb{K} other than the unit character. The mapping

$$\mathbb{K}^2 \to \mathbb{K}^2; \quad x \mapsto \chi_x, \quad \chi_x(y) = \chi(\langle x, y \rangle)$$

is a topological group isomorphism. Here for $x = (x_1, x_2)$ and $y = (y_1, y_2)$ in \mathbb{K}^2 , $\langle x, y \rangle = x_1 y_1 + x_2 y_2$. Under this identification, the dual action of a matrix $h \in \mathrm{SL}_2(\mathbb{K})$ on \mathbb{K}^2 corresponds to the inverse transpose of the standard action on \mathbb{K}^2 , namely, $\mathbb{K}^2 \ni x \mapsto {}^t h^{-1} x$. We also recall from Subsection 2.5.2 for the definitions of means.

Next, we prove the following proposition:

Proposition 2.6.3. If m is an $SL_2(\mathbb{K})$ -invariant mean on the class of Borel subsets of \mathbb{K}^2 , then m is the Dirac measure at 0.

Proof. Let $|\cdot|$ be an absolute value which defines the topology of K. Set

$$\Omega := \left\{ \left(\begin{array}{c} x \\ y \end{array} \right) \in \mathbb{K}^2 \setminus \{0\} : |y| \ge |x| \right\}.$$

We claim that $m(\Omega) = 0$. Indeed, choose a sequence $(\lambda_n)_{n\geq 1}$ in \mathbb{K} with $|\lambda_{n+1}| \geq |\lambda_n| + 2$ for all n and $|\lambda_1| \geq 2$, and set $\Omega_n = h_n \Omega$, where

$$h_n = \begin{pmatrix} 1 & \lambda_n \\ 0 & 1 \end{pmatrix} \in \mathrm{SL}_2(\mathbb{K}).$$

Then by $SL_2(\mathbb{K})$ -invariance of m, for any $n \ge 1$ $m(\Omega_n) = m(\Omega)$. Also, since

$$\Omega_n \subseteq \left\{ \begin{pmatrix} x \\ y \end{pmatrix} \in \mathbb{K}^2 \setminus \{0\} : \frac{|x|}{|\lambda_n| + 1} \le |y| \le \frac{|x|}{|\lambda_n| - 1} \right\},\$$

and $0 < |\lambda_n| + 1 < |\lambda_m| - 1$ for any $1 \le n < m$, the sequence $(\Omega_n)_{n \ge 1}$ is pairwise disjoint. Therefore by finite additivity, for any $n \ge 1$, $n \cdot m(\Omega) \le m(\mathbb{K}^2) = 1$. This shows our claim.

2.6. EXAMPLES OF KAZHDAN GROUPS

Also, there exists $h \in SL_2(\mathbb{K})$ such that $h\Omega = \mathbb{K}^2 \setminus (\Omega \cup \{0\})$ Explicitly, for instance,

$$h = \left(\begin{array}{cc} 0 & 1\\ -1 & 0 \end{array}\right).$$

Therefore $m(\{0\}) = 1$, and this proves the proposition.

Proof. (Theorem 2.6.1) Set $H = \mathrm{SL}_2(\mathbb{K})$, $N = \mathbb{K}^2$, and $G = H \ltimes N$. Suppose $G \succeq N$ does not have relative (T). Then there exists a unitary representation (π, \mathfrak{H}) of G such that $\pi \mid_N \not\supseteq 1_N$ and $\pi \succeq 1_G$. In particular, there exists a unitary G-representation (π, \mathfrak{H}) such that the following two conditions hold:

- (a) The space $\mathfrak{H}^{\pi(N)} = 0$.
- (b) The representation $\pi \mid_H$ has almost invariant vectors.

By spectral theorem for the abelian group N, the representation $\pi \mid_N : N \to U(\mathfrak{H})$ yields the corresponding spectral measure

$$E: \mathcal{B}(N) \to \mathbb{B}(\mathfrak{H}),$$

where $\mathcal{B}(\hat{N})$ denotes the σ -algebra of all Borel subsets of \hat{N} . Namely, E satisfies the following: for any $r \in N$ and $\xi \in \mathfrak{H}$,

$$\langle \pi(r)\xi|\xi\rangle = \int_{\hat{N}} \chi(r)d\langle E(\chi)\xi|\xi\rangle.$$

By the uniqueness of the spectral measure and $\pi(h)\pi(r)\pi(h^{-1}) = \pi(hrh^{-1})$ $(h \in H, r \in N)$, we have the following:

for any
$$h \in H$$
 and any $\hat{C} \in \mathcal{B}(\hat{N})$, $\pi(h)E(\hat{C})\pi(h^{-1}) = E(h \cdot \hat{C})$.

Here h acts on \hat{N} by $h\chi(x) = \chi(h^{-1}x)$. From this point, we identify \hat{N} with N by the correspondence explained in this subsection. Then the equality above means:

for any
$$h \in H$$
 and any $C \in \mathcal{B}(N)$, $\pi(h)E(C)\pi(h^{-1}) = E({}^th^{-1}C)$.

Here on the left hand side, the action is given by the matrix multiplication.

Note that H is compactly generated, and fix a concact generating set $S \subseteq H$ for H. By condition (b) for any $n \in \mathbb{N}$, there exists a unit (S, 1/n)-invariant vector ξ_n . Thus we obtain a sequence of unit vectors $(\xi_n)_{n\in\mathbb{N}}$ such that

for any
$$h \in H$$
, $\lim_{n} ||\xi_n - \pi(h)\xi_n|| = 0$

(see Corollary 2.1.14). Now consider the product space $K = \prod_{T \in \mathbb{B}(\mathfrak{H})} D_T$, where for each $T \in \mathbb{B}(\mathfrak{H})$, D_T is the closed disc in \mathbb{C} of radius ||T||. We endow K with the product topology, and then Tychnoff's theorem shows K is compact. Since

 $(\langle T\xi_n|\xi_n\rangle)_{T\in\mathbb{B}(\mathfrak{H})}$ is in K, there exists a subnet $(\xi_j)_{j\in J}$ such that for all $T\in\mathbb{B}(\mathfrak{H})$ the limit

$$\lim_{j\in J} \langle T\xi_j | \xi_j \rangle$$

exists. We define a map $M \colon \mathbb{B}(\mathfrak{H}) \to \mathbb{C}$ by setting M(T) as the limit above. Then M is a positive linear functional, sending I to 1, and $\pi(H)$ -bi invariant, that means

for any
$$h \in H$$
 and $T \in \mathbb{B}(\mathfrak{H})$, $M(\pi(h)T) = M(T\pi(h)) = M(T)$

Indeed, it is straightforward to see M is a positive linear functional and sends I to 1. For $\pi(H)$ -bi invariance, it follows from the following inequalities: for all $j \in J$,

$$\begin{aligned} |\langle \pi(h)T\xi_j|\xi_j\rangle - \langle T\xi_j|\xi_j\rangle| &= |\langle T\xi_j|\pi(h^{-1})\xi_j - \xi_j\rangle| \le ||T|| ||\pi(h^{-1})\xi_j - \xi_j||,\\ |\langle T\pi(h)\xi_j|\xi_j\rangle - \langle T\xi_j|\xi_j\rangle| &= |\langle T(\pi(h)\xi_j - \xi_j)|\xi_j\rangle| \le ||T|| ||\pi(h)\xi_j - \xi_j||. \end{aligned}$$

Finally, we obtain a mean m on $N = \mathbb{K}^2$ by

$$\mathcal{B}(N) \ni C \mapsto m(C) := M(E(C)) \in \mathbb{R}_{\geq 0}$$

(recall that (orthogonal) projections are positive operators). By $\pi(H)$ -invariance of M and the equality $\pi(h)E(C)\pi(h^{-1}) = E({}^{t}h^{-1}C)$, it follows this m is invariant under the inverse transpose multiplication of $H = \text{SL}_2(\mathbb{K})$. This means m is an $\text{SL}_2(\mathbb{K})$ -invariant mean. By Proposition 2.6.3, this m must coincide with the Dirac measure at $0 \in \mathbb{K}^2$. This in particular means $E(\{0\}) \neq 0$. However this contradicts condition (a) because $E(\{0\})$ is a projection onto $\mathfrak{H}^{\pi(N)}$.

Remark 2.6.4. (*i*) In fact, we prove the following:

Theorem 2.6.5. Let \mathbb{K} be any local field. Let $H = \mathrm{SL}_2(\mathbb{K})$, $N = \mathbb{K}^2$, and $G = H \ltimes N$. Then the pair $H \ltimes N \supseteq N$ has strong relative property (T), in the sense of Bader–Furman–Gelander–Monod [BFGM]. That means, for any unitary G-representation (π, \mathfrak{H}) , whenever $\pi \mid_{H \succeq 1_H}, \pi \mid_N \supseteq 1_N$ follows.

This observation is a key to the proof of property $(F_{\mathcal{L}_p})$ for totally higher rank lattice in [BFGM]. For details, we will see in Subsection 3.3.3.

We warn that this strong relative property (T) does *not* pass to lattices. Indeed, tha pair $G = \operatorname{SL}_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2 = N$ does *not* have this property. For the proof of this observation, consider the unitary representation $(\pi, \ell^2(N))$ induced by affine action of G on $N: (W, v_1) \cdot v_2 = Wv_2 + v_1$ ($W \in H = \operatorname{SL}_2(\mathbb{Z})$ and $v_1, v_2 \in N$. Then $\pi \mid_{H} \succeq 1_H$: in fact, $\pi \mid_{H} \supseteq 1_H$ (the Dirac function δ_0 on the origin of N is $\pi(H)$ -invariant). Nevertheless, trivially $\pi \mid_N \not\supseteq 1_N$.

(ii) By Theorem 2.6.1 (and Corollary 2.6.2), the following is easily deduced:

2.6. EXAMPLES OF KAZHDAN GROUPS

Proposition 2.6.6. Let $m \geq 3$. Then for any local field \mathbb{K} , the pair $SL_m(\mathbb{K}) \ltimes \mathbb{K}^m \supseteq \mathbb{K}^m$ has relative (T). The pair $SL_m(\mathbb{Z}) \ltimes \mathbb{Z}^m \supseteq \mathbb{Z}^m$ has relative (T).

Proof. Set $G = \operatorname{SL}_m(\mathbb{K}) \ltimes \mathbb{K}^m$ and $N = \mathbb{K}^m$. Let π be any unitary G-representation, and c be any π -cocycle. Two points here are the following: firstly, by Theorem 2.4.15, the pair $\operatorname{SL}_2(\mathbb{K}) \ltimes \mathbb{K}^2 \supseteq \mathbb{K}^2$ has relative (FH). Secondly, G contains an isomorphic copy of $\operatorname{SL}_2(\mathbb{K}) \ltimes \mathbb{K}^2$ with \mathbb{K}^2 part included in N: for instance,

$$G = \left\{ \begin{pmatrix} W & v \\ 0 & 1 \end{pmatrix} : W \in \mathrm{SL}_m(\mathbb{K}), v \in \mathbb{K}^m \right\}$$
$$\geqslant \left\{ \begin{pmatrix} W' & 0 & v' \\ 0 & I_{m-2} & 0 \\ \hline 0 & 0 & 1 \end{pmatrix} : W' \in \mathrm{SL}_2(\mathbb{K}), v' \in \mathbb{K}^2 \right\}$$
$$\supseteq \left\{ \begin{pmatrix} I_2 & 0 & v' \\ 0 & I_{m-2} & 0 \\ \hline 0 & 0 & 1 \end{pmatrix} : v' \in \mathbb{K}^2 \right\} \cong \mathbb{K}^2 \leqslant N.$$

For any $1 \leq i < j \leq m$, set $N_{i,j}$ as an isomorphic copy of \mathbb{K}^2 in N which consits of all elements in N whose all but *i*-th and *j*-th entries are zero. Here we identify N with the additive group of column vectors of size m. Thanks to the two observations above, we have that c is bounded on each $N_{i,j}$. Note that N is boundedly generated by $(N_{i,j})_{i,j}$, namely, there exists global $M \geq 1$ such that any $r \in N$ can be written as a product of at most M elements in $\bigcup_{i,j} N_{i,j}$ (one can take M above with $M \leq m/2 + 1$). Since by cocycle identity, for any $g_1, g_2 \in G$

$$||c(gh)|| \le ||c(g)|| + ||c(h)||,$$

we conclude that c is bounded on N.

By Corollary 2.3.7, this means the pair $G \leq N$ has relative (FH). Again by Theorem 2.4.15, $G \leq N$ has relative (T). The latter assertion in the proposition follows from the same argument.

2.6.2 $\operatorname{SL}_m(\mathbb{R})$ and $\operatorname{SL}_m(\mathbb{Z})$ have (T) for $m \geq 3$

We represent how to deduce property (T) for $SL_{m\geq 3}(\mathbb{K})$ from Theorem 2.6.1. One way is to appeal to the Howe–Moore ergordicity theorem [HoMo], although in this case it seems we use too deep theorem for the task.

Theorem 2.6.7. (Howe–Moore's theorem) Let G be a group of the form $\prod_{i=1}^{m} \mathbf{G}_i(k_i)$, where k_i are local fields, $\mathbf{G}_i(k_i)$ are k_i -points of Zariski connected simple k_i -algebraic groups. Let π be a unitary G-representation such that for each $G_i = \mathbf{G}_i(k_i) \pi |_{G_i} \not\supseteq$ 1_{G_i} . Then π is a C_0 representation, namely, for any vector ξ and η , the matrix coefficient

$$G \ni g \mapsto \langle \pi(g)\xi | \eta \rangle \in \mathbb{C}$$

is a C_0 -function (recall item (i) of Definition 2.5.1).

In particular, every Zariski connected simple algebraic group G over a local field enjoys the Howe–Moore property in the sense of [CCLTV]. Namely, any unitary G-representation with $\pi \not\supseteq 1_G$ is C_0 .

The proof of this theorem is involved. For instance, see the original paper [HoMo], or Chapter 2 of a book of R. J. Zimmer [Zim]. For studies on the Howe–Moore property, we refer to [CCLTV]. The following corollary is surprising, and of importance.

Corollary 2.6.8. Let G be a Zariski connected simple algebraic group G over a local field. Then for any unitary G-representation (π, \mathfrak{H}) , there is an equality:

for any non-compact subgroup $H \leq G$, $\mathfrak{H}^{\pi(H)} = \mathfrak{H}^{\pi(G)}$.

Proof. The inclusion $\mathfrak{H}^{\pi(H)} \supseteq \mathfrak{H}^{\pi(G)}$ is trivial. For the inverse inclusion, suppose there exists a unit vector $\xi \in \mathfrak{H}^{\pi(H)} \setminus \mathfrak{H}^{\pi(G)}$. Then thanks to the Howe–Moore property, $g \mapsto \langle \pi(g)\xi | \xi \rangle$ is C_0 . This contradicts the observation that the function above takes value 1 on H.

Theorem 2.6.9. For any local field \mathbb{K} and $m \geq 3$, $SL_m(\mathbb{K})$ has property (T).

Compare with the fact that $SL_2(\mathbb{K})$ has the Haagerup property (this follows from the group above contains F_2 as a lattice).

Proof. Set $G = \operatorname{SL}_m(\mathbb{K})$. Take any unitary *G*-representation π with $\pi \succeq 1_G$. Suppose $\pi \not\supseteq 1_G$. Firstly observe *G* contains an isomorphic copy of $H \ltimes N$, where $H = \operatorname{SL}_2(\mathbb{K})$ and $N = \mathbb{K}^2$. Then by relative (T) for $H \ltimes N \trianglerighteq N$ (Theorem 2.6.1), $\pi \mid_N \supseteq 1_N$. (Note that $\pi \succeq 1_G$ implies $\pi \mid_{H \ltimes N} \succeq 1_{H \ltimes N}$.) Secondly, note that *G* and π satisfies the assumption of the Howe–Moore theorem. By Corollary 2.6.8, $\pi \mid_N \supseteq 1_N$ implies $\pi \supseteq 1_G$ (because *N* is noncompact). This is a contradiction. Therefore *G* has (T).

Theorem 2.6.10. For $m \geq 3$, $SL_m(\mathbb{Z})$ has property (T).

Corollary 2.6.11. Let \mathbb{K} be a local field and $m \geq 3$. Then $SL_m(\mathbb{K}) \ltimes \mathbb{K}^m \supseteq \mathbb{K}^m$ has property (T). Also $SL_m(\mathbb{Z}) \ltimes \mathbb{K}^m \supseteq \mathbb{Z}^m$ has (T).

Proof. This follows from Theorem 2.6.9, Propsotion 2.6.6 and Proposition 2.2.7. It is also not difficult to deduce directly from Theorem 2.6.9 in a similar argument to that in Proposition 2.6.6. \Box

2.6.3 General totally higher rank lattices

Firstly, we consider the case of $\operatorname{Sp}_{2m}(\mathbb{K})$. We recall the definition. In this subsection, let A be a commutative ring (recall from Chapter 0 we always assume a ring is associative and has unit). As mentioned in Chapter 0, in this thesis, we use the following alternating matrix:

$$J_m := \left(\begin{array}{cc} 0 & I_m \\ -I_m & 0 \end{array}\right)$$

The symplectic group over A, written as $\text{Sp}_{2m}(A)$, is defined as the multiplicative group of symplectic matrices in matrix ring $M_{2m}(A)$ associated with the alternating matrix J_m , namely,

$$\operatorname{Sp}_{2m}(A) = \{ g \in \operatorname{M}_{2m}(A) : {}^{t}gJ_{m}g = J_{m} \}.$$

For $m \ge m_0 \ge 2$, by $\operatorname{SL}_{m_0}(A) \le \operatorname{Sp}_{2m}(A)$ (or, $\operatorname{SL}_{m_0}(A) \hookrightarrow \operatorname{Sp}_{2m}(A)$) we mean the inclusion is realized in the following way:

$$\left\{ \begin{pmatrix} W & 0 & 0 & 0 \\ 0 & I_{m-m_0} & 0 & 0 \\ 0 & 0 & {}^t W^{-1} & 0 \\ 0 & 0 & 0 & I_{m-m_0} \end{pmatrix} : W \in \mathrm{SL}_{m_0}(A) \right\} \leqslant \mathrm{Sp}_{2m}(A).$$

Also, we denote by $S^{m*}(A^m)$ the additive group of all symmetric matrices in $M_m(A)$. Finally, by $SL_m(A) \ltimes S^{m*}(A^m) \succeq S^{m*}(A^m)$, we identify these groups respectively with

$$\left\{ (W,v) := \left(\begin{array}{c|c} W & v \\ \hline 0 & tW^{-1} \end{array} \right) : W \in \mathrm{SL}_m(A), \ v \in S^{m*}(A^m) \right\}$$
$$\ge \left\{ \left(\begin{array}{c|c} I_m & v \\ \hline 0 & I_m \end{array} \right) : v \in S^{m*}(A^m) \right\}.$$

Thus the action of $SL_m(A)$ on $S^{m*}(A^m)$ is:

$$(W,0)(I_m,v)(W^{-1},0) = (I_m, Wv^t W) \quad (W \in \mathrm{SL}_m(A), v \in S^{m*}(A^m)).$$

Note that these groups are included in $\text{Sp}_{2m}(A)$. The following theorem is a symplectic version of relarive (T) result, which is parallel to Theorem 2.6.1:

Theorem 2.6.12. For a local field \mathbb{K} , the pair $SL_2(\mathbb{K}) \ltimes S^{2*}(\mathbb{K}^2) \trianglerighteq S^{2*}(\mathbb{K}^2)$ has relative (T).

The proof is basically the same as that of Theorem 2.6.1, which is based on the spectral theory for the abelian group $S^{2*}(\mathbb{K}^2)$. However in this case it is more complicated. See Section 1.5 in [BHV]. In fact, this pair has strong relative property (T), see item (*i*) of Remark 2.6.4. **Corollary 2.6.13.** The pair $SL_2(\mathbb{Z}) \ltimes S^{2*}(\mathbb{Z}^2) \trianglerighteq S^{2*}(\mathbb{Z}^2)$ has relative (T).

The following theorem can be deduced from Theorem 2.6.12 and Corollary 2.6.8:

Theorem 2.6.14. Let \mathbb{K} be a local field and $m \geq 2$. Then $\operatorname{Sp}_{2m}(\mathbb{K})$ and $\operatorname{Sp}_{2m}(\mathbb{Z})$ have (T).

We note that the case of SL_3 and Sp_4 are the basis of property (T) for totally higher rank algebraic groups. For more precise meaning, see below.

The following theorem is the main goal in this chapter. Recall from Chapter 0 that we define a *totally higher rank* (algebraic) group as a group of the following form: $G = \prod_{i=1}^{m} \mathbf{G}_{i}(k_{i})$, where k_{i} are local fields, $\mathbf{G}_{i}(k_{i})$ are k_{i} -points of Zariski connected simple k_{i} -algebraic groups (with finite center), and each simple factor $\mathbf{G}_{i}(k_{i})$ has (local) rank ≥ 2 . We define *totally higher rank lattice* as a lattice in a totally higher rank algebraic group. Note that the original paper of Kazhdan [Kaz], the assumption of each local rank ≥ 3 was needed. This result is extended to the case of each local rank ≥ 2 by Delaroche–Kirillov, Vaserstein, and Wang.

Theorem 2.6.15. (based on Kazhdan [Kaz]) Any totally higher rank algebraic group and totally higher rank lattice have (T).

Proof. We only treat the case of a connected simple algebraic group **G** over a local field \mathbb{K} of rank ≥ 2 . Then the following is well-known:

Lemma 2.6.16. For the group **G** in the setting above, there exists a almost \mathbb{K} -simple algebraic group **H** over \mathbb{K} whose simply connected covering is isomorphic over \mathbb{K} to either SL_3 or Sp_4

This lemma follows from the classification of root systems. For details, see Chapter I (1.6.2) in [Mar2].

Let **H** be as in Lemma 2.6.16. Let **H** be its simply connected covering, and $\phi \colon \tilde{\mathbf{H}} \to \mathbf{H}$ be the canonical homomorphism. Set $H = \mathbf{H}(\mathbb{K})$. Since $\tilde{\mathbf{H}}(\mathbb{K})$ has (T), so does $\phi(\tilde{\mathbf{H}}(\mathbb{K}))$. Therefore *H* has (T), because $\phi(\tilde{\mathbf{H}}(\mathbb{K}))$ is a (normal) subgroup of finite index. Corollary 2.6.8 ends our proof.

Corollary 2.6.17. Every totally higher rank lattice is finitely generated, has finite abelianization, and has property (FA) (recall Definition 2.5.13).

Proof. Follows from Theorem 2.6.15, Theorem 2.2.1, Corollary 2.2.6, and Theorem 2.5.14 $\hfill \Box$

2.6.4 Other examples

As we stated in Subsection 2.5.3, $SO_{m,1}$ and $SU_{m,1}$ have the Haagerup property. For rank 1 groups, B. Kostant shown the following result:

Theorem 2.6.18. (Kostant [Kos]) For $m \ge 2$, $\operatorname{Sp}_{m,1}$ has (T).

Kostant also shown that $F_{4(-20)}$ has (T). For the proof of these results, see Section 3.3 in [BHV] for geometric proof.

As some discrete generalization of rank 1 groups, there is a well-known conception of *hyperbolic groups*, in the sense of Gromov [Gro2].

Definition 2.6.19. Let $\delta \ge 0$.

(i) A geodesic metric space (X, d) is said to be δ -hyperbolic if for any geodesic triangle $\Delta \alpha \beta \gamma$, the following three conditions are satisfied:

$$\alpha \subseteq N_{\delta}(\beta) \cup N_{\delta}(\gamma);$$

$$\beta \subseteq N_{\delta}(\gamma) \cup N_{\delta}(\alpha);$$

$$\gamma \subseteq N_{\delta}(\alpha) \cup N_{\delta}(\beta).$$

Here for $S \subseteq X$, $N_{\delta}(S)$ is the δ -neighborhood of S, namely, the set of all points $x \in X$ such that $\inf_{s \in S} d(x, s) \leq \delta$ holds. The space X is said to be *hyperbolic* if there exists $\delta \geq 0$ such that X is δ -hyperbolic.

(*ii*) A finitely generated group G is said to be (*word*) hyperbolic if there exists a finite generating set S and $\delta \geq 0$ such that the Cayley graph Cay(G; S) (recall definition 2.5.9) is δ -hyperbolic.

Let G be a finitely generated group. It is known that whether Cay(G; S) is a hyperbolic metric space does *not* depend on the choice of finite generating sets. Therefore, we can consider hyperbolicity of finitely generated groups as a property of groups. One typical example of hyperbolic groups is a finitely generated free group, because a tree is 0-hyperbolic.

Therefore it might seem to be reasonable to ask whether all hyperbolic groups have the Haagerup property. However it is *far from* being true: for instance, any cocompact lattice in $\text{Sp}_{n,1}$ is a hyperbolic group, and by Theorem 2.6.18 this has property (T). In fact, there are plenty of (in some sense, "generic") examples of (infinite) hyperbolic groups with (T). The main tool for producing such examples is initiated by Gromov [Gro4], and is to consider so-called *random groups*. Here we briefly explain one model of random groups, so-called *density model*.

Definition 2.6.20. (*Density model*) Let F_n be the free group with $n \ge 2$ free generators a_1, \ldots, a_n . For any integet l, let $S_l \subseteq F_n$ be the set of reduced words in those generators (, namely, words with each letter being in $\{a_1^{\pm}, \ldots, a_n^{\pm}\}$ without cancellation occurring) of length l.

Let $0 \le d \le 1$.

(i) A random set of relators at density d at length l is a $(2n-1)^{dl}$ -tuple of elements of S_l , randomly picked among all elements of S_l uniformly and independently.

- (ii) A random group at density d at length l is the group G presented by $\langle a_1, \ldots, a_n | R \rangle$, where R is a random set of relators at density d at length l. Namely, G is a group quotient of F_n by the normal closure of R in F_n .
- (iii) For a property for groups, the property is said to occur with overwhelming probability at density d if the probability of G having the property tends to 1 as $l \to \infty$.

On random groups, we refer to papers of Gromov [Gro4], [Gro6]; and surveys of Y. Ollivier [Oll] and L. Silberman [Sil].

We collect results on property (T) and the Haagerup property for random groups, which are respectively due to Gromov, and Olliver–Wise.

Theorem 2.6.21. (*Gromov* [Gro4]; *Olliver–Wise*[OlWi]) Let $0 \le d \le 1$.

(i) If d > 1/2, then with overwhelming probability a random group at density d is either trivial or isomorphic to $\mathbb{Z}/2\mathbb{Z}$. In particular, is finite.

If d < 1/2, then with overwhelming probability a random group at density d is (infinite) torsion free hyperbolic group.

- (ii) If d > 1/3, then with overwhelming probability a random group at density d has (T).
- (iii) If d < 1/6, then with overwhelming probability a random group at density d has the Haagerup property.

This theorem in particular implies there exists uncountably many (non-isomorphic) Kazhdan groups. We mention that it is also known if d < 1/5, then with overwhelming probability a random group at density d does not have (T). Little is known on what happens at critical points, or at intermediate phase.

We note that A. Zuk [Zuk] found a powerful local criterion for (finitely generated) groups having (T), and that he shown in another model of random groups (so-called *triangular model*) with overwhelming probability either a random group at density d is free (d < 1/3); or a random group at density d has (T) (d > 1/3).

Another example of Kazhdan groups are some class of Kac–Moody groups. We refer to papers of Dymara–Januszkiewicz [DyJa]; Ballmann–Swiatkowski [BaSw2]; and Cartwright–Mlotkowski–Steger [CMS]. We note that the last two papers above examine groups acting on \tilde{A}_2 buildings. Also we refer to Chapter 5 of [BHV].

2.7 Expander graphs – as an application

In this section, we briefly explain expander graphs, which can be seen as one important application of property (T). In this section, we regard edges of a graph as a nonoriented object, unless otherwise stated, and assume all graphs are connected.

Definition 2.7.1. Let X is a finite graph, with the vertex set V and the edge set E. Suppose X is d-regular for some $d \in \mathbb{N}$, and X does not have multiple edges.

(a) The expansion constant (or, the isoperimetric constant) of X is the following real number:

$$h(X) := \inf \left\{ \frac{|\partial S|}{|S|} : \emptyset \neq S \subseteq V, \ |S| \le \frac{1}{2} |V| \right\}.$$

Here for $S \subseteq V$, $\partial S \subseteq E$ is the set of all edges which connect a vertex in S and a vertex in $V \setminus S$.

(b) The spectral gap of X, written as $\lambda_1(X)$, is the smallest positive eigenvalue of the Laplacian operator Δ_X associated with X. Here the Δ_X is defined as the matrix $\in M_{|V|}(\mathbb{C})$ of the form

$$\Delta_X = d - A,$$

where A is the adjacency matrix, namely for $v, w \in V$, A(v, w) is 1 if $(v, w) \in E$ and 0 otherwise.

Remark 2.7.2. The original definition of Δ_X is

$$\Delta_X = \frac{1}{2} \partial^* \partial \in \mathbb{B}(\ell^2(V)).$$

Here $\partial: \ell^2(V) \to \ell^2(E_{\pm})$ means the differential operator,

$$\partial f((v, w)) := f(w) - f(v) \ (f \in \ell^2(V)).$$

Here the symbol E_{\pm} means the set of edges with orientation.

Since X is connected, the space of constant function in $\ell^2(V)$ coincides with the eigenspace of Δ_X associated with the eigenvalue 0. Therefore, for $\lambda > 0$, $\lambda_1(X) \ge \lambda$ is equivalent to the following condition (recall Δ_X is a positive operator):

for any
$$\xi \in \ell^2(V)$$
 with $\xi \perp 1$, $\lambda \|\xi\|^2 \leq \langle \Delta_X \xi |\xi \rangle$

(this argument is called a Rayreigh quotient argument).

Definition 2.7.3. A sequence $(X_n)_{n\geq 1}$ of finite graphs X_n , with vertex sets V_n and edges sets E_n without multiple edges is called a *family of expanders* if the following conditions are satisfied:

- (i) There exists $d \in \mathbb{N}$ such that for all n, X_n is d-regular.
- (*ii*) The number $|V_n|$ increases to infinity as $n \to \infty$.
- (*iii*) Either of the following two equivalent conditions holds:

- (a) (expander condition) there exists $\epsilon > 0$ such that for any $n \in \mathbb{N}$, $h(X_n) \ge \epsilon$;
- (b) (spectral expander condition) there exists $\lambda > 0$ such that for any $n \in \mathbb{N}$, $\lambda_1(X_n) \ge \lambda$.

Note that the equivalence between item (a) and item (b) in Definition 2.7.3 follows from the following inequality:

for any *d*-regular graph,
$$\frac{\lambda_1(X)}{2} \le h(X) \le \sqrt{2d\lambda_1(X)}$$
.

For the proof of the inequality above and details of expanders, we refer to a survey of Hoory–Linial–Wigderson [HLW]. From the expander condition, a family of expanders has the following property, which is of extreme importance in network theory: the diameter of $X_{n\geq 1}$ is of polylog order with respect to $|V_{n\geq 1}|$.

With the aid of random graphs, it was known a family of expanders exists, but this method is not constructive. The first explicit construction is due to Margulis [Mar1], and he employs property (T). Before seeing this, we need some definitions.

Definition 2.7.4. Let G be a discrete group.

- (i) The group G is said to be residually finite if for any $g \in G \setminus \{e_G\}$, there exists a finite group quotient $p: G \to H$ such that $p(g) \neq e_H$. Equivalently, if there exists a sequence of normal subgroups $G_n \leq G$ of finite index such that $\bigcap_n G_n = \{e_G\}$.
- (*ii*) The group G is called a *linear group* if it is finitely generated and if it has a faithful linear representation $G \to \operatorname{GL}_m(K)$ for some $m \ge 2$ and K a field.

In other words, by *linear groups* we mean a finitely generated (*not necessarily* closed or discrete) subgroup in a group of the form $GL_m(K)$.

For instance, it is easy to see $SL_{m\geq 2}(\mathbb{Z})$ is residually finite. The following theorem of A. I. Malcev is well-known:

Theorem 2.7.5. (Malcev) Every linear group is residually finite.

For the proof, see for instance Theorem 6.4.12 in a book of Brown–Ozawa [BrOz]. Recall the definition of the Kazhdan constant $\mathcal{K}(G; S)$ in Definition 2.1.10 and the Cayley graph $\operatorname{Cay}(G; S)$ inDefinition 2.5.9.

Theorem 2.7.6. Suppose G is a discrete group with (T) and is residually finite. Then one can construct a family of expanders from finite group quotients of G. More precisely, the following holds true:

Let S be a symmetric finite generating set without containing e_G , and $\mathcal{K} = \mathcal{K}(G; S)$ be the Kazhdan constant. Let $(G_n)_{n\geq 1}$ be a sequence of finite index normal subgroups of G with $|G/G_n|$ increasing to infinity as $n \to \infty$ and with $p_n: G \to G/G_n$

being injective on S. Consider the sequence of Cayley graphs $X_n = \text{Cay}(p_n(G); p_n(S))$. Then $(X_n)_{n\geq 1}$ is a family of expanders, with satisfying

$$\lambda_1(X_n) \ge \mathcal{K}^2.$$

Note that in fact what is needed here is relative (T) instead of (T). Margulis' explicit construction uses relative (T) for the pair $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$. However in that case we deal with graphs with multiple edges, and we need some modification (because the differential operator cannot be defined in this case).

Proof. Existence of such $(G_n)_n$ is from residual finiteness. Fix n and set $H = G/G_n$ (a finite group), set $X = X_n$, and identify $p_n(S)$ with S. Consider the right regular representation $\pi: H \to \ell^2(H)$. By assumption, we can choose subset $S_0 \subseteq S$ such that $S = S_0 \sqcup S_0^{-1}$. Now consider the positive operator D on $\ell^2(V)$ defined by

$$D = \sum_{s \in S_0} (1 - \pi(s))^* (1 - \pi(s)).$$

Then D coincides with Δ_X . Indeed

$$D = 2|S_0| - \sum_{s \in S_0} (\pi(s) + \pi(s^{-1})) = |S| - \sum_{s \in S} \pi(s),$$

and this equals Δ_X by definition.

Note that $\mathcal{K}(H; S_0) = \mathcal{K}(H; S)$ and $\mathcal{K}(H; S) \geq \mathcal{K}(G; S) = \mathcal{K}$ hold (these follow from the definition of Kazhdan constant). Therefore we have the following estimate: for any $\xi \in (\ell^2(V)_{\pi(G)})^{\perp}$ (note that this orthogonal complement is identical to $\{\xi \in \ell^2(V) : \xi \perp 1\}$),

$$\|\xi\|^{2} \leq (\mathcal{K}(H; S_{0})^{-1} \sup_{s \in S_{0}} \|\xi - \pi(s)\xi\|)^{2}$$

$$\leq \mathcal{K}^{-2} \sup_{s \in S_{0}} \|(1 - \pi(s))\xi\|^{2}$$

$$\leq \mathcal{K}^{-2} \sum_{s \in S_{0}} \|(1 - \pi(s))\xi\|^{2} = \mathcal{K}^{-2} \langle D\xi|\xi\rangle.$$

By Remark 2.7.2, we obtain the conclusion.

Therefore, estimation of Kazhdan constants has significant meaning in particular for theory of expander graphs. Note that our proof of property (T) for $SL_3(\mathbb{Z})$ does *not* give such estimation. We shall see some estimation in Subsection 4.2.3.

Finally, we note an important role of expander graphs on coarse geometry as below. The concept of uniform embeddability into a Hilbert space is paid attention in a paper of Gromov [Gro5].

Definition 2.7.7. Let (X, d_X) and (Y, d_Y) be a metric space, and G be a finitely generated group.

- (i) Let $f: X \to Y$ be a (not necessarily continuous) map. We say f is a uniformly embedding into Y, if there exist two monotone increasing functions $\rho_{\pm} \colon \mathbb{R}_{\geq 0} \to \mathbb{R}_{>0}$ such that the following two conditions hold:
 - (a) For any $x_1, x_2 \in X$, $\rho_-(d_X(x_1, x_2)) \le d_Y(f(x_1), f(x_2)) \le \rho_+(d_X(x_1, x_2))$.
 - (b) $\lim_{t\to\infty} \rho_-(t) = \infty$.

We say X is uniformly embeddable into Y if there exists a uniform embedding $f: X \to Y$.

- (ii) We say G is uniformly embeddable into Y if there exists a uniform embedding $f: \operatorname{Cay}(G; S) \to Y$ for some (equivalently any) symmetric finite generating set S of G.
- (*iii*) We say X is with bounded geometry if for any r > 0 there exist N > 0 such that for any $x \in X$, $|B(x,r)| \leq N$. Here B(x,r) is the closed ball with center x and radius r.

We refer to a book of J. Roe [Roe] for comprehensive treatment on coarse geometry. In coarse geometry, we consider a finitely generated group G as a metric space Cay(G; S). Usually the symbol |G| is used for an associated metric space with G(since for any pair of symmetric finite generating sets, the word lengths associated to those are bi-Lipschitz equivalent, |G| is uniquely determined as a coarse equivalence class). However we will not use this symbol in this thesis, because we use $|\cdot|$ for the number of finite sets. Note that G, viewed as a metric space, is with bounded geometry.

The following celebrated theorem of G. Yu [Yu1] and Kasparov–Yu [KaYu] suggests importance of uniform embeddability into Hilbert spaces and into uniformly convex Banach spaces. Although we will not go into details around the statement, we state the theorem. For the definition of uniform convexity, see Subsection 3.1.1.

Theorem 2.7.8. (Yu [Yu1]; Kasparov–Yu [KaYu]) Suppose X is a metric space with bounded geometry. If X is uniformly embeddable into a Hilbert space, then X satisfies the coarse Baum–Connes conjecture. In particular, if a finitely generated group G is uniformly embeddable into a Hilbert space, then G, as a group, satisfies the injectivity side of the Baum–Connes conjecture and satisfies the Novikov conjecture.

In above, the same statements hold true if a Hilbert space is replaced with any uniform convex Banach space.

For a family of expanders $(X_n)_n$, we regard it as a metric space as follows: for any $v \in X_n$ and $w \in X_m$, we define the distance d by $d(v, w) := d_n(v, w)$ if n = m; and $d(v, w) := \operatorname{diam}(X_n) + \operatorname{diam}(X_m) + 1$ otherwise. Here d_n is the graph metric on X_n and $\operatorname{diam}(X)$ is the diameter of the graph X (, namely, the maximum of distances of two vertices).

2.7. EXPANDER GRAPHS – AS AN APPLICATION

Lemma 2.7.9. Any family of expander, viewed as a single metric space in the manner above, is not uniformly embeddable into a Hilbert space.

Proof. Take any family of expanders $(X_n)_n = ((V_n, E_n)_n)$. Let d be the regularity of each X_n . Suppose there exists a uniform embedding f into a Hilbert space ℓ^2 . Restrict f on each X_n , and by translation of the origin respectively we get a sequence of uniform embeddings $f_n: X_n \to \ell^2$ such that the following holds:

- (a) For any n, $\sum_{v \in V_n} f_n(v) = 0$.
- (b) There exists a constant C (independent of n) such that $||f_n(v) f_n(w)|| \ge C ||v w||$ whenever $(v, w) \in E_n$.
- (c) For any r > 0 there exists a constant M_r (independent of n) such that for any n, at most M_r points of V_n are mapped by f_n into any ball of radius r in ℓ^2 .

Indeed, for item (b), set $C = \rho_+(1)$; and for item (c), set $M_r = \rho_-^{-1}(r)$, in the definition of a uniform embedding. We use the definition of spectral expanders, and take $\lambda > 0$ such that for all $n \lambda_1(X_n) \ge \lambda$. Then in the view of Remark 2.7.2, for n the following holds:

for any
$$\xi \in \ell^2(V_n)$$
 with $\xi \perp 1$, $\lambda \sum_{v \in V_n} |\xi(v)|^2 \le \sum_{(v,w) \in E_{n,\pm}} |\xi(v) - \xi(w)|^2$.

Here $E_{n,\pm}$ means the set of all oriented edges. By summing the inequality above up associated with the various components of the vector-valued function f_n , we obtain the following:

$$\lambda \sum_{v \in V_n} \|f_n(v)\|^2 \le \sum_{(v,w) \in E_{n,\pm}} \|f_n(v) - f_n(w)\|^2$$

Here we use item (a). By using item (b) and noticing $|E_{n,\pm}| = d|V_n|$, we have

$$\sum_{v \in V_n} \|f_n(v)\|^2 \le \lambda^{-1} dC^2 |V_n|.$$

In particular, for all $v \in V_n$, $||f_n(v)|| \leq C' |V_n|^{1/2}$ holds, where C' is a universal constant, independent of n and v.

As $n \to \infty$, this contradicts item (c) because $|V_n|^{-1/2} \to 0$. Therefore, no uniform embeddings exist.

The following observation implies that a family of expanders represents strong rigidity in coarse setting. This is an analogue to that property (T) represents strong rigidity in group setting. In comparison, we state the following surprising theorem of Guentner–Higson–Weinberger. Note that there is a concept of *property* A of Yu [Yu1] for metric spaces (and for finitely generated groups), and that property A implies uniform embeddability into a Hilbert space (a slogan is the following: *uniform embeddability is coarse version of the Haagerup property; and property* A *is coarse version of amenability.* Compare with Lemma 2.5.8).

Theorem 2.7.10. (Guentner-Higson-Weinberger [GHW]) Every linear group G has property A. In particular, G admits uniformly embedding into a Hilbert space.

Therefore although $SL_3(\mathbb{Z})$ has (T), it satisfies the injectivety part of the Baum– Connes conjecture and thus satisfies the Novikov conjecture. We note that the surjectivity part of the Baum–Connes conjecture for $SL_3(\mathbb{Z})$ is a big open problem, and (T) for this group is one of obstructions in accessing the surjectivity side. We also note that in a recent work of Guentner–Tessera–Yu [GTY], they show all linear groups have *finite geometric complexity*, which is stronger than having property A.

Chapter 3

Property (T_B) and (F_B)

In this chapter, we treat property (T_B) and property (F_B) , which are respectively variants of property (T) and property (FH), introduced by a paper of Bader– Furman–Gelander–Monod [BFGM]. We see in general (F_B) is a stronger property than (T_B) . We will examine the case of $B = \mathcal{L}_p$, the class of all L^p spaces (for fixed $p \in (1, \infty)$), on details and see the proof in [BFGM] of the fact any totally higher rank lattice enjoys $(F_{\mathcal{L}_p})$ for all p. Basic reference in this chapter is the original paper of Bader–Furman–Gelander–Monod [BFGM]. Also, we refer to a book of Baniyamini–Lindenstrauss [BeLi] for preliminaries on Banach spaces.

3.1 Preliminaries on Banach spaces

In this section, we recall basic facts on uniform convexity, uniform smoothness, and superreflexivity of Banach spaces. Also, we see some useful facts in [BFGM] on uniformly bounded representations on a superreflexive Banach space, which are employed in this thesis.

3.1.1 Uniform convexity, uniform smoothness, and superreflexivity

Definition 3.1.1. Let E be a Banach space (recall from Chapter 0, B(E) denotes the closed unit ball at the origin in E).

(i) The space E is said to be uniformly convex (or uc) if for all $0 < \epsilon < 2$, $d_{\parallel \cdot \parallel}(\epsilon) > 0$ holds. Here for $0 < \epsilon < 2$, we define

$$d_{\|\cdot\|}(\epsilon) = \inf\left\{1 - \frac{\|\xi + \eta\|}{2} : \xi, \eta \in B(E), \text{ and } \|\xi - \eta\| \ge \epsilon\right\}.$$

(*ii*) The space E is said to be uniformly smooth (or us) if $\lim_{\tau \to 0} r_{\parallel \cdot \parallel}(\tau)/\tau = 0$ holds. Here for $\tau > 0$, we define

$$r_{\|\cdot\|}(\tau) = \sup\left\{\frac{\|\xi + \eta\| + \|\xi - \eta\|}{2} - 1 : \|\xi\| \le 1, \, \|\eta\| \le \tau\right\}.$$

(*iii*) The space E is said to be *ucus* if E is uc and us.

(iv) The space E is said to be *superreflexive* if it admits a compatible ucus norm.

We call d and r the modulus of convexity and that of smoothness respectively.

We refer to Appendix A of [BeLi] for details on ucus Banach spaces.

Example 3.1.2. (i) Any Hilbert space \mathcal{H} is ucus. Indeed, by the parallelogram law,

$$d_{\|\cdot\|_{\mathcal{H}}}(\epsilon) = 1 - \sqrt{1 - (\epsilon/2)^2}$$

and

$$r_{\|\cdot\|_{\mathcal{H}}}(\tau) = \sqrt{1+\tau^2} - 1.$$

(*ii*) It is a theorem that any L^p space is ucus if $1 . However <math>L^1$ spaces and L^∞ spaces (infinite dimensional) are not superreflexive (the latter fact follows from the fact that superreflexivity implies reflexivity, which is also nontrivial in this definition of superreflexivity).

The following lemma is due to Lindenstrauss, and gives a new characterization of uniform smoothness:

Lemma 3.1.3. Let $(B, \|\cdot\|)$ be a Banach space. Then for any $\tau > 0$,

$$r_{\|\cdot\|}(\tau) = \sup_{0 < \epsilon < 2} \left\{ \frac{\epsilon \tau}{2} - d_{\|\cdot\|_*}(\epsilon) \right\}.$$

In particular, B is us if and only if B^* is uc.

The duality mapping, defined below, plays a key role in this chapter. Recall from Chapter 0 we use the symbol $\langle \cdot, \cdot \rangle$ for the duality $B \times B^* \to \mathbb{C}$. We mention that the usual order is $B^* \times B \to \mathbb{C}$, but that for our purpose the order above is more convenient.

Definition 3.1.4. (Duality mapping) Let B be a us Banach space. For $\xi \in S(B)$, we define $\xi^* \in S(B^*)$ as the unique element in $S(B^*)$ such that $\langle \xi, \xi^* \rangle = 1$ holds. We call this map $S(B) \to S(B^*); \xi \mapsto \xi^*$ the duality mapping.

3.1. PRELIMINARIES ON BANACH SPACES

Note that although we can extend duality mapping to a map $B \to B^*$ by real multiplication, the resulting map is in general *not* conjugate linear. Namely, $(\xi + \eta)^*$ does *not* coincide with $\xi^* + \eta^*$, unless *B* is a Hilbert space. Therefore, we restrict the duality mapping on the unit sphere.

One merit of restricting the duality mapping on the unit sphere is that then it is uniformly continuous. The following proposition is Proposition A.5 in [BeLi], but we note that there seems to be a misprint in the original version. Here we present a corrected one.

Lemma 3.1.5. Let B be us. Suppose $0 < \kappa < 2$. Then for any $\xi, \eta \in S(B)$ with $\|\xi - \eta\| \leq \kappa$, there is an inequality

$$\|\xi^* - \eta^*\|_* \le 2 \cdot \frac{r_{\|\cdot\|}(2\kappa)}{\kappa}.$$

In particular, the duality mapping $S(B) \to S(B^*)$ is uniformly continuous.

We deal with a (uniformly bounded) representation on a (superreflexive) Banach space. Here we see a representation yields a corresponding representation on the dual Banach space.

Definition 3.1.6. Let ρ be a representation of a group G on a Banach space B. Then we define the *contragredient representation* ρ^{\dagger} of G on B^* as follows:

for any $g \in G$, $\phi \in B^*$ and $\xi \in B$, $\langle \xi, \rho^{\dagger}(g)\phi \rangle = \langle \rho(g^{-1})\xi, \phi \rangle$.

Lemma 3.1.7. Let ρ be a representation of a group G on a Banach space B.

- (i) For any $g \in G$, $\|\rho(g^{-1})\| = \|\rho^{\dagger}(g)\|$. Here the norms are operator norms with repsect to respectively $(B, \|\cdot\|_B)$ and $(B^*, \|\cdot\|_{B^*})$.
- (ii) Suppose B is us and ρ is isometric. Then there is an equality

for any $\xi \in S(B)$ and any $g \in G$, $(\rho(g)\xi)^* = \rho^{\dagger}(g)\xi^*$.

Here $\xi \mapsto \xi^*$ denotes the duality mapping on the unit sphere.

Proof. Item (i) is a direct consequence of the Hahn–Banach theorem. Item (ii) holds true because $\|\rho^{\dagger}(g)\xi^*\|_{B^*} = \|\xi^*\|_{B^*} = 1$ and $\langle \rho(g)\xi, \rho^{\dagger}(g)\xi^* \rangle = \langle \xi, \xi^* \rangle = 1$.

Finally, we state the following lemma, which plays a basic role in theory of fixed point properties:

Lemma 3.1.8. Let X be a bounded subset in a uc Banach space B. Then there exists a unique closed ball with the minimum radius which includes X. We define the Chebyshev center of X as the center of this ball.

Proof. Firstly, we show existence. Set $r = \inf_{\eta \in B} \sup_{\xi \in X} ||\xi - \eta||$. Since X is bounded, $r < \infty$. For any t > r, set

$$C_t = \{\eta \in B : \sup_{\xi \in X} \|\xi - \eta\| \le t\} \neq \emptyset.$$

Then C_t is a closed and bounded convex subset. Now we employ the fact that a uc Banach space is reflexive. Therefore, each C_t is weak*-compact, and clearly $(C_t)_{t>r}$ satisfies the finite intersection property. Thus the set

$$C = \bigcap_{t > r} C_t$$

is nonempty, and an element in C attains the minimum r.

Secondly, we show uniqueness, but this directly follows form uniform convexity of B.

In the view of item (i) of Example 3.1.2, Lemma 2.1.13 is a special case of Lemma 3.1.8.

3.1.2 Changing norms and space decomposition

In the next section, we mainly consider an isometric (linear) representation on a superreflexive Banach space. By Definition 3.1.1, this a priori means that this representation can be seen as a uniformly bounded representation on a ucus Banach space, in the following sense.

Definition 3.1.9. A representation ρ of Γ on B is said to be uniformly bounded if $|\rho| := \sup_{q \in \Gamma} \|\rho(g)\|_{\mathbb{B}(B)} < +\infty.$

One observation of Bader–Furman–Gelander–Monod states that in fact one can take a norm in above such that the representation is *isometric*.

Proposition 3.1.10. (Proposition 2.3 [BFGM]) Let ρ be a uniformly bounded representation ρ of a group G on a superreflexive Banach space B. Then there exists a compatible norm on B with respect to which B is ucus and ρ is isometric.

Proof. By the definition of superreflexivity we may assume B is ucus. Put $M = |\rho|$.

Firstly, we show there exists a compatible uc norm on B with respect to which ρ is isometric. Indeed, for an original norm $\|\cdot\|_B$, set a new norm $\|\cdot\|$ by

$$\|\xi\| := \sup_{g \in G} \|\rho(g)\xi\|_B.$$

Then by assumption this is compatible with $\|\cdot\|_B$, and $\rho(G)$ -invariant. We show the uniform convexity. Let $\alpha > 0$ and $\xi, \eta \in B$ with $\|\xi\| \le 1, \|\eta\| \le 1$ and $\|(\xi + \eta)/2\| > 1 - \alpha$. Then there exists $g \in G$ such that

$$\left\|\frac{\rho(g)\xi + \rho(g)\eta}{2}\right\|_{B} > 1 - \alpha \text{ while } \|\rho(g)\xi\|_{B} \le \|\xi\| \le 1 \text{ and } \|\rho(g)\eta\|_{B} \le \|\eta\| \le 1.$$

3.1. PRELIMINARIES ON BANACH SPACES

By the definition of the modulus of convexity of $\|\cdot\|_B$, we have

$$\alpha \ge d_{\|\cdot\|_B}(\|\rho(g)\xi - \rho(g)\eta\|) \ge d_{\|\cdot\|_B}\left(\frac{\|\xi - \eta\|}{M}\right).$$

This exactly shows

for any
$$\epsilon > 0$$
, $d_{\|\cdot\|}(\epsilon) \ge d_{\|\cdot\|_B}(\epsilon/M) > 0$.

This implies that $\|\cdot\|$ is also uc.

Now let N(B) be the set of all compatible norms on B equipped with the metric

$$d(\|\cdot\|_1, \|\cdot\|_2) := \log(\text{the norm ratio}).$$

Here the *norm ratio* between two compatible norms is defined by

$$\max\{\sup_{x\neq 0} \|x\|_1 / \|x\|_2, \sup_{x\neq 0} \|x\|_2 / \|x\|_1\}.$$

Note that N(B) becomes a complete metric space. Let $N(G)^{\rho(G)}$ stands for the closed subspace of norms on B invariant under $\rho(G)$. Define the subset $N_{\rm uc}(V)^{\rho(G)}$ as the set of all uc $\rho(G)$ -invariant norms. Then this is given by the following countable intersection:

$$N_{\rm uc}(V)^{\rho(G)} = \bigcap_{n=1}^{\infty} O_n, \text{ where } O_N := \{ \| \cdot \| \in N(V)^{\rho(G)} : d_{\| \cdot \|}(1/n) > 0 \}.$$

Recall that $d_{\|\cdot\|}$ denotes the modulus of convexity. Observe the sets O_n are open. Also, we claim $N_{\rm uc}(V)^{\rho(G)}$ is dense in $N(B)^{\rho(G)}$. Indeed, let $\|\cdot\|_0$ be an element in $N_{\rm uc}(V)^{\rho(G)}$ (we see in the first paragraph of this proof that this set is non-empty). Then any $\|\cdot\| \in N(V)^{\rho(G)}$ can be viewed as a limit of uniformly convex norms $\|\cdot\| + \epsilon \|\cdot\|_0$ as $\epsilon \to 0$.

By duality between $N_{\rm uc}(B^*)^{\rho^{\dagger}(G)}$ and $N_{\rm us}(B)^{\rho(G)}$ (see Lemma 3.1.3 and Definition 3.1.6), we also show $N_{\rm uc}(V)^{\rho(G)}$ is an intersection of open subsets and is dense. Here $N_{\rm us}(B)^{\rho(G)}$ denotes the set of all uniformly smooth $\rho(G)$ -invariant norms in N(B).

Therefore the set $N_{\rm uc}(B)^{\rho(G)} \cap N_{\rm us}(B)^{\rho(G)}$ is an intersection of open dense subsets in $N(B)^{\rho(G)}$. Since $N(B)^{\rho(G)}$ is a complete metric space, the Baire category theorem applies. Thus $N_{\rm uc}(B)^{\rho(G)} \cap N_{\rm us}(B)^{\rho(G)}$ is also dense in $N(B)^{\rho(G)}$, and in particular is not empty. This ends our proof.

Thanks to this proposition, as long as treating uniformly bounded representation on a superreflexive Banach space, we may assume the representation is isometric and the Banach space is ucus. Finally, we need a construction of natural complement of subspaces, associated with an isometric representation ρ of a group G on a Banach space B. If a representation (π, \mathfrak{H}) is unitary, then the canonical orthogonal complement does the job, and there is a decomposition of \mathfrak{H} as $\pi(G)$ -spaces

$$\mathfrak{H} = \mathfrak{H}^{\pi(G)} \oplus \mathfrak{H}_{\pi(G)}^{\perp}$$

In general case, it is also possible to find a canonical candidate of complement. Since we need the relative version, we define in the following setting:

Definition 3.1.11. ([BFGM]) Let G be a group, $N \leq G$ be a normal subgroup. Let ρ be a uniformly bounded G-representation on a Banach space B. Then we define a closed subspace $B'_{\rho(N)}$ as the annihilator of $(B^*)^{\rho^{\dagger}(N)}$. Here $(B^*)^{\rho^{\dagger}(N)}$ means the subspace of all $\rho^{\dagger}(N)$ -invariant vectors in B^* .

Since N is normal, $B'_{\rho(N)}$ is $\rho(G)$ -invariant space.

In general, it is not necessarily true that $B = B^{\rho(N)} \oplus B'_{\rho(N)}$. However, Bader– Furman–Gelander–Monod show this holds true if B is superreflexive (note that for the $B'_{\rho(N)}$ is independent of the choice of compatible norms on B).

Proposition 3.1.12. (Proposition 2.6 and Proposition 2.10, [BFGM]) Let G be a group, $N \leq G$ be a normal subgroup. Let ρ be a uniformly bounded G-representation on a superreflexive Banach space B. Then there is a decomposition of B as $\rho(G)$ -spaces

$$B = B^{\rho(N)} \oplus B'_{\rho(N)}$$

Moreover, let any $\xi = \xi_0 + \xi_1$ where $\xi \in B$, $\xi_0 \in B^{\rho(N)}$ and $\xi_1 \in B'_{\rho(N)}$. Then for any uc compatible norm $\|\cdot\|'$ on B, there are inequalities

$$\|\xi_0\|' \le \|\xi\|'$$
 and $\|\xi_1\|' \le 2\|\xi\|'$.

In particular, the G-representation ρ restricted on $B'_{\rho(N)}$ is isomorphic to the G-representation ρ' on the quotient Banach space $B/B^{\rho(N)}$ induced by ρ .

Proof. With the aid of Proposition 3.1.10, take a ucus $\rho(G)$ -invariant norm $\|\cdot\|$.

By item (*ii*) of Lemma 3.1.7, note that for any $\xi \in S(B^{\rho(N)})$ and any $\eta \in B'_{\rho(N)}$, $\langle \eta, \xi^* \rangle = 0$ holds. Hence,

$$1 = \langle \xi, \xi^* \rangle = \langle \xi - \eta, \xi^* \rangle \le \|\xi - \eta\|.$$

Therefore $B^{\rho(N)} \cap B'_{\rho(N)} = \{0\}$. We then claim $B^{\rho(N)} \oplus B'_{\rho(N)}$ (, which is closed,) is dense in *B*. Indeed, suppose there exists $\phi \in B^* \setminus \{0\}$ which vanishes on $B^{\rho(N)}$ and on $B'_{\rho(N)}$. We may assume ϕ is a unit vector in B^* . By the latter condition, the Hahn–Banach shows $\phi \in (B^*)^{\rho^{\dagger(N)}}$. With noting that *B* is reflexive and B^* is us, we can consider the duality mapping $S(B^*) \to S(B)$. Then $\phi^* \in S(B)$ in fact lies in

3.2. DEFINITIONS OF (T_B) AND (F_B)

 $S(B^{\rho(N)})$ for the reason above. This contradicts our assumption of ϕ vanishing on $B^{\rho(N)}$. Therefore, we conclude $B = B^{\rho(N)} \oplus B'_{\rho(N)}$.

The inequalities in the proposition follow from our first argument, because at that point only uniform smoothness of the norm on B is needed.

3.2 Definitions of (T_B) and (F_B)

In this section, we define property (T_B) and property (F_B) for groups, for a given Banach space (or a class of Banach spaces). Also, we define relative property (T_B) for a group pair $G \geq N$; and property (F_B) for a group pair $G \geq N$ (we will explain below why normality is needed for the definition of relative (T_B)). Finally, we see (F_B) does not necessarily imply (T_B) , whereas (T_B) always implies (F_B) .

3.2.1 Definitions and examples

Firstly, we define a representation having almost invariant vectors.

Definition 3.2.1. Let ρ be a representation of a group G on a Banach space.

(i) For a subset $S \subseteq G$ and $\kappa > 0$, a vector $\xi \in \mathfrak{H}$ is said to be (S, κ) -invariant if

$$\sup_{s\in S} \|\xi - \rho(s)\xi\| < \kappa \|\xi\|.$$

(ii) We say ρ has almost invariant vectors if for any compact subset $S \subseteq G$ and $\kappa > 0$, there exists an (S, κ) -invariant vector. We write $\rho \succeq 1_G$ if ρ has almost invariant vectors.

Note that the concept of a representation having almost invariant vectors is determined independently of the choices of compatible norms on B. Also note that the symbol $\rho \succeq 1_G$ is artificial, because in general there is no analogue of diagonal matrix coefficients (note that if B is us, then the map

$$G \ni g \mapsto \langle \rho(g)\xi, \xi^* \rangle \in \mathbb{C}$$

works, where $\xi \in S(B)$).

Recall that property (T) for a group is defined by the condition "for any unitary G-representation, $\pi \not\supseteq 1_G \Rightarrow \pi \not\succeq 1_G$." However, in general setting, the straight generalization of above does not give information for the case of $\pi \supseteq 1_G$. More precisely, if we consider unitary representation, then we can restrict our representation on the orthogonal complement of the space of invariant vectors. However even in the case of considering isometric representation on L^p spaces, the canonical complement, defined in Definition 3.1.11 is a subspace of L^p space, and usually *not* realizable as an L^p space on any measure space. Therefore, the following definition is appropriate:

Definition 3.2.2. ([BFGM]) Let B be a Banach space.

(i) A pair $G \geq N$ of groups is said to have relative property (T_B) if for any isometric representation ρ of G on B, the isometric representation ρ' on the quotient Banach space $B/B^{\rho(N)}$, naturally induced by ρ , satisfies

$$\rho' \not\succeq 1_G.$$

By Proposition 3.1.12, if *B* is superreflexive, then the definition above is equivalent to the following: for any isometric representation ρ of *G* on *B*, the restriction of ρ on $B'_{\rho(N)}$ (see Definition 3.1.11) does not have almost invariant vectors.

(*ii*) A group G is said to have property (T_B) if $G \succeq G$ has relative (T_B) .

Note that here normality of the subgroup N is needed, otherwise $B^{\rho(N)}$ may not be $\rho(G)$ -invariant.

We move to the definition of property (F_B) . First, we define the first group cohomology with isometric Banach coefficient.

Definition 3.2.3. Let *B* be a Banach space. Let *G* be a group, and ρ be an isometric *G*-representation on *B*.

(i) A continuous map $c: G \to B$ is called a ρ -cocycle if the following holds:

For any $g, h \in G$, $c(gh) = c(g) + \rho(g)c(h)$.

This equality is called the *cocycle identity*.

(*ii*) A continuous map $c: G \to B$ is called a ρ -coboundary if the following holds:

There exists $\xi \in B$ such that for any $g \in G$, $c(g) = \xi - \pi(g)\xi$.

(*iii*) The space $Z_{c}^{1}(G; \rho, B)$ (or shortly, $Z_{c}^{1}(G; \rho)$) denotes the vector space of all ρ -cocycles. The space $B_{c}^{1}(G; \rho, B)$ (or shortly, $B_{c}^{1}(G; \rho)$) denotes the vector space of all ρ -coboundaries, which is a subspace of $Z_{c}^{1}(G; \rho)$. The quotient vector space

$$H^{1}_{c}(G;\rho,B) := Z^{1}_{c}(G;\rho)/B^{1}_{c}(G;\rho)$$

(or shortly, $H^1_{\rm c}(G;\rho)$) is called the *first cohomology group* with ρ -coefficient.

(iv) A cocycle c is said to be proper if it tends to infinity as $g \to \infty$, namely, if for any real M > 0 the subset $\{g \in G : ||c(g)|| \le M\}$ in G is compact.

3.2. DEFINITIONS OF (T_B) AND (F_B)

If G is discrete, we use the symbols $Z^1(G;\rho), B^1(G;\rho), H^1(g;\rho)$ instead.

In the same way as one in Proposition 2.3.3, we have the following: there is a oneto-one correspondence (as sets) between affine isometric actions α on B and pairs (ρ, c) of isometric G-representation ρ on B and ρ -cocycles c. The correspondence is:

$$\alpha(g) \cdot \xi = \rho(g)\xi + c(g) \quad (g \in G, \ \xi \in B).$$

In the correspondence above, we call ρ the *linear part* of α , and call c the cocycle part (or transition part) of α .

Definition 3.2.4. Let *B* be a Banach space.

- (i) A group pair $G \ge H$ is said to have relative property (\mathbf{F}_B) if for any isometric representation ρ of G and for every ρ -cocycle c, the restriction of c on H is a coboundary (with $\rho \mid_H$ coefficient). Equivalently, if every affine isometric G-action on B has an H-fixed point.
- (*ii*) A group $G \ge H$ is said to have *property* (F_B) if $G \ge G$ has relative (T_B). Equivalently, if for any isometric G-representation ρ on B,

$$H^1_{\mathbf{c}}(G;\rho,B) = 0.$$

By Lemma 3.1.8, there is a useful characterization of (relative) property (F_B) in the case of that B is superreflexive.

Lemma 3.2.5. Let B a superreflexive Banach space. Then for any affine isometries action α of a group G on B, the following are all equivalent:

- (i) The action α has a G-fixed point.
- (*ii*) Any G-orbit is bounded.
- (iii) Some G-orbit is bounded.
- (iv) The cocycle part of α is bounded.

In particular, a group pair $G \ge H$ has relative (F_B) if and only if for any isometric G-representation on B and for any ρ -cocycle c, c is bounded on H.

Note that we define these properties also for a given class of Banach spaces as we mentioned in Chapter 0. Namely, for a class of Banach space C, property (P_C) is defined as having property (P_B) for all $B \in C$. Here are classes of Banach spaces of our main concern:

- \mathcal{H} denotes the class of all Hilbert spaces;
- \mathcal{L}_p denotes the class of all L^p spaces on any measure spaces;

• [H] denotes the class of all Banach spaces which admit compatible norms to those of Hilbert spaces.

Recall from Chapter 0 that we always assume $p \in (1, \infty)$ (and that we always assume groups are σ -compact, unless otherwise stating). Hence by Example 3.1.2, all elements in those classes are superreflexive. We will explain some reasons why we are mainly interested in superreflexive cases in the example below:

- **Example 3.2.6.** (*i*) Property $(T_{\mathcal{H}})$ is identical to property (T), and property $(F_{\mathcal{H}})$ is identical to property (FH). Therefore, in the case of $B = \mathcal{H}$, (relative) property (T_B) is equivalent to (relative) property (F_B) , thanks to the Delorme–Guichardet theorem (Theorem 2.4.13). In some sense, thus one can define relative property (T) for a pair $G \ge H$ where H is not normal.
- (*ii*) Consider the closed subspace $B = L_0^1(G) \subseteq L^1(G)$ of all elements ξ in $L^1(G)$ with $\int_G \xi(x) d\mu(x) = 0$. Then whenever G is noncompact, G fails to have (F_B). To see this, choose an element $\eta \in L^1(G)$ with $\int_G \eta(x) d\mu(x) = 1$ and construct a λ_G -cocycle $c: G \to B$ by

$$c(g) = \eta - \lambda_G(g)\eta,$$

where λ_G is the left regular representation on $L^1(G)$. Note that this *c* ranges into *B*, although $\eta \notin B$. However this *c* is *not* a coboundary in *B*. Indeed, if *c* is coboundary in *B*, namely, there exists $\zeta \in B$ such that $c(g) = \zeta - \lambda_G(g)\zeta$ holds, then this the vector $\eta - \zeta$ is $\lambda_G(G)$ -invariant, and hence it equals 0 because *G* is noncompact. This is a contradiction.

This example shows Lemma 3.2.5 is *false* for general Banach space *B*. Note that $L_0^p(G)$ is no longer closed in $L^p(G)$ for $p \in (1, \infty)$, where $L_0^p(G) := \{\xi \in L^p(G) : \int_G \xi(x) d\mu(x) = 1\}$.

(*iii*) Let G be an infinite discrete group. Then G does not have property $(F_{\ell^{\infty}(G)})$. Indeed, let d be an unbounded left-invariant metric on G (for instance, a word length with respect to a (possibly infinite) symmetric generated set). Define a cocycle $c: G \to \ell^{\infty}(G)$ by

$$c(g)(x) = d(g, x) - d(e_G, x)$$

where ρ_G is the right regular representation on $\ell^{\infty}(G)$. It is straightforward to check this is indeed a cocycle. This is not a coboundary, because c is unbounded (c(g)(e) = d(g, e)).

(*iv*) Even strictly convex Banach space cases, there exists an example similar to above two. Haagerup–Przybyszewska [HaPr] show any group G admits a *proper* cocycle on the strictly convex Banach space $\bigoplus_{n\geq 1} L^{2n}(G)$. Here the direct sum is taken in ℓ^2 -sense.

3.2. DEFINITIONS OF (T_B) AND (F_B)

Finally, we define the relative Kazhdan constant for property (T_B) .

Definition 3.2.7. ([Mim1]) Let *B* be a superreflexive Banach space, $G \geq N$, and $S \subseteq G$ be a compact subset. Let ρ be an isometric *G*-representation on *B*. We define the *relative Kazhdan constant for property* (T_B) for $(G, N; S, \rho)$ by the following equality:

$$\mathcal{K}(G,N;S,\rho,B) = \inf_{\xi \in S(B_1)} \sup_{s \in S} \|\xi - \rho(s)\xi\|.$$

Here $B_1 = B'_{\rho(N)}$ as in Definition 3.1.11. We also write $\mathcal{K}(G, N; S, \rho, B)$ as $\mathcal{K}(G, N; S, \rho)$ for short.

We note that the relative Kazhdan constant for (T_B) can be defined even for the case of B is not superreflexive (we only need to replace B_1 with $B/B^{\rho(N)}$). However in the case of that B is us, these two definitions differ up to at most double multiple.

By definition, if $G \geq N$ has relative (T_B) and G is compactly generated, then for any ρ and any compact generating set $S \subseteq G$, $\mathcal{K}(G, N; S, \rho)$ is strictly positive.

Note that for the proof of Lemma 2.1.7, we need direct sum argument. Therefore, in general case, even if $G \geq N$ has relative (T_B) and a compact subset $S \subseteq G$ generates G, this does *not* necessarily imply $\inf_{\rho} \mathcal{K}_B(G, N; S, \rho) > 0$, where ρ moves among all isometric G-representation on B. For instance, the class $[\mathcal{H}]$ is *not* stable under (at least L^2) direct sum. This is because there exists a sequence of elements $(B_n)_n$ in $[\mathcal{H}]$ with the norm ratio of between B_n and a Hilbert space tends to infinity.

However, if a class \mathcal{C} of Banach space admits direct-sum operation inside \mathcal{C} , then in the setting above, $\inf_{\rho} \mathcal{K}_{\mathcal{C}}(G, N; S, \rho) > 0$ holds. Main examples are \mathcal{L}_p , including $\mathcal{H} = \mathcal{L}_2$, with L^p -direct sum operation. In the case of $B = [\mathcal{H}]$, if we instead consider a subclass of $[\mathcal{H}]$ with uniformly bounded norm ratio (to Hilbert space norms), then L^2 -sum works.

The following lemma is a generalization of Lemma 2.1.16.

Lemma 3.2.8. ([Mim1]) Suppose B is us, G is a compactly generated group and S is a compact generating set of G. Let $N \leq G$. Let ρ be any isometric representation of G on B, ξ be any vector in B and set $\delta_{\xi} := \sup_{s \in S} ||\xi - \rho(s)\xi||$. If a pair $G \geq N$ has relative (T_B), then there exists a $\rho(N)$ -invariant vector $\xi_0 \in B$ with

$$\|\xi - \xi_0\| \le 2\mathcal{K}^{-1}\delta_{\xi}.$$

Here \mathcal{K} stands for the relative Kazhdan constant $\mathcal{K}(G, N; S, \rho)$ for (T_B) . In particular, there is an inequality

for any
$$l \in N$$
, $\|\xi - \rho(l)\xi\| \leq 4\mathcal{K}^{-1}\delta_{\xi}$

Proof. Decompose B as $B = B_0 \oplus B_1$, where $B_0 = B^{\rho(N)}$ and $B_1 = B'_{\rho(N)}$. Take the associated decomposition of $\xi \ \xi = \xi_0 + \xi_1 \ (\xi_0 \in B_0, \xi_1 \in B_1)$. Then

$$\xi - \rho(s)\xi = (\xi_0 - \rho(s)\xi_0) + (\xi_1 - \rho(s)\xi_1)$$

is the decomposition of $\xi - \rho(s)\xi$. By Proposition 3.1.12 and the definition of relative Kazhdan constant for (T_B) , there are the following inequalities:

$$2\delta_{\xi} = 2\sup_{s \in S} \|\xi - \rho(s)\xi\|$$
$$\geq \sup_{s \in S} \|\xi_1 - \rho(s)\xi_1\| \ge \mathcal{K}\|\xi_1\|$$

This shows $\|\xi_1\| \leq 2\mathcal{K}^{-1}\delta_{\xi}$, as desired.

For the latter part, observe for any $l \in N$,

$$\xi - \rho(l)\xi = (\xi_0 - \rho(l)\xi_0) + (\xi_1 - \rho(l)\xi_1) = \xi_1 - \rho(l)\xi_1.$$

Therefore, the norm is at most $2\|\xi_1\| \leq 4\mathcal{K}^{-1}\delta_{\xi}$.

3.2.2 Property (F_B) implies (T_B)

Our proof of Guichardet's direction "(FH) \Rightarrow (T)" does not use any special property for Hilbert spaces. Thus one can extend it to a general setting, and obtain the following theorem of Bader–Furman–Gelander–Monod:

Theorem 3.2.9. ([BFGM]) For any Banach space B, property (F_B) implies property (T_B).

Note that here even superreflexivity assumption for B is unnecessary.

3.2.3 Property (T_B) does not implies (F_B)

We note, in contrast to Theorem 3.2.9, property (T_B) does *not* imply property (F_B) in general. An extremely intriguing example is the case of $B = \mathcal{L}_p$ for p > 2 is sufficiently large. We will see in Subsection 3.3.3.

3.3 Property $(T_{\mathcal{L}_p})$ and $(F_{\mathcal{L}_p})$

In this section, we examine property $(T_{\mathcal{L}_p})$ and $(F_{\mathcal{L}_p})$. The main goals in this section are the following two: firstly, we show property (T) implies $(T_{\mathcal{L}_p})$ (in fact, they show these are equivalent), a theorem due to Bader–Furman–Gelander–Monod; secondly, we see results of P. Pansu, Bourdon–Pajot and Yu which indicate that $(F_{\mathcal{L}_p})$ is strictly stronger than (T) if $p \gg 2$.

3.3.1 Facts for L^p spaces

We collect needed facts on L^p spaces. Recall the definition of Bernstein function from Definition 2.4.10.

3.3. PROPERTY $(T_{\mathcal{L}_{P}})$ AND $(F_{\mathcal{L}_{P}})$

Lemma 3.3.1. For $0 < \alpha < 1$, the function $\mathbb{R}_{\geq 0} \to \mathbb{R}_{\geq 0}$; $x \mapsto x^{\alpha}$ is a Bernstein function.

Proof. It can be shown there exists some positive constant $c_{\alpha} > 0$ such that

$$x^{\alpha} = c_{\alpha} \int_0^\infty (1 - e^{-tx}) t^{-\alpha - 1} dt.$$

Corollary 3.3.2. Let p be $1 . Then for any <math>L^p$ space B on a measure space, the kernel

 $B \times B \to \mathbb{R}; \ (\xi_1, \xi_2) \mapsto \|\xi_1 - \xi_2\|^p$

is conditionally negative definite.

Proof. If p = 2, we see in item (i) of Example 2.4.6. For 1 , the conclusion follows from the case of <math>p = 2, Theorem 2.4.10, and Lemma 3.3.1.

Remark 3.3.3. Note that this fact itself is true even for $0 . Also, there is a theorem of Bretagnolle–Dacunha-Castelle–Krivine [BDK] that states the following: let <math>1 \le p \le 2$. Then for a Banach space *B*, the kernel defined in Corollary 3.3.2 is conditionally negative definite if and only if *B* is isometrically isomorphic to a closed subspace of some L^p space.

We warn that the statement of Corollary 3.3.2 is no longer true for p > 2. Next, we need so-called the *Mazur map*. See Theorem 9.1 of [BeLi] for details.

Theorem 3.3.4. (the Mazur map) Let ν be a σ -finite measure. Then for any $1 < p, q < \infty$, the Mazur map $M_{p,q}: L^p(\nu) \to L^q(\nu)$, defined by

$$M_{p,q}(f) = \operatorname{sign}(f)|f|^{p/q},$$

is a (nonlinear) map which induces a uniformly continuous homeomorphism between the unit spheres $M_{p.q}: S(L^p(\nu)) \to S(L^q(\nu)).$

The statement of this theorem is true also for p = 1.

Finally, we need the classification theorem for linear isometries of L^p related spaces, which is called the *Banach–Lamperti theorem*. For a measure space (X, ν) and Banach space B, we let $L^p(X, B, \nu)$ denote the *p*-integrable *B*-valued function space by means of the Bochner integral (this space is needed for *p*-inductions. See Subsection 3.4.2).

Theorem 3.3.5. (Banach–Lamperti) For any $p \neq 2$, any linear isometry U of $L^p(X, E, \nu)$ has the form

$$Uf(x) = f(T(x))h(x)\left(\frac{dT_*\nu}{d\nu}(x)\right)^{1/p}$$

Here T is a measurable, measure-class preserving map of (X, ν) , and h is a measurable function with |h(x)| = 1 almost everywhere.

As a corollary of the Banach–Lamperti theorem, we obtain the following.

Corollary 3.3.6. Let $p \neq 2$. Then for any linear isometry U on a L^p space on a measure space, the conjugation

$$U \mapsto M_{p,2} \circ U \circ M_{2,p}$$

is a linear map. Here $M_{p,q}$ denotes the Mazur map.

Note that the Mazur map itself is nonlinear.

3.3.2 Property (T) implies $(T_{\mathcal{L}_p})$

We will show the following theorem of Bader–Furman–Gelander–Monod. Recall $p \in (1, \infty)$, unless otherwise stated.

Theorem 3.3.7. ([BFGM]) For any p, (T) implies $(T_{\mathcal{L}_p})$.

Remark 3.3.8. In fact, Bader–Furman–Gelander–Monod show the converse is also true. See Section 4.c in [BFGM].

Proof. We will sketch two ways in proving Theorem 3.3.7.

- (1) The first way is imitate the proof of Delorme's direction $(T) \Rightarrow (FH)$. Thanks to Corollary 3.3.2, it is possible if $1 . Thus we obtain <math>(T) \Rightarrow (F_{\mathcal{L}_p})$ for 1 . (For details see Section 3.b of [BFGM].) By Theorem 3.2.9, $in particular <math>(T) \Rightarrow (T_{\mathcal{L}_p})$ holds for $1 . Next, observe property <math>(T_B)$ concerns only linear representations. Therefore with the aid of contragredient representation and duality mapping, for a superreflexive Banach space B, (T_B) is equivalent to (T_{B^*}) . Therefore $(T) \Rightarrow (T_{\mathcal{L}_p})$ holds for any p.
- (2) The second way is valid for L^p spaces on a σ -finite measure. Theorem 3.3.4 and Theorem 3.3.5 are the main tools here. It is the basic idea of Theorem C (and the previous work of Puschnigg [Pus]) for property (T_B) for noncommutative L^p spaces.

Let $p \neq 2$. Set $B = L^p(\nu)$ and $\mathfrak{H} = L^2(\nu)$, where μ is a σ -finite measure. Suppose G does not have (\mathbf{T}_B) . Then there exists an isometric representation ρ on G such that the restriction of ρ on B'_{ρ} has almost invariant vectors. Set $\pi = M_{p,2} \circ \pi \circ M_{2,p}$. Then by Corollary 3.3.6, this π becomes a unitary representation $G \to U(\mathfrak{H})$. Thanks to Theorem 3.3.4 (uniform continuity), we also see that the restriction of π on $\mathfrak{H}^{\perp}_{\pi(G)}$ has almost invariant vectors. Therefore G does not have (T). This ends the proof for the case where $B = L^p(\nu)$ on a σ -finite measure. 3.3. PROPERTY $(T_{\mathcal{L}_P})$ AND $(F_{\mathcal{L}_P})$

Remark 3.3.9. The proof(s) of Theorem 3.3.7 in fact show the following:

Theorem 3.3.10. For any p, relative property (T) implies relative property $(T_{\mathcal{L}_p})$. More precisely, for any pair of a group and a normal (possibly non-proper) subgroup $G \ge N$, if it has relative (T), then it has relative $(T_{\mathcal{L}_p})$.

This observation shall play a key role in Section 8.1.

3.3.3 Property (T) does not imply $(F_{\mathcal{L}_p})$

In this subsection, we see (T) does *not* imply $(F_{\mathcal{L}_p})$ for $p \gg 2$. Recall from Subsection 2.6.4 we see plenty of examples of (infinite) hyperbolic groups with (T).

The following result is due to P. Pansu [Pan], deduced by means of study of L^{p} -cohomology, which is constructed from differential forms:

Theorem 3.3.11. (*Pansu* [Pan]) For any $m \ge 2$, $\operatorname{Sp}_{m,1}$ fails to have $(F_{\mathcal{L}_p})$, provided p > 4m + 2.

Recall Kostant's theorem that $\operatorname{Sp}_{m\geq 2,1}$ has (T) (Theorem 2.6.18). Therefore, at least p > 10, $(F_{\mathcal{L}_p})$ is strictly stronger than (T).

The following definition is not common, but it is convenient for our theory:

Definition 3.3.12. Let *B* be a Banach space or a class of Banach spaces. A group *G* is said to have the *B*-Haagerup property (or *G* is a-F_B-menable) if *G* admit a proper cocycle with some isometric *B* coefficient.

The following theorem of Cluckers–Cornulier–Louvet–Tessera–Valette is interesting.

Theorem 3.3.13. ([CCLTV]) Any simple connected algebraic group G over a local field has the "bounded or proper" property. This means, any cocycle of G with isometric (superreflexive) Banach coefficient is either bounded or proper.

By combining Theorem 3.3.11 and Theorem 3.3.13, we obtain the following theorem:

Theorem 3.3.14. For any $m \ge 2$, $\operatorname{Sp}_{m,1}$ has the \mathcal{L}_p -Haagerup property, provided p > 4m + 2.

For general (discrete) hyperbolic group, first Bourdon–Pajot [BoPa] show for each group, there exists $p \gg 2$ such that it fails to have $(F_{\mathcal{L}_p})$ (with some estimate of such p). Later, Yu [Yu2] shows in fact they have the \mathcal{L}_p -Haagerup property for respectively large p: **Theorem 3.3.15.** (Yu [Yu2]) For every hyperbolic group H, there exists p > 2 such that H admits a proper cocycle on $B = \ell^p(H \times H)$, where the isometric representation of G on B is the diagonal translation.

In particular, for each H there exists p > 2 such that H has the \mathcal{L}_p -Haagerup property.

We make remark that to the best knowledge of the author, there is no known example of p > q > 2 in which $(F_{\mathcal{L}_p})$ implies $(F_{\mathcal{L}_q})$.

3.4 $SL_3(\mathbb{Z})$ has $(F_{\mathcal{L}_p})$

In this section, we see the following result of Bader–Furman–Gelander–Monod: any totally higher rank algebraic group has $(F_{\mathcal{L}_p})$ for any p. We pay a special attention for the case of $\mathrm{SL}_{m\geq 3}(\mathbb{R})$. Next we examine theory of p-induction of cocycles, which enables us to deduce $(F_{\mathcal{L}_p})$ for certain lattices.

3.4.1 SL₃(\mathbb{R}) has (F_{L_p})

In this subsection, the following is the main goal.

Theorem 3.4.1. Let \mathbb{K} be a local field. For any $m \geq 3$, $SL_m(\mathbb{K})$ has $(F_{\mathcal{L}_p})$ for any p. For any $m \geq 2$, $Sp_{2m}(\mathbb{K})$ has $(F_{\mathcal{L}_p})$ for any p.

To show Theorem 3.4.1, we observe the following: firstly, by combining Theorem 3.3.10 and Theorem 2.6.5, we have the following result.

Theorem 3.4.2. ([BFGM]) For any local field \mathbb{K} , the pair $G = \mathrm{SL}_2(\mathbb{K}) \ltimes \mathbb{K}^2 \supseteq \mathbb{K}^2 = N$ has strong relative $(\mathrm{T}_{\mathcal{L}_p})$ in the following sense: for any isometric representation (ρ, B) with $B \in \mathcal{L}_p$, $\rho \mid_H$ on $B'_{\rho(N)}$ does not have almost invariant vectors. Here $H = \mathrm{SL}_2(\mathbb{K})$.

Next, there is a generalization of the Howe–Moore theorem (Theorem 2.6.7). This is an unpublished result of Shalom, and stated in Section 9 in [BFGM] with proof.

Theorem 3.4.3. (Shalom) Let G be a group of the form $\prod_{i=1}^{m} \mathbf{G}_{i}(k_{i})$, where k_{i} are local fields, $\mathbf{G}_{i}(k_{i})$ are k_{i} -points of Zariski connected simple k_{i} -algebraic groups.

Let B be a ucus Banach space. Let ρ be an isometric G-representation on B such that for each $G_i = \mathbf{G}_i(k_i) \pi |_{G_i} \not\supseteq \mathbf{1}_{G_i}$. Then ρ is a C_0 -representation, namely, for any vector $\xi \in B$ and $\phi \in B^*$, the matrix coefficient

$$G \ni g \mapsto \langle \rho(g)\xi, \phi \rangle \in \mathbb{C}$$

is a C_0 -function.

3.4. $SL_3(\mathbb{Z}) HAS(F_{\mathcal{L}_P})$

Corollary 3.4.4. Let G be a Zariski connected simple algebraic group G over a local field. Let B be a ucus Banach space. Then for any isometric G-representation ρ on B, there is an equality:

for any non-compact subgroup $H \leq G$, $B^{\rho(H)} = B^{\rho(G)}$.

The difficulty in proving Theorem 3.4.1 is to deduce some form of property (F_B) from property (T_B). This direction is in general *not* true, and therefore we need some technique.

Proof. (Theorem 3.4.1) For simplicity, we only consider the case of $SL_3(\mathbb{K})$. Let $G = SL_3(\mathbb{K})$, and find a copy of $H = SL_2(\mathbb{K})$ in the left upper corner and of $N = \mathbb{K}$ in the (1,3)-(2,3)-th unipotent part. Also set A be the multiplicative abelian group of diagonal elements, with its diagonal entries $[t, t, t^{-2}]$ $(t \in \mathbb{K}^{\times})$. Then for any isometric G-representation ρ on B, we have the following:

- (1) The group H commutes with A.
- (2) The restricted representation $\rho \mid_H$ on $B'_{\rho(G)}$ does not have almost invariant vectors.

Indeed, item (1) is obvious. Item (2) follows from Theorem 3.4.2 and Corollary 3.4.4 (we use $B^{\rho(N)} = B^{\rho(G)}$).

Let c be any ρ -cocycle. We project c to $B_1 = B'_{\rho(G)}$, and write it c_1 . By $\rho(G)$ invariance of B_1 , c_1 is also a ρ -cocycle on B_1 . Then by the cocycle identity, for any $h \in H$ and $a \in A$,

$$c_1(h) + \rho(h)c_1(a) = c_1(ha) = c_1(ah) = c_1(a) + \rho(a)c_1(h).$$

Here we use item (1). Hence we have

$$c_1(a) - \rho(h)c_1(a) = c_1(h) - \rho(a)c_1(h).$$

By item (b), there exist a compact subset $S \subseteq H$ and $\kappa > 0$ such that for any $\xi \in B_1$, $\sup_{s \in S} \|\xi - \rho(s)\xi\| \ge \kappa \|\xi\|$. Set $R = \sup_{s \in S} \|c_1(s)\| < \infty$. Then from the equality above, we have,

for any
$$a \in A$$
, $\kappa \|c_1(a)\| \le \sup_{s \in S} \|c_1(s) - \rho(a)c_1(s)\| \le 2R$.

Therefore c_1 is bounded on A. By Lemma 3.2.5, there exists an A-fixed point for the affine isometric G-action α_1 on B_1 associated with (ρ, c_1) . Note that this fixed point is unique, because by Corollary 3.4.4 B_1 has no non-zero $\rho(A)$ -invariant vectors. Define $\eta \in B_1$ the unique $\alpha_1(A)$ -fixed point.

Finally, set $T \leq G$ be the diagonal maximal split torus. Since T and A commute, any element in set $\alpha_1(T) \cdot \eta$ is $\alpha_1(A)$ -fixed. Again by the uniqueness, $\alpha_1(T) \cdot \eta$ equals the single point η . This means η is $\alpha_1(T)$ -fixed. Now consider the Cartan decomposition of G

$$G = KTK.$$

where K is the maximal compact subgroup. Through this decomposition, it follows from the argument above that any $\alpha_1(G)$ orbit is bounded. Since G has trivial abelianization (in general, finite abelianization follows from (T)), any affine isometric action on $B_0 = B^{\rho(G)}$ with linear part $\rho(= 1_G \text{ on } B_0)$ has bounded orbit. By combining these two boundedness, we obtain the boundedness of the affine isometric G-action on B with linear part ρ and cocycle part c. By Lemma 3.2.5, this means c is a coboundary. Therefore, we establish ($F_{\mathcal{L}_p}$) for G.

Note that the decomposition $B = B_0 \oplus B_1$ in this proof is essential for the argument.

In a similar manner to the proof above, we have the following theorem of Bader– Furman–Gelander–Monod (for more details, see Section 5.c in [BFGM]):

Theorem 3.4.5. ([BFGM]) Any totally higher rank algebraic group has $(F_{\mathcal{L}_p})$ for all p.

3.4.2 *p*-inductions

We explain *p*-induction of cocycles in Section 8 of [BFGM]. Recall the induction of cocycles in unitary setting in Subsection 2.3.3. Let $\Gamma \leq G$ be a lattice, and \mathcal{D} be a Borel fundamental domain for Γ . We identify \mathcal{D} with G/Γ . Recall the definition of the Borel cocycle $\beta \colon G \times \mathcal{D} \to \Gamma$:

 $\beta(g, x) = \gamma$ if and only if $g^{-1}x\gamma \in \mathcal{D}$.

Definition 3.4.6. A lattice Γ in a group G is said to be *p*-integrable if either of the following two conditions is satisfied:

- (1) the lattice Γ is cocompact;
- (2) the lattice Γ is finitely generated, and for some (equivalently any) symmetric finite generating set S of Γ , there exists a Borel fundamental domain $\mathcal{D} \subseteq G$ such that

for any
$$g \in G$$
, $\int_D l_S(\beta(g, x))^p d\mu(x) < \infty$.

Here $l_S \colon \Gamma \to \mathbb{Z}_{\geq 0}$ denotes the word length on Γ with respect to S.

It is shown in $\S2$ in a paper of Shalom [Sha3] that higher rank lattices, in the sense in Chapter 0 is *p*-integrable.

Suppose the lattice $\Gamma \leq G$ is *p*-integrable. Take a cocycle on a Banach space B, then consider the space $E = L^p(\mathcal{D}, B)$ of all *p*-integrable vector valued functions by means of the Bochner integral. A result of Figiel–Pisier implies the following: Lemma 3.4.7. (Figiel-Pisier [FiPi]) If B is ucus, then so is E.

Finally, define an induced cocycle \tilde{c} of c on E as follows:

$$\tilde{c}(g)(x) := c(\beta(g, x)) \quad (g \in G, \ x \in \mathcal{D}).$$

The *p*-integrability condition ensures the well-definedness (namely, $\tilde{c} \in E$). With the aid of these theories, the following theorem is deduced in a similar manner to one in Theorem 2.3.16:

Proposition 3.4.8. ([BFGM]) Keep the notation of this subsection. Suppose Γ is *p*-integrable. Then if G has (F_E), then Γ has (F_B).

Thus finally, we obtain the following theorem of Bader–Furman–Gelander–Monod:

Theorem 3.4.9. ([BFGM]) Any totally higher rank lattice has $(F_{\mathcal{L}_p})$ for all p. In particular, $SL_{m\geq 3}(\mathbb{Z})$ has $(F_{\mathcal{L}_p})$ for all p.

Proof. This follows from Theorem 3.4.5 and Proposition 3.4.8 because the class \mathcal{L}_p is stable under *p*-inductions with the same *p*.

3.5 Property $(T_{[\mathcal{H}]})$ and $(F_{[\mathcal{H}]})$

Recall the class $[\mathcal{H}]$ consists of all Banach spaces B admitting compatible norms to those of Hilbert spaces. Therefore, by considering compatible Hilbert norms on B, we have the following (see also the proof of Proposition 3.1.10):

- Property $(T_{[\mathcal{H}]})$ for a group G is equivalent to having the following condition: for any *uniformly bounded* representation ρ of G on a Hilbert space, $\rho \succeq 1_G$ implies $\rho \supseteq 1_G$.
- Property $(F_{[\mathcal{H}]})$ for a group G is equivalent to having the following condition: for any *uniformly bounded* representation ρ on a Hilbert space, every ρ -cocycle is a coboundary (equivalently, is bounded). Also, this is equivalent to the following: any affine *uniformly bi-Lipschitz* action ρ of G on a Hilbert space has a G-fixed point (equivalently, has bounded orbits).

For relative versions, similar results hold.

In [BFGM], these properties are respectively called $(\overline{T}_{\mathcal{H}})$ and $(\overline{F}_{\mathcal{H}})$. It seems to be an open problem whether $(T_{[\mathcal{H}]})$ implies $(F_{[\mathcal{H}]})$.

The following is an unpublished result of Shalom (see item (3) of Remark 1.7 in [BFGM]):

Theorem 3.5.1. (Shalom)

- (i) Any totally higher rank algebraic group and lattice have $(F_{[\mathcal{H}]})$. In particular, they have $(T_{[\mathcal{H}]})$.
- (ii) Any rank 1 algebraic group and lattices therein fail to have $(T_{[\mathcal{H}]})$. In particular, they also fail to have $(F_{[\mathcal{H}]})$.

Item (*ii*) of this theorem shows $(T_{[\mathcal{H}]})$ (and hence $(F_{[\mathcal{H}]})$) are strictly stronger than (T). For instance, consider $Sp_{m,1}$ ($m \geq 2$).

To the best knowledge of the author, it is unknown whether infinite hyperbolic group can have $(T_{\mathcal{H}})$.

Finally, we define a generalization of Kazhdan constants in this setting. It is more convenient to consider uniformly bounded representations on Hilbert spaces than isometric representations on $[\mathcal{H}]$. Therefore we define in the following situation. With noting that a class of uniformly uniformly bounded representations are stable under L^2 direct sum, we define in the following manner.

Definition 3.5.2. ([Mim1]) Let $G \geq N$ be a pair of groups, and S be a compact subset of Γ . For $M \geq 1$, we define \mathcal{A}_M as the class of all pairs (ρ, \mathfrak{H}) of a uniformly bounded G-representation and a representation Hilbert space that satisfy $|\rho| \leq M$. We define the relative Kazhdan constant for uniformly bounded representations by the following equality:

$$\overline{\mathcal{K}}(\Gamma, N; S; M) := \inf_{(\rho, \mathfrak{H}) \in \mathcal{A}_M} \inf_{\xi \in S\left(\mathfrak{H}'_{\rho(N)}\right)} \sup_{s \in S} \|\xi - \rho(s)\xi\|_{\mathfrak{H}}.$$

Chapter 4

Universal lattices

In this chapter, we define universal lattices, via the definition of elementary groups and the Suslin stability theorem. Next, we observe a celebrated method of Shalom [Sha1] to prove property (T) (and (FH)) from bounded generation. Finally, we consider bounded generation of elementary groups over certain rings, and state the difficulty in proving property (T) for universal lattices, which is a theorem of Shalom [Sha5] and Vaserstein [Vas2] (Theorem 1.0.1). This theorem shall be shown in Chapter 5. A main reference of this chapter is a paper [Sha1] of Shalom. See also Chapter 4 of [BHV] for details on Section 4.2 in this thesis.

4.1 Definition

We define elementary group over a (possibly noncommutative) ring.

4.1.1 Elementary group

Recall from Chapter 0 we always assume all rings are associative and unital. For a ring R (possibly noncommutative), let $M_m(R)$ denote the $m \times m$ matrix ring over R.

Definition 4.1.1. Let R be a ring and $m \ge 2$.

- (i) An elementary matrix in $M_m(R)$ is a matrix of the form $E_{i,j}(r) = I_m + r \cdot e_{i,j}$ $(1 \le i \le m, 1 \le j \le m, i \ne j, \text{ and } r \in R)$. Here $e_{i,j}$ denotes the matrix with the (i, j)-th entry 1 and the other entries 0.
- (ii) The elementary group over R, written as $E_m(R)$, is the multiplicative group in $M_m(R)$ generated by all elementary matrices in $M_m(R)$.
- (*iii*) The general linear group over R, written as $GL_m(R)$, is the multiplicative group of all invertible elements in $M_m(R)$.

(iv) Suppose A = R is commutative. Then the special linear group over A, written as $SL_m(A)$, is the subgroup of $GL_m(A)$ consisting of all elements of determinant 1.

We note in some literature, $E_m(R)$ is written as $EL_m(R)$ in order to distinguish it from exceptional groups. Also observe for a commutative ring A, $E_m(A)$ is a priori a subgroup of $SL_m(A)$. It depends on the ring A and m whether these two coincide.

Remark 4.1.2. We note that in many cases, situations are completely different between $m \ge 3$ and m = 2.

(i) Suppose $m \ge 3$. Then it is straightforward to see the following commutator relation, which plays an important role:

for any $i \neq j$, $j \neq k$, $k \neq i$, $[E_{i,j}(r_1), E_{j,k}(r_2)] = E_{i,k}(r_1r_2)$.

Here $1 \leq i, j, k \leq m$ and $r_1, r_2 \in R$, and our commutator convention is $[g, h] = ghg^{-1}h^{-1}$. In particular, the following holds:

Lemma 4.1.3. Let $m \ge 3$. Let R be a ring and S be a generating (as a ring) subset. Then the group $E_m(R)$ is generated by the subset $\{E_{i,j}(s) : 1 \le i, j \le m, i \ne j, s \in S\}$.

In particular, if R is a finitely generating ring and S is a finite generating set, then the group $E_m(R)$ is finitely generated with a finite generating subset

$$\{E_{i,j}(s): 1 \le i, j \le m, i \ne j, s \in S\}.$$

- (*ii*) The commutator relation above shows that for any ring R and $m \ge 3$, $E_m(R)$ is perfect, namely, $[E_m(R), E_m(R)] = E_m(R)$. As we will see in Lemma 4.1.6, for any $m E_m(\mathbb{Z}) = SL_m(\mathbb{Z})$. Hence $SL_{m\ge 3}(\mathbb{Z})$ is perfect.
- (*iii*) Suppose m = 2. Then the commutator relation above is empty, because there are no i, j, k with $i \neq j, j \neq k, k \neq i$. For instance, it is known that the group $\operatorname{SL}_2(\mathbb{F}_q[x])$ is not finitely generated, where q is a positive power of a prime. As we will see in Subsection 4.1.2, $\operatorname{SL}_2(\mathbb{F}_q[x])$ coincides with $\operatorname{E}_2(\mathbb{F}_q[x])$. Thus $\operatorname{E}_2(\mathbb{F}_q[x])$ is not finitely generated, although the ring $\mathbb{F}_q[x]$ is finitely generated.
- (*iv*) Suppose A and A' are commutative rings and there is a ring homomorphism $\phi: A \to A'$. Let $m \geq 2$. Then ϕ induces group homomorphisms $\Phi_1: E_m(A) \to E_m(A')$ and $\Phi_2: SL_m(A) \to SL_m(A')$. If ϕ is surjective, then by the very definition, Φ_1 is surjective for all m. However, it is *not* necessarily true that Φ_2 is surjective. This argument for Φ_1 also holds in noncommutative ring settings.

One corollary is that $E_2(\mathbb{Z}[x])$ is *not* finitely generated. Indeed, for a prime $q, \mathbb{Z}[x]$ maps onto $\mathbb{F}_q[x]$. Therefore there is a surjection $E_2(\mathbb{Z}[x]) \twoheadrightarrow E_2(\mathbb{F}_q[x])$. Then item (*ii*) above ends the proof.

4.1.2 The Suslin stability theorem

It is a problem of high interest and importance in algebraic K-theory to determine whether $E_m(A)$ coincides with $SL_m(A)$ for $m \in \mathbb{Z}_{\geq 2}$ and a commutative ring A. Note that there is a concept of the general elementary group $GE_m(R)$ for a noncommutative group. It is reasonable to think whether $GE_m(R)$ coincides with $GL_m(R)$ in noncommutative ring settings. See a paper of P. M. Cohn [Coh] for instance.

One tool to consider this problem is the *stable range* of H. Bass [Bas].

Definition 4.1.4. Let R be a (possibly noncommutative) ring.

(i) A sequence a_1, \ldots, a_n in R is said to be *left unimodular* if $Ra_1 + \cdots + Ra_n = R$. If $n \ge 2$, we say this sequence is *left reducible* if there exists $r_1, \ldots, r_{n-1} \in R$ such that

$$R(a_1 + r_1 a_n) + \dots + R(a_{n-1} + r_{n-1} a_n) = R.$$

(ii) Let $n \ge 1$. The ring R is said to have left stable range $\le n$ if any sequence of length $\ge n+1$ is reducible. We define the left stable range of R as the infimum of such n (if there exists no such n, then we set the stable range as ∞).

There is a parallel notion of *right stable range*. However, Vaserstein [Vas1] shows these two coincide. Therefore, we call them *stable range* of R, and write as sr(R).

We warn that this definition differs 1 from the original definition of Bass. Our definition of stable range implies that $\operatorname{sr}(R) = 1$ for any euclidean ring R. For instance, $\operatorname{sr}(\mathbb{Z}) = 1$. We note that Grunewald–Mennicke–Vasestein [GMV2] shows $\operatorname{sr}(\mathbb{Z}[x]) = 3$. In general, it is known $\operatorname{sr}(\mathbb{Z}[x_1, \ldots, x_k]) \leq k + 2$. It seems that it is unknown whether $\operatorname{sr}(\mathbb{Z}\langle x_1, \ldots, x_k \rangle) < \infty$ for $k \geq 2$. Here $\mathbb{Z}\langle x_1, \ldots, x_k \rangle$ denotes the noncommutative polynomial ring over \mathbb{Z} .

The following theorem is due to Dennis–Vaserstein [DeVa], based on a result of Bass [Bas] in the case of m - r = 2:

Proposition 4.1.5. ([Bas], [DeVa]) Let A be a commutative ring with $sr(a) \ge r$. Then for any m > r, we have

$$\operatorname{SL}_m(A) = ULUL\operatorname{SL}_{r+1}(A).$$

Here $U = U_m A$ and $L = L_m A$ respectively means the group of upper triangular unipotent matrices and that of lower triangular unipotent matrices.

Proof. We only give a proof in the case of r = 2 and m = 4. This proof is based on a pair of U. Hadad [Had], the proof of Lemma 3.2. Let $g \in SL_3(A)$. Consider the (1, i)-th entries a_i of g (i = 1, 2, 3, 4). Also consider (i, 1)-th entries b_i of $g^{-1} \in SL_3(A)$ (i = 1, 2, 3, 4). Then we have

$$b_1a_1 + b_2a_2 + b_3a_3 + b_4a_3 = 1$$

Therefore, the sequence $(a_1, a_2, b_3a_3 + b_4a_4)$ is unimodular. By stable range condition, there exists $t_1, t_2 \in A$ such that

$$A(a_1 + t_1(b_3a_3 + b_4a_4)) + A(a_2 + t_2(b_3a_3 + b_4a_4)) = A.$$

Therefore, there exists $x_1 \in A$ and $x_2 \in A$ such that

$$x_1(a_1 + t_1(b_3a_3 + b_4a_4)) + x_2(a_2 + t_2(b_3a_3 + b_4a_4)) = 1.$$

This implies, in terms of elementary matrices, that the following matrix has (4, 4)-th entry 1:

$$E_{4,2}((a_4-1)x_2)E_{4,1}((a_4-1)x_1)E_{2,3}(t_2b_3)E_{2,4}(t_2b_4)E_{1,3}(t_1b_3)E_{1,4}(t_1b_4)g.$$

For details, we refer to the proof of Lemma 3.2 in [Had].

Therefore by some other mulplications of U and L by left, we can eliminate all (i, 4)-th and (4, j)-th entries for i = 1, 2, 3. This shows

$$\operatorname{SL}_4(A) = ULU\operatorname{SL}_3(A).$$

as desired.

Also, it can be done by euclidean algorithm to express any element in $SL_2(A)$ as a product of elementary matrices, provided that the ring A is euclidean. Thus combining this with Proposition 4.1.5, we obtain the following:

Lemma 4.1.6. For any euclidean domain A and for any $m \ge 2$, $E_m(A) = SL_m(A)$.

Note that in the lemma above, the minimum number of expressing each element $g \in$ $SL_2(A)$ as a product of elementary matrices in $M_2(A)$ is usually unbounded on $g \in SL_2(A)$ (if A is a field, then it is bounded). However, if we consider the case of $m \geq 3$, then the situation may change. For details, see Subsection 4.2.2; and Subsection 4.3.1. Also we note that Lemma 4.1.6 itself can be deduced without Proposition 4.1.5, but that Proposition 4.1.5 gives more precise information about elimination procedures from $SL_m(A)$ to $SL_2(A)$.

Consider the case of $A = \mathbb{Z}[x_1, \ldots, x_k]$. As soon as $k \ge 1$, this ring becomes *noneuclidean*. The stable range of A will increase if k increases. Nevertheless, A. S. Suslin shows the following celebrated stability theorem:

Theorem 4.1.7. (Suslin [Sus]) Let $k \in \mathbb{N}$. Then for any $m \geq 3$, there are equalities

In fact, Suslin shows that these equalities hold true even if \mathbb{Z} is replaced with any euclidean domain. This proof employs high techniques in algebraic K-theory and is not algorithmic, but later Park–Woodburn [PaWo] gave an algorithmic proof for the case of $SL_m(F[x_1, \ldots, x_k])$, where F is any (commutative) field.

Remark 4.1.8. It is worth stating that the situation is completely different if m = 2. Indeed, the following is shown by Cohn.

Proposition 4.1.9. The group $E_2(\mathbb{Z}[x])$ is a proper subgroup of $SL_2(\mathbb{Z}[x])$: $E_2(\mathbb{Z}[x]) \leq SL_2(\mathbb{Z}[x])$.

For instance, he shown the following matrix is in $SL_2(\mathbb{Z}[x]) \setminus E_2(\mathbb{Z}[x])$:

$$\left(\begin{array}{cc} 1+2x & x^2\\ -4 & 1-2x \end{array}\right) \in \mathrm{SL}_2(\mathbb{Z}[x]).$$

It is an open problem whether $E_2 = SL_2$ for $\mathbb{Z}[x, x^{-1}]$.

The following terminology universal lattice is due to Shalom [Sha1], and it is the main object in this thesis. Recall from Chapter 0 (, thanks to Theorem 4.1.7,) we let k be any natural number, and do not distinguish each other, unless otherwise stating.

Definition 4.1.10. Let $m \geq 3$.

A universal lattice of degree m denotes a group

$$\operatorname{SL}_m(\mathbb{Z}[x_1,\ldots,x_k]) (= \operatorname{E}_m(\mathbb{Z}[x_1,\ldots,x_k])).$$

A noncommutative universal lattices of degree m denotes a group

$$\mathbf{E}_m(\mathbb{Z}\langle x_1,\ldots,x_k\rangle).$$

We note before Shalom, interests in universal lattices from aspects of rigidity date back to those of A. Lubotzky.

Remark 4.1.11. (i) Universal lattices are universal in the following sense: by item (ii) of Remark 4.1.2, for $m \geq 3$ and any finitely generated commutative ring A, some universal lattice of degree m maps onto $E_m(A)$. Therefore, for a property P which passes to group quotients, if all universal lattices of degree m has P, then it implies for any commutative finitely generated ring A the group $E_m(A)$ fulfills the property. Such examples of $E_m(A)$ includes $SL_m(\mathcal{O})$ for a ring of integers \mathcal{O} of a local field, which is a lattice in a semi-simple algebraic group. However they also include groups not coming from these:

Lemma 4.1.12. No universal lattice (with $k \ge 1$) can be realized as an arithmetic lattice in a semisimple algebraic group.

We note that if universal lattice can be realized as a lattice in a semisimple algebraic group, then it will be automatically arithmetic. This follows from the Margulis arithmeticity theorem.

Proof. The Margulis finiteness theorem states the following:

any normal subgroup of a higher rank (and hence arithmetic) lattice is either finite or of finite index.

However, $\operatorname{SL}_m(\mathbb{Z}[x])$ has an infinite normal subgroup of infinite index. Indeed, for instance, consider the congruence subgroup H associated with the ideal $(x) \leq \mathbb{Z}[x]$, namely, H is the kernel of a map

$$\operatorname{SL}_m(\mathbb{Z}[x]) \twoheadrightarrow \operatorname{SL}_m(\mathbb{Z})$$
, which sends x to 0.

One of the main motivations of Shalom to deal with universal lattices is a relation to property (T). As we see in the paragraph above, if universal lattices above have (T), then this result will be able to regarded as the universal reseults among all groups of the form $E_m(A)$, such as $SL_m(\mathbb{Z})$, $SL_m(\mathbb{Z}[1/2])$, $SL_m(\mathbb{Z}[\sqrt{2},\sqrt{3}])$, $SL_m(\mathbb{F}_q[x])$, and $SL_m(\mathbb{Z}[x,x^{-1}])$. Note that all but last one examples are arithmetic lattices, however the last example is *not*, for the same reason as in above. Also, since Kazhdan constant does not decrease under group quotients, if universal lattices above have (T), then this will imply Kazhdan constants of $E_m(A)$ (with a finite generating set in Lemma 4.1.3) is bounded below from zero, and a lower bound only depends on m and the number of generators of A.

As we mentioned repeatedly, property (T) for universal lattices is proven [Sha5], [Vas2]. Also, a problem on property (T) for noncommutative universal lattices is of great significance, and Ershov–Jaikin-Zapirain have answered positively in [ErJa]. We will also see this in Subsection 9.2.2.

(*ii*) Although structures of universal lattices have not been well-studied, universal lattices are linear groups in the sense in Definition 2.7.4. This is verified by sending each x_i to algebraically independent transcendental numbers. Therefore by Theorem 2.7.10, these groups have property A of Yu, and the Novikov conjectures for these hold true.

On the other hand, it is shown by Kassabov–Sapir [KaSa] that noncommutative universal lattices (with $k \ge 1$) are *non*-linear. It is an open problem to determine these has property A, or these admit uniform embeddings into Hilbert spaces (Definition 2.7.7). **Definition 4.1.13.** Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ or $R = \mathbb{Z}\langle x_1, \ldots, x_k \rangle$. For $m \ge 2$, a *unit* elementary matrix in $\mathbb{E}_m(A)$ means a matrix of the form $E_{i,j}(\pm x_l)$ with $i \ne j$ and $0 \le l \le m$. Here x_0 means 1.

The set of unit elementary matrices is finite. By Lemma 4.1.3, unit elementary matrices generate the whole group, provided $m \geq 3$.

4.2 Bounded generation and (T)

In this section, we see the argument of Shalom [Sha1] in proving property (T) for certain elementary group $E_m(R)$ with an explicit estimate for Kazhdan constants. An attempt to make an estimation of a Kazhdan constant for $SL_3(\mathbb{Z})$ dates back to a work of M. Burger [Bur], and Shalom [Sha1]. Shalom employs bounded generation for this.

4.2.1 Relative (T) for $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$

First, we show the following theorem due to Shalom, which is a quantitative version of Corollary 2.6.2. Throughout this subsection, we keep the following setting.

Set $G = H \ltimes N$ with $H = \mathrm{SL}_2(\mathbb{Z})$ and $N = \mathbb{Z}^2$. We identify $G \supseteq N$ with

$$\left\{ \left(\begin{array}{c|c} W & v \\ \hline 0 & 1 \end{array} \right) : W \in \mathrm{SL}_2(\mathbb{Z}), \ v \in \mathbb{Z}^2 \right\} \quad \succeq \quad \left\{ \left(\begin{array}{c|c} I_2 & v \\ \hline 0 & 1 \end{array} \right) : v \in \mathbb{Z}^2 \right\}.$$

Set a finite generating set S of G as the set of unit elementary matrices in $SL_3(\mathbb{Z})$ which lie in G with the identification above. Also, we set $S_0 = S \cap H$ and $S_1 = S \cap N$. Then $S = S_0 \cup S_1$. Namely, with the identification above,

$$S_0 = \{h_1^{\pm}, h_2^{\pm}\}, \ S_1 == \{l_1^{\pm}, l_2^{\pm}\}.$$

Here $h_1 = E_{1,2}(1)$, $h_1 = E_{2,1}(1)$; and $l_1 = E_{1,3}(1)$, $l_2 = E_{2,3}(1)$.

Theorem 4.2.1. (Shalom [Sha1]) The pair $SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 \supseteq \mathbb{Z}^2$ has relative (T). Moreover, with the setting in this subsection, there is an inequality

$$\mathcal{K}(G,N;S) > \frac{1}{10}$$

Proof. Suppose (S, 1/10) is not a relative Kazhdan pair. Then there exists a unitary G-representation (π, \mathfrak{H}) such that the following two conditions are satisfied:

- (a) The restriction of π on N satisfies $\pi \mid_N \not\supseteq 1_N$.
- (b) The representation π admist an (S, 1/10)-invariant vector, namely, there exists $\xi \in S(\mathfrak{H})$ such that the folloiwing holds:

$$\epsilon := \sup_{s \in S} \|\xi - \pi(s)\xi\| < 1/10.$$

We fix this π and ξ . Thanks to spectral theory on the additive group $N, \pi \mid_N : N \to U(\mathfrak{H})$ yields a *-homomorphism $\sigma : C(\hat{N}) \to \mathbb{B}(\mathfrak{H})$. We identify \hat{N} with $\mathbb{T}^2 \cong [-1/2, 1/2)$ with the identification

$$\mathbb{T}^2 \ni t \longleftrightarrow \chi_t \in \hat{N}; \quad \chi_t(n) = \exp(2\pi\sqrt{-1}n \cdot t),$$

where for $t = {}^{t}(t_1, t_2) \in \mathbb{T}^2$ and $n = {}^{t}(n_1, n_2) \in N$, $n \cdot t = n_1 t_1 + n_2 t_2$. With this identification, we obtain a *-homomorphism

$$\sigma\colon C(\mathbb{T}^2)\to \mathbb{B}(\mathfrak{H}),$$

which sends each z_i (i = 1, 2) to $\pi(l_i) \in U(\mathfrak{H})$, where $z_i(t) := \exp(2\pi\sqrt{-1}t_i)$. Consider a positive linear functional on $C(\mathbb{T}^2)$, $f \mapsto \langle \sigma(f)\xi | \xi \rangle$, where ξ is taken by item (b). Then by the Riesz-Markov-Kakutani theorem, there exists a (positive) regular Borel measure ν such that

for any
$$f \in C(\mathbb{T}^2)$$
, $\int_{\mathbb{T}^2} f d\nu = \sigma(f)\xi |\xi\rangle$

Note that by pointwisely approximating a characteristic function by positive elements in $C(\mathbb{T}^2)$ monotone decreasing way (here we consider the convergence in $B(\mathfrak{H})$ in the strong operator topology), we obtain the spectral projection

$$E: \mathcal{B}(\mathbb{T}^2) \to \mathbb{B}(\mathfrak{H}),$$

where $\mathcal{B}(\mathbb{T}^2)$ denotes the Borel σ -algebra of \mathbb{T}^2 . In particular, for any $Z \in \mathcal{B}(\mathbb{T}^2)$ the following holds:

$$\nu(Z) = \int \chi_Z d\nu = \langle E(Z)\xi, \xi \rangle = \langle \sigma(\chi_Z)\xi, \xi \rangle.$$

Here χ_Z is the characteristic function.

We take the following decomposition of \mathbb{T}^2 :

$$\{0\}; D_0 = \{|t_1| \ge 1/4 \text{ or } |t_2| \ge 1/4\}; D_1 = \{|t_2| \le |t_1| < 1/4 \text{ and } t_1 t_2 > 0\}; D_2 = \{|t_1| < |t_2| < 1/4 \text{ and } t_1 t_2 \ge 0\}; D_3 = \{|t_1| \le |t_2| < 1/4 \text{ and } t_1 t_2 < 0\}; D_4 = \{|t_2| < |t_1| < 1/4 \text{ and } t_1 t_2 \le 0\}.$$

Firstly, we claim that $\nu(D_0) \leq \epsilon^2$. Indeed, for i = 1, 2, set $D_{0,i} = D_0 \cap \{|t_i| \geq 1/4\}$. Then $t \in D_{0,i}$ implies $|1 - z_i|^2 \geq 2$ for i = 1, 2. Therefore, for each i = 1, 2, by item (b),

$$2\nu(D_{0,i}) \leq \int \chi_{D_{0,i}} |1 - z_i|^2 d\nu \leq \int |1 - z_i|^2 d\nu$$

=
$$\int (1 - \overline{z_i})(1 - z_i) d\nu = \langle \sigma((1 - \overline{z_i})(1 - z_i))\xi |\xi \rangle$$

=
$$\langle (I - \pi(l_i)^*)(I - \pi(l_i))\xi |\xi \rangle$$

=
$$\|\xi - \pi(l_i)\xi\|^2 \leq \epsilon^2.$$

Here $\overline{z_i}$ means the complex conjugation of z_i in $C(\mathbb{T}^2)$. By observing $D_0 = D_{0,1} \cup D_{0,2}$, we confirm the claim.

Secondly, as we have seen in Subsection 2.6.1, the associated action of H =SL₂(\mathbb{Z}) on ($\hat{N} \cong$) \mathbb{T}^2 is the multiplication of inverse transpose. In other words,

for any $h \in H$, and $Z \in \mathcal{B}(\mathbb{T}^2)$, $E({}^t h^{-1} \cdot Z) = \pi(h)E(Z)\pi(h^{-1})$.

We now claim that for any $Z \in \mathcal{B}(\mathbb{T}^2)$, the following holds:

for any
$$h \in S_0(=\{h_1^{\pm}, h_2^{\pm}\}), \quad |\nu({}^t h^{-1} \cdot Z) - \nu(Z)| < 2\epsilon.$$

Indeed, we have for any Z and any $h \in S_0$,

$$\begin{aligned} |\nu({}^{t}h^{-1} \cdot Z) - \nu(Z)| &= |\langle \pi(h)E(Z)\pi(h^{-1})\xi|\xi\rangle - \langle E(Z)\xi|\xi\rangle| \\ &= |(\langle \pi(h)E(Z)\pi(h^{-1})\xi|\xi\rangle - \langle E(Z)\pi(h^{-1})\xi|\xi\rangle) \\ &+ (\langle E(Z)\pi(h^{-1})\xi|\xi\rangle - \langle E(Z)\xi|\xi\rangle)| \\ &\leq |\langle E(Z)\pi(h^{-1})\xi|\pi(h^{-1})\xi - \xi\rangle| + |\langle E(Z)(\pi(h^{-1})\xi - \xi)|\xi\rangle| \\ &\leq 2\epsilon. \end{aligned}$$

Here in the last line of the inequalities in above, we employ item (b), the Cauchy–Schwartz inequality, and the fact E(Z) is projection (and hence $||E(Z)|| \le 1$).

Finally, note that ${}^{t}h_{1}^{-1} \cdot (D_{1} \sqcup D_{2}) \subseteq D_{2} \sqcup D_{0}$. By the two claims above, this inclusion implies $\nu(D_{1}) \leq 2\epsilon + \epsilon^{2}$. Similarly, we have the same estimate for any $\nu(D_{i})$ (i = 1, 2, 3, 4). Therefore, we conclude

$$\mu(\{0\}) = 1 - \sum_{i=0}^{4} \nu(D_i) \ge 1 - 8\epsilon - 5\epsilon^2.$$

At the beginning we set $\epsilon < 1/10$, and thus we have

$$\mu(\{0\}) > 0.$$

However this means $E(\{0\})$, the projection onto $\mathfrak{H}^{\pi(N)}$, is non-zero. This contradicts item (a). Therefore we have established the assertion.

By Lemma 2.1.12, we obtain the following:

Corollary 4.2.2. Keep the setting in this subsection. Let (π, \mathfrak{H}) be any unitary representation of $G = H \ltimes N$. For a vector ξ , set $\delta_{\xi} := \sup_{s \in S} \|\xi - \pi(s)\xi\|$. Then there is an inequality:

for any
$$l \in N$$
, $\|\xi - \pi(l)\xi\| \le 20\delta_{\xi}$.

Shalom shows this argument also works if \mathbb{Z} is replaced with $\mathbb{Z}[x]$, or with $\mathbb{Z}[x_1, \ldots x_k]$, with some estimation of relative Kazhdan constant. Kassabov [Kas1] improved the estimate. In [Kas2], he noticed in fact it works even noncommutative ring setting, namely in the case of $\mathbb{Z}\langle x_1, \ldots x_k \rangle$, because for N part in our setting we only consider additive group structure. Thus we see the following result:

Theorem 4.2.3. (Shalom [Sha1]; Kassabov [Kas2]) Let $R_k = \mathbb{Z}\langle x_1, \ldots, x_k \rangle$. Then the pair $E_2(R_k) \ltimes R_k^2 \supseteq R_k^2$ has relative property (T). Moreover, for the finite generating set S consisting of all unit elementary matrices in $E_2(R_k) \ltimes R_k^2$ (inside $E_3(R_k)$), there is a following estimate of the relative Kazhdan constant:

$$\mathcal{K}(\mathcal{E}_2(R_k) \ltimes R_k^2, R_k^2; S) > \frac{1}{3\sqrt{2}(\sqrt{k}+3)}.$$

In particular, the statements above hold if R_k is replaced with the commutative ring $A_k = \mathbb{Z}[x_1, \ldots x_k]$.

For the precise estimate, we refer to the original paper [Kas2] of Kassabov.

4.2.2 Bounded generation of Carter–Keller

We define the following concepttion of high importance:

Definition 4.2.4. Let G be a group and S be a subset of G which contains e_G . Then we say G is *boundedly generated* by S if there exists $n \in \mathbb{N}$ such that $G = S^n$, namely, if there exists a universal constant n such that any $g \in G$ can be expressed as a product of n elements in S.

Let $(S_j)_{j \in J}$ are subsets of G containing e_G . Then we say G is boundedly generated by $(S_j)_{j \in J}$ if $S = \bigcup_{j \in J} S_j$ boundedly generates G.

Note that G always boundedly generates G itself, and therefore properties for the generating set S matter in study if bounded generation. Also, we warn in other literture, the bounded generation is defined in the following *confined* sense: J is a finite set, and each S_i is a cyclic group.

The following theorem, due to Carter–Keller [CaKe], plays a fundamental role on bounded generation and its application.

Theorem 4.2.5. (Carter-Keller [CaKe]) Let $m \ge 3$. Then $SL_m(\mathbb{Z})$ is boundedly generated by the set of all elementary matrices. Moreover, every $g \in SL_m(\mathbb{Z})$ can be written as a product of at most

$$v_m := \frac{1}{2}(3m^2 - m) + 36$$

elementary matrices.

4.2. BOUNDED GENERATION AND (T)

Note that they also obtain a similar result for the case of $SL_m(\mathcal{O})$. Here \mathcal{O} is the ring of integers in a local field. For the proof of this theorem, see Section 4.1 in [BHV].

We warn that this result is deep in the following sense. Firstly, the proof employs a special property for \mathbb{Z} , precisely, Dirichlet's arithmetic progression theorem. Secondly, there exists an example of a euclidean domain with respect to which, SL_m is *not* boundedly generated by elementary matrices. For the latter part, see Subsection 4.3.1.

4.2.3 Property (T) for non-algebraic matrix groups

First we deduce property (T) for $\mathrm{SL}_{m\geq 3}(\mathbb{Z})$, without appealing to the fact that they are lattices in $\mathrm{SL}_m(\mathbb{R})$. The keys are relative property (T) (Theorem 4.2.1; Corollary 4.2.2) and bounded generation by elementary matrices (Theorem 4.2.5).

Theorem 4.2.6. (Shalom [Sha1]) Let $m \ge 3$. Then $SL_m(\mathbb{Z})$ has property (T). Moreover, for the finite generating set S consisting of all unit elementary matrices there is an inequality

$$\mathcal{K}(\mathrm{SL}_m(\mathbb{Z}); S) > \frac{1}{30m^2 - 20m + 720}$$

Proof. We present two proofs, one gives the estimate above and the other does not. Both ways have a sprit in common to the proof(s) of Proposition 2.6.6.

(1) The first proof gives an estimate for the Kazhdan constant. Let $G_0 = \operatorname{SL}_2(\mathbb{Z}) \ltimes \mathbb{Z}^2$ and $N = \mathbb{Z}^2$ as abstract groups. Observe that G contains many copies of G_0 in the following sense: for any group $Z_{i,j}$ of elementary matrices with fixed (i, j) $(i \neq j)$ (, which is isomorphic to \mathbb{Z} ,) there exists an embedding $\phi: G_0 \hookrightarrow G$ such that $\phi(N)$ includes $Z_{i,j}$. Set $\mathcal{S} = \bigcup_{i \neq j} Z_{i,j}$, which is the set of all elementary matrices.

Suppose (π, \mathfrak{H}) is a unitary representation and take any $\xi \in S(\mathfrak{H})$. Set $\delta_{\xi} = \sup_{s \in S} \|\xi - \pi(s)\|$. Then by Corollary 4.2.2 and the observation above, we have

for any
$$\gamma \in \mathcal{S}$$
, $\|\xi - \pi(\gamma)\xi\| < 20\delta_{\xi}$.

By employing bounded generation of G by \mathcal{S} (Theorem 4.2.5), we obtain:

for any
$$g \in G$$
, $\|\xi - \pi(g)\xi\| < 20v_m \cdot \delta_{\xi}$

Recall the argument in item (ii) of Remark 2.1.2.

Finally, recall (G, 1) is a Kazhdan pair (see Corollary 2.1.14). Therefore, as soon as there exists $\xi \in S(\mathfrak{H})$ with $\delta_{\xi} \geq (20v_m)^{-1}$, π contains 1_G . This means, $(S, (20v_m)^{-1})$ is a Kazhdan pair for G, as desired. (2) This is the other proof, which shows that G has (FH). Consider an arbitrary cocycle c with unitary coefficient. By Theorem 4.2.1 and an argument in proof (1) (with $S = \bigcup_{i \neq j} Z_{i,j}$), we have the following:

c is bounded on \mathcal{S} .

Finally through bounded generation, it follows c is bounded on G. This implies that c is a coboundary. Therefore G has (FH).

We stated proof (2), because we have known that in general case, (F_B) is stronger than (T_B) . Proof (2) indicates, that if relative property (F_B) together with bounded generation (in appropriate sense) implies property (F_B) .

We note that Kassabov [Kas1] estimated the optimal order of Kazhdan constant for $SL_m(\mathbb{Z})$ with respect to m: it is $O(m^{-1/2})$.

Finally, we state the following application. Recall from Chapter 0 that the symbol S^1 denotes the unit circle.

Theorem 4.2.7. (Shalom [Sha1]) Let $m \geq 3$. Then the loop group of $SL_m(\mathbb{C})$, identified the $SL_m(C(S^1))$ with the topology of uniform convergence on compact subset (it is a non locally compact group), has property (T).

This theorem follows from the observation that $C(S^1)$ contains a dense subring which is finitely generated and satisfies bounded generation by elementary matrices for E_m . For details, we refer to [Sha1].

4.3 Bounded generation for universal lattices

In this section, we state some facts on bounded generation for elementary groups over certain rings.

4.3.1 A result of van der Kallen

A question of high importance is the following: Is a universal lattice boundedly generated by elementary matrices. The affirmative answer will immediately imply property (T) for universal lattices, in the argument in this Section. However, even for the case of k = 1, this question is completely open. In fact, it is open whether $SL_m(\mathbb{Q}[x])$ is boundedly generated by elementary matrices, for $m \geq 3$, although $\mathbb{Q}[x]$ is a euclidean domain and the treatment is much easier than that of $\mathbb{Z}[x]$.

The following theorem, due to W. van der Kallen, is surprising: it states even for euclidean domains, bounded generation of SL_m by elementary matrices can be *false*. **Theorem 4.3.1.** (van der Kallen [vdKa]) Let K a (commutative) field of infinite transcendence degree over a subfield. Then for any $m \ge 3$, $SL_m(K[x])$ is not boundedly generated by elementary matrices.

Here are examples of fields satisfying the condition of this theorem: firstly, any uncountable fields (such as $\mathbb{C}, \mathbb{R}, \mathbb{Q}_p$ here p is a prime); secondly, it follows from a counting argument that there exists a countable field with this property. Also, by theorems on transcendental numbers, the field generated by $e^{\sqrt{p_n}}$'s, where $(p_n)_{n\geq 1}$ is a strictly increasing sequence of primes, has this property.

The proof of van der Kallen is based on the fact that for such field, $K_2(K)$ is not boundedly generated by the Steinberg symbols. We do not go further on algebraic K-theory in this thesis. We remark that I. V. Erovenko [Ero] found a constructive proof of this theorem. Namely, he found a sequence of explicit elements in $SL_m(K[x])$ of which the word lengths (with respect to the set of elementary matrices) tends to infinity.

4.3.2 Vaserstein's bounded generation

As we see in the subsection above, bounded generation by elementary matrices is not known for universal lattices. However certain weak form of this bounded generation has been established by Vaserstein [Vas2]. This result, together with Shalom's machinery (Subsection 5.3) are the keys to proving property (T) for universal lattices:

Theorem 4.3.2. (Vaserstein [Vas2]) Let $A_k = \mathbb{Z}[x_1, \ldots, x_k]$ for any k. Then for any $m \geq 3$, $\mathrm{SL}_m(A_k)$ is boundedly generated by the set of elementaly matrices and the subgroup $\mathrm{SL}_2(A_k)$. Here $\mathrm{SL}_2(A_k) \leq \mathrm{SL}_m(A_k)$ sits in the left upper corner.

Chapter 5

Reduced cohomology and ultraproduct

By taking a closure of the coboundary space $B^1(G; \rho, B)$ (with respect to the Fréchet topology on the cocycle space $Z^1(G; \rho, B)$), we can consider the reduced cohomology $\overline{H}^1(G; \rho, B)$. We see a relation between this concept and ultraproduct of Banach spaces. As an application, we introduce Shalom's machinery, which enables us to approach property (T) for universal lattices. References on reduced cohomology with unitary coefficients are the original paper of Shalom [Sha2], and Chapter 3 of [BHV]. References on ultraproducts on metric spaces are a book of Aksoy–Khamsi [Ak], a book of Bridson–Haefliger [BrHa], and a paper of S. Heinrich [Hei]. For Shalom's machinery, we refer to the original paper [Sha5] of Shalom.

In this chapter, we assume all groups are discrete (and hence countable by σ -compactness assumption) in order to avoid continuity arguments on ultraproduct of actions.

5.1 Ultraproduct of metric spaces

Before proceeding into theory of reduced cohomology, we consider the concept of ultraproducts of metric spaces.

5.1.1 Definition of ultralimits and properties

We begin with the definition of ultrafilters on \mathbb{N} :

Definition 5.1.1. A non-principal ultrafiler ω on \mathbb{N} is a collection of subsets of \mathbb{N} that satisfies the following properties:

- (i) The collection ω satisfies $\omega \not\ni \emptyset$, and $\omega \ni \mathbb{N}$.
- (*ii*) For any $A, B \in \omega, A \cap B \in \omega$.

- (*iii*) For any $A \in \omega$ and $A \subseteq C, C \in \omega$.
- (iv) For any $A \subseteq \mathbb{N}$, either $A \in \omega$ or $A^c \in \omega$. Here A^c means the complement.
- (v) No finite subset of \mathbb{N} is in ω

The existence of non-principal ultrafiler based on Zorn's lemma. Also, each non-principal ultrafilter can be seen as an element of Stone–Čech boundary of \mathbb{N} .

Throughout this chapter, we fix a non-principal ultrafilter and write it ω .

Definition 5.1.2. Let $(a_n)_n$ be a sequence in \mathbb{C} . We say a_n converges to $a \in \mathbb{C}$ with respect to ω , if the following holds:

for any
$$\epsilon > 0$$
, $\{n \in \mathbb{N} : |a_n - a| < \epsilon\} \in \omega$.

In this case we call a the *ultralimit* of (a_n) , and write $a = \lim_{\omega} a_n$.

It is easy to see if an ultralimit exists, then it is unique. Also, ultralimits preserves sums: indeed, suppose $(a_n)_n$, $(b_n)_n$ in \mathbb{C} such that $\lim_{\omega} a_n = a$ and $\lim_{\omega} b_n = b$ for some $a, b \in \mathbb{C}$. Then for any $\epsilon > 0$,

$$\{n \in \mathbb{N} : |(a_n + b_n) - (a + b)| < 2\epsilon\} \supseteq \{n \in \mathbb{N} : |a_n - a| < \epsilon\} \cap \{n \in \mathbb{N} : |b_n - b| < \epsilon\}.$$

This ensures $\lim_{\omega} (a_n + b_n) = a + b$. Also, it is clear that the ultralimit preserves scalar multiplication.

The following is one of the most important features of ultralimits.

Proposition 5.1.3. The ultralimit $\lim_{\omega} a_n$ exists for any bounded sequence $(a_n)_n$ in \mathbb{C} .

Proof. Thanks to the observations above, we may assume $(a_n)_n$ lies in \mathbb{R} . Choose positive M such that $|a_n| \leq M$ for all n. Consider subsets of $\mathbb{N} S_0^+ := \{n : a_n \geq 0\}$ and $S_0^- := \{n : a_n \leq 0\}$. Since $(S_0^+)^c \subseteq S_0^-$, either $S_0^+ \in \omega$ or $S_0^- \in \omega$. We may assume $S_0^+ \in \omega$. Then consider $S_1^+ := \{n : a_n \geq M/2\}$ and $S_1^- := \{n : a_n \leq M/2\}$ and either of these is in ω . By diminishing intervals, we obtain $a \in \mathbb{R}$ such that for any $m \geq 1$, $\{n : |a_n - a| \leq M/2^m\} \in \omega$. This shows $\lim_{\omega} a_n = a$.

Now we restrict ourselves to consider bounded sequence in \mathbb{C} . This corresponds to consider an element of $\ell^{\infty}(\mathbb{N})$. It means, we consider the *ultralimit* with respect to ω as a map $\ell^{\infty}(\mathbb{N}) \to \mathbb{C}$. Note that both spaces $\ell^{\infty}(\mathbb{N})$ and \mathbb{C} are equipped with *-operations (, namely, complex conjugations).

Lemma 5.1.4. The ultralimit with respect to $\omega \lim_{\omega} : \ell^{\infty}(\mathbb{N}) \to \mathbb{C}$ satisfies the following properties:

(i) The map \lim_{ω} is a linear functional.

5.1. ULTRAPRODUCT OF METRIC SPACES

- (ii) The map \lim_{ω} is unital: sends $1_{\mathbb{N}}$ to 1.
- (iii) The map \lim_{ω} is positive: sends nonnegative functions to nonnegative reals.
- (iv) The map \lim_{ω} preserves *-structure: $\lim_{\omega} (\overline{a_n}) = \overline{\lim_{\omega} a_n}$.
- (v) If $(a_n)_n$ is a converging sequence in usual sense, then $\lim_{\omega} a_n = \lim_{n \to \infty} a_n$.
- (vi) The map \lim_{ω} is multiplicative: $\lim_{\omega} (a_n \cdot b_n) = (\lim_{\omega} a_n)(\lim_{\omega} b_n)$.

Proof. Items (i)-(v) are directly shown. For item (vi), let $\lim_{\omega} a_n = a$ and $\lim_{\omega} b_n = b$. Observe

$$a_n b_n - ab = (a_n - a)b_n + a(b_n - b)$$

and $(a_n)_n, (b_n)_n \in \ell^{\infty}(\mathbb{N})$. We have the conclusion.

Therefore, we can regard an ultralimit as a unital positive *multiplicative* functional (*-preserving)

$$\lim : \ell^{\infty}(\mathbb{N}) \to \mathbb{C}$$

which coincides with the usual limit on converging sequences.

5.1.2 Ultraproduct of metric spaces, and scaling limits

Although we treat Banach spaces, for treatments of affine actions, it is convenient to consider Banach space as affine Banach spaces. Therefore, here we define an ultraproduct of metric spaces with base points.

Definition 5.1.5. (i) Let $(X_n, d_n, z_n)_n$ be a sequence of triples of metric spaces, metrics, and base points. We define the ultraproducts of pointed metric spaces $(X_n, d_n, z_n)_n$ with respect to ω as follows: set

$$\tilde{X} := \{ (x_n)_n : x_n \in X_n, \sup_n d_n(x_n, z_n) < \infty \}.$$

We define a metric d_{ω} on X by

$$d_{\omega}((x_n)_n, (y_n)_n) := \lim_{n \to \infty} d_n(x_n, y_n)$$

(observe that by triangle inequality, $(d_n(x_n, y_n))_n \in \ell^{\infty}(\mathbb{C})$). Finally, set a equivalence relation

$$(x_n)_n \simeq (y_n)_n$$
 if $d_{\omega}((x_n)_n, (y_n)_n) = 0.$

The ultraproduct of $(X_n, d_n, z_n)_n$ with respect to ω is defined as the space \tilde{X}/\simeq , equipped with the canonical quotient metric d_{ω} and with the canonical base point. We write this as $(X_{\omega}, d_{\omega}, z_{\omega})$, or $\lim_{\omega} (X_n, d_n, z_n)$.

(*ii*) Let (X, d) be a single metric space. Then we say a *scaling limit* of (X, d) is a metric space of the following form

$$\lim_{\dots} (X, a_n \cdot d, z_n),$$

where $(a_n)_n$ is a positive increasing sequence tending to infinity as $n \to \infty$.

(*iii*) Let \mathcal{X} be a class of metric spaces. Then we say \mathcal{X} is stable under ultraproducts if any sequence $(X_n, d_n)_n$ in \mathcal{X} and base points $(z_n)_n$ $(z_n \in X_n)$, the ultrapower is an element in \mathcal{X} .

We say \mathcal{X} is stable under scaling limits if any element (X, d) in \mathcal{X} and any base points $(z_n)_n$ $(z_n \in X)$ and any sequence $(a_n)_n$ of positive reals with $a_n \nearrow \infty$, the resulting scaling limit is an element in \mathcal{X} .

We note that if we start from affine Banach spaces, then the resulting ultraproduct is also an affine Banach space (in particular, it is closed). Also, in that case, the choices of base points does not affect the resulting Banach space as affine spaces. In general, ultraproduct of complete metric spaces is also complete.

Definition 5.1.6. Let $(B_n, \|\cdot\|_n, 0)_n$ be a sequence of Banach spaces (with the base point as origin). Let $(T_n)_n, T_n \in \mathbb{B}(B_n)$ be a sequence of bounded linear operators which satisfies

$$\sup \|T_n\|_{\mathbb{B}(B_n)} < \infty.$$

Then the *ultraproduct of* $(T_n)_n$ with respect to ω , written as $T_{\omega} = \lim_{\omega} T_n$, is defined as the operator on $\lim_{\omega} (B_n, \|\cdot\|_n, 0)_n$ induced by the following operation:

$$(\xi_n)_n \mapsto (T_n\xi_n)_n.$$

In the definition above, it is a routine to check that this is well-defined, and that there is a following formula:

$$\|\lim_{\omega} T_n\|_{\mathbb{B}(B_{\omega})} = \lim_{\omega} \|T_n\|_{\mathbb{B}(B_n)}.$$

Example 5.1.7. (i) Consider a single Banach space B with a fixed norm $\|\cdot\|$ and a base point 0. We claim that whenever B is infinite dimensional, the ultraproduct $\lim_{\omega} B = \lim_{\omega} (B, \|\cdot\|, 0)$ is not separable. Indeed, from infinite dimensionality, there exists a sequence $(\xi_n)_{n\geq 0}$ in S(B) such that for any $n \neq$ $m, \|\xi_n - \xi_m\| \geq 1/2$. For positive real number t, set [t] be the maximum of integers which is at most t. For any real number $\alpha > 0$, we consider the limit $\xi_{\omega}^{\alpha} := \lim_{\omega} (\xi_{[n^{\alpha}]})_n$. Then it is easy to see the following: for any distinct $\alpha, \beta > 0, \xi_{\omega}^{\alpha}$ and ξ_{ω}^{β} has difference at least 1/2. This implies that $\lim_{\omega} B$ is not separable. In particular, if the original B is separable, then B and $\lim_{\omega} B$ are never isomorphic.

5.1. ULTRAPRODUCT OF METRIC SPACES

- (*ii*) The class \mathcal{H} , consisting of all Hilbert spaces, is stable under ultraproducts. This follows from the fact that (with setting all base points as origins), the ultralimit of inner products is again an inner product, after separation. For the class \mathcal{L}_p , it is unclear whether it is stable under ultraproducts. See Subsection 5.1.3.
- (*iii*) If we consider a class C of Banach spaces, then scaling limits inside C are special case of ultrapowers inside C. Therefore, if C is stable under ultrapowers, then it is stable under scaling limits. We note that the converse in *not* true. We will see it below.

Firstly, the class $[\mathcal{H}]$ is stable under scaling limits. Indeed, any single element $X \in [\mathcal{H}]$ has a compatible norm to that of a Hilbert space \mathfrak{H} . Thanks to the inner product on \mathfrak{H} , one can show $\lim_{\omega} X$ also has a compatible Hilbert norm (recall that $\lim_{\omega} \mathfrak{H}$ is a Hilbert space and Definition 5.1.6). Secondly, we see $[\mathcal{H}]$ is *not* stable under ultraproducts. Indeed, in this case we can choose a sequence $(X_n)_n$ whose norm ratios to norms of Hilbert spaces tends to infinity. Hence the resulting ultrapower no longer has a compatible norm in general.

Among metric spaces (with base points), we consider the *ultraproduct* of isometric actions of a (fixed) group. Recall that we always assume groups are discrete in this chapter. Therefore, there is no worry about continuity on resulting isometric action.

Definition 5.1.8. Let G a group. Let (X_n, d_n, z_n) be a sequence of metric spaces with base points. Let $(\alpha_n)_n$ be a sequence of isometric G-actions on (X_n, d_n, z_n) (, namely, for each $n \alpha_n \colon G \to \text{Isom}(X_n)$, where Isom(X) denotes the group of isometries on X). Suppose $(\alpha_n)_n$ satisfies the following condition:

for any
$$g \in G$$
, $\sup_{n} d_n(z_n, \alpha_n(g)z_n) < \infty$.

Then we define the *ultraproduct of* (α_n) with respect to ω , written as $\alpha_{\omega} = \lim_{\omega} \alpha_n$ as the isometric action induced by the following map: for each $g \in G$

$$(x_n)_n \mapsto (\alpha(g)x_n)_n.$$

Remark 5.1.9. Here are remarks on the definition above.

- 1. If $(X_n, d_n)_n = (B_n, \|\cdot\|_n)$ is a sequence of affine Banach spaces and α_n 's are *affine* isometric action, then the ultrapower α_{ω} is also affine on the affine Banach space B_{ω} .
- 2. If the group G is finitely generated and S is a finite generating set, then it suffices to check the following condition on $(\alpha)_n$ for the well-definedness of α_{ω} :

for any
$$s \in S$$
, $\sup_{n} d_n(z_n, \alpha_n(s)z_n) < \infty$.

In next section, we focus on the case of finitely generated groups.

5.1.3 Some facts on ultraproducts

We collect some facts needed for our theory. Firstly, we state stability results under ultraproducts for certain classes, which is due to Heinrich [Hei]. Recall we always assume p is a real in $(1, \infty)$, as in Chapter 0.

Theorem 5.1.10. (*Heinrich* [Hei]) *The following classes of Banach spaces are stable under ultraproducts:*

- (i) The class \mathcal{H} .
- (ii) For any $M \ge 1$, the subclass \mathcal{H}_M of $[\mathcal{H}]$ which is defined as follows: \mathcal{H}_M is the class of all Banach spaces which admits compatible norms to those of Hilbert spaces, with norm ratio $\le M$.
- (*iii*) The class \mathcal{L}_p .

In particular, these classes are stable under scaling limit. The class $[\mathcal{H}]$ itself is stable under scaling limit.

Proof. The cases of item (i) and item (ii) are confirmed in Example 5.1.7. The proof of the case of item (iii) is deep, and this is what Heinrich did. He appealed to the Theorem of Bohnenblust and Nakano that a Banach space B is in \mathcal{L}_p if (and only if) B is an *abstract* L^p space. Here an abstract L^p space is a Banach lattice E which satisfies the following condition:

for any
$$\xi, \eta \in E$$
 with $\xi \wedge \eta = 0$, $\|\xi + \eta\|^p = \|\xi\|^p + \|\eta\|^p$.

Here $\xi \wedge \eta$ is the greatest lower bound, as in the definition of Banach lattices. Heinrich shown that the class of all Banach lattices is stable under ultraproducts, and that the condition above is also stable. Therefore, \mathcal{L}_p is stable under ultraproducts. For details, we refer to [Hei] and Chapter 1 and Chapter 5 of a book of H. E. Lacey [Lac].

The second half is obvious by Example 5.1.7.

Next, we need some stability result for modulus of convexity. Recall from Definition 3.1.1 that the modulus of convexity of a Banach space $(E, \|\cdot\|)$ is defined by the following formula: for $0 < \epsilon < 2$,

$$d_{\|\cdot\|}(\epsilon) = \inf\left\{1 - \frac{\|\xi + \eta\|}{2} : \xi, \eta \in B(E), \text{ and } \|\xi - \eta\| \ge \epsilon\right\}.$$

Theorem 5.1.11. Let $(B_n, \|\cdot\|_n)_n$ be a sequence of Banach spaces. Suppose for any $0 < \epsilon < 2$, there exists $t_{\epsilon} > 0$ such that

for all
$$n$$
, $d_{\|\cdot\|_n}(\epsilon) \ge t_{\epsilon}$.

Then there is an inequality: for any $0 < \epsilon < 2$,

 $d_{\|\cdot\|_{\omega}}(\epsilon) \ge t_{\epsilon}.$

In particular, in the setting above, the ultraproduct B_{ω} is uniformly convex.

For the proof, see (and imitate) the proof of Theorem 4.4 of a book in Aksoy–Khamsi [Ak].

5.2 Reduced cohomology

In this section, we define reduced (1-)cohomology with Banach coefficients. We prove one proposition of high importance, so-called the *Proposition of scaling limits*. This proposition gives a deep corollary on reduced cohomology of a *finitely generated* groups.

5.2.1 Definition and uniform actions

Recall groups in this chapter are assumed to be discrete (and hence countable). Consider an isometric G-representation ρ on a Banach space B. In the proof of Guchardet's direction in Theorem 2.4.13 or its generalization 3.2.9, we consider the Fréchet topology on the cocycle space $Z^1(G; \rho, B)$ induced by the separating seminorms

$$Z^1(G;\rho,B) \ni c \mapsto \sup_{s \in S} \|c(s)\|,$$

where S moves in a countable family of finite subsets of G, whose union is G. In general the subspace of coboundaries $B^1(G; \rho.B)$ is *not* closed with respect to this topology, and that was a key to proving these theorems.

Definition 5.2.1. Let G be a group. Let ρ be an isometric G-representation on a Banach space B.

- (i) The space $\overline{B}^1(G; \rho, B)$ is the closure of $B^1(G; \rho, B)$ in $Z^1(G; \rho, B)$ with respect to the Fréchet topology defined as the paragraph above.
- (*ii*) The reduced (1-)cohomology $\overline{H}^{1}(G; \rho, B)$ is defined as the following quotient space:

$$\overline{H}^{1}(G;\rho,B) = Z^{1}(G;\rho,B)/\overline{B}^{1}(G;\rho,B)$$

In [Sha2], Shalom defined the notion of *uniform* affine isometric action on a Hilbert space. We extend this notion to isometric group action on a metric space.

Definition 5.2.2. Let G be a group and (X, d) be a metric space. Let $\alpha \colon G \to \text{Isom}(X)$ be an isometric G-action on X.

(i) We define the *displacement function* associated with S by the following formula:

$$\delta_S \colon X \to \mathbb{R}_{\geq 0}; \delta_S(x) := \sup_{s \in S} d(x, \alpha(s)x)$$

(ii) We say α is a *uniform action* if there exists a finite subset $S \subseteq G$ and a positive number $\epsilon > 0$ such that for any $x \in X$,

$$\delta_S(x) \ge \epsilon$$

holds.

Remark 5.2.3. In the definition above, α is not uniform if and only if for any finite subset $S \subseteq G$ and any $\epsilon > 0$ there exists $x \in B$ with

$$\delta_S(x) < \epsilon.$$

Suppose G is finitely generated. Then for above condition, it is easy to see that we only have to consider the case of that S is a generating set. Then since S is generating, we have the following: α is not uniform if and only if for some (equivalently, any) finite generating subset $S \subseteq G$, there exists $(x_n) x_n \in B$ with

$$\lim_{n \to \infty} \delta_S(x_n) = 0.$$

In the setting above, choice of finite generating sets S is not essential. Therefore if G is finitely generated, then we also write a displacement function as δ for short.

Lemma 5.2.4. Let G be a group and ρ be an isometric G-representation on a Banach space B. Then for any ρ -cocycle c, the following are equivalent:

- (i) The cocycle c belongs in $\overline{B}^1(G; \rho)$.
- (ii) The affine isometric action associated with (ρ, c) is not uniform.

Proof. Let α be isometric action associated with (ρ, c) and let $S \subseteq G$ be a finite subset. Then for any $\xi \in B$,

$$\delta_S(\xi) = \sup_{s \in S} \|(\xi - \rho(s)\xi) - c(s)\| = \sup_{s \in S} \|c(s) - (\xi - \rho(s)\xi)\|.$$

Recall that any coboundary $b \in B^1(G; \rho)$ has the form $b(g) = \xi - \rho(g)\xi$. Now by definition the equivalence holds true.

5.2.2 Proposition of scaling limits

The following proposition is so-called the *Proposition of scaling limits*, and plays a key role in this chapter. Here we note the assumption of *finite generation* is necessary.

Proposition 5.2.5. (Proposition of scaling limits) Let G be a finitely generated group. Let (X, d) be a complete metric space, and $\alpha \colon G \to \text{Isom}(X)$ be an isometric action.

Suppose α is not uniform, but has no *G*-fixed point. Then there exist a sequence (z_n) in X and a sequence of positive reals (b_n) with $b_n \nearrow \infty$ such that the resulting isometric action from the scaling limit $\lim_{\omega} (X, b_n \cdot d, z_n)$ is uniform.

Note that the existence of scaling limit action is a part of the statement.

The proof which we will present here seems to be due to R. Schoen. See also, [Gro4] and [Gro6]. For the proof, we need the following lemma. From the assumption of *finite generation* for G, we take a finite generating set $S \subseteq G$ and consider the displacement function $\delta = \delta_S \colon X \to \mathbb{R}$ for the action α (recall Remark 5.2.3). By assumption, δ satisfies the following two conditions:

(1) There exists a sequence $(x_n)_n$ $(x_n \in X)$ such that

$$\lim_{n \to \infty} \delta(x_n) = 0$$

(2) There does not exist x satisfying $\delta(x) = 0$.

For the proof of the proposition above, we need the following lemma:

Lemma 5.2.6. Keep the setting as in Proposition 5.2.5, and let S and δ be as in above.

Let t > 0. Then for any $x \in X$, there exists $x' \in B(x, t\delta(x))$ such that the following holds:

for any
$$x'' \in B(x', \frac{t}{2}\delta(x')), \quad \delta(x'') \ge \frac{1}{2}\delta(x')$$
 holds.

Here B(x,r) denotes the closed ball centered at x of radius r.

We make a remark that for the proof of the lemma above, condition (2) in above is only needed and condition (1) is not employed.

Proof. (Lemma 5.2.6) Suppose the contrary. Then there exists $x \in X$ such that for any $x' \in B(x, t\delta(x))$,

for some
$$x'' \in B(x, \frac{t}{2}\delta(x')), \ \delta(x'') < \frac{1}{2}\delta(x')$$
 holds.

Firstly, we apply this to the case of x' = x and obtain $x_1 \in B(x, t/2 \cdot \delta(x))$ such that $\delta(x_1) < 1/2 \cdot \delta(x)$. Observe that

$$d(x_1, x) \le \frac{t}{2}\delta(x) < t\delta(x).$$

Therefore, secondly we apply the assumption above to the case of $x' = x_1$ and get $x_2 \in B(x_1, t/4\delta(x))$ (recall the condition above) such that

$$\delta(x_2) < \frac{1}{2} \cdot \delta(x_1) < \frac{1}{4} \cdot \delta(x).$$

Also observe that

$$d(x_2, x) \le d(x_2, x_1) + d(x_1, x) < \frac{3t}{4}\delta(x) < t\delta(x).$$

By iterating this procedure, we obtain a sequence $(x_n)_{n\geq 1}$ in X such that the following two condition are satisfied:

- For any $n, x_{n+1} \in B(x_n, (1/2)^n t \delta(x)).$
- For any $n, \delta(x_n) < (1/2)^n \delta(x)$.

This sequence is a Cauchy sequence, and from the completeness of X, there exists a convergent point x_{∞} . Then $\delta(x_{\infty}) = 0$, and this contradicts condition (2) in Proposition 5.2.5.

Proof. (Proposition 5.2.5) By condition (1), there exists a sequence (y_n) in Y such that $\delta(y_n) \searrow 0$ as $n \to \infty$. Choose a sequence (c_n) of positive reals satisfying $c_n \nearrow \infty$ and $c_n \delta(y_n) \nearrow 0$ (for instance, $c_n = (\delta(y_n))^{-1/2}$). Then by Lemma 5.2.6, for each n there exists $z_n \in B(y_n, c_n \delta(y_n))$ such that the following holds:

for any
$$x_n \in B(z_n, \frac{c_n}{2}\delta(z_n)), \quad \delta(x_n) \ge \frac{1}{2}\delta(z_n)$$
 holds.

Set a sequence (b_n) of positive reals with $c_n \nearrow \infty$ by

$$b_n = \frac{2}{\delta(z_n)}.$$

We will show in below that the scaling limit of the action α with respect to the sequence $(X, b_n \cdot d, z_n)$ exists, and this scaling limit action α_{ω} is uniform. Firstly,

$$\sup_{n} \sup_{s \in S} b_n d(z_n, \alpha(s)z_n) = \sup_{n} b_n \cdot \delta(z_n) = 2 < \infty.$$

Hence it follows the scaling limit isometric action α_{ω} exists (see also Remark 5.1.9). Secondly, we see α_{ω} is uniform. For any sequence (x_n) in X which satisfies $\sup d(x_n, z_n) < \infty$, exists sufficiently large n_x such that the following holds:

for any
$$n \ge n_x$$
, $d(x_n, z_n) \le \frac{c_n}{2}\delta(z_n)$

(this is because $c_n \nearrow \infty$). Therefore, if we consider the displacement function (associated with S) on $(X, b_n \cdot d)$ and write it δ_n , then we have

for any
$$n \ge n_x$$
, $\delta_n(x_n) \ge 1$.

This means for $n \ge n_x$, there exists $s_n \in S$ such that $d_n(x_n, \alpha(s)x_n) \ge 1$, where we set $d_n = b_n \cdot d$. For $s \in S$, set

$$I(s) = \{n : d_n(x_n, \alpha(s)x_n) \ge 1\},\$$

and consider $I(S) = \bigcup_{s \in S} I(s)$. Then I(S) includes $\{n : n \geq n_x\}$ and hence is in ω . Therefore $\lim_{\omega} \chi_{I(S)} = 1$, where χ is characteristic unction (on \mathbb{N}). Here we consider the unital positive *-preserving linear functional $\lim_{\omega} : \ell^{\infty}(\mathbb{N}) \to \mathbb{C}$, as we argued in Subsection 5.1.1. Since $\chi^2 = \chi$ for any characteristic function and \lim_{ω} is multiplicative (see Lemma 5.1.4), $\lim_{\omega} \chi_{I(s)}$ is either 0 or 1. Finally, by the inequality

$$\lim_{\omega} \sum_{s \in S} \chi_{I(s)} \ge \lim_{\omega} \chi_{I(S)},$$

there exists $s \in S$ such that $\lim_{\omega} \chi_{I(s)} = 1$. With this s, we have

$$d_{\omega}((x_n), (\alpha(s)x_n)) \ge 1.$$

This shows that for any $x_{\omega} \in X_{\omega}$,

$$\delta_{\omega}(x_{\omega}) := \sup_{s \in S} d_{\omega}(x_{\omega}, \alpha_{\omega}(s)x_{\omega}) \ge 1.$$

Hence α is uniform.

In the view of Definition 5.2.1 and Lemma 5.2.4, we establish the following theorem.

Theorem 5.2.7. Suppose C be a class of Banach spaces which is stable under scaling limit. Then the following two conditions for a finitely generated group G are equivalent:

- (i) The group G has $(F_{\mathcal{C}})$. Namely, for any isometric G-representation ρ on $B \in \mathcal{C}$, $H^1(G; \rho, B) = 0$.
- (ii) The group G satisfies the reduced cohomology vanishing with all isometric coefficients in C. That means, for any isometric G-representation ρ on $B \in C$, $\overline{H}^1(G; \rho, B) = 0.$

Note that a priori condition (ii) is much weaker than condition (i) in general setting. Also, we note that even among finitely generated groups, $\overline{H}^1(G;\rho,B) = 0$ does not imply $H^1(G;\rho,B) = 0$ for a single pair (ρ,B) .

We note Shalom [Sha2] shows in fact we can consider the cases of locally compact and *compactly generated* groups with $C = \mathcal{H}$ (unitary coefficients), in Theorem 5.2.7. His argument is based on study of conditionally negative definite kernels, and is completely different from the argument above. See [Sha2] or Section 3.2 in [BHV].

5.2.3 Applications

There are some direct applications of Theorem 5.2.7:

Theorem 5.2.8. The following two conditions for a finitely generated group G are equivalent:

- (i) The group G has (T).
- (ii) For any irreducible unitary G-representation π , $H^1(G; \pi) = 0$.

This follows from the fact that reduced cohomology is decomposable according to (even for infinite) L^2 -direct sums.

Next application is on finite presentability of Kazhdan groups. As we have seen in Remark 2.2.4 it is not always true, Shalom, nevertheless, show the following:

Theorem 5.2.9. (Shalom [Sha2]) Suppose a group G has (T). Then there exists a finitely presented group \tilde{G} with (T) and normal subgroup $H \leq \tilde{G}$ such that \tilde{G}/H is isomorphic to G.

Proof. By assumption, G is finitely generated. Let S be a finite generating set of G and set n = |S|. Let

$$\phi \colon F_n \twoheadrightarrow G$$

be a homomorphism and set N be the kernel. Let w_1, w_2, \ldots be an enumeration of the elements in N. For $k \in \mathbb{N}$, let N_k is the normal closure in F_n of the group generated by w_1, \ldots, w_k . Set $\Gamma_k = F_k/N_k$. Thus we have

$$F_n \twoheadrightarrow \Gamma_1 \twoheadrightarrow \Gamma_2 \twoheadrightarrow \cdots \twoheadrightarrow G.$$

We claim there exists $k \in \mathbb{N}$ such that Γ_k has $(T)(\Leftrightarrow (FH))$. Suppose the contrary. Then by Proposition 5.2.5, for any Γ_k there exists an affine isometric action α_k of Γ_k on a Hilbert space which is uniform. Consider the ultralimit of α_k (with an appropriate scaling), then this yields a uniform action on a Hilbert space of G. Indeed, consider the ultraproduct of affine isometric actions of F_n . Then the kernel of resulting action contains N so that this action factors through an action of G. This contradicts (FH) for G.

5.3 Shalom's machinery

We introduce a powerful machinery, invented by Shalom [Sha5]. Shalom employed this machinery to establish property (T) for universal lattices.

In [Mim1], the author extended the conception of the *Shalom property*, which is found in Definition 12.1.13 of book [BrOz] of Brown–Ozawa.

Definition 5.3.1. (Definition 5.4 [Mim1]) Let *B* be a Banach space and *G* be a finitely generated group. A triple of subgroups (H, N_1, N_2) of *G* is said to have the *Shalom property for* (F_B) if all of the following four conditions hold:

- (i) The group G is generated by N_1 and N_2 together.
- (*ii*) The subgroup H normalizes N_1 and N_2 .
- (*iii*) The group G is boundedly generated by H, N_1 , and N_2 (in the sense in Definition 4.2.4).
- (*iv*) For both $i \in \{1, 2\}$, $N_i \leq G$ has relative (F_B).

The following theorem is called Shalom's machinery in a paper [Mim1] of the author. Shalom [Sha5] shows in the case of $\mathcal{C} = \mathcal{H}$, and it is extended to the following general case by the author [Mim1].

Theorem 5.3.2. ([Sha5]; Shalom's Machinery, Theorem 5.5 [Mim1]) Let C be a class of superreflexive Banach spaces which is stable under ultraproducts. Let G be a finitely generated group with finite abelianization. Suppose there exist subgroups H, N_1 , and N_2 of G such that (H, N_1, N_2) has the Shalom property for (F_c) . Then G has property (F_c) .

We note that in this theorem, we need the stability of the class under ultraproducts, not only under scaling limits.

For the proof below, recall the canonical decomposition of a superreflexive Banach space B as

$$B = B^{\rho(G)} \oplus B_{\rho(G)}' = B_0 \oplus B_1.$$

associated with an isometric representation ρ of a group G (Proposition 3.1.12). Also, for a given affine isometric G-action α with linear part ρ , there is a canonical decomposition of α associated with the decomposition of B above. More precisely, it is constructed in the following manner: consider the cocycle part c of α , and decompose $c = c_0 + c_1$ according to the decomposition $B = B_0 \oplus B_1$. Namely, for $i = 0, 1, c_i$ is the image of c by the projection from B onto B_i associated with the decomposition $B = B_0 \oplus B_1$. By $\rho(G)$ -invariance of these spaces, each c_i is also a ρ -cocycle on B_i . The decomposition $\alpha = \alpha_0 + \alpha_1$ is the associated affine isometric action with the decomposition $c = c_1 + c_2$ of the cocycle.

Proof. Suppose the contrary. Then by Theorem 5.2.7, there must exist an affine isometric action α_0 on some $B_0 \in \mathcal{C}$ which is uniform. For simplicity, we may assume that B_0 is uc by choosing an appropriate norm. Fix a finite generating set S of G. We define \mathcal{A} as the class of all pairs (α, E) of an affine isometric G-action (on E) and a uc Banach space which satisfy the following conditions:

(a) For any $\xi \in E$, $\sup_{s \in S} \|\xi - \alpha(s)\xi\|_E \ge 1$.

(b) For all $0 < \epsilon < 2$, $d_{\|\cdot\|_E}(\epsilon) \ge d_{\|\cdot\|_0}(\epsilon)$. Here $d_{\|\cdot\|}$ is the modulus of convexity, and $\|\cdot\|_0$ be the (uc) norm of B_0 .

Firstly, note that this class \mathcal{A} is non-empty, because (α_0, B_0) , with an appropriate rescaling, is an element. Secondly, we claim \mathcal{A} is stable under ultraproducts. Indeed, condition (b) is stable by Theorem 5.1.11. To see condition (a) is also stable, imitate the last part of the proof of Proposition 5.2.5.

Next we define a real number D by the following formula:

$$D = \inf\{\|\xi^1 - \xi^2\| : (\alpha, E) \in \mathcal{A}\},\$$

where for $i \in \{1, 2\}$, ξ^i moves through all $\alpha(N_i)$ -fixed points in E. By condition (iv) of the Shalom property, the set is non-empty and hence D is well-defined. By taking an ultraproduct, one can show that D is in fact a minimum. Let ξ^1_{∞} and ξ^2_{∞} be vectors which attain the minimum D. Also let $(\alpha_{\infty}, E_{\infty}) \in \mathcal{A}$ be the associated affine isometric G-action and ρ_{∞} be the linear part of α_{∞} .

Decompose the action α_{∞} into $\alpha_{\infty,\text{triv}}$ and α_{∞}' , where the former takes values in $E_{\infty}^{\rho_{\infty}(G)}$ and the latter tales values in $E'_{\infty, \rho_{\infty}(G)}$. Firstly, we consider orbits by $\alpha_{\infty,\text{triv}}$. This part is easy: by the assumption of finite abelianization, $\alpha_{\infty,\text{triv}}$ is indeed a trivial action. Hence every $\alpha_{\infty,\text{triv}}(G)$ orbit is bounded. This is the first goal in this proof.

Next, we deal with orbits by the action α_{∞}' . For each i = 1, 2, we decompose the vector ξ_{∞}^i according to the decomposition

$$E = E_{\infty}^{\rho_{\infty}(\Gamma)} \oplus E_{\infty,\rho_{\infty}(\Gamma)}'.$$

and write as $\xi_{\infty}^{i} = \xi_{\infty,\text{triv}}^{i} + \xi_{\infty}^{i'}$. Firstly, we will see in below that each $\xi_{\infty}^{i'}$ is an $\alpha_{\infty}'(H)$ -fixed point. To see this, let $h \in H$ be any element. For each i = 1, 2, set $\eta_{\infty}^{i} = \alpha_{\infty}(h)\xi_{\infty}^{i}$. Since α_{∞} is an isometric action, we have

$$\|\eta_{\infty}^{1} - \eta_{\infty}^{2}\| = \|\xi_{\infty}^{1} - \xi_{\infty}^{2}\| = D.$$

We now claim that for each $i = 1, 2, \eta_{\infty}^{i}$ is an $\alpha_{\infty}(N_{i})$ -fixed point. This follows from condition (*ii*) of the Shalom property because for any $l_{i} \in N_{i}$,

$$\alpha_{\infty}(l_i)\alpha_{\infty}(h)\xi_{\infty}^i = \alpha_{\infty}(h)\alpha_{\infty}(h^{-1}l_ih)\xi_{\infty}^i = \alpha_{\infty}(h)\xi_{\infty}^i.$$

For i = 1, 2, set ζ_{∞}^{i} as the midpoint between ξ_{∞}^{i} and η_{∞}^{i} , namely,

$$\zeta_{\infty}^{i} = \frac{\xi_{\infty}^{i} + \eta_{\infty}^{i}}{2}.$$

Since α_{∞} is affine, for each $i = 1, 2, \zeta_{\infty}^{i}$ is also an $\alpha_{\infty}(N_{i})$ -fixed point. By the triangle inequality, we have

$$\|\zeta_{\infty}^{1} - \zeta_{\infty}^{2}\| \le \frac{\|\xi_{\infty}^{1} - \xi_{\infty}^{2}\| + \|\eta_{\infty}^{1} - \eta_{\infty}^{2}\|}{2} = D.$$

5.4. UNIVERSAL LATTICES HAVE (T)

Recall that D is a minimum and the Banach space E_{∞} is uniformly convex (in particular strictly convex). Therefore, the inequality above forces that the four points ξ_{∞}^1 , η_{∞}^1 , η_{∞}^2 , ξ_{∞}^2 form a rectangle with this order. Namely, one has

$$\xi^1_\infty - \eta^1_\infty = \xi^2_\infty - \eta^2_\infty$$

Set this vector Ω .

Since for each $i = 1, 2, \xi_{\infty}^{i}$ and η_{∞}^{i} are $\alpha_{\infty}(N_{i})$ -fixed, the difference $\xi_{\infty}^{i} - \eta_{\infty}^{i}$ is $\rho_{\infty}(N_{i})$ -invariant. By condition (1) of the Shalom property, the equality above implies Ω is a $\rho_{\infty}(G)$ -invariant vector. However, this forces $\Omega = 0$. Indeed, recall the decomposition

$$\xi^i_{\infty} = \xi^i_{\infty, \text{triv}} + {\xi^i_{\infty}}'.$$

Then from the decomposition of α_{∞} into $\alpha_{\infty,\text{triv}}$ and α_{∞}' , we have the following: for each i = 1, 2

$$\eta_i = \xi^i_{\infty, \text{triv}} + \alpha_{\infty}'(h) {\xi^i_{\infty}}'$$

Here we use the observation above that $\alpha_{\infty,\text{triv}}$ is a trivial action. This implies for each i = 1, 2,

$$\Omega = {\xi_{\infty}^{i}}' - {\alpha_{\infty}}'(h) {\xi_{\infty}^{i}}' \in E'_{\infty, \rho_{\infty}(G)}$$

However the space $E'_{\infty,\rho_{\infty}(G)}$ has no non-zero $\rho_{\infty}(G)$ -invariant vectors by construction. Hence Ω must be zero. Again by the equality above, this implies each ξ^{i}_{∞} ' is $\alpha_{\infty}'(H)$ -fixed.

Secondly, the existence of $\alpha_{\infty}'(H)$ -fixed points implies that every $\alpha_{\infty}'(H)$ -orbit is bounded. Condition (4) already shows for i = 1, 2 that every $\alpha_{\infty}'(N_i)$ -orbit is bounded. With the use of condition (*iii*) (bounded generation), this implies that every $\alpha_{\infty}'(G)$ -orbit is bounded. This is the second goal in this proof.

Finally, boundedness of $\alpha_{\infty,\text{triv}}(G)$ -orbits and $\alpha_{\infty}'(G)$ -orbits means that indeed every $\alpha_{\infty}(G)$ -orbit is bounded. This implies that α_{∞} has a *G*-fixed point, and in particular α_{∞} is *not* uniform. This contradicts the construction of $(\alpha_{\infty}, E_{\infty})$. This ends our proof.

5.4 Universal lattices have (T)

We restate Theorem 1.0.1, a deep theorem due to Shalom and Vaserstein.

Theorem 5.4.1. (Shalom [Sha5], Vaserstein [Vas2]) Universal lattice

$$G = \mathrm{SL}_{m \ge 3}(\mathbb{Z}[x_1, \dots, x_k])$$

has property (T).

Therefore, for any commutative finitely generated ring A, $E_{m\geq 3}(A)$ has property (T).

Proof. We have prepared all needed facts for the proof.

Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Set $G = \mathrm{SL}_m(A)$, $H \cong \mathrm{SL}_{m-1}(A)$, and $N_1, N_2 \cong A^{n-1}$. Here in G we realize H as in the left upper corner (, namely, the $((1-(m-1)) \times (1-(m-1)))$ -th parts), realize N_1 as in the $((1-(m-1)) \times m)$ -th unipotent parts, and realize N_2 as in the $(m \times (1-(m-1)))$ -th unipotent parts. Then we claim that Shalom's machinery applies with $\mathcal{C} = \mathcal{H}$.

Indeed, (H, N_1, N_2) is has the Shalom property for (FH). Conditions (i) and (ii) are confirmed directly. Condition (iv) follows from Theorem 4.2.3 (also recall the proof of Proposition 2.6.6). Condition (iii) follows from Theorem 4.3.2, a deep result of Vasestein on bounded generation. Also G has trivial abelianization (see Remark 4.1.2).

Thus we have established property (FH), which is equivalent to (T).

We have one historical remark. Shalom's original argument in [Sha5] uses stable range condition for the ring $\mathbb{Z}[x_1, \ldots, x_k]$ (recall Definition 4.1.4 and Proposition 4.1.5), because at that moment Vaserstein's bounded generation had not been proven. Therefore, in [Sha5], there is some condition on m in terms of the number of generators k.

Remark 5.4.2. We note that at the moment, Shalom's machinery does not apply to the case of noncommutative universal lattices. The obstruction in applying this is the lack of bounded generation theorem (condition (iii)). It is not known whether an analogue of Vaserstein's bounded generation holds for noncommutative universal lattices. Nevertheless Ershov–Jaikin-Zapirain have established property (T) for noncommutative universal lattices [ErJa].

Corollary 5.4.3. Let $m \geq 3$.

Let A be a commutative finitely generated ring with a finite generating set S. Set S be the following finite generating set of $E_m(A)$:

$$\mathcal{S} = \{ E_{i,j}(s) : 1 \le i \le m, 1 \le j \le m, i \ne j; s \in S \}.$$

Then there exists a constant $\kappa_{m,k} > 0$, only depending on m and k = |S| such that the following estimate for the Kazhdan constant holds true:

$$\mathcal{K}(\mathcal{E}_m(A);\mathcal{S}) \ge \kappa_{m,k}$$

The Kazhdan constant for universal lattices is estimated in Theorem 12.1.14 in [BrOz] (although they considered the case of $SL_3(\mathbb{Z})$, their proof applies to general case). Also, Ershov–Jaikin-Zapirain [ErJa] has given estimation even for noncommutative universal lattices.

Chapter 6

Property $(\mathbf{F}_{\mathcal{L}_p})$ and $(\mathbf{F}_{[\mathcal{H}]})$ for universal lattices

This part is one of the main parts in this thesis. In this chapter, we shall establish property $(F_{\mathcal{L}_p})$ (for all $p \in (1, \infty)$) and property $(F_{[\mathcal{H}]})$ for universal lattices with degree at least 4 (this is Theorem A). There is one key trick to deduce them, and we will see that in Section 6.2. The main reference is the original paper [Mim1] of the author.

Here we sketch the philosophy of the proof of Theorem A. In the view of Shalom's machinery (Theorem 5.3.2), to establish property ($F_{\mathcal{C}}$) for a class of superreflexive Banach spaces being stable under ultraproducts, it suffices to establish *relative property* ($F_{\mathcal{C}}$) for the pair $E_2(A) \ltimes A^2 \supseteq A^2$ (here $A = \mathbb{Z}[x_1, \ldots, x_k]$). Therefore, we consider the following two steps:

- Step1: establish relative property $(T_{\mathcal{C}})$ for the pair $E_2(A) \ltimes A^2 \supseteq A^2$.
- Step2: deduce relative property $(F_{\mathcal{C}})$ for $E_2(A) \ltimes A^2 \succeq A^2$, from relative property $(T_{\mathcal{C}})$ obtained in Step1.

We have succeeded in Step1 for special case, but have failed in the original Step2. Nevertheless, we have succeeded in proving that, if we consider the "larger" pair $SL_3(A) \ltimes A^3 \succeq A^3$, then Step2 always works. Thus we need the assumption of degree ≥ 4 in establishing (F_C) for the class $C = \mathcal{L}_p$ and $C = [\mathcal{H}]$.

6.1 Relative property $(T_{[\mathcal{H}]})$

First we show the following theorem, which amounts to Step1 in the philosophy explained in this chapter page.

Recall from Chapter 0 we always assume $p \in (1, \infty)$, and we use the symbol k for representing any nonnegative integer.

Theorem 6.1.1. (Theorem 1.4 [Mim1]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Then the pair $E_2(A) \ltimes A^2 \supseteq A^2$ has relative property (T_c) . Here C stands for \mathcal{L}_p and $[\mathcal{H}]$.

By Theorem 3.3.10, relative (T) implies relative $(T_{\mathcal{L}_p})$. Therefore in the statement above, the case of $\mathcal{C} = \mathcal{L}_p$ is trivial from Theorem 4.2.3.

Hence in this section, we deal with the case of $\mathcal{C} = [\mathcal{H}]$. In other words, we examine *uniformly bounded* representations on Hilbert spaces.

6.1.1 Dixmier's unitarization of uniformly bounded representation

The key to proving Theorem 6.1.1 is the following proposition by J. Dixmier [Dix], which states any uniformly bounded representation on a Hilbert space of an *amenable* group is unitarizable. For amenable groups, see Subsection 2.5.2.

Proposition 6.1.2. (Dixmier [Dix]) Suppose Λ is amenable. Then for any uniformly bounded representation ρ on a Hilbert space \mathfrak{H} of Λ , there exists an invertible operator $T \in \mathbb{B}(\mathfrak{H})$ such that

$$\pi := \operatorname{Ad}(T) \circ \rho = T \circ \rho \circ T^{-1}$$

is a unitary representation. Moreover, one can choose T such that

 $||T|| ||T^{-1}|| \le |\rho|^2.$

Here ||T|| means the operator norm of T in $\mathbb{B}(\mathfrak{H}, ||\cdot||_{\mathfrak{H}})$.

Later we will change norms on \mathfrak{H} , and then the operator norm of T in $\mathbb{B}(\mathfrak{H}, \|\cdot\|')$ will be changed according to norms $\|\cdot\|'$ on \mathfrak{H}' .

Proof. We use the characterization (ii) in Theorem 2.5.7 for amenability. Let $E: UCB(\Lambda) \to \mathbb{C}$ be a Λ invariant mean. We define the following map

$$\Phi \colon (\xi,\eta) \in \mathfrak{H} \times \mathfrak{H} \mapsto E(\phi_{\xi,\eta}),$$

where $\phi_{\xi,\eta} \in UCB(\Lambda)$ is defined as

$$\phi_{\xi,\eta}(g) = \langle \rho(g)\xi | \rho(g)\eta \rangle.$$

Note that the map Φ is well-defined because ρ is uniformly bounded. Then Φ gives a *new* (positive-definite) inner product on \mathfrak{H} , and $\rho(G)$ -invariant by (right) G-invariance of E. Therefore ρ is unitary with respect to this Hilbert space structure on \mathfrak{H} . Again since $|\rho| < \infty$, this structure is compatible to the original Hilbert space structure in certain quantitative sense.

We note that in general, there exists a uniformly bounded representation on a Hilbert space which is *not* unitarizable. It is known that any group which includes a subgroup isomorphic to F_2 admits such representation. It is a long-standing open problem to determine whether nonexistence of such representations implies amenability.

6.1.2 Proof of relative $(T_{[\mathcal{H}]})$

The proof of Theorem 6.1.1 is based on the argument in the proof of Theorem 4.2.1. Therefore here we give a sketched proof. Also, there is a quantitative version of this theorem (Proposition I) as we mentioned in Chapter 1.

Proof. (Theorem 6.1.1, Outlined) For simplicity, we shall show the case of k = 0. Namely, we will prove relative property $(T_{[\mathcal{H}]})$ for $N = \mathbb{Z}^2 \leq SL_2(\mathbb{Z}) \ltimes \mathbb{Z}^2 = G$. Set $H = SL_2(\mathbb{Z})$. We also use other notation as in the proof of Theorem 4.2.1, such as, $h_1 = E_{1,2}(1), h_2 = E_{2,1}(1), l_1 = E_{1,3}(1), l_2 = E_{2,3}(1), S_0 = \{h_1^{\pm}, h_2^{\pm}\}, S_1 = \{l_1^{\pm}, l_2^{\pm}\},$ and $S = S_0 \cup S_1$.

Suppose that there exist a ucus Banach space $B \in [\mathcal{H}]$ and an isometric representation (ρ, B) of G such that ρ admits almost invariant vectors in $B'_{\rho(N)}$. We may assume that $B'_{\rho(N)} = B$ because $B'_{\rho(N)}$ is also an element in $[\mathcal{H}]$. Consider the compatible Hilbert norm on B, and we write \mathfrak{H} as the space B equipped with this norm. Then the representation ρ , viewed as a representation on \mathfrak{H} , is uniformly bounded. Thanks to the amenability of N and Proposition 6.1.2, we may also assume (ρ, \mathfrak{H}) is a *unitary* representation on N (by replacing ρ with $\operatorname{Ad}(T) \circ \rho$, where T is defined as in Proposition 6.1.2). We choose any vector $\xi \in S(B)$ and fix it. We let $\delta_{\xi} = \sup_{s \in S} \|\xi - \rho(s)\xi\|_{B}$ and $\delta^{*}_{\xi} = \sup_{s \in S} \|\xi^{*} - \rho^{\dagger}(s)\xi^{*}\|_{B^{*}}$. Here recall that ρ^{\dagger} denotes the contragredient representation on B^{*} and $\xi \mapsto \xi^{*}$ is the duality mapping (see Definition 3.1.6 and Definition 3.1.4).

Then from a chosen vector ξ and the duality on B, we can construct a spectral measure $\nu = \nu_{\xi}$ on the unitary dual $\widehat{\mathbb{Z}^2} \cong \mathbb{T}^2 \cong \left[-\frac{1}{2}, \frac{1}{2}\right)^2$. The method for constructing the measure is similar to one in the proof of Theorem 4.2.1. More precisely, we set ν on \mathbb{T}^2 such that the following holds: for any $f \in C(\mathbb{T}^2)$,

$$\int_{\mathbb{T}^2} f d\nu = \langle \sigma(f)\xi, \xi^* \rangle$$

Here $\sigma \colon C(\mathbb{T}^2) \to \mathbb{B}(\mathfrak{H})$ is the *-homomorphism induced by the *unitary* representation $\rho \mid_N$.

Note that we use the duality on B, not the inner product on \mathfrak{H} in the construction above. Therefore unlike the original case, the measure ν is complex-valued in general. However, we obtain the positive part ν_+ by taking the Hahn–Jordan decomposition of ν . We can also verify the following three facts in an argument similar to one in the original proof of Theorem 4.2.1:

- (i) The inequality $\nu_+(\mathbb{T}^2) \ge 1$ holds.
- (*ii*) For any Borel set Z being far from the origin 0 of \mathbb{T}^2 (in certain quantitative sense),

$$\nu_+(Z) = O(\delta_{\xi} \cdot \delta^*_{\xi}),$$

as $\delta_{\xi}, \, \delta^*_{\xi} \to 0.$

(*iii*) For any Borel subset $Z \subset \mathbb{T}^2$ and $h \in S_0$,

$$|\nu_{+}({}^{t}h^{-1} \cdot Z) - \nu_{+}(Z)| = O(\delta_{\xi} + \delta^{*}_{\xi}),$$

as $\delta_{\xi}, \, \delta^*_{\xi} \to 0.$

Now let $\xi \in S(B)$ move among almost invariant vectors with $\delta_{\xi} \to 0$. Then by uniform continuity of the duality mapping (Lemma 3.1.7), δ^*_{ξ} also tends to 0. Hence there must exist some vector $\xi \in S(B)$ such that the associated positive measure ν_+ has a non-zero value on $\{0\}(\subset \mathbb{T}^2)$. This contradicts our assumption that $B^{\rho(N)} = 0$.

We refer to the Appendix for details and a certain quantitative treatment.

6.2 A key trick

The following theorem amounts to Step 2 in the philosophy explained at the beginning of this chapter, and is the key in the paper [Mim1] of the author.

Theorem 6.2.1. (Theorem 1.3 [Mim1]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Suppose B is any superreflexive Banach space. Then, if the pair $\mathbb{E}_2(A) \ltimes A^2 \supseteq A^2$ has relative property (\mathbb{T}_B) , then the pair $\mathbb{SL}_3(A) \ltimes A^3 \ge A^3$ has relative property (\mathbb{F}_B) .

We note that in the statement of this theorem, we use the symbol $G \ge N$ for relative (F_B) because normality of N is not necessarily to define relative (F_B) (compare with the symbol $G \ge N$ for relative (T_B)).

We state the following elementary lemma, which shall be used in the proof of Theorem 6.2.1:

Lemma 6.2.2. Let ρ be an isometric *G*-representation and *c* be a ρ -cocycle. Then the following hold:

- (i) For any $g, h \in G$, $||c(gh)|| \le ||c(g)|| + ||c(h)||$.
- (*ii*) For any $g \in G$, $c(g^{-1}) = -\rho(g^{-1})c(g)$.
- (iii) For any $g, l \in G$, $c(glg^{-1}) = \rho(g)c(l) + (I \rho(glg^{-1}))c(g)$. In particular, if $\rho(glg^{-1})c(g) = c(g)$, then

$$c(glg^{-1}) = \rho(g)c(l).$$

Before proceeding to the main part of proof, in below we consider some setting in the proof of the theorem above. Let $H = SL_3(A)$, $N = A^3$ and $G = H \ltimes N$. We identify G with the following subgroup of $SL_4(A)$:

$$\left\{ \left(\frac{W \mid v}{0 \mid 1} \right) : W \in \mathrm{SL}_3(A), \ v \in A^3 \right\} \ \supseteq \ \left\{ \left(\frac{I_3 \mid v}{0 \mid 1} \right) : v \in A^3 \right\}.$$

We also write an element $g \in G$ as (W, v) $(W \in SL_3(A), v \in A^3)$ associated with in the identification above. Also, we identify elements of the form $(I, v) \in N$ with column vectors in A^3 .

We let N_1 be the subgroup of $N \subseteq SL_4(A)$ of all elements whose (2, 4)-th and (3, 4)-th entries are 0. Namely, N_0 is identified with the additive groups of column vectors ${}^t(v_1, 0, 0) \in A^3$ as in the identification above. Take any isometric *G*-representation ρ on *B* and any ρ -cocycle *c*. Fix one ucus and $\rho(G)$ -invariant norm on *B* as in Proposition 3.1.10. Take a decomposition of *B*:

$$B = B^{\rho(N)} \oplus B'_{\rho(N)} =: B_0 \oplus B_1,$$

and obtain the associated decomposition of the cocycle c

$$c = c_0 + c_1 \ (c_0(g) \in B_0, \ c_1(g) \in B_1).$$

From the $\rho(G)$ -invariance of B_0 and B_1 , each c_j , $j \in \{0, 1\}$ is a ρ -cocycle. For any elements $h = (W, 0) \in H$ and $r = (I, v) \in N$, $hrh^{-1} = (I, Wv) = \in N$ holds, where Wv means the multiplication of the matrix W to the column vector v. In particular, by noting that $\rho \mid_N = \text{id on } B_0$, we have the following equality: for any $h = (W, 0) \in H$ and $r = (I, v) \in N$,

$$c_0((I, Wv)) = \rho((W, 0))c_0((I, v))$$

(see item (ii) in Lemma 6.2.2).

Finally, we claim the following:

- (a) The set $c_0(N)$ is bounded (and hence actually is equal to 0).
- (b) If $c_1(N_1)$ is bounded, then $c_1(N)$ is bounded.

Indeed, for item (a), the key is that any vector $v = {}^{t}(v_1, v_2v_3) \in N$ is decomposed as

$${}^{t}(v_{1}, v_{2}v_{3}) = {}^{t}(1, v_{2} - 1, v_{3} - 1) + {}^{t}(v_{1} - 1, 1, 0) + {}^{t}(0, 0, 1)$$

Then for each u of the three terms in the right hand side, there exists $h = (W, 0) \in H = SL_3(A)$ such that $Wu = {}^t(1, 0, 0) =: e_1$. Therefore, in the view of the formula on $c_0((I, Wv))$ above, we have

$$||c_0((I,v))|| \le 3||c_0((I,e_1))||.$$

This gives the uniform bound of $c_0(N)$. For item (b), consider some alternating matrix $W \in SL_3(A)$, which sends first column to the second (, namely, for instance, take W satisfying $W^t(v_1, v_2, v_3) = {}^t(v_2, v_1, -v_3)$). By assumption, we know that c is uniformly bounded on $N_1 = \{v = {}^t(v_1, 0, 0) : v_1 \in A\}$. Then since $(W, 0)(I, v)(W^{-1}, 0) = (I, Wv)$, we have

$$||c_1((I, Wv))|| \le ||c_1((I, v))|| + ||c_1((W, 0))|| + ||c_1((W^{-1}, 0))||$$

Since W is taken independently of v, this shows c_1 is also bounded on the group $\{{}^t(0, v_2, 0) : v_2 \in A\} \leq N$. In a similar way, c_1 is also bounded on the group $\{{}^t(0, 0, v_3) : v_3 \in A\} \leq N$. Finally, observe

$$||c_1({}^t(v_1, v_2, v_3))|| \le ||c_1({}^t(v_1, 0, 0))|| + ||c_1({}^t(0, v_2, 0))|| + ||c_1({}^t(0, 0, v_3))||,$$

and get item (b).

Proof. (Theorem 6.2.1) Keep the setting as in above. Thanks to item (a) and item (b) above and Lemma 3.2.5, for our proof it suffices to verify the boundedness of $c_1(N_1)$. We define a *finite* subset S_0 and two subgroups G_1 , G_2 of G by the following expressions respectively:

$$\left\{ \left(\begin{array}{cccc} 1 & * & * & * \\ 0 & 1 & * & * \\ 0 & * & 1 & * \\ 0 & 0 & 0 & 1 \end{array} \right) \right\}, \left\{ \left(\begin{array}{cccc} 1 & {}^{t}v' & 0 \\ 0 & W' & 0 \\ 0 & 0 & 1 \end{array} \right) \right\} \text{ and } \left\{ \left(\begin{array}{cccc} 1 & 0 & 0 \\ 0 & W' & v' \\ 0 & 0 & 1 \end{array} \right) \right\}.$$

Here in the first definition, the expression means that for each element in S_0 , only one of the above *'s is $\pm x_l$ ($0 \le l \le k$) and the others are 0. Also in the second and the third expressions, W' moves among all elements in $E_2(A)$ and v' moves among all elements in A^2 . We let $C = \sup_{s \in S_0} ||c_1(s)|| < \infty$. We set L ($\le G_1$) as the group of all elements in G_1 with R' = I and N_2 ($\le G_2$) as the group of all elements in G_2 with W' = I. A crucial point here is that N_1 commutes with S_0 : therefore for any $r_1 \in N_1$ and any $s \in S_0$, we have the following inequalities:

$$\|\rho(s)c_1(r_1) - c_1(r_1)\| = \|c_1(sr_1) - c_1(r_1) - c_1(s)\|$$

$$\leq \|c_1(r_1s) - c_1(r_1)\| + \|c_1(s)\|$$

$$= \|\rho(r_1)c_1(s)\| + \|c_1(s)\|$$

$$= 2\|c_1(s)\| \leq 2C.$$

We set a number \mathcal{K} as the minimum of the two relative Kazhdan constants $\mathcal{K}(G_1, L; S_0 \cap G_1, \rho \mid_{G_1})$ and $\mathcal{K}(G_2, N_2; S_0 \cap G_2, \rho \mid_{G_2})$ (recall definition 3.2.7). Then by relative (T_B) for the pair $E_2(A) \ltimes A^2 \succeq A^2$, \mathcal{K} is strictly positive. Hence by Lemma 3.2.8, for any $\xi \in c_1(N_1)$ there are following inequalities:

for any
$$l \in L$$
, $\|\xi - \rho(l)\xi\| \leq 8\mathcal{K}^{-1}C$,
and for any $r_2 \in N_2$, $\|\xi - \rho(r_2)\xi\| \leq 8\mathcal{K}^{-1}C$.

Next, note that the group N_1 is obtained by single commutators between L and N_2 : for any $r \in N$, there exist $r_1 \in N_1$, $r_2 \in N_2$, $r' \in N_2$, and $l \in L$ such that $r = r_1 r_2$ and $r_1 = lr' l^{-1} r'^{-1}$. Hence for any $\xi \in c_1(N_1)$ and $r \in N$, the following inequality holds:

$$\|\xi - \rho(r)\xi\| = \|\xi - \rho(lr'l^{-1}r'^{-1}r_2)\xi\|$$

< 32 $\mathcal{K}^{-1}C$.

Note that the upper bound of the inequality above is *independent* of the choices of $\xi \in c_1(N_1)$ and of $r \in N$.

Finally, suppose that $c_1(N_1)$ is not bounded. Then one can choose $\xi \in c_1(N_1)$ such that the following holds:

for all
$$r \in N$$
, $\|\xi - \rho(r)\xi\| < \|\xi\|$.

However, in a similar argument as in the proof of Corollary 2.1.14, this inequality implies existence of a non-zero $\rho(N)$ -invariant vector in $B_1 = B_{\rho(N)}'$. This is a contradiction.

6.3 Proof of Theorem A

Here we restate Theorem A in Chapter 1 for convenience:

Theorem 6.3.1. (Theorem 1.2, [Mim1]) Let $A = \mathbb{Z}[x_1, \ldots x_k]$ ($k \in \mathbb{N}$). Then for $m \geq 4$, the universal lattice $G = \mathrm{SL}_m(A)$ has property ($F_{\mathcal{C}}$). Here \mathcal{C} stands for either the class \mathcal{L}_p ($1) or the class [<math>\mathcal{H}$].

Proof. We will see that Shalom's machinery (Theorem 5.3.2) works for $SL_m(A)$ in the following setting: set $H \cong SL_{m-1}(A)$, and $N_1, N_2 \cong A^{n-1}$. Here in G we realize H as in the left upper corner (, namely, the $((1-(m-1)) \times (1-(m-1)))$ -th parts), realize N_1 as in the $((1-(m-1)) \times m)$ -th unipotent parts, and realize N_2 as in the $(m \times (1-(m-1)))$ -th unipotent parts. Namely, we take

$$H = \operatorname{SL}_{m-1}(A) := \left\{ \left(\frac{W \mid 0}{0 \mid 1} \right) : W \in \operatorname{SL}_{m-1}(A) \right\},$$
$$N_1 := \left\{ \left(\frac{I_{m-1} \mid v}{0 \mid 1} \right) : v \in A^{m-1} \right\} \cong A^{m-1},$$
$$N_2 := \left\{ \left(\frac{I_{m-1} \mid 0}{{}^t v' \mid 1} \right) : v' \in A^{m-1} \right\} \cong A^{m-1},$$

We have seen that this triple satisfies conditions (i), (ii), and (iii) of the Shalom property (for any class C) in the proof of Theorem 5.4.

Firstly we claim that Shalom's machinery applies with $C = \mathcal{L}_p$. The stability of \mathcal{L}_p under ultraproducts follows from item (*iii*) of Theorem 5.1.10. It only remains to check that (H, N_1, N_2) satisfies condition (*iv*) of the Shalom property for $(F_{\mathcal{L}_p})$. This part is the main work of this chapter (and the paper [Mim1] of the author): Theorem 6.2.1 together with Theorem 6.1.1 insures the conclusion. Therefore, Shalom's machinery works.

In the case of $\mathcal{C} = [\mathcal{H}]$, we need an additional care, because $[\mathcal{H}]$ itself is not stable under ultraproducts. What needs here is to just replace $[\mathcal{H}]$ with \mathcal{H}_M , the class defined in Theorem 5.1.10, for every $M \geq 1$. Again Theorem 6.2.1 together with Theorem 6.1.1 shows (H, N_1, N_2) satisfies condition (iv) of the Shalom property for $(\mathcal{F}_{\mathcal{H}_M})$ for each M. Therefore for any $M \geq 1$, Shalom's machinery works with $\mathcal{C} = \mathcal{H}_M$, and hence G has property $(\mathcal{F}_{\mathcal{H}_M})$. Finally, observe

$$[\mathcal{H}] = \bigcup_{M \ge 1} \mathcal{H}_M.$$

Thus we have established property $(F_{\mathcal{H}})$ for $SL_{m\geq 4}(A)$.

We note that for the case of $\mathcal{C} = \mathcal{L}_p$, there is a shortcut in proving this theorem. Namely, we have a way to prove property $(F_{\mathcal{L}_p})$ without appealing to Shalom's machinery. We will discuss it in Section 8.1. With the aid of this shortcut, we have established the fixed point property on *p*-Schatten class operators in the paper [Mim3] of the author.

Finally, we state one immediate corollary of Theorem 6.3.1, which follows from the fact that (F_B) always implies (T_B) (Theorem 3.2.9).

Corollary 6.3.2. ([Mim1]) Let $A = \mathbb{Z}[x_1, \ldots x_k]$ $(k \in \mathbb{N})$. Then for $m \ge 4$, the universal lattice $SL_m(A)$ has property $(T_{[\mathcal{H}]})$.

Chapter 7

Property (TT) and (FF_B)

In this chapter, we introduce a notion of property (FF_B) in the paper [Mim1] of the author, which is an extension of property (TT) of Monod (we refer to Chapter 12 of a book of Monod [Mon1]). Also, we explain some connection to the concept of bounded cohomology, whose theory has been invented by Gromov [Gro1]. Finally, we state some previously known facts on property (TT), which accounts to two papers [BuMo1], [BuMo2] of Burger–Monod; two papers of Monod–Shalom [MoSh1], [MoSh2]; and a paper [MMS] of Mineyev–Monod–Shalom. Some results of the author on (FF_B) (and (TT)) shall be stated and proven in Chapter 8.

The main references in this section are a book [Mon1] and an ICM proceedings paper [Mon2] of Monod.

7.1 Property (FF_B) and bounded cohomology

In this section, we give the definition of property (FF_B) and explain relation to bounded cohomology. Property (FF_B) is a quasification of property (F_B) . That means, we consider a map which is a cocycle up to bounded error, and we call this map a quasi-cocycle. Property (FF_B) is defined as the boundedness property of such quasi-cocycles.

7.1.1 Quasi-cocycles and (FF_B)

Definition 7.1.1. Let B be a Banach space and G be a group.

(i) Let ρ be an isometric *G*-representation on *B*. A continuous map $b: G \to B$ is called a *quasi-\rho-cocycle* if the following holds:

$$\sup_{g,h\in G} \|b(gh) - b(g) - \rho(g)b(h)\| < \infty.$$

(*ii*) A (strongly) continuous map β from G to the set of all affine isometries on B is called a *quasi-action* if the following holds:

$$\sup_{g,h\in G}\sup_{\xi\in B}\|\beta(gh)\xi-\beta(g)\beta(h)\xi\|<\infty.$$

In the definition of quasi-actions, one can decompose the map β into the linear part ρ and the transition part b, namely, $\beta(g)\xi = \rho(g)\xi + b(g)$ for any $g \in G$ and $\xi \in B$. Then the map β is a quasi-action if and only if ρ is a group representation and b is a quasi- ρ -cocycle. Indeed, "if" part is trivial. For "only if" part, what we need to show is that the linear part ρ is indeed a representation (a group homomorphism). Suppose there exist $g, h \in G$ such that $\rho(g)\rho(h) - \rho(gh) \neq 0$. Then by taking η with $(\rho(g)\rho(h) - \rho(gh))\eta \neq 0$ and considering $t\eta$ with $t \to \infty$, we have that this β is not a quasi-action. Therefore ρ must be a group homomorphism, and we are done.

Remark 7.1.2. We warn that the terminology of *quasi-actions* is *not* a standard one, although the terminology *quasi-cocycle* is standard. Usually, the word "quasi-action" is used in relation to *quasi-isometry* among spaces (compare with Definition 11.4.3). On this topic, see [Roe], [Gro2], [Gro4], and [Gro5].

The following terminology is due to the author [Mim1]:

Definition 7.1.3. ([Mim1]) Let B be a Banach space.

- (i) A pair $G \ge N$ of groups is said to have *relative property* (FF_B) if for any isometric representation ρ of G on B and any quasi- ρ -cocycle b, b(N) is bounded. Equivalently, if for any quasi-action on B, some (or equivalently, any) N-orbit is bounded.
- (*ii*) A group G is said to have property (FF_B) if $G \ge G$ has relative (FF_B).

This notion has its origin to property (TT) of Monod [Mon1]. He defined property (TT) as boundedness property of all quasi-cocycles with unitary coefficients. This property is identical to property (FF_{\mathcal{H}}) in our definition. We use the terminology (FF_B) because this property is a quasification of (F_B), not of (T_B), and in general setting, (F_B) is not identical to (T_B) (we have seen that it is usually much stronger).

This is an obvious observation from Lemma 3.2.5.

Lemma 7.1.4. If a Banach space B is superreflexive, then property (FF_B) implies property (F_B) .

This implication may happen to be false in the case of that B is non-superreflexive.

7.1.2 Relation to bounded cohomology

Bounded cohomology is initiated around a book of B. E. Johnson [Joh]. By an inventional paper [Gro1] of Gromov, this theory has been expanded. In [Gro1], much attention has been paid on the trivial coefficient $((1_G, \mathbb{R}))$ case. Here we consider bounded cohomology with isometric Banach coefficient (we shall pay attention to the trivial coefficient case as well in Chapter 10). For details of bounded cohomology with Banach coefficients, we refer to [Mon1], [Mon2]. We give a definition of bounded cohomology, which is based on *inhomogeneous* standard cochain complex:

Definition 7.1.5. Let G be a group, and ρ be an isometric G-representation on a Banach space B. We define the (continuous) bounded cohomology $H^{\bullet}_{cb}(G; \rho, B)$ in the following way:

- (i) The cochain $C^n_{\rm b}(G;\rho,B)$ consists of all continuous bounded functions $c: G^{(n)} \to B$. Here we define $C^0_{\rm b}(G;\rho,B) := B$.
- (*ii*) The coboundary map $\delta = \delta^n \colon C^n_b \to C^{n+1}_b$ is the restriction of the coboundary map in the ordinary cohomology theory on C^n_b : namely, we define δ above by the following formula:

$$\delta c(g_0, \dots, g_n) := \rho(g_0) c(g_1, \dots, g_n) + \sum_{i=0}^{n-1} (-1)^{i+1} c(g_0, \dots, g_i g_{i+1}, \dots, g_n) + (-1)^{n+1} c(g_0, \dots, g_{n-1}).$$

(*iii*) The bounded cohomology is defined as

 $H^n_{\rm cb}(G;\rho,B) := Z^n_{\rm cb}(G;\rho,B) / B^n_{\rm cb}(G;\rho,B).$

Here $Z_{cb}^n(G;\rho,B)$ denotes $\operatorname{Ker}\delta^n$ and $B_{cb}^n(G;\rho,B)$ denotes $\operatorname{Im}\delta^{n-1}$.

If ρ is obvious in context, then we write $H^{\bullet}_{cb}(G; \rho, B)$ shortly as $H^{\bullet}_{cb}(G; B)$. Also if (ρ, B) is the trivial representation on reals (, namely, $(1_G, \mathbb{R})$,), then we write it as $H^{\bullet}_{cb}(G, \mathbb{R})$, or just $H^{\bullet}_{cb}(G)$.

Similar to the definition of group (1-)cohomology, we use the symbol $H^{\bullet}_{\rm b}(G;\rho,B)$ if G is discrete.

In below, we will explain relation between bounded cohomology and quasicocycles. Let (ρ, B) is an isometric representation of a group G, and consider the space of all quasi- ρ -cocycles. This space itself is ridiculously large because it contains *all* continuous map $G \to B$ with *bounded* range, and it contains all ρ -cocycles. In order to get a reasonable space, we need to *mod out* these maps.

Definition 7.1.6. Let G be a group and (ρ, B) be an isometric G-representation.

- (i) Define $QH_{c}(G; \rho, B)$ as the space of all quasi- ρ -cocycles.
- (*ii*) Define $QH_{c}(G; \rho, B)$ as the following quotient space:

$$\widetilde{QH}_{c}(G;\rho,B)$$

:= $QH_{c}(G;\rho,B)/(\{\rho\text{-cocycles}\} + \{\text{continuous maps } G \to B \text{ with bounded range}\})$

We call $QH_{c}(G; \rho, B)$ the actual space of quasi- ρ -cocycles.

If ρ is obvious in context, then we write $QH_{c}(G;\rho,B)$ shortly as $QH_{c}^{\bullet}(G;B)$ and $\widetilde{QH}_{c}(G;\rho,B)$ shortly as $\widetilde{QH}_{c}^{\bullet}(G;B)$. Also if (ρ,B) is $(1_{G},\mathbb{R})$, then we write them respectively as $QH_{c}(G)$ and $\widetilde{QH}_{c}(G)$.

We also omit the symbol c if G is discrete. We warn that the terminology actual space of quasi-cocycles and the symbols $QH_c(G; \rho, B)$, $\widetilde{QH}_c(G; \rho, B)$ are not standard. This QH is named after Quasi-Homomorphisms (see Chapter 10), and hence the symbols $QH_c(G)$ and $\widetilde{QH}_c(G)$ are standard.

Next, we define the *comparison map*.

Definition 7.1.7. Let G be a group and (ρ, B) be an isometric G-representation. Let $n \ge 1$. Then a natural homomorphism (between vector spaces)

$$\Psi_{\rm cb}^n \colon H^n_{\rm cb}(G;\rho,B) \to H^n_{\rm c}(G;\rho,B)$$

is induced by the natural inclusion of complexes. Here $H^n(G; \rho, B)$ denotes the ordinary group cohomology. This map is called the *comparison map* in degree n.

Again, we omit the symbol c if G is discrete.

We note that comparison map is *neither* injective *nor* surjective in general. The following lemma connects bounded cohomology in degree 2 and the actual space of quasi-cocycles. Note that if B is superreflexive, then $H^1_{cb}(G; \rho, B) = 0$ for any group G and isometric representation ρ on B (this amounts to Lemma 7.1.4, namely, any bounded cocycle on B is a coboundary). Therefore, the case of our main concern and high importance is dimension 2 case.

Lemma 7.1.8. Let G be a group and (ρ, B) be an isometric G-representation. Then there is a natural isomorphism among vector spaces,

$$\widetilde{QH}_{c}(G;\rho,B)\cong \mathrm{Ker}\Psi^{2}_{\mathrm{cb}}$$

Here $\operatorname{Ker} \Psi_{cb}^2$ denotes the comparison map in degree 2

$$\Psi^2_{\rm cb} \colon H^2_{\rm cb}(G;\rho,B) \to H^2_{\rm c}(G;\rho,B)$$

Proof. Consider the following homomorphism among vector spaces:

$$QH_{\rm c}(G;\rho,B) \to {\rm Ker}\Psi_{\rm cb}^2; \ b \mapsto [\delta b]_{\rm cb},$$

where $[\cdot]_{cb}$ is the bounded cohomology class. Then it is a routine to check that it is surjective, and that the kernel of this map is the space of all maps of the form of " ρ -cocycles + maps with bounded range." Therefore, this homomorphism induces the isomorphism in the lemma above.

Corollary 7.1.9. Let G be a group and B be a Banach space. If G has (FF_B) , then for any isometric G-representation ρ on B, $H^2_{cb}(G; \rho, B)$ naturally injects into $H^2_c(G; \rho, B)$.

Some examples on property $(TT)(=(FF_{\mathcal{H}}))$ will be examined in Section 7.2. Also, we will see some examples of quasi-cocycle for the case of $(\rho, B) = (1_G, \mathbb{R})$ in Subsection 10.2.2.

7.2 Property (TT) and some facts

Recall that property (TT) of Monod is identical to property (FF_{\mathcal{H}}) in our definition, and it means that every quasi-cocycle with any unitary coefficient is bounded. The theorem of Monod–Shalom [MoSh1], [MoSh2] and Mineyev–Monod–Shalom [MMS] in below implies that property (TT) is strictly stronger than property (T).

- **Definition 7.2.1.** (i) A discrete group is said to be *virtually abelian* if it contains an abelian subgroup of finite index (we regard $\{e\}$ as an abelian group as well).
- (*ii*) A hyperbolic group (recall Definition 2.6.19) is said to be *non-elementary* if it is not virtually abelian.

It is known that hyperbolic groups cannot contain \mathbb{Z}^2 (see [Gro2]). Therefore a hyperbolic group is elementary if and only if it is finite or it is virtually \mathbb{Z} (, namely, it contains \mathbb{Z} as a finite index subgroup). All virtually abelian groups are amenable. On the other hand, it is known that any non-elementary hyperbolic group contains F_2 , and hence is not amenable. We note that it is a theorem of Gromov–Cheeger that for amenble groups, bounded cohomology with reasonable coefficients always vanishes. In contrast, I. Mineyev [Min] shows that for non-elementary hyperbolic groups, comparison maps in degree ≥ 2 with any reasonable coefficient are all surjective.

Theorem 7.2.2. ([MoSh1], [MoSh2], [MMS]) Let G be a non-elementary hyperbolic group. Then $\widetilde{QH}(G; \ell^2(G))$ is infinite dimensional. Here the isometric Grepresentation on $\ell^2(G)$ is the left regular representation. Note that F_2 is hyperbolic, and also that $H^2(F_2; \ell^2(F_2)) = 0$ because it is a free group. Therefore, $H^2_b(F_2; \ell^2(F_2))$ has infinite dimension, and the comparison map in degree 2 sends it to the zero-space. This example shows that sometimes behavior of bounded cohomology is pathological.

On the contrary, Burger–Monod [BuMo1], [BuMo2] show that for higher rank lattices, this sort of pathology does not occur, at least in degree 2. Recall our terminology *totally higher rank lattices* in Chapter 0.

Theorem 7.2.3. (Burger-Monod [BuMo1], [BuMo2]) Any totally higher rank algebraic group and totally higher rank lattice have (TT).

Remark 7.2.4. They show this theorem first for totally higher rank algebraic group. In [BuMo1], they show (TT) passes to cocompact lattices. In [BuMo2], they deal with noncocompact cases. Note that (p-, and)2-induction of quasi-cocycles is available for (p-, and)2-integrable lattices in a similar manner to that in Subsection 3.4.2. However that even for cocompact case, there is a gap in just imitating the argument of heredity of $((F_{\mathcal{L}_p}), \text{ and})$ (T) to lattices. The problem lies in how to deduce boundedness of the original quasi-cocycle from the boundedness of the induced quasi-cocycle. For details, see [BuMo1], [BuMo2], and also a paper [Oza] of N. Ozawa.

However, as long as considering discrete case (, namely, the case of finite index subgroups), induction has no difficulties: the boundedness of the induced quasi-cocycle obviously implies the boundedness of the original quasi-cocycle. Therefore, it is easy to see that property (TT) passes to finite index subgroups.

Finally, we briefly state the strong negation of property (TT), which is weakening of the Haagerup property:

Definition 7.2.5. A group G is said to have the *a*-TT-manablity (or the ha-haagerup property) if there exists a proper quasi-cocycle on G with some unitary coefficient.

The following result is implicitly stated in [Min]. See also [Oza].

Theorem 7.2.6. (*Mineyev* [Min]) Every hyperbolic group is a-TT-menable.

Chapter 8

Property $(FF_B)/T$

In this chapter, we introduce a notion of property $(FF_B)/T$, which is a priori a weaker version of property (FF_B) . This property means the following: any quasicocycle with any isometric coefficient on B is bounded modulo trivial part. This notion is needed, at the present, in deducing the property for universal lattices. More precisely, we deduce property $(FF_{\mathcal{L}_p})/T$ for all p and property $(FF_{[\mathcal{H}]})/T$ for universal lattices of degree at least 4. These are one of the main results in the paper [Mim1] of the author, and stated as Theorem B in this thesis. However at the moment, it seems to be open to determine whether $(FF_{\mathcal{L}_p})$ (or $(FF_{[\mathcal{H}]})$) is satisfied for universal lattices. It is worth noting that even $(FF_{\mathcal{H}})/T$ for universal lattices of degree at least 4 is a new result.

Also, we introduce a shortcut to deduce $(FF_B)/T$ for universal lattices (with degree ≥ 4) which does *not* appeal to Shalom's machinery. This implies in particular that there is a possibility to apply this shortcut to the case of that the class B of Banach spaces (or, even a single Banach space) is *not* stable under ultraproducts. We note that, however, this shortcut is deeply based on the fact that universal lattices have (T), and hence philosophically this is not free from Shalom's machinery. Nevertheless, this shortcut applies to the case of p-Schatten class spaces C_p . This shows property $(FF_{C_p})/T$ for universal lattices (with degree ≥ 4), and this result can be seen as a generalization of Theorem B in (some) noncommutative setting. This is one of the main results in the paper [Mim3] of the author, and is stated as Theorem C in this thesis.

Finally, we study more on property $(FF_{\mathcal{H}})/T$, which is also called *property* (TT)/T. By employing a theorem of Ozawa, we prove universal lattice of degree 3 in fact enjoys this property. As we will see in Chapter 11, (TT)/T has intriguing application to homomorphism rigidity into mapping class groups (of surfaces) and into automorphism groups of free groups.

Works in this chaper are based on the papers [Mim1], [Mim2]. On the fixed point property (F_{C_p}) on the *p*-Schatten classes C_p , it is also inspired by a previous work of Puschnigg [Pus]. Also, we refer to a book [PiXu] of Pisier–Xu for comprehensive treatment for noncommutative spaces. For property (TTT) of Taka Ozawa, see the original paper [Oza].

8.1 A shortcut without ultraproducts

In this section, we see a shortcut of the proof of prperty $(F_{\mathcal{L}_p})$ for universal lattices of degree ≥ 4 , which is based on the Shalom–Vaserstein's theorem of property (T) for this group. Also, by employing this shortcut, we show property $(FF_{\mathcal{L}_p})/T$ for this group, which is stronger than property $(F_{\mathcal{L}_p})$. We introduce this notion of property $(FF_B)/T$ in this section.

8.1.1 Definition of property $(FF_B)/T$

Let G be a group, and ρ is an isometric G-representation on a Banach space B. Recall if B is superreflexive, then there is a canonical decomposition as $\rho(G)$ -spaces:

$$B = B^{\rho(G)} \oplus B_{\rho(G)}' =: B_0 \oplus B_1$$

(see Proposition 3.1.12). Let $b: G \to B$ be a quasi- ρ -cocycle. Then for the decompsition

$$b = b_0 + b_1$$

associated with the decomposition of B, b_0 and b_1 are also quasi- ρ -cocycles. This follows from $\rho(G)$ -invariance of spaces and a norm estimate in Proposition 3.1.12.

In [Mim1], the author has come up with the following technical notion:

Definition 8.1.1. ([Mim1]) Let *B* be a Banach space. A group *G* is said to have property (FF_B)/T (, which is called "property (FF_B) modulo trivial part",) if for any isometric *G*-representation ρ on *B* and any quasi- ρ -cocycle *b*, b'(G) is bounded. Here $b': \Gamma \to B/B^{\rho(G)}$ is the natural quasi-cocycle constructed from the projection of *b* associated the canonical quotient map $B \twoheadrightarrow B/B^{\rho(G)}$. If *B* is superreflexive, then this definition is equivalent to the following condition: for any isometric representation ρ of Γ on *B* and any quasi- ρ -cocycle *b*, $b_1(G)$ is bounded. Here we decompose *b* as $b_0 + b_1$ such that b_0 takes values in $B_0 = B^{\rho(\Gamma)}$ and b_1 takes values in $B_1 = B'_{\rho(\Gamma)}$.

In particular, the following holds: if G has $(FF_B)/T$, then for any isometric G-representation ρ on B satisfying $\rho \not\supseteq 1_G$, every quasi- ρ -cocycle is bounded. Property $(FF_B)/T$ is a priori a weaker notion than that of property (FF_B) , but at the moment we have no example which distinguishes these. As we will see in below, universal lattices of degree ≥ 4 have $(FF_{\mathcal{L}_p})/T$ but it is not known whether they enjoy $(FF_{\mathcal{L}_p})$. Therefore these groups are possible candidates. The following is easy to see:

Lemma 8.1.2. Let \mathcal{C} be \mathcal{L}_p or $[\mathcal{H}]$. Then property $(FF_{\mathcal{C}})/T$ implies property $(F_{\mathcal{C}})$.

Proof. Note that spaces in such \mathcal{C} are superreflexive, and hence that all bounded cocycle with isometric coefficients on these are coboundary. Therefore for the proof, it suffices to show the abelianization $H = G/\overline{[G,G]}$ is compact. Suppose not. Then since H is abelian, H has unbounded λ_H -cocycle on $L^p(H)$ (it can be constructed from the fact that $\lambda_H \succeq 1_H$, where λ_H denotes the left-regular representation). Here if $\mathcal{C} = [\mathcal{H}]$, then set p = 2. Pull-back this cocycle to a cocycle of G. Also, since H is non-compact, the pull-back of the representation does not contain 1_G . This contradicts (FF_C) for G.

This proof shows Lemma 8.1.2 can be true in much more general situations. Also, the proposition below follows from Lemma 3.2.5, and the proof of Theorem 3.2.9:

Proposition 8.1.3. Let B be a superreflexive Banach space. Then property $(FF_B)/T$ implies property (T_B) .

Next, we see some permanence properties.

Proposition 8.1.4. Let G be a group, and B be a Banach space. Suppose G has $(FF_B)/T$.

- (i) Suppose there is a continuous homomorphism $G \to H$ to a group H with dense range. Then H has $(FF_B)/T$.
- (ii) Suppose G is discrete. Let $\Gamma \leq G$ be a finite index subgroup. Let $p \in (1, \infty)$, and consider p-induction $E = \ell^p(G/\Gamma, B)$ (recall Subsection 3.4.2). Then Γ has $(FF_E)/T$.

In particular, $(FF_B)/T$ passes to group quotients; and property $(FF_{\mathcal{L}_p})/T$ and property $(FF_{\mathcal{H}_l})/T$ pass to finite index subgroups.

Proof. Item (i) is obvious. For item (ii), recall Remark 7.2.4 and also check on the invariant spaces. Induction shows the assertions.

With more effort, it might be possible to obtain a similar result to cocompact lattices (and even more to *p*-integrable lattices under some condition). However we do not proceed in this direction, because our groups of main concern are discrete groups.

We warn that there is no reason to expect that $(FF_{\mathcal{L}_p})/T$, for instance, is stable under extensions. In fact, there is no reason to expect this even for direct product (the reason is that $(G_1 \times G_2)$ -invariant space is in general much smaller than each G_i -invariant spaces).

8.1.2 Property $(FF_B)/T$ for universal lattices

Here we introduce a shortcut in proving (F_B) or strongly, $(FF_B)/T$ without taking ultraproducts. Firstly we see the following generalization of Theorem 6.2.1, which can be directly obtained by imitating the original proof:

Theorem 8.1.5. (Theorem 6.4 [Mim1]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Suppose B is any superreflexive Banach space. Then, if the pair $\mathbb{E}_2(A) \ltimes A^2 \supseteq A^2$ has relative property (\mathbb{T}_B) , then the pair $\mathbb{SL}_3(A) \ltimes A^3 \ge A^3$ has relative property (\mathbb{FF}_B) .

The following is the shortcut theorem. It is a generalization of Proposition 6.6 in [Mim1], and plays a key role to establishing (TT)/T for symplectic universal lattices (Section 9.3). For the proof, we extend our definition of relative (FF_B) for (closed) subsets.

Definition 8.1.6. Let G be a group and B be a Banach space. Let $Q \subseteq G$ be a closed *subset*.

- (i) A pair $G \supseteq Q$ is said to have *relative* (FF_B) if for any isometric *G*-representation ρ on *B*, every ρ -cocycle is bounded on *Q*.
- (ii) A pair $G \supseteq Q$ is said to have relative $(FF_B)/T$ if for any isometric G-representation ρ on B and any quasi- ρ -cocycle b, b'(S) is bounded, where $b': \Gamma \to B/B^{\rho(G)}$ is the natural quasi-cocycle constructed from the projection of b associated the canonical quotient map $B \to B/B^{\rho(G)}$. If B is superreflexive, then this definition is equivalent to the following condition: for any isometric representation ρ of Γ on B and any quasi- ρ -cocycle b, $b_1(S)$ is bounded. Here we decompose b as $b_0 + b_1$ such that b_0 takes values in $B_0 = B^{\rho(\Gamma)}$ and b_1 takes values in $B_1 = B'_{\rho(\Gamma)}$.

In this definition, if Q is a subgroup of G, then we prefer to use the symbol " $G \ge Q$ has relative (FF_B)/T."

We note in the definition of relative $(FF_B)/T$, "T" always means the $\rho(G)$ -trivial part, even if we consider subsets or subgroups.

Theorem 8.1.7. ([Mim4]) Let G be a group and B be a Banach space. Suppose the group G and a pair (H, Q), where $H \leq G$ is a subgroup and $Q \subseteq G$ is a closed subset, satisfies the following five conditions:

- (i) The group G has (T_B) .
- (ii) The set Q generates G.
- (iii) The set Q is invariant under the conjugation of elements in H. Namely, for any $h \in H$, $hQh^{-1} \subseteq Q$.

8.1. A SHORTCUT WITHOUT ULTRAPRODUCTS

- (iv) The pair $G \supseteq Q$ has relative $(FF_B)/T$.
- (v) The group G is boundedly generated by Q and H.

Then G has property $(FF_B)/T$.

Even without condition (v), we have that $G \ge H$ has relative $(FF_B)/T$. Namely, if (G, H, Q) $(H \le G, Q \subseteq G)$ satisfies conditions (i), (ii), (iii) and (iv), then $G \ge H$ has relative $(FF_B)/T$.

Proof. We will show the latter assertion. Then the former assertion is confirmed by the bounded generation (condition (v)). Let ρ be an isometric *G*-representation, and *b* be a quasi- ρ -cocycle. Consider the representation $\rho' \colon G \to O(B/B^{\rho(G)})$ and the ρ' -cocycle $b' \colon G \to B/B^{\rho(G)}$, which are naturally determined by the quotient map $B \twoheadrightarrow B/B^{\rho(G)}$. Since *b'* is a quasi-cocycle, we set

$$C_1 := \sup_{g_1, g_2 \in G} \|b'(g_1g_2) - b'(g_1) - \rho(g_1)b'(g_2)\| < \infty.$$

Also by condition (iv), b' is bounded on Q. Set

$$C_2 := \sup_{q \in Q} \|b'(q)\| < \infty.$$

Set $C = \max\{C_1, C_2\} < \infty$.

Take any $h \in H$ and any $q \in Q$. Then we have the following inequalities:

$$\begin{aligned} \|\rho'(q)b'(h) - b'(h)\| &\leq \|b'(qh) - b'(q) - b'(h)\| + C \\ &\leq \|b'(qh) - b'(h)\| + 2C \\ &= \|b'(h(h^{-1}qh)) - b'(h)\| + 2C \\ &\leq \|b'(h) + \rho'(h)b'(h^{-1}qh) - b'(h)\| + 3C \\ &\leq \|b'(h^{-1}qh)\| + 3C \\ &\leq 4C. \end{aligned}$$

Here in the last line we use condition (*iii*) (and condition (*iv*)). Note that the last dominating term 4C is *independent* of the choices of $h \in H$ and $q \in Q$.

Now suppose b'(H) is not bounded. Then by the inequalities above, this means for any ϵ , ρ' admits a (Q, ϵ) -invariant vector. By condition (ii), this means $\rho' \succeq 1_G$ (see the proof of Lemma 2.1.3). However this contradicts condition (i).

Also, the proof shows that there is an (F_B) -version of this theorem. Note that unless B is superreflexive, boundedness of cocycle does not necessarily imply the cocycle being a coboundary (there are some examples of Banach spaces *beyond* superreflexive ones with respect to which the implication above is true: for instance, any separable reflexive Banach space is known to satisfy the implication above). **Theorem 8.1.8.** ([Mim4]) Let G be a group and B be a Banach space. Suppose the group G and a pair (H, Q) where $H \leq G$ be a subgroup and $Q \subseteq G$ be a closed subset satisfies the following five conditions:

- (i) The group G has (T_B) .
- (ii) The set Q generates G.
- (iii) The set Q is invariant under the conjugation of elements in H.
- (iv) The pair $G \supseteq Q$ has relative (F_B) .
- (v) The group G is boundedly generated by Q and H.

Suppose in addition B is superreflexive and G has compact abelianization. Then G has property (F_B) .

Even without condition (v), we have that $G \ge H$ has relative (F_B) . Namely, if (G, H, Q) $(H \le G, Q \subseteq G)$ satisfies conditions (i), (ii), (iii) and (iv); B is superreflexive; and G has compact abelianization, then $G \ge H$ has relative (F_B) .

Remark 8.1.9. Here are remarks on Theorem 8.1.7 and Theorem 8.1.8.

(i) The proof of Theorem 8.1.7 is very simple, as we have seen in above. We note that this can*not* apply to establishing property (T) because a property having form (T_B) *itself* is contained in the conditions of the theorem (condition (i)). Nevertheless, Theorem 8.1.7 is very powerful: once property (T_B) is confirmed (by some other means), then existence of such pair (H, Q) implies $(FF_B)/T$; and thus in particular (F_B) if B is superreflexive and G has finite abelianization.

In particular, recall that property (T) implies $(T_{\mathcal{L}_p})$ but that $(F_{\mathcal{L}_p})$ $(p \gg 2)$ is strictly stronger than (T). However, Theorem 8.1.7 implies that if a group G has a pair (H, Q) in the theorem, then (T) for G is sufficient to deduce $(F_{\mathcal{L}_p})$ and even $(FF_{\mathcal{L}_p})/T$. We note that condition (iv) contains information on a(relative) $(FF_{\mathcal{L}_p})$ and hence Theorem 8.1.7 is not free from conditions concerning (FF_B) -side. However, in general, it is much easier to find a subset (or a subgroup) with relative $(FF_B)/T$ (or relative (F_B)), than to verify the whole group has $(FF_B)/T$ (or (F_B)).

(*ii*) One example of such (G, H, Q) is the following: let G is a finitely generated group and a triple of subgroups in $G(H, N_1, N_2)$ has the Shalom property for (F_B) (see Definition 5.3.1). Then if G has (T_B) , then $(H, Q) = (H, N_1 \cup N_2)$ satisfies the conditions (i)-(v) in Theorem 8.1.8. If moreover for each i = $1, 2 \ G \ge N_i$ has relative $(FF_B)/T$, then $(H, Q) = (H, N_1 \cup N_2)$ satisfies the conditions (i)-(v) in Theorem 8.1.7. Note that in Shalom's machinery, it is essential that Q in above is the union of two subgroups, and H normalizes each of them.

Therefore, merits of introducing Theorem 8.1.7 are the following, in comparison to considering Shalom's machinery:

- (A) One can consider subsets $Q \subseteq G$, free from subgroup structures.
- (B) The stability of a Banach space (or a class) B under ultraproducts is *not* needed.
- (C) One can apply to deduce not only (F_B) , but also $(FF_B)/T$ (with approriate assumption).

On the other hand, the following point is the obstruction in applying Theorem 8.1.7 in general cases:

(T) One needs to have property (T_B) for G in advance.

Note that Shalom's machinery is a deep theorem, but the proof of Theorem 8.1.7 is quite easy: because the main point of this theorem is to reduce $(FF_B)/T$ to (T_B) , and (T_B) is in advance assumed. From this point of view, we can consider property (T_B) is the *origin* of these rigidities on B, although in general (T_B) is much weaker than (F_B) or $(FF_B)/T$ (if B is superreflexive). Therefore, Shalom's machinery is significant because this machinery provides us with a method to deduce $((F_B)$ and hence) (T_B) from relative (F_B) .

- (iii) Item (B) in the remark above is essential in proving Theorem C. Item (A) is essential in the proof of Theorem D. These theorems are based on Theorem 8.1.7.
- (iv) The proof of Theorem 8.1.7 indicates that in fact even for H, we only need the assumption $h^{-1}Qh \subseteq Q$ for all $h \in H$, and then H can be taken as a *subset* of G. We have stated in these theorems that H is assumed to be a subgroup simply because in practically use that case is a main case.
- (v) As we have mentioned, the proof of Theorem 8.1.7 is based on property (T_B) , namely, the non-existence of almost invariant vectors for induced representations on the quotient Banach space modulo the space of invariant vectors. Therefore, the situation is completely different for quasi-cocycles with trivial coefficients (this is the reason why we introduce $(FF_B)/T$). The study of cocycles with trivial coefficient is not difficult: it only depends on the abelianization of the group. However, the study of quasi-cocycles with trivial coefficient is much involved. A quasi-cocycles with trivial real coefficient $(1_G, \mathbb{R})$ is called a

quasi-homomorphism, and this object has intensively studied by many mathematicians. We will see some part of theories on quasi-homomorphisms in Chapter 10.

We note that it seems open to determine whether all quasi-homomorphisms on universal lattices are bounded. See Subsection 10.5.3 for details.

8.1.3 Proof of Theorem B (i)

Here we show item (i) of Theorem B. We restate it of the extracted form:

Theorem 8.1.10. (Theorem 1.5 [Mim1]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Then if $m \ge 4$, then for any p, universal lattice $G = SL_m(A)$ has property $(FF_{\mathcal{L}_p})/T$ and property $(FF_{[\mathcal{H}]})/T$.

In particular for any ρ , an isometric G-representation on an L^p space or a uniformly bounded G-representation on a Hilbert space which satisfies $\rho \not\supseteq 1_G$, then the comparison map in degree $2 H_b^2(G; \rho) \to H^2(G; \rho)$ is injective.

Proof. Let (H, N_1, N_2) be the triples of subgroups of G which is defined as in the proof of Theorem 6.3.1 (see Section 6.3). Set $(H, Q) = (H, N_1 \cup N_2)$. Then as we see in item (*ii*) Remark 8.1.9, this pair of a subgroup and a subset of G satisfies conditions (*ii*), (*iii*) and (v) in Theorem 8.1.7. We set $B = \mathcal{L}_p$ or $B = [\mathcal{H}]$ in Theorem 8.1.7.

By Theorem 5.4.1 and Theorem 3.3.7, G has $(T_{\mathcal{L}_p})$. By Corollary 6.3.2, G also has $(T_{[\mathcal{H}]})$. Hence G fulfills condition (i). Finally, by combining Theorem 6.1.1 and Theorem 8.1.5, we have that (G, Q) enjoys condition (iv). Therefore Theorem 8.1.7 applies, and we have accomplished the proof of Theorem 8.1.10.

By examining the proof above carefully, we obtain the following criterion for a Banach space (or a class of Banach spaces) that $G = \operatorname{SL}_{m \geq 4}(\mathbb{Z}[x_1, \ldots, x_k])$ has $(FF_B)/T$. This is one of the main results in the paper [Mim3] of the author. Here we need one definition.

Definition 8.1.11. ([Mim3]) Let E_1 be a Banach space (or a class of Banach spaces). Let E_2 be a Banach space (or a class of Banach spaces). Then we say *relative* property (T_{E_1}) implies relative property (T_{E_2}) if the following holds true: for a pair $G \ge N$ of a group and a (possibly non-proper) normal subgroup, whenever $G \ge N$ has relative (T_{E_1}), it has (T_{E_2}).

We say relative property (T) implies relative property (T_{E_2}) if for a pair $G \ge N$ of a group and a (possibly non proper) normal subgroup, whenever $G \ge N$ has relative (T), it has (T_{E_2}) .

Note that in above definition, if relative (T_{E_1}) implies relative (T_{E_2}) and if a group G has (T_{E_1}) , then G also has (T_{E_2}) (consider $G \succeq G$). Theorem 3.3.10 exactly means that relative (T) implies relative $(T_{\mathcal{L}_p})$ in the sense above.

Theorem 8.1.12. ([Mim3]) Let k is a non-negative integer and $A = \mathbb{Z}[x_1, \ldots, x_k]$. Let B be a class of superreflexive Banach spaces (or a superreflexive Banach space). Suppose B satisfies either of the following conditions:

- (i) the class B is stable under ultraproducts, and the pair $E_2(A) \ltimes A^2 \supseteq A^2$ has relative (T_B) ;
- (ii) relative $(T_{\mathcal{H}})$ implies relative (T_B) , in the sense in Definition 8.1.11.

Then for any $m \ge 4$, universal lattices $SL_m(A)$ has $(FF_B)/T$. In particular, $SL_m(A)$ has (F_B) .

In particular, if relative (T) implies relative (T_B), then for any $m \ge 4$, $SL_m(A)$ has $(FF_B)/T$ and (F_B) .

This observation is the key to proving Theorem C.

Remark 8.1.13. Here is a remark for condition (i) (Shalom's machinery) of the theorem above.

It is a problem of high interest to determine whether totally higher rank lattices and universal lattices have property $(F_{\mathcal{B}_{uc}})$ for \mathcal{B}_{uc} being the class of all uniformly convex Banach spaces. One of the main motivations for studying this problem is this relates to uniform (non-)embeddability of expander graphs constructed by the group into a uniformly convex Banach space (see Section 2.7. We also refer to the original argument of Gromov [Gro6] for existence of groups which are not uniformly embeddabile into Hilbert spaces. For details of this argument, also see a paper [ArDe] of Arzhantseva–Delzant). That relates the coarse geometric Novikov conjecture [KaYu] and (possible counter example of the surjectivity-side of) the Baum–Connes conjecture with coefficient (we refer to [HLS] in the connection of this). There is a breakthrough by V. Lafforgue [Laf1], [Laf2], and his results imply $\mathrm{SL}_{n>3}(\mathbb{K})$ (K is a non-archimedean local field) and cocompact lattices therein have this property (in fact, he shown property (F_B) for these groups where B is the class of all Banach spaces with type > 1. For details, see his original papers [Laf1], [Laf2]). For archimedean local field cases or noncocompact lattice cases, however, it seems no result is known for this problem.

In [BFGM], Bader–Furman–Gelander–Monod observed that for higher rank groups and lattices, in order to verify property $(F_{\mathcal{B}_{uc}})$ it suffices to show that pairs $SL_2(\mathbb{R}) \ltimes \mathbb{R}^2 \supseteq \mathbb{R}^2$ and $SL_2(\mathbb{R}) \ltimes S^{2*}(\mathbb{R}^2) \supseteq S^{2*}(\mathbb{R}^2)$ have relative property $(T_{\mathcal{B}_{uc}})$ (here the latter pair relates to symplectic groups, see Subsection 2.6.3). Thanks to Theorem 8.1.12, we have the following analogue of this observation for universal lattices: "if $E_2(A) \ltimes A^2 \supseteq A^2$ has relative property $(T_{\mathcal{B}_{uc}})$, then $SL_{m\geq 4}(A)$ has property $(F_{\mathcal{B}_{uc}})$." This is because the family of uniformly convex Banach spaces with uniform lower bounds for modulus of convexity is stable under ultraproducts (see Theorem 5.1.11), and hence a similar argument to one in the proof of Theorem 6.3.1 for the case of $\mathcal{C} = [\mathcal{H}]$ applies.

8.2 The case of *p*-Schatten spaces C_p

In this section, we prove Theorem C. This is an extension of a work of Puschnigg [Pus].

8.2.1 Basics on noncommutative L^p spaces

Firstly, we state some basic facts on noncommutative L^p spaces. We refer to [PiXu] for details. Since the *p*-Schatten class space is of our interest, we state in this case.

Let \mathfrak{H} be a separable Hilbert space. Then there exists a countable orthonormal basis $(\xi_n)_{n\in\mathbb{N}}$, and we fix it. For a *positive* operator $A \in \mathbb{B}(\mathcal{H})$, we define the *trace* of A by the following formula:

$$\operatorname{Tr}(A) := \sum_{n \in \mathbb{N}} \langle A\xi_i | \xi_i \rangle \in [0, \infty].$$

Note that this value is independent of the choices of orthonormal basis $(\xi_n)_n$.

Set the following subspace of $\mathbb{B}(\mathfrak{H})$:

$$C_1 := \{T \in \mathbb{B}(\mathfrak{H}) : \operatorname{Tr}(|T|) < \infty\}.$$

Here $|T| := (T^*T)^{1/2}$ denotes the absolute value (operator) of T. For $T \in C_1$, $\operatorname{Tr}(T) := \sum_{n \in \mathbb{N}} \langle T\xi_i | \xi_i \rangle \in \mathbb{C}$ is well-defined, and is independent of the choices of $(\xi_n)_n$. The space C_1 is closed with respect to the norm $||T||_1 := \operatorname{Tr}(|T|)$. For $p \in (1, \infty)$, similarly we define the space

$$C_p := \{ T \in \mathbb{B}(\mathfrak{H}) : \mathrm{Tr}(|T|^p) < \infty \},\$$

which is closed with respect to the norm

$$||T||_p := (\operatorname{Tr}(|T|^p))^{1/p}.$$

We call $(C_p, \|\cdot\|_p)$ the space of p-Schatten class operators.

Note that C_2 is in fact a Hilbert space, equipped with the following inner product:

$$\langle T|S\rangle := \operatorname{Tr}(S^*T).$$

This space C_2 is also called the space of *Hilbert–Schmidt operators* (we note that noncommutative L^p spaces are constructed even from type *III* von-Neumann algebras with normal a semi-finite weight (the construction is due to U. Haagerup and H. Kosaki). See Section 3 in [PiXu]). For duality, the following holds:

Theorem 8.2.1. Let $p, p' \in (1, \infty)$ with $p^{-1} + p'^{-1} = 1$. Then

$$(C_p)^* \cong C_{p'}$$

The following inequality is called the *Clarkson type inequality*. See Section 5 in [PiXu].

Theorem 8.2.2. (Clarkson type inequality) Let p, p' in $(1, \infty)$ with $p^{-1} + p'^{-1} = 1$. Then the following hold:

(i) If $p \leq 2$, then for any $S, T \in C_p$,

$$\left[\frac{1}{2}(\|S+T\|_p^{p'}+\|S-T\|_p^{p'})\right]^{1/p'} \le (\|S\|_p^p+\|T\|_p^p)^{1/p}.$$

(ii) If $p \geq 2$, then for any $S, T \in C_p$,

$$\left[\frac{1}{2}(\|S+T\|_p^p + \|S-T\|_p^p)\right]^{1/p} \le (\|S\|_p^{p'} + \|T\|_p^{p'})^{1/p'}.$$

By this theorem and the duality in above, it is straightforward that C_p is uniformly convex and uniformly smooth:

Corollary 8.2.3. For $p \in (1, \infty)$, C_p is ucus.

Finally, we state the following inequality of T. Ando:

Theorem 8.2.4. (Ando [Ando]) Let $0 < \alpha < 1$ and $p \in (1, \infty)$. Then for any positive operators A, B in C_p , there is an inequality:

$$||A^{\alpha} - B^{\alpha}||_{r}^{r} \le ||A - B||_{p}^{p}$$

Here $r = p/\alpha$.

We note that all of these three theorems are extended to general noncommutative L^p setting (the extension of last one is due to Kosaki).

Remark 8.2.5. (i) Although there are many similarities between noncommutative L^p spaces and commutative ones, it is known that any (infinite dimensional) noncommutative L^p spaces can*not* be isometrically embedded into commutative one, whenever $p \neq 2$ (for instance, see a book [HRS] of Haagerup-Rosenthal–Sukochev). In the view of Remark 3.3.3, this implies the kernel

$$C_p \times C_p \to \mathbb{R} \colon (S,T) \mapsto \|S - T\|^p$$

is not conditionally positive definite if $p \neq 2$, even when $p \in (1, 2)$.

(ii) The space C_p is not stable under ultraproducts because C_p itself is separable (see Example 5.1.7). In fact, even if we consider the class of all noncommutative L^p spaces (for fixed p) associated with normal semifinite trace on type I or II von Neumann algebras, it is not stable. If we consider ultraproducts, then we have to deal with the case of type III von Neumann algebras. See [Ray] for details.

8.2.2 Proof of Theorem C

We restate Theorem C in below, which is a main theorem in the paper [Mim3] of the author. Recall from Chapter 0 that we always assume p is in $(1, \infty)$.

Theorem 8.2.6. ([Mim3]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ $(k \in \mathbb{N})$ and $m \ge 4$. Take any p. Then the following hold true:

- (i) The universal lattice $G = SL_m(A)$ has $(FF_{C_p})/T$. In particular, G has (F_{C_p}) .
- (ii) Any finite index subgroup Γ of $SL_m(A)$ has $(FF_{C_p})/T$. In particular, Γ has (F_{C_p}) .
- (iii) Any totally higher algebraic rank group and totally higher rank lattice have (F_{C_p}) .

Firstly, we will prove item (i) in below. Recall that the criterion in Theorem 8.1.12 show there are two ways in establishing $(FF_B)/T$: either to use Shalom's machinery (consider ultraproducts); or verify (for instance) relative (T) implies relative (T_B) in the sense in Definition 8.1.11. Consider the case of C_p . By item (ii) of Remark 8.2.5, the former way does *not* work. Hence we aim to show that relative (T) implies relative (T_{C_p}) , as in the case of commutative L^p spaces. To sum up, our main goal is the following theorem:

Theorem 8.2.7. ([Mim3]) Relative (T) implies relative (T_{C_p}) , in the sense in Definition 8.1.11.

We note that item (iii) of Theorem 8.2.7 is essentially obtained by Puschnigg [Pus]. More precisely, he has proven the following theorem.

Theorem 8.2.8. (Theorem 5.8, Corollary 5.10 [Pus]) Let $p \in (1, \infty)$ and \mathfrak{H} be a separable Hilbert space. Denote by C_p the space of p-Schatten class operators.

(i) Let G be a totally higher rank algebraic group. Let π be a unitary G-representation on \mathfrak{H} , and let ρ be the isometric G-representation on C_p induced by π , namely, for $g \in G$ and $T \in C_p$ define

$$\rho(g)T = \pi(g)T\pi(g)^{-1}.$$

Then for any π the following holds:

$$H^1_{\mathbf{c}}(G;\rho,C_p) = 0.$$

 (ii) Let Γ be a totally higher rank lattice. Let π be a unitary Γ-representation on *n*, and let ρ be the isometric Γ-representation on C_p induced by π. Then for any π the following holds:

$$H^1(\Gamma; \rho, C_p) = 0.$$

Our proof has its origin in his proof of Theorem 8.2.8, and we will explain his strategy. Recall that in the case of commutative L^p spaces on σ -finite measure, there are two ways in proving Theorem 3.3.10:

- (1) utilize conditionally negative definite kernel $\|\xi \eta\|^p$;
- (2) combine the Mazur map and the Banach–Lamperti theorem.

By item (i) of Remark 8.2.5, strategy (1) does not work for the case of C_p . The strategy of Puschnigg [Pus] is to construct an analogue of the Mazur map between (unit spheres of) C_p 's. We note that in [Pus], he considered the case of that isometric representation is induced by unitary representations on \mathfrak{H} . This means, in his case, an analogue of the Banach-Lamperti theorem is not needed.

We explain his construction of the noncommutative analogue of the Mazur map. Recall that for any $T \in \mathbb{B}(\mathfrak{H})$, there is a *polar decomposition*:

$$T = U|T|.$$

Here $|T| = (T^*T)^{1/2}$ is the canonical positive operator, and U is a unitary operator (in general, we have many choices for U).

Definition 8.2.9. (Puschnigg [Pus]) Let $p, q \in (1, \infty)$. Define a noncommutative Mazur map by the following formula:

$$M_{p.q} \colon S(C_p) \to S(C_q); \ T \mapsto U|T|^{p/q}.$$

Here T = U|T| is a polar decomposition of T.

Note that although there are many choices for unitary U, $U|T|^{p/q}$ defines the same operator. Therefore the definition above is well-defined.

Puschinigg deduced uniform continuity of the noncommutative Mazur map as in Definition 8.2.9 from Theorem 8.2.4:

Theorem 8.2.10. (Corollary 5.6 [Pus]) For any $p, q \in (1, \infty)$, the noncommutative Mazur map $M_{p,q}: S(C_p) \to S(C_q)$ is uniformly continuous.

For the proof, we refer to the original paper [Pus].

Proof. (Theorem 8.2.7) We need an analogue of the Banach–Lamperti theorem, namely, a classification of linear isometries on C_p ($p \neq 2$). This is obtained by J. Arazy [Ara]. Since our Hilbert space \mathfrak{H} is separable, by choosing a (countable) orthogonal normal basis $(\xi_n)_n$, we can identify \mathfrak{H} with a square integrable sequence space $\ell^2 = \ell^2(\mathbb{N})$. Through this identification, we can consider the transpose map (associated with $(\xi_n)_n$); $S \mapsto {}^tS$ on $\mathbb{B}(\mathfrak{H}) \cong \mathbb{B}(\ell^2)$. Note that transpose maps are not canonical: it depends on the choice of $(\xi_n)_n$. However, it is easy to see the following: **Lemma 8.2.11.** Stick to the setting in above. Let $S \mapsto {}^{t}S$ be the transpose map on $\mathbb{B}(\ell^{2})$.

- (i) The transpose map is linear.
- (ii) The transpose map is compatible with the adjoint operation. Namely, for any $T \in \mathbb{B}(\ell^2)$,

$$(^{t}T)^{*} = {}^{t}(T^{*})$$

- (iii) For any $S, T \in \mathbb{B}(\ell^2)$, ${}^t(ST) = {}^tT^tS$.
- (iv) If U is a unitary, then so is ${}^{t}U$.
- (v) If T is positive, then so is ${}^{t}T$.
- (vi) If T is positive and $\alpha > 0$, then ${}^{t}(T^{\alpha}) = ({}^{t}T)^{\alpha}$.
- (vii) The transpose map is a (linear) isometry on each C_p .

The following is the classification theorem of Arazy:

Theorem 8.2.12. (Arazy [Ara]) Let $1 with <math>p \neq 2$ and C_p be the space of p-Schatten class operators on ℓ^2 . Then every linear isometry Z on C_p is either of the following two forms:

(1) there exist unitaries $W, V \in \mathcal{U}(\ell^2)$ such that

$$Z\colon C_p\to C_p;\ T\mapsto WTV;$$

(2) there exist unitaries $W, V \in \mathcal{U}(\ell^2)$ such that

$$Z \colon C_p \to C_p; \ T \mapsto W^t T V.$$

Thanks to Theorem 8.2.10 and Theorem 8.2.12, by following footsteps of Bader– Furman–Gelander–Monod (option (1) of the proofs of Theorem 3.3.7), we accomplish the conclusion of Theorem 8.2.7. More precisely, we follow the following argument:

Suppose a group pair $G \geq N$ does not have prelative (\mathcal{T}_{C_p}) . We identify the Hilbert space with ℓ^2 in the way explained in above. Then there exists an isometric G-representation ρ on C_p such that $\rho' \mid_N \succ 1_N$ Here $\rho' \mid N$ is the restriction of ρ on the subspace $C'_p := C'_{p,\rho(N)}$ (here we use Corollary 8.2.3 that C_p is ucus). By employing the noncommutative Mazur map in the sense in Definition 8.2.9, define a (possibly nonlinear) map $\pi: C_2 \to C_2$ by

$$\pi(g) = M_{p,2} \circ \rho(g) \circ M_{2,p} \quad (g \in G).$$

We claim that this $\pi(g)$ is in fact a linear map for $g \in G$. Indeed, take any linear isometry Z on C_p , and show $\widetilde{Z} := M_{p,2} \circ Z \circ M_{2,p}$ is a linear map. By Theorem 8.2.12,

Z is either of form (1) or of form (2) in the statement of the theorem. With use of Lemma 8.2.11, we treat each case as follows (we take unitaries W, V as in the statement):

(i) in the case of (1), for any $T \in C_2$ with a polar decomposition T = U|T|, a polar decomposition of $WU|T|^{2/p}V$ is $(WUV)(V^*|T|^{2/p}V)$. Therefore, we have

$$\widetilde{Z} \cdot T = WUV(V^*|T|^{2/p}V)^{p/2}$$
$$= WUVV^*|T|V$$
$$= WU|T|V = WTV.$$

Hence \widetilde{Z} is linear;

(*ii*) in the case of (2), for any $T \in C_2$ with a polar decomposition T = U|T|, a polar decomposition of $W^t(U|T|^{2/p})V = W({}^t|T|)^{2/pt}UV$ is

$$(W^{t}UV)\{V^{*}(^{t}U)^{*}(^{t}|T|)^{2/p}(^{t}U)V\}.$$

Therefore, we have

$$\widetilde{Z} \cdot T = W^{t}UV\{V^{*}({}^{t}U)^{*}({}^{t}|T|)^{2/p}({}^{t}U)V\}^{p/2} = W^{t}UVV^{*}({}^{t}U)^{*}({}^{t}|T|)^{t}UV = W^{t}|T|^{t}UV = W^{t}(U|T|)V = W^{t}TV.$$

Hence \widetilde{Z} is linear.

Therefore the π constructed in above is indeed a unitary representation (recall that C_2 is equipped with a natural inner product). Finally, by uniform continuity of the noncommutative Mazur maps (Theorem 8.2.10), it is not difficult to see that $\rho' \succ 1_{\Lambda}$ implies $\pi' \succ 1_{\Lambda}$ (, where π' is the restriction of π on the orthogonal complement of $(\ell^2)^{\pi(N)}$). This means $G \succeq N$ does not have relative property (T). This ends our proof.

We note that linear isometries on a general noncommutative L^p space (with $p \neq 2$) have been classified in terms of *Jordan homomorphisms*. For details on these results, see [She] and [Yea]. Also we refer to [Sto] for a study of Jordan homomorphisms.

By Theorem 8.2.7 and Theorem 8.1.12, item (i) of Theorem 8.2.6 is proven.

Proof. (Item (*ii*) of Theorem 8.2.6) We employ *p*-induction in this setting. This is also done by Puschnigg as follows: let (X, μ) is a Borel space with finite measure (in mind, we consider $X = \mathcal{D} = G/\Gamma$, for a general G group and *p*-integrable lattice

 $\Gamma \leq G$). He constructed $L^p(X, C_p)$ as the space of all measurable $\mathbb{B}(\mathfrak{H})$ -valued functions with finite norms, where we define the norm on $L^p(X, C_p)$ by

$$\|\xi\|_p^p := \int_X \operatorname{Tr}((\xi(x)^*\xi(x))^{p/2})d\mu(x).$$

Observe that again $L^2(X, C_2)$ becomes a Hilbert space.

Puschnigg shown the following:

Theorem 8.2.13. (Corollary 5.7, Puschnigg [Pus])

- (i) For any p, $L^p(X, C_p)$ is ucus.
- (ii) Let $p, q \in (1, \infty)$. In the setting above, a noncommutative Mazur map, defined by

 $M_{p,q}: S(L^p(X, C_p)) \to S(L^q(X, C_q)); \xi(x) \mapsto U_x |\xi(x)|^{p/q}$

is uniformly continuous. Here for each x, $\xi(x) = U_x |\xi(x)|$ is a polar decomposition.

Now suppose a finite index subgroup Γ of $G = \operatorname{SL}_m(A)$ does not have $(\operatorname{FF}_{C_p})/\operatorname{T}$. Then there exists an isometric Γ -representation σ on C_p and a quasi- σ -cocycle c such that $c' \colon \Gamma \to C'_{p,\sigma(\Gamma)}$ is unbounded. Take a p-induction of c (see Subsection 3.4.2 and Proposition 8.1.4). Then we can construct the induced representation $\rho = \operatorname{Ind}_{\Gamma}^G \sigma$ and the induced quasi- ρ -cocycle b. Then since Γ is of finite index in $G, b' \colon G \to L^p(G/\Gamma, C_p)'_{\rho(G)}$ is unbounded (here we really use the assumption of finite index. For general setting of $\Gamma \leq G$ being a cocompact lattice, there is a gap to fix). However in the view of Theorem 8.1.7 and Theorem 6.2.1, this implies that there exists a copy of $G_0 = \operatorname{E}_2(A) \ltimes A^2 \supseteq A^2 = N$ such that the relative Kazhdan constant for $\rho \mid_{G_0}$,

$$\mathcal{K}(G_0, N; S, \rho \mid_{G_0})$$

is zero (see Definition 3.2.7). Here S is some finite generating set of G_0 (to see the assertion above, take negations in the arguments in the proofs of these theorems).

However, in a similar argument in the proof of Theorem 8.2.7, by interpolating by the noncommutative Mazur map (Theorem 8.2.13), we deduce that then $G_0 \geq N$ does not have relative (T). This contradicts Theorem 4.2.3. Therefore, Γ has $(FF_{C_p})/T$, as desired.

Item (iii) of Theorem 8.2.6 can be shown in the similar manner to one in the proof above, together with the arguments in Subsection 3.4.1 (recall that induction of a (genuine) cocycle has no problem in deducing that the original cocycle is a coboundary from the induced cocycle being a coboundary).

Thus we complete our proof of Theorem 8.2.6, which is Theorem C.

8.3 Property (TT)/T

In this section, we show item (ii) of Theorem B. The key to proving this is a theorem of Ozawa [Oza], in a connection to *property* (TTT), which is invented by him.

8.3.1 Ozawa's property (TTT)

Definition 8.3.1. (Ozawa, [Oza]) Let G be a group.

- (i) A measurable map $b: G \to \mathfrak{H}$ from G to a Hilbert space \mathfrak{H} is called a *wq-cocycle* if the following two conditions are satisfied:
 - (a) There exists a (measurable) map $\pi: G \to U(\mathfrak{H})$, which is not assumed to be a group homomorphism, such that

$$\sup_{g,h\in G} \|\tilde{b}(gh) - \tilde{b}(g) - \pi(g)\tilde{b}(h)\| < \infty.$$

- (b) The map \tilde{b} is *locally bounded*. This means, for any compact set $S \subseteq G$, the set $\{\|\tilde{b}(s)\| : s \in S\} \subseteq \mathbb{R}_{\geq 0}$ is relatively compact.
- (*ii*) The group G is said to have *property* (TTT) if any wq-cocycle on G is bounded.
- (*iii*) Let $N \leq G$ is a subgroup. We say $G \geq N$ has relative property (TTT) if any wq-cocycle on G is bounded on N.

Recall that property (TT), which is identical to $(FF_{\mathcal{H}})$, is the boundedness property of all quasi-cocycles with unitary coefficients. A notion of wq-cocycles are weakening of that of quasi-cocycles. Therefore, (relative) property (TTT) is a strengthening of (relative) property (TT). At the present, there is no known example of groups with (TT) but without (TTT). For his motivations of inroducing (TTT) and application of that property, we refer to the original paper [Oza] and a paper [BOT] of Burger–Ozawa–Thom.

The following result is one of the main results in [Oza]:

Theorem 8.3.2. (Ozawa [Oza]) Let \mathbb{K} be any local field and $m \geq 3$. Then the group $SL_m(\mathbb{K})$ and lattices therein have property (TTT). In particular, $SL_{m\geq 3}(\mathbb{Z})$ has (TTT).

In the proof of item (ii) of Theorem B, we employ the following theorem. This is used by Ozawa to deduce the theorem above.

Theorem 8.3.3. (Ozawa [Oza]) Let $G = G_0 \ltimes A$ be the semidirect product of a abelian group by a continuous action of G_0 . Then the following are equivalent:

(i) The pair $G \ge A$ has relative (T).

(ii) The pair $G \ge A$ has relative (TTT).

In particular, if $G \ge A$ has relative (T), then it has relative (TT).

For the proof, see Proposition 3 in [Oza].

8.3.2 Universal lattices have (TT)/T

We restate the definition of property (TT)/T, because it has an interesting application (see Section 11.5).

Definition 8.3.4. ([Mim1], [Mim4]) A group G is said to have property (TT)/T if for any unitary G-representation π with $\pi \not\supseteq 1_G$, every π -cocycle is bounded.

By definition, property (TT)/T is identical to property $(FF_{\mathcal{H}})/T$.

By utilizing Theorem 8.3.3, we show item (ii) of Theorem B. We restate the assertion:

Theorem 8.3.5. (*Remark* 6.7, [Mim1]) Let $k \in \mathbb{N}$ and $A = \mathbb{Z}[x_1, \ldots, x_k]$. Then for $m \geq 3$, $G = SL_{m\geq 3}(A)$ possesses property (TT)/T.

Proof. Take the following subgroups (H, N_1, N_2) : $H \cong \operatorname{SL}_{m-1}(A)$, and $N_1, N_2 \cong A^{n-1}$. Here in G we realize H as in the left upper corner (, namely, the $((1-(m-1)) \times (1-(m-1)))$ -th parts), realize N_1 as in the $((1-(m-1)) \times m)$ -th unipotent parts, and realize N_2 as in the $(m \times (1-(m-1)))$ -th unipotent parts. And set $Q = N_1 \cup N_2$.

Then thanks to Theorem 8.3.3, this pair (H, Q) fulfills all conditions of Theorem 8.1.7 for $\mathcal{C} = \mathcal{H}$. Here the crucial point is on condition (iv). Therefore Theorem 8.1.7 applies, and we obtain property $(FF_{\mathcal{H}})/T(=(TT)/T)$.

We have accomplished the proof(s) of Theorem B. Note that by item (i) of Theorem B (Theorem 8.1.10), we are already done for the case of $m \ge 4$, and the essential part of Theorem 8.3.5 is the case of m = 3.

Chapter 9

Symplectic universal lattices

In this chapter, we consider an analogue of the elementary group over a ring (Chapter 4) in the symplectic setting. In this case, it is called the *elementary symplectic* group. Note that this concept is only defined over a *commutative* ring.

We see an analogue of the Suslin stability theorem, which is due to Grunewald– Mennicke–Vaserstein. We define the notion of *symplectic universal lattices* in a similar manner to one in the definition of universal lattices. We see a symplectic version of Vaserstein's bounded generation for symplectic universal lattices, which is also due to Vaserstein. Note that although it looks similar to that for universal lattices, their properties are different from the viewpoint of Shalom's machinery. Namely, Shalom's machinery does *not* work for symplectic universal lattices.

Next, we see a celebrated result of Ershov–Jaikin-Zapirain that noncommutative universal lattices have (T), and its extension, which is a recent work of Ershov– Jaikin-Zapirain–Kassabov. Specially, we see their result that symplectic universal lattices as well have (T).

Finally, we appeal to this result, and deduce Theorem D. In particular, we see that symplectic universal lattices possess (TT)/T.

9.1 Definition

In this section, we introduce the notion of *symplectic universal lattices*, which is parallel to that of universal lattices.

9.1.1 Elementary symplectic groups– one realization

Firstly, for $m \ge 1$, we take the 2m by 2m alternating matrix as

$$L_m = \begin{pmatrix} L & 0 & 0 & 0 \\ 0 & L & 0 & 0 \\ 0 & 0 & \ddots & 0 \\ 0 & 0 & 0 & L \end{pmatrix} \in \mathcal{M}_{2m}.$$

Here

$$L = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix} \in \mathcal{M}_2.$$

This choice of alternating matrix is standard, but in connection to property (T), this choice is not good. Hence this alternating matrix L_m later shall be replaced with another one J_m in this thesis. See the next subsection for details.

Let A be a *commutative* ring. Then the *symplectic group* over A of degree 2m is the following group:

$$\operatorname{Sp}_{2m}(A) = \{ g \in \operatorname{M}_2m(A) : {}^tgL_mg = L_m \}.$$

Here ${}^{t}g$ is the transpose matrix of g. We use the following permutation symbol on \mathbb{Z} :

$$(2i)' := 2i - 1; \ (2i - 1)' = 2i \quad (i \in \mathbb{N}).$$

For any (i, j) with $1 \le i \le 2m$, $1 \le i \le 2m$ and $i \ne j$; and any $a \in A$, we define the *elementary symplectic matrix* as follows:

$$SE_{i,j}(a) := \begin{cases} I_2m + ae_{i,j} & \text{if } i = j'; \\ I_2m + ae_{i,j} - (-1)^{i+j}ae_{j',i'} & \text{if } i \neq j'. \end{cases}$$

Here $e_{i,j}$ is the matrix with (i, j)-th entry 1 and the other entries 0. The *elementary* symplectic group over A of degree 2m is the group generated by all elementary symplectic matrices, and it is written as $\operatorname{Ep}_{2m}(A)$. A priori, $\operatorname{Ep}_{2m}(A)$ is a (possibly non-proper) subgroup of $\operatorname{Sp}_{2m}(A)$.

Then there are some basic commutator relations between elementary symplectic matrices. They are so many, and here we only state some important relations. For the all precise relations, we refer to Lemma 4.1 of the paper [GMV1] of Grunewald–Menniche–Vaserstein.

Lemma 9.1.1. Let $a, b \in A$ are any element. Then there are the following formulae:

- (i) (1) $[SE_{i,i'}(a), SE_{i',l}(b)] = SE_{i,l}(ab)SE_{l',l}((-1)^{i+l}ab^2),$ if $i \neq l, i' \neq l$.
 - (2) $[SE_{i,i'}(a), SE_{k,i'}(b)] = SE_{i,i'}(2ab + (-1)^{i+k}ab^2),$ if $i \neq k, i' \neq k.$

- (3) $[SE_{i,i'}(a), SE_{k,i}(b)] = SE_{k,i'}(-ab)SE_{k,k'}(-(-1)^{i+k}ab^2),$ if $i \neq k, i' \neq k.$
- (4) $[SE_{i,i'}(a), SE_{i,l}(b)] = SE_{l',l}(-2ab (-1)^{i+l}ab^2),$ if $i \neq l, i' \neq l.$

(*ii*) (1)
$$[SE_{i,j}(a), SE_{j,l}(b)] = SE_{i,l}(ab),$$

if $i' \neq j, j' \neq l, i \neq l, i' \neq l.$

- (2) $[SE_{i,j}(a), SE_{j,l}(b)] = SE_{i,l}(2ab),$ if $i' \neq j, j' \neq l, i \neq l, i' = l.$
- (3) $[SE_{i,j}(a), SE_{k,j'}(b)] = SE_{k,i'}((-1)^{i+j}ab),$ if $i' \neq j, j' \neq k, i' \neq k, i \neq k.$
- (4) $[SE_{i,j}(a), SE_{k,j'}(b)] = SE_{i,i'}(2(-1)^{i+j}ab),$ if $i' \neq j, j' \neq k, i' \neq k, i = k.$

These relation implies that if $m \ge 2$, then whenever A is finitely generated (as a ring), $\operatorname{Ep}_{2m}(A)$ is a finitely generated group. However, we warn that we need some care on a finite generating set of $\operatorname{Ep}_{2m}(A)$. If one looks Lemma 9.1.1 carefully, then one will notice that some rules are different between elementary symplectic matrices of the form $SE_{i,j}(a)$ $(i' \ne j)$; and those of the form $SE_{i,i'}(a)$. Structures of the latter in terms of commutators is much more complicated than that of the former. Indeed, let $m \ge 2$ and suppose A is finitely generated and $S = \{s_1, \ldots, s_k\}$ is a finite generating set for A (as a ring). Then the following finite set is a generating set for $\operatorname{Ep}_{2m}(A)$:

$$S := \{ SE_{i,j}(\pm s_l) : 1 \le i, j \le 2m, i \ne j, i' \ne j; 0 \le l \le k \} \\ \cup \{ SE_{i,i'}(\pm s_1^{\epsilon_1} s_2^{\epsilon_2} \cdots s_k^{\epsilon_k}) : 1 \le i \le 2m; \epsilon_1, \dots, \epsilon_k \in \{0, 1\} \}.$$

Here we regard $s_0 = 1$ and $s_l^0 = 1$.

Note that if $m \ge m_0$, then there is a natural inclusion

$$\operatorname{Ep}_{2m_0}(A) \hookrightarrow \operatorname{Ep}_{2m}(A),$$

sending to the left upper corner.

Remark 9.1.2. Some reasons why we need the *commutativity* of a ring are the following: firstly, if a ring R is not commutative, then the transpose map $M_{2m}(R) \rightarrow M_{2m}(R)$ is not well-defined (more precisely, if we transpose just entries, then this map is *not* an anti-homomorphism). Therefore the group $\operatorname{Sp}_{2m}(R)$ is *not* well-defined. Secondly, if R is noncommutative, then the commutator relation stated above (Lemma 9.1.1) does *not* hold.

If R itself admits the transpose map (for instance let $R = \mathbb{Z}G$ for a group G and define ${}^{t}\delta_{g} := \delta_{g^{-1}}$), then it is possible to define $\operatorname{Sp}_{2m}(R)$. However, definition of

elementary symplectic matrices needs an appropriate modification. It is still possible to define $\operatorname{Ep}_{2m}(R)$ over such a ring after some modification, but there is no reason to expect finite generation for $\operatorname{Ep}_{2m}(R)$ even if R is finitely generated.

As we have mentioned in above, we use *another* realization for Sp(A) and Ep(A) in this thesis. That means, we choose *different* $2m \times 2m$ alternating matrix, as in the next subsection.

9.1.2 Another realization

Although the realization in the previous subsection is natural, it has some difficulty to examine (T)-type properties. Therefore, we use the following *another* realization throughout this thesis, even though it may look somewhat awkward.

In the rest part of this thesis, we basically use the following alternating matrix:

$$J_m := \begin{pmatrix} 0 & I_m \\ -I_m & 0 \end{pmatrix} \in \mathcal{M}_{2m}$$

Note that J_m is always conjugate to L_m , and that hence the following definitions are equivalent to ones in the previous subsection.

Definition 9.1.3. Let $m \ge 1$ and A be a commutative ring.

(i) The symplectic group over A of degree 2m, written as $\text{Sp}_{2m}(A)$, is defined as the multiplicative group of symplectic matrices in matrix ring $M_{2m}(A)$ associated with the alternating matrix J_m , namely,

$$\operatorname{Sp}_{2m}(A) := \{ g \in \operatorname{M}_{2m}(A) : {}^{t}gJ_{m}g = J_{m} \}.$$

- (*ii*) Matrices of the following form are called *elementary symplectic matrices*:
 - (1) For $1 \le i, j \le m$ with $i \ne j$, define

$$B_{i,j}(a) := I_{2m} + a(e_{i,m+j} + e_{j,m+i}),$$

$$C_{i,j}(a) := I_{2m} + a(e_{m+j,i} + e_{m+i,j})(= {}^{t}B_{i,j}(a)),$$

$$D_{i,j}(a) := I_{2m} + ae_{i,j} - ae_{m+j,m+i}.$$

(2) For $1 \le i \le m$, define

$$B_{i,i}(a) := I_{2m} + ae_{i,m+i},$$

$$C_{i,i}(a) := I_{2m} + ae_{m+i,i} (= {}^{t}B_{i,i}(a)).$$

Here $a \in A$ is any element.

Namely, for $i \neq j$,

$$B_{i,j}(a) := \begin{pmatrix} I_m & a(e_{i,j} + e_{j,i}) \\ 0 & I_m \end{pmatrix}, C_{i,j}(a) := \begin{pmatrix} I_m & 0 \\ a(e_{i,j} + e_{j,i}) & I_m \end{pmatrix},$$
$$D_{i,j}(a) := \begin{pmatrix} I_m + ae_{i,j} & 0 \\ 0 & I_m - ae_{j,i} \end{pmatrix};$$

and for i = j,

$$B_{i,i}(a) := \begin{pmatrix} I_m & ae_{i,i} \\ 0 & I_m \end{pmatrix}, C_{i,i}(a) := \begin{pmatrix} I_m & 0 \\ ae_{i,i} & I_m \end{pmatrix}$$

(*iii*) The elementary symplectic group $\operatorname{Ep}_{2m}(A)$ over A of degree 2m is the subgroup of $\operatorname{Sp}_{2m}(A)$ generated by all elementary symplectic matrices in the sense above

Note that in item (*ii*) above, the case of $i \neq j$ corresponds to that of $SE_{l,k}(a)$ $(l' \neq k)$; and the case of i = j corresponds to that of $SE_{l,l'}(a)$ in the previous chapter. However, the indices are permuted.

The following is mere interpretation of an observation in the previous subsection to this realization:

Lemma 9.1.4. Let $m \ge 2$ and suppose A is finitely generated and $S = \{s_1, \ldots, s_k\}$ is a finite generating set for A (as a ring). Then the following finite set is a generating set for $\text{Ep}_{2m}(A)$:

$$S := \{ B_{i,j}(\pm s_l), C_{i,j}(\pm s_l), D_{i,j}(\pm s_l) : 1 \le i, j \le m, i \ne j; 0 \le l \le k \} \\ \cup \{ B_{i,i}(\pm s_1^{\epsilon_1} s_2^{\epsilon_2} \cdots s_k^{\epsilon_k}), C_{i,i}(\pm s_1^{\epsilon_1} s_2^{\epsilon_2} \cdots s_k^{\epsilon_k}) : 1 \le i \le m; \epsilon_1, \dots, \epsilon_k \in \{0, 1\} \}.$$

Here we regard $s_0 = 1$ and $s_l^0 = 1$.

Finally, we define the following identification for certain subgroups of $\text{Sp}_{2m}(A)$ (or $\text{Ep}_{2m}(A)$):

Definition 9.1.5. Let *A* be a commutative ring.

(i) For $m \ge m_0 \ge 2$, by $\operatorname{SL}_{m_0}(A) \leqslant \operatorname{Sp}_{2m}(A)$ (or, $\operatorname{SL}_{m_0}(A) \hookrightarrow \operatorname{Sp}_{2m}(A)$) we mean the inclusion is realized in the following way:

$$\left\{ \begin{pmatrix} W & 0 & 0 & 0 \\ 0 & I_{m-m_0} & 0 & 0 \\ 0 & 0 & {}^t W^{-1} & 0 \\ 0 & 0 & 0 & I_{m-m_0} \end{pmatrix} : W \in \mathrm{SL}_{m_0}(A) \right\} \leqslant \mathrm{Sp}_{2m}(A).$$

(ii) Let $m \geq 2$. We denote by $S^{m*}(A^m)$ the additive group of all symmetric matrices in $M_m(A)$. By $E_m(A) \ltimes S^{m*}(A^m) \trianglerighteq S^{m*}(A^m)$, we identify these groups respectively with

$$\left\{ (W,v) := \left(\begin{array}{c|c} W & v \\ \hline 0 & t(W^{-1}) \end{array} \right) : W \in \mathcal{E}_m(A), \ v \in S^{m*}(A^m) \right\}$$
$$\ge \left\{ \left(\begin{array}{c|c} I_m & v \\ \hline 0 & I_m \end{array} \right) : v \in S^{m*}(A^m) \right\}.$$

Thus the action of $E_m(A)$ on $S^{m*}(A^m)$ is:

$$(W,0)(I_m,v)(W^{-1},0) = (I_m, Wv^t W) \quad (W \in \mathcal{E}_m(A), v \in S^{m*}(A^m)).$$

Here $E_m(A)$ denotes the elementary group over A (see Section 4.1).

(*iii*) Let $m \ge m_0 \ge 2$. Then by $\operatorname{Sp}_{2m_0}(A) \le \operatorname{Sp}_{2m}(A)$ (or, $\operatorname{Sp}_{2m_0}(A) \hookrightarrow \operatorname{Sp}_{2m}(A)$), we mean the inclusion is realized as

$$\left\{ \begin{pmatrix} P & 0 & Q & 0 \\ 0 & I_{m-m_0} & 0 & 0 \\ R & 0 & S & 0 \\ 0 & 0 & 0 & I_{m-m_0} \end{pmatrix} : \begin{pmatrix} P & Q \\ R & S \end{pmatrix} \in \operatorname{Sp}_{2m_0}(A) \right\} \leqslant \operatorname{Sp}_{2m}(A).$$

The advantage of this realization of $\text{Sp}_{2m}(A)$ (, namely the choice of J_m) is that then it is easy to express the pair in item (*ii*). This pair is the base point for study on property (T) for symplectic groups, as we have seen in Subsection 2.6.3.

Remark 9.1.6. It is easy to see that if m = 1, then for any commutative ring A, $\operatorname{Sp}_2(A) = \operatorname{SL}_2(A)$, and $\operatorname{Ep}_2(A) = \operatorname{E}_2(A)$. Nevertheless, in this thesis, (for $m \ge 2$) we distinguish $\operatorname{Sp}_2(A) \leq \operatorname{Sp}_{2m}(A)$ and $\operatorname{SL}_2(A) \leq \operatorname{Sp}_{2m}(A)$. Namely, in this thesis the former is realized as in item (iii) in the definition above, and the latter is realized as in item (i) in the definition above. Similarly, we distinguish $\operatorname{Ep}_2(A) \leq \operatorname{Sp}_{2m}(A)$ and $\operatorname{E}_2(A) \leq \operatorname{Sp}_{2m}(A)$. These distinctions have important meaning on study of property (T) on symplectic universal lattices in connection to Vaserstein's bounded generation for Sp. These shall be examined in Subsection 9.2.1.

9.1.3 The Grunewald–Mennicke–Vaserstein stability theorem

As is parallel to elementary group cases, it is natural to ask in which case $\operatorname{Ep}_{2m}(A)$ coincides with $\operatorname{Sp}_{2m}(A)$. Situations are extremely different between the case of m = 1 and the case of $m \geq 2$. In the paper [GMV1], Grunewald–Mennicke–Vaserstein proven the following stability theorem:

Theorem 9.1.7. ([GMV1]) Let $k \in \mathbb{N}$. Then for any $m \ge 2$, there are equalities

$$\operatorname{Ep}_{2m}(\mathbb{Z}[x_1,\ldots,x_k]) = \operatorname{Sp}_{2m}(\mathbb{Z}[x_1,\ldots,x_k]);$$

$$\operatorname{Ep}_{2m}(\mathbb{Z}[x_1^{\pm},\ldots,x_k^{\pm}]) = \operatorname{Sp}_{2m}(\mathbb{Z}[x_1^{\pm},\ldots,x_k^{\pm}]).$$

In fact, they shown that these equalities hold true even if \mathbb{Z} is replaced with any euclidean domain.

In parallel to the definition of universal lattices of Shalom, in forthcoming paper [Mim4], the author defined a notion of *symplectic universal lattices* as follows.

Definition 9.1.8. ([Mim4]) Let $m \ge 2$. Take any $k \in \mathbb{N}$. A symplectic universal *lattice* of degree 2m denotes a group

$$\operatorname{Sp}_{2m}(\mathbb{Z}[x_1,\ldots,x_k]) (= \operatorname{Ep}_{2m}(\mathbb{Z}[x_1,\ldots,x_k])).$$

9.2 Symplectic universal lattices have (T)

In this section, we see the fact that symplectic universal lattices have (T). First, we see some analogue of Vaserstein's bounded generation in this case (this is also due to Vaserstein [Vas2]). However, we then see this is *not* sufficient to deduce (T) (in other words, Shalom's machinery does *not* work). Finally, we state a theorem of Ershov–Jaikin-Zapirain–Kassabov on property (T) for (twisted) Steinberg groups, from which (T) for symplectic universal lattices follows.

9.2.1 Vaserstein's bounded generation for Sp

Before stating Vaserstein's generation, we state a fundamental relative property (T) concerning $\text{Sp}_{2m}(A)$. This is due to M. Neuhauser [Neu], and is obtained by idea inspired by Shalom's theorem [Sha1] for elementary group case (Theorem 4.2.1, Theorem 4.2.3):

Theorem 9.2.1. (Neuhauser, Theorem 3.3, [Neu]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ for any integer $k \in \mathbb{Z}$. Then for $m \ge 2$, the pair $\mathbb{E}_m(A) \ltimes S^{m*}(A^m) \trianglerighteq S^{m*}(A^m)$, which is defined as in item (ii) of Definition 9.1.5, has relative (T).

Here we state Vaserstein's bounded generation for symplectic universal lattices:

Theorem 9.2.2. (Vaserstein [Vas2]) Let $A_k = \mathbb{Z}[x_1, \ldots l_k]$ for any k. Then for any $m \ge 2$, $\operatorname{Sp}_{2m}(A_k)$ is boundedly generated by the set of elementally symplectic matrices and the subgroup $\operatorname{Sp}_2(A_k)$.

Here $\operatorname{Sp}_2(A_k)$ is realized in $\operatorname{Sp}_{2m}(A_k)$ as in item (iii) of Definition 9.1.5, namely,

$$\operatorname{Sp}_{2}(A_{k}) := \left\{ \begin{pmatrix} a & 0 & b & 0 \\ 0 & I_{m-1} & 0 & 0 \\ \hline c & 0 & d & 0 \\ 0 & 0 & 0 & I_{m-1} \end{pmatrix} : \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{Sp}_{2}(A_{k}) \right\} \leqslant \operatorname{Sp}_{2m}(A_{k}).$$

Now we explain why Shalom's machinery does *not* work for symplectic universal lattices. Consider for instance the case of $G = \text{Sp}_4(A)$ $(A = \mathbb{Z}[x_1, \ldots x_k])$. Then in the view of Theorem 9.2.1, the following triple (H, N_1, N_2) is a standard candidate for pairs with the Shalom property (for (T)):

$$H = \mathrm{SL}_{2}(A) := \left\{ \begin{pmatrix} W & | & 0 \\ 0 & | & {}^{t}W^{-1} \end{pmatrix} : W \in \mathrm{SL}_{2}(A) \right\}$$
$$= \left\{ \begin{pmatrix} a & b & | & 0 & 0 \\ c & d & 0 & 0 \\ \hline 0 & 0 & | & d & -c \\ 0 & 0 & | & -b & a \end{pmatrix} : \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \mathrm{SL}_{2}(A) \right\}$$
$$N_{1} := \left\{ \begin{pmatrix} I_{2} & | & v \\ \hline 0 & | & I_{2} \end{pmatrix} : v \in S^{2*}(A^{2}) \right\} \cong S^{2*}(A^{2}),$$
$$N_{2} := \left\{ \begin{pmatrix} I_{2} & | & 0 \\ \hline v' & | & I_{2} \end{pmatrix} : v' \in S^{2*}(A^{2}) \right\} \cong S^{2*}(A^{2}).$$

The point here is the following, as in Remark 9.1.6: even though the group $\operatorname{Sp}_2(A)$ and $\operatorname{SL}_2(A)$ are isomorphic as abstract groups, their realization inside $\operatorname{Sp}_{2m}(A)$ are completely different. Moreover, a group relating to Vaserstein's bounded generation is $\operatorname{Sp}_2(A)$; but a group relating to relative (T) is $\operatorname{SL}_2(A)$. Therefore at the present, it seems impossible to apply Shalom's machinery.

9.2.2 A theorem of Ershov–Jaikin-Zapirain

As we have mentioned in Remark 5.4.2, property (T) for noncommutative lattices is shown by Ershov–Jaikin-Zapirain [ErJa] (with some estimation for Kazhdan constant).

Theorem 9.2.3. (Ershov–Jaikin-Zapirain [ErJa]) Noncommutative universal lattice

$$G = \mathcal{E}_{m \ge 3}(\mathbb{Z}\langle x_1, \dots, x_k \rangle)$$

has property (T).

In particular, for any finitely generated ring and any $m \ge 3$, $E_m(R)$ has (T).

Their argument is inspired by a work of Dymara–Januszkiewicz [DyJa]. Main idea is for given unitary representation (π, \mathfrak{H}) of a group G and finite family of subgroups $N_1, \ldots N_n$ of G that generate G, estimates angles of spaces $(\mathfrak{H}^{\rho(N_i)})_{1 \leq i \leq n}$. If each $G \geq N_i$ has relative (T) and the angles above are sufficiently big (close to orthogonal, far from coinciding), then G itself has (T). The actual proof is much more involved, and we refer to the original paper [ErJa]. One of the point of this result is that they does *not* appeal to bounded generation. They moreover have proven property (T) for *Steinberg groups*.

Definition 9.2.4. Let $m \ge 3$ and R be a ring. The *Steinberg group* over R of degree m is defined as follows: the group generated by $\{x_{i,j}(r) : 1 \le i, j \le m, i \ne j\}$ (the set of formal generators) which subjects to the following commutator relations:

$$\begin{aligned} x_{i,j}(r)x_{i,j}(s) &= x_{i,j}(r+s), \\ [x_{i,j}(r), x_{k,l}(s)] &= e & \text{if } i \neq l, j \neq k, \\ [x_{i,j}(r), x_{j,k}(s)] &= x_{i,k}(rs) & \text{if } i \neq k. \end{aligned}$$

The Steinberg group is written as $St_m(R)$.

Steinberg groups are generalization of elemetary groups. More precisely, there is a natural surjective homomorphism:

$$\operatorname{St}_m(R) \twoheadrightarrow \operatorname{E}_m(R); \ x_{i,j}(r) \mapsto E_{i,j}(r).$$

Therefore, the following result of Ershov–Jaikin-Zapirain is more universal than Theorem 9.2.3:

Theorem 9.2.5. (Ershov–Jaikin-Zapirain [ErJa]) For any finitely generated ring and any $m \geq 3$, the Steinberg group $St_m(R)$ has (T).

Also, there is a concept of the *twisted Steinberg group* (over a commutative ring) associated with a reduced irreducible classical root system. For instance, commutator relations as in Lemma 9.1.1 (and more) yields one example of twisted Steinberg group, and in this case it maps onto elementary symplectic groups (in this case, associated root system is so-called of type C_n). In a recent work of Ershov–Jaikin-Zapirain–Kassabov, they have established property (T) for such twisted Steinberg groups, whenever the rank of the associated root system is at least 2:

Theorem 9.2.6. ([EJK]) Let Φ be a reduced irreducible classical root system of rank at least 2 and A be a finitely generated commutative ring. Then $St_{\Phi}(A)$, the twisted Steinberg group over A, has property (T).

In particular, every symplectic universal lattice $(m \ge 2)$,

$$\operatorname{Sp}_{2m}(\mathbb{Z}[x_1,\ldots,x_k]) (= \operatorname{Ep}_{2m}(\mathbb{Z}[x_1,\ldots,x_k]))$$

has property (T).

In the next section, we show property (TT)/T for symplectic universal lattices, by appealing to this theorem. Observe that Theorem 9.2.6 amounts to condition (i)in Theorem 8.1.7. **Remark 9.2.7.** The proof of Theorem 9.2.3 does not directly apply to the case of property $(F_{\mathcal{L}_p})$ because angle estimate is special to Hilbert spaces. Therefore at the present, it seems open to determine whether noncommutative lattices have $(F_{\mathcal{L}_p})$. However, Theorem 8.1.7 provides us with some relative property:

Proposition 9.2.8. ([Mim4]) Let $R = \mathbb{Z}\langle x_1, \ldots, x_k \rangle$ $(k \in \mathbb{N})$. Then for any $m \ge 4$, $\mathbb{E}_m(R) \ge \mathbb{E}_{m-1}(R)$ has relative $(FF_{\mathcal{L}_p})/T$. In particular, this pair has relative $(F_{\mathcal{L}_p})$. Here the inclusion above is realized as the subgroup sitting in the left upper corner.

Also, for any $m \ge 3$, $\mathbb{E}_m(R) \ge \mathbb{E}_{m-1}(R)$ has relative (TT)/T.

Proof. This is a mere application of Theorem 8.1.7, and can be done in a similar argument to ones in the proof(s) of Theorem B. Note that all conditions but condition (v) (bounded generation) in Theorem 8.1.7 are valid. For condition (iv), observe that Theorem 8.1.5 can be extended to noncommutative universal lattice cases without any change in the proof.

9.3 Symplectic universal lattices have (TT)/T

The goal of this section is to prove the following theorem, which is item (ii) of Theorem D (see Definition 8.3.4 for the definition of (TT)/T):

Theorem 9.3.1. ([Mim4]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ $(k \in \mathbb{N})$. Then for any $m \ge 2$, symplectic universal lattice $\operatorname{Sp}_{2m}(A)$ has $(\mathrm{TT})/\mathrm{T}$.

We will see that Vaserstein's bounded generation (Theorem 9.2.2) works in connection to Theorem 8.1.7, although it is not enough for Shalom's machinery.

Proof. We only prove in the case of m = 2 (other case also follows from Theorem 9.2.2 and the proof below). Set $G = \text{Sp}_4(A)$. We take the subgroup $\text{Sp}_2(A) \leq G$, as in item (*iii*) in Definition 9.1.5, and the following (infinite) subset $P \subseteq G$. Namely,

154

we take the following subgroup H and subset P of G:

$$H = \left\{ \begin{pmatrix} a & 0 & | & b & 0 \\ 0 & 1 & 0 & 0 \\ \hline c & 0 & | & d & 0 \\ 0 & 0 & | & 0 & 1 \end{pmatrix} : \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{Sp}_2(A) \right\} \leq \operatorname{Sp}_{2m}(A),$$

$$P = \{B_{1,2}(r), C_{1,2}(r), D_{1,2}(r), D_{2,1}(r), B_{2,2}(r), C_{2,2}(r) : r \in A\}$$

$$= \left\{ \begin{pmatrix} 1 & 0 & | & 0 & r \\ 0 & 1 & r & 0 \\ \hline 0 & 0 & | & 1 & 0 \\ 0 & 0 & | & 1 & 0 \\ \hline 0 & 0 & | & 1 & 0 \\ \hline 0 & 0 & | & 1 & 0 \\ \hline 0 & 0 & | & 1 & 0 \\ \hline 0 & 0 & | & -r & 1 \end{pmatrix}, \begin{pmatrix} 1 & 0 & | & 0 & 0 \\ 0 & 1 & 0 & 0 \\ \hline 0 & 0 & | & 1 & 0 \\ \hline 0 & 0 & | & -r & 1 \end{pmatrix}, \begin{pmatrix} 1 & 0 & | & 0 & 0 \\ 0 & 1 & 0 & 0 \\ \hline 0 & 0 & | & 0 & 1 \\ \hline 0 & 0 & | & 0 & 1 \\ \hline 0 & 0 & | & 0 & 1 \\ \hline 0 & 0 & | & 0 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 0 & | & 0 & 0 \\ \hline 1 & 0 & 0 \\ \hline 0 & 0 & | & 0 & 1 \\ \hline 0 & 0 & | & 0 & 1 \end{pmatrix}, \begin{pmatrix} 1 & 0 & | & 0 & 0 \\ \hline 0 & 1 & 0 & 0 \\ \hline 0 & 0 & | & 0 & 1 \\ \hline 0 & 0 & | & 0 & 1 \end{pmatrix} : r \in A \right\}$$

$$\subseteq \operatorname{Sp}_4(A).$$

Define a new subset $Q \subseteq G$ as

$$Q := \bigcup_{h \in H} h P h^{-1}.$$

We will show that this pair (H, Q) in G fulfills all conditions of Theorem 8.1.7 for $\mathcal{C} = \mathcal{H}$. Firstly, condition (i), namely, property (T) for G follows from Theorem 9.2.6 (here we are appealing to this deep theorem)! Condition (iii) (invariance of Q by conjugation of H) is trivial by construction. Next, we will see condition (ii) (generation of G by Q). For each element in P, if we rewrite them in the realization by another alternating matrix L_m as in Subsection 9.1.1, then we have the following correspondences:

alternating matrix:	J_2	\longleftrightarrow	L_2
	$B_{1,2}(r)$	\longleftrightarrow	$SE_{1,4}(r),$
	$C_{1,2}(r)$	\longleftrightarrow	$SE_{4,1}(r),$
	$D_{1,2}(r)$	\longleftrightarrow	$SE_{1,3}(r),$
	$D_{2,1}(r)$	\longleftrightarrow	$SE_{3,1}(r),$
	$B_{2,2}(r)$	\longleftrightarrow	$SE_{3,4}(r),$
	$C_{2,2}(r)$	\longleftrightarrow	$SE_{4,3}(r),$
	$B_{1,1}(r)$	\longleftrightarrow	$SE_{1,2}(r),$
	$C_{1,1}(r)$	\longleftrightarrow	$SE_{2,1}(r).$

Therefore by Lemma 9.1.1, we have following equalities:

$$B_{1,1}(r) = D_{2,1}(-r)[B_{2,2}(r), C_{1,2}(1)],$$

$$C_{1,1}(r) = D_{1,2}(r)[C_{2,2}(r), B_{1,2}(1)].$$

These equalities show every elementary symplectic matrix is a product of at most 5 elements in P (and hence Q). This shows condition (*ii*). Note that Theorem 9.2.2 states G is boundedly generated by H and elementary symplectic matrices. Therefore, condition (v) (bounded generation by H and Q) holds true.

It remains to check condition (iv) (, namely, that $G \supseteq Q$ has relative (TT)/T). We will prove in fact $G \subseteq Q$ has relative (TT). Theorem 9.2.1 (relative (T)) together with Theorem 8.3.3 implies relative (TTT) (hence in particular relative (TT)) for the pair $E_2(A) \ltimes S^{2*}(A^2) \trianglerighteq S^{2*}(A^2)$. Thus for any copy of $E_2(A) \ltimes S^{2*}(A^2)$ in $G = \text{Sp}_4(A), S^{2*}(A^2)$ -part has relative (TT) (with respect to G). This implies the following: "let $Y \subseteq G$ be the set of all elementary symplectic matrices. Then $G \supseteq Y$ has relative (TT)". Finally, for each

$$H \ni h = \left(\begin{array}{cc} a & b \\ c & d \end{array}\right)$$

(the identification is as in above) and for each element l in P, we compute hlh^{-1} . Then we have the following:

$$hB_{1,2}(r)h^{-1} = B_{1,2}(ar)D_{2,1}(-cr),$$

$$hC_{1,2}(r)h^{-1} = C_{1,2}(dr)D_{1,2}(br),$$

$$hD_{1,2}(r)h^{-1} = C_{1,2}(cr)D_{1,2}(ar),$$

$$hD_{1,2}(r)h^{-1} = B_{1,2}(-br)D_{2,1}(dr),$$

$$hB_{2,2}(r)h^{-1} = B_{2,2}(r),$$

$$hC_{2,2}(r)h^{-1} = C_{2,2}(r).$$

Therefore, every element in Q is a product of at most 2 elements in Y. This shows that $G \supseteq Q$ has relative (TT), and hence condition (iv) is fulfilled.

We have checked all conditions (i)-(v), and thus Theorem 8.1.7 ends our proof.

9.4 Property $(FF_B)/T$ for symplectic universal lattices

Here we show item (i) of Theorem D. We restate it here:

Theorem 9.4.1. ([Mim4]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ $(k \in \mathbb{N})$. Then for any $m \ge 3$, a symplectic universal lattice $\operatorname{Sp}_{2m}(A)$ has $(\operatorname{FF}_{\mathcal{L}_p})/\operatorname{T}$ and $(\operatorname{FF}_{\mathcal{L}_p})/\operatorname{T}$. In particular, it possesses $(\operatorname{F}_{\mathcal{L}_p})$ and $(\operatorname{F}_{\mathcal{L}_p})$.

156

9.4. PROPERTY $(FF_B)/T$ FOR SYMPLECTIC UNIVERSAL LATTICES 157

For the proof, we employ the following result, which is parallel to Theorem 8.1.5.

Theorem 9.4.2. ([Mim4]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$. Suppose B is any superreflexive Banach space. If the pairs $\mathbb{E}_2(A) \ltimes A^2 \supseteq A^2$; and $\mathbb{E}_2(A) \ltimes S^{2*}(A^2) \supseteq S^{2*}(A^2)$ have relative property (\mathbb{T}_B) , then the pair $\mathrm{SL}_3(A) \ltimes S^{3*}(A^3) \ge S^{3*}(A^3)$ has relative property (FF_B) .

Proof. The proof of Theorem 9.4.2 goes along the line of the base case: that of Theorem 8.1.5, but it requires more care. Set $G = \operatorname{SL}_3(A) \ltimes S^{3*}(A^3)$ and $N = S^{3*}(A^3) \leq G$, and consider G (and N) as subgroup(s) in $\operatorname{Sp}_6(A)$, as in item (*ii*) of Definition 9.1.5. Let ρ be an isometric G-representation on B and $b: G \to B$ be an (arbitrary) quasi- ρ -cocycle. Take a ucus ρ -invariant norm on B and fix it. Take a decomposition $B = B_0 \oplus B_1 := B^{\rho(N)} \oplus B'_{\rho(N)}$, and decompose b as

$$b = b_0 + b_1,$$

where $b_0: G \to B_0$ and $b_1: G \to B_1$. Then for i = 1, 2, by $\rho(G)$ -invariance of B_i, b_i becomes a quasi- ρ -cocycle. Set the following two subgroups N_1, N_2 of N:

$$N_{1} := \left\{ B_{1,1}(r) = \begin{pmatrix} |r & 0 & 0 \\ I_{3} & 0 & 0 & 0 \\ 0 & 0 & 0 \\ \hline 0 & I_{3} \end{pmatrix} : r \in A \right\},$$
$$N_{2} := \left\{ B_{1,2}(r) = \begin{pmatrix} |0 & r & 0 \\ I_{3} & r & 0 & 0 \\ \hline 0 & I_{3} \end{pmatrix} : r \in A \right\}.$$

Then in a similar way to one in the proof of Theorem 8.1.5, the following two can be verified:

- (1) The quasi-cocycle b_0 is bounded on N.
- (2) If b_1 is bounded both on N_1 and on N_2 , then it is bounded on N.

Therefore for the proof of the theorem, it suffices to show the following two assertions:

- (A1) The set $b_1(N_1)$ is bounded.
- (A2) The set $b_1(N_2)$ is bounded.

In below, we shall prove assertions (A1) and (A2). Recall from Definition 9.1.3 the definitions of $B_{i,j}(r)$ and $D_{i,j}(r)$. Also recall from Lemma 9.1.4 a finite generating set of $\text{Sp}_6(A)$. Set

$$S = \{B_{i,j}(\pm x_l), D_{i,j}(\pm x_l) : 1 \le i, j \le 3, i \ne j, 0 \le l \le k\} \\ \cup \{B_{i,i}(\pm x_1^{\epsilon_1} \cdots x_k^{\epsilon_k}) : 1 \le i \le 3, \epsilon_1, \dots, \epsilon_k \in \{0, 1\}\},\$$

where $x_0 = 1$ and $x_l^0 = 1$. Then \mathcal{S} is a finite generating set of $G(\leq \text{Sp}_6(A))$.

Firstly, we verify assertion (A1). Define a *finite* subset S_0 of G as follows:

$$\begin{split} \mathcal{S}_{1} &:= \left\{ B_{1,2}(\pm x_{l}), B_{1,3}(\pm x_{l}), B_{2,3}(\pm x_{l}), D_{1,2}(\pm x_{l}), D_{1,3}(\pm x_{l}), D_{2,3}(\pm x_{l}), D_{3,2}(\pm x_{l}) : 0 \leq l \leq k \right\} \\ &\cup \left\{ B_{i,i}(\pm x_{1}^{\epsilon_{1}} \cdots x_{k}^{\epsilon_{k}}) : 1 \leq i \leq 3, \epsilon_{1}, \dots, \epsilon_{k} \in \{0, 1\} \right\} \\ &= \left\{ \begin{pmatrix} 1 & * & * & * & * \\ 0 & 1 & * & * & * & * \\ 0 & * & 1 & * & * & * \\ 0 & * & 1 & * & * & * \\ & & & 1 & 0 & 0 \\ 0 & & & * & 1 & * \\ & & & & * & * & 1 \end{pmatrix} \right\} \cap \mathcal{S}. \end{split}$$

 Set

$$C := \max\{\sup_{s \in S_1} \|b_1(s)\|, \sup_{g,h \in G} \|b_1(gh) - b_1(g) - \rho(g)b_1(h)\|\} < \infty.$$

It is a key observation to this proof that N_1 commutes with S_1 . Therefore for any $l \in N_1$ and $s \in S_1$, the following inequalities hold:

$$\begin{aligned} \|b_1(l) - \rho(s)b_1(l)\| &\leq \|b_1(l) - b_1(sl)\| + \|b_1(s)\| + C \\ &\leq \|b_1(l) - b_1(sl)\| + 2C \\ &= \|b_1(l) - b_1(ls)\| + 2C \\ &\leq \|b_1(l) - b_1(l) - \rho(l)b_1(s)\| + 3C \\ &\leq 4C. \end{aligned}$$

Note that the term in the very below is independent of the choice of $l \in N_1$. Now we define the following pairs of subgroups (G', N'); (H, H') of G:

$$G' := \left\{ \begin{cases} 1 & 0 & 0 & 0 \\ 0 & W' & 0 & v' \\ \hline 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & t(W')^{-1} \end{cases} : W' \in \mathcal{E}_2(A), v' \in S^{2*}(A^2) \right\}$$
$$\ge \{g \in G' : W' = I_2\} := N';$$
$$H := \left\{ \begin{cases} 1 & tu & 0 & 0 \\ 0 & W' & 0 & 0 \\ \hline 0 & 0 & 1 & 0 \\ 0 & 0 & t(W')^{-1}u & t(W')^{-1} \\ 0 & 0 & t(W')^{-1}u & t(W')^{-1} \\ \end{bmatrix} : W' \in \mathcal{E}_2(A), u \in A^2 \right\}$$
$$\ge \{h \in H : W' = I_2\} := H'.$$

Set a real number \mathcal{K} by the following formula:

$$\mathcal{K} := \min\{\mathcal{K}(G', N'; \mathcal{S} \cap G', \rho \mid_{G'}), \mathcal{K}(H, H'; \mathcal{S} \cap H, \rho \mid_{H})\}$$

(recall the definition of the relative Kazhdan constant for property (T_B) from Definition 3.2.7). Then by the assumptions of relative property (T_B) for $E_2(A) \ltimes S^{2*}(A^2) \supseteq S^{2*}(A^2)$; and $E_2(A) \ltimes A^2 \supseteq A^2$, \mathcal{K} is strictly positive. Therefore, we have the following inequalities by the inequalities above and Lemma 3.2.8:

for any
$$\xi \in b_1(N_1)$$
, $\|\xi - \rho(n')\| \le 16\mathcal{K}^{-1}C$,
and $\|\xi - \rho(h')\| \le 16\mathcal{K}^{-1}C$.

Here $n' \in N'$ and $h' \in H'$ are arbitrary elements. Note the following:

$$N' = \{ B_{2,3}(r), B_{2,2}(r), B_{3,3}(r) : r \in A \},\$$

$$H' = \{ D_{1,2}(r), D_{1,3}(r) : r \in A \}.$$

Now we observe the following correspondences among elementary symplectic matrices according to choices of alternating matrices $(J_3 \leftrightarrow L_3)$:

alternating matrix:	J_3	\longleftrightarrow	L_3
	$B_{2,3}(r)$	\longleftrightarrow	$SE_{3,6}(r),$
	$D_{1,2}(r)$	\longleftrightarrow	$SE_{1,3}(r),$
	$D_{1,3}(r)$	\longleftrightarrow	$SE_{1,5}(r),$
	$B_{2,2}(r)$	\longleftrightarrow	$SE_{3,4}(r),$
	$B_{3,3}(r)$	\longleftrightarrow	$SE_{5,6}(r),$
	$B_{1,2}(r)$	\longleftrightarrow	$SE_{1,4}(r),$
	$B_{1,3}(r)$	\longleftrightarrow	$SE_{1,6}(r),$
	$B_{1,1}(r)$	\longleftrightarrow	$SE_{1,2}(r).$

By Lemma 9.1.1 (item (*ii*) (1) and item (*i*) (3)), we have the following equalities: for any $r \in A$,

$$B_{1,3}(r) = [D_{1,2}(1), B_{2,3}(r)], \quad B_{1,1}(r) = B_{1,3}(-r)[B_{3,3}(-r), D_{1,3}(1)].$$

These equalities together with the inequalities in the paragraph above imply that for any $\xi \in b_1(N_1)$ and any $r \in A$,

$$\|\xi - \rho(B_{1,3}(r))\xi\| \le 64\mathcal{K}^{-1}C$$
, and $\|\xi - \rho(B_{1,1}(r))\xi\| \le 128\mathcal{K}^{-1}C$.

In a similar way, we also have $\|\xi - \rho(B_{1,2}(r))\xi\| \le 64\mathcal{K}^{-1}C$ in the setting above. Therefore we conclude the following:

for any
$$\xi \in b_1(N_1)$$
 and any $n \in N$, $\|\xi - \rho(n)\xi\| \leq 304\mathcal{K}^{-1}C$.

Note that the right hand side is independent of the choices of $\xi \in b_1(N_1)$ and $n \in N$. By (a generalization to uc Banach space cases of) Corollary 2.1.14 and the trivial fact that $(B'_{\rho(N)})^{\rho(N)} = 0$, this inequality forces that

$$\sup_{\xi \in b_1(N_1)} \|\xi\| \le 304 \mathcal{K}^{-1} C.$$

This means that $b_1(N_1)$ is bounded. Thus we have shown assertion (A1).

Finally, we will confirm assertion (A2). Define a *finite* subset S_2 of G as follows:

$$S_{2} := \{B_{1,2}(\pm x_{l}), B_{1,3}(\pm x_{l}), B_{2,3}(\pm x_{l}), D_{1,2}(\pm x_{l}), D_{2,1}(\pm x_{l}), D_{1,3}(\pm x_{l}), D_{2,3}(\pm x_{l}) : 0 \le l \le k\}$$

$$\cup \{B_{i,i}(\pm x_{1}^{\epsilon_{1}} \cdots x_{k}^{\epsilon_{k}}) : 1 \le i \le 3, \epsilon_{1}, \dots, \epsilon_{k} \in \{0, 1\}\}$$

$$= \left\{ \begin{pmatrix} 1 & * & * & * & * \\ * & 1 & * & * & * & * \\ 0 & 0 & 1 & * & * & * \\ \hline & 1 & * & 0 \\ 0 & & * & 1 & 0 \\ & & & * & * & 1 \end{pmatrix} \right\} \cap S.$$

Then the following holds true:

for any
$$s \in S_2$$
 and $\gamma \in N_2$, there exists $l \in N_1$ such that $s\gamma = \gamma sl$.

More precisely, l = e unless s is of the form of $D_{1,2}$ or $D_{2,1}$. Since we have already verified assertion (A1), we know that $b_1(N_1)$ is bounded. Therefore

$$C' := \max\{\sup_{s \in \mathcal{S}_2} \|b_1(s)\|, \sup_{g,h \in G} \|b_1(gh) - b_1(g) - \rho(g)b_1(h)\|, \sup_{l \in N_1} \|b_1(l)\|\}$$

is a finite real number. We have the following inequalities for any $\gamma \in N_2$ and $s \in \mathcal{S}_2$:

$$\begin{aligned} \|b_{1}(\gamma) - \rho(s)b_{1}(\gamma)\| &\leq \|b_{1}(\gamma) - b_{1}(s\gamma)\| + \|b_{1}(s)\| + C' \\ &\leq \|b_{1}(\gamma) - b_{1}(s\gamma)\| + 2C' \\ &= \|b_{1}(\gamma) - b_{1}(\gamma sl)\| + 2C' \\ &\leq \|b_{1}(\gamma) - b_{1}(\gamma) - \rho(\gamma)b_{1}(s) - \rho(\gamma s)b_{1}(l)\| + 4C' \\ &\leq 6C'. \end{aligned}$$

Here $l \in N_1$ is chosen such that $s\gamma = \gamma sl$ as in above. Note that the very below term in these inequalities is independent of the choice of $\gamma \in N_2$. Therefore we can verify assertion (A2) in a similar way to one in the proof of assertion (A1).

We have verified assertions (A1) and (A2), and thus have completed the proof of the theorem.

Proof. (Theorem 9.4.1) For simplicity, we give a proof for the case of m = 3. Let

160

 $G = \text{Sp}_6(A)$, we take the following subgroup H and subset P of G:

$$H = \operatorname{Sp}_{2}(A) := \left\{ \begin{pmatrix} a & 0 & b & 0 \\ 0 & I_{2} & 0 & 0 \\ \hline c & 0 & d & 0 \\ 0 & 0 & 0 & I_{2} \end{pmatrix} : \begin{pmatrix} a & b \\ c & d \end{pmatrix} \in \operatorname{Sp}_{2}(A) \right\} \leqslant \operatorname{Sp}_{6}(A)$$
$$P = \{B_{1,2}(r), C_{1,2}(r), D_{1,2}(r), D_{2,1}(r), B_{2,2}(r), C_{2,2}(r), B_{i,3}(r), C_{i,3}(r), D_{j,3}(r), D_{3,j}(r) : r \in A, 1 \le i \le 3, 1 \le j \le 2 \}.$$

Define $Q \subseteq G$ as

$$Q = \bigcup_{h \in H} hPh^{-1}$$

It is the same as in the proof of Theorem 9.3.1 that the pair (H, Q) in G satisfies conditions (ii), (iii), and (v). For condition (i), we use the facts that relative (T)implies relative $(T_{\mathcal{L}_p})$ and relative (T_{C_p}) (Theorem 3.3.10 and Theorem 8.2.7).

For condition (iv), we utilize Theorem 9.4.2 together with Theorem 9.2.1 and Theorem 4.2.3. Theorem 8.1.7 ends our proof.

Thus we have accomplished the proofs of Theorem D.

Remark 9.4.3. We warn that we have *no* result on property $(FF_{[\mathcal{H}]})/T$. This is because in the proof of the theorem above, it is crucial that relative (T) implies relative $(T_{\mathcal{L}_p})$ and relative $(T_{\mathcal{C}_p})$. Since Shalom's machinery does not work in symplectic setting, we do not have the statement that implies $(T_{[\mathcal{H}]})$ for universal symplectic lattices. This means, if we try to apply Theorem 8.1.7, then condition (1) becomes an obstruction.

Chapter 10

Quasi-homomorphisms and stable commutator length

In this chapter, all groups are assumes to be discrete, and we allow uncountable discrete groups.

Until this point, we have examined quasi-cocycles modulo trivial linear part. In this chapter, we turn into studies on quasi-cocycles with trivial (real) coefficient, which are called *quasi-homomorphisms*. Firstly, we start from defining apparently different object, called the *scl* (Stable Commutator Length) on the commutator subgroup of a group.

Secondly, we define this celebrated concept, and see some nontrivial examples on F_2 . In the views of theorems of Epstein–Fujiwara [EpFu] and Burger– Monod [BuMo1], [BuMo2], this concept has "higher rank versus rank 1" phenomena. Thirdly, we see the homogenization procedure for quasi-homomorphism, which is special for trivial coefficient cases. By utilizing this, we state the celebrated *Bavard duality theorem*, which connects (homogeneous) quasi-homomorphisms and scl.

Finally, we state a question of M. Abért and N. Monod on (un)boundedness of commutator length and scl. Then we see some answers to this question, one by A. Muranov [Mur] and another by the author [Mim2]. Specially, we prove Theorem E.

A comprehensive reference in this chapter is a book [Cal2] of D. Calegari, which has the name "*scl*."

10.1 Scl

Recall the assumption that in this chapter all groups are discrete, and that we allow uncountable discrete groups. Also recall that our commutator convention is: $[g,h] = ghg^{-1}h^{-1}$.

Definition 10.1.1. Let G be a group.

(i) The set of single commutators means

$$\{[g,h]:g,h\in G\}.$$

The commutator subgroup, written as [G, G], is the subgroup of G generated by (single) commutators.

- (ii) The commutator length cl: $[G, G] \to \mathbb{Z}_{\geq 0}$ is the word length on [G, G] with respect to the set of single commutators. Namely, for $g \in [G, G]$, cl(g) is the minimum of numbers which is needed to express g as a product of single commutators. We set cl(e) = 0.
- (*iii*) The stable commutator length, also written as scl, is

$$\operatorname{scl}: [G,G] \to \mathbb{R}_{\geq 0}; \quad \operatorname{scl}(g) := \lim_{n \to \infty} \frac{\operatorname{cl}(g^n)}{n}.$$

- (iv) The group G is said to be *perfect* if [G, G] = G.
- (v) Suppose G is perfect. Then the *commutator width* is defined as

$$\sup_{g\in G} \operatorname{cl}(g) \in \mathbb{N} \cup \{\infty\}.$$

Note that cl is subadditive, and hence that the limit in the definition of scl always exists.

Example 10.1.2. (i) Let $F_2 = \langle a, b \rangle$ be a free group of rank 2. Then

$$\operatorname{cl}([a,b]) = 1,$$

but there is a fact that

$$\operatorname{scl}([a,b]) = \frac{1}{2}.$$

We will sketch the proof of this fact in below (see Subsection 10.3.2). From the viewpoint of rigidity, what is important here is this value of scl is *strictly positive*. Namely, there exists $g \in [F_2, F_2]$ such that scl(g) > 0 (in fact, in the case of F_2 , any nontrivial element $g \in [F_2, F_2]$ has $scl(g) \ge 1/2$).

(*ii*) Consider $G = SL_3(\mathbb{Z})$. Then commutator relation

$$[E_{i,j}(r), E_{j,k}(s)] = E_{i,k}(rs) \quad (i \neq j, j \neq k, k \neq i; r, s \in \mathbb{Z})$$

shows G is perfect. For this G, it is known that

$$\operatorname{scl} \equiv 0 \quad \text{on } [G, G] = G.$$

10.2. QUASI-HOMOMORPHISMS

One reason to extend the definition of cl to that of scl is that in many cases it is much more computable than cl (the meaning of this will be more clear after stating the Bavard duality theorem. see Subsection 10.3.2). Also, by considering scl (not cl itself), we become able to consider the comparison as in the example above: namely, "vanishing identically" versus "non-vanishing" in terms of scl. It has been revealed by works of several mathematicians that this concept scl has deep consequences and background. In this thesis, we are unable to see these beautiful theories due to the limit of contents. We refer to [Cal2] for this direction. For instance, Calegari [Cal1] shown that scl takes only rational values if G is a free group.

10.2 Quasi-homomorphisms

Here we define apparently quite distinct concept from scl, quasi-homomorphisms. As we have seen in Chapter 8 and Chapter 9, quasi-cocycles on trivial coefficient are in some sense the most difficult cases, because property (T_B) -type property gives no information on trivial linear part. In theories of quasi-homomorphisms, we usually consider the trivial real coefficient.

10.2.1 Definition

We state the definition of *quasi-homomorphisms* on a group. We note that it is (more) common to call them *quasimorphisms*. Compare with Definition 7.1.6 for the general definitions of quasi-cocycles and actual space of them. Also recall in this chapter we assume all groups are discrete, and we allow uncountable discrete groups.

Definition 10.2.1. Let G be a group.

(i) A map $\psi: G \to \mathbb{R}$ is called a *quasi-homomorphism* (or *quasimorphism*) if

$$\sup_{g,h\in G} |\psi(gh) - \psi(g) - \psi(h)| < \infty.$$

We write the left hand side of the formula above as $D(\psi)$, and call it the *defect* of ψ .

- (ii) Define QH(G) as the space of all quasi-homomorphisms on G.
- (*iii*) Define QH(G) as the following quotient space:

 $\widetilde{QH}(G)$:= $QH(G)/(\{\text{homomorphisms on } G\} + \{\text{maps } G \to \mathbb{R} \text{ with bounded range}\}).$

We call QH(G) the actual space of quasi-homomorphisms.

(iv) We say a quasi-homomorphism ψ is trivial (or cohomologically trivial) if the image of ψ by the projection $QH(G) \twoheadrightarrow \widetilde{QH}(G)$ is zero. Namely, if ψ can be written as the sum of a homomorphism and a bounded map $G \to \mathbb{R}$.

Note that since bounded homomorphism (to \mathbb{R}) is a zero map, in the definition of $\widetilde{QH}(G)$ the sum of vector spaces is in fact a direct sum.

The following lemma is the special case of Lemma 7.1.8. We note that second bounded cohomology with trivial (real) coefficient is one of the most important object in the theory of bounded cohomology (for the definitions of bounded cohomology and of comparison map, see respectively Definition 7.1.5 and Definition 7.1.7).

Lemma 10.2.2. Let G be a group. Then there is a natural isomorphism among vector spaces,

$$\widetilde{QH}(G) \cong \operatorname{Ker} \Psi_{\mathrm{b}}^2.$$

Here $\mathrm{Ker}\Psi^2_b$ denotes the comparison map in degree 2

$$\Psi^2 \colon H^2_{\mathbf{b}}(G) \to H^2(G),$$

where $H^2_{\rm b}(G) = H^2_{\rm b}(G; \mathbf{1}_G, \mathbb{R})$ denotes the second bounded cohomology of G with trivial coefficient $(\mathbf{1}_G, \mathbb{R})$.

10.2.2 Examples on free groups

Let F_2 be a free group of rank 2, and a, b be free generators. In below, we will see that the actual space of quasi-homomorphisms $\widetilde{QH}(F_2)$ is enormous. In the view of Lemma 10.2.2 (recall that $H^2(F_2) = 0$), this implies behavior of bounded cohomology is quite different from ordinary cohomology (as we have seen with left regular representation coefficients in Subsection 7.1.2). First example of nontrivial quasi-homomorphism on F_2 is due to B. E. Johnson [Joh] (the author acknowledges P. de la Harpe for pointing this out to him). Here we see different examples, which are respectively due to R. Brooks [Bro] and P. Rolli [Rol].

Example 10.2.3. (i) (Brooks [Bro]) Consider a reduced word w in F_2 and fix it. Since F_2 is free, any element $g \in F_2$ has a canonical form as a reduced word. Set

 $|g|_w :=$ maximum of numbers of disjoint w's appearing in the reduced word g.

Here we considering the order. For instance, $|ababab^{-1}|_{aba} = 1$, $|ab^2aba^{-3}|_{ab} = |abbaba^{-1}a^{-1}a^{-1}|_{ab} = 2$, $|b^{-1}a|_{ab^{-1}} = 0$, and $|(ab)^{-1}|_{ab} = |b^{-1}a^{-1}|_{ab} = 0$.

Then this $|\cdot|$ itself is *not* a quasi-homomorphism because in the multiplication process

$$(g,h) \mapsto gh,$$

10.2. QUASI-HOMOMORPHISMS

there may be a plenty of cancellations. However, if we set

$$\psi_w(g) := |g|_w - |g|_{w^{-1}}$$

Then it can be shown that this ψ_w is a quasi-homomorphism. Indeed, $D(\psi_w) \leq 3$.

Next we see that there exists a reduced word w for which $\psi_w \in QH(F_2)$ is nontrivial. For example, w = [a, b] does the job. Indeed, suppose ψ_w is trivial, namely, there exists a homomorphism f and a bounded map c such that

$$\psi_w = f + c$$

Then by considering $[a, b]^n$ $(n \in \mathbb{N})$, we have

$$n = f([a, b]^n) + c([a, b]^n) = c([a, b]^n).$$

Here the last equality follows from the assumption that f is a homomorphism. Since c is bounded, this contradicts as we let $n \to \infty$.

We note in below that there is a geometric interpretation for this construction: consider the Cayley graph Cay $(F_2; S)$ of F_2 with respect to the generating set $S = \{a^{\pm}, b^{\pm}\}$ (recall Definition 2.5.9). This is a 4-regular tree. Then a reduced word w can be seen as a directed path from e to w (as a group element). Recall F_2 acts on Cay $(F_2; S)$ isometrically by left multiplication (, namely, $h \in F_2$ acts by $F_2 \ni l \mapsto hl$), and this induces *copies* of a directed path $w = \{h \cdot w : h \in F_2\}$. Now observe that since F_2 is free, for any $g \in G$ there exists a *unique* geodesic path from e to g. With this setting, we have the following natural interpretation of $|\cdot|_w$:

> $|g|_w$ =maximum of numbers of disjoint copies of wwhich appears in the directed path from e to q.

- (*ii*) (Rolli [Rol]) Take any function $\sigma \colon \mathbb{Z} \to \mathbb{R}$ with following two conditions:
 - (1) Anti-symmetry: for any $n \in \mathbb{Z}$, $\sigma(-n) = -\sigma(n)$.
 - (2) Boundedness: $\|\sigma\|_{\infty} := \sup\{|\sigma(n)| : n \in \mathbb{Z}\} < \infty$.

Note that condition (1) forces $\sigma(0) = 0$. Now note that any element in G can be expressed as the following "almost reduced word":

$$g = a^{e_1} b^{f_1} \cdots a^{e_k} b^{f_k},$$

where e_j , f_j $(1 \le j \le k)$ except e_1 and f_k must be non-zero. Then Rolli's construction $\sigma \mapsto \psi_{\sigma}$ is defined by the following formula:

$$\psi_{\sigma}(g) = \sum_{j=1}^{k} (\sigma(e_j) + \sigma(f_j)).$$

It is straightforward to check that $\psi_{\sigma} \in QH(F_2)$, which is left to the readers (indeed, $D(\psi_{\sigma}) \leq 3 \|\sigma\|_{\infty}$).

We next show that there exists a nontrivial $\psi_{\sigma} \in QH(F_2)$. In fact, it is always the case, unless $\sigma \equiv 0$. To see this observe that if $\sigma \not\equiv 0$, then there exists $n_1, n_2 > 0$ such that $\sigma(n_2) \neq 0$ and

$$\frac{n_1}{n_2} \neq \frac{\sigma(n_1)}{\sigma(n_2)}.$$

We then claim ψ_{σ} is non-trivial. Indeed, suppose ψ_{σ} is trivial. Then there exists a homomorphism f and a bounded map c such that $\psi_{\sigma} = f + c$. For any $k \in \mathbb{N}$, consider $g^k = (a^{n_1}b^{n_1})^k$ and $h^k = (a^{n_1}b^{n_1})^k$. Then we have

$$\frac{\sigma(n_1)}{\sigma(n_2)} = \frac{2k\sigma(n_1)}{2k\sigma(n_2)} = \frac{\psi_\sigma(g^k)}{\psi_\sigma(h^k)} = \frac{kn_1f(ab) + c_1}{kn_2f(ab) + c_2}$$

If we let $k \to \infty$, then we see that f(ab) cannot be zero and that the very right hand side in the equality tends to n_1/n_2 . Contradiction occurs.

In fact, both of the examples above implies the following:

Proposition 10.2.4. The space $QH(F_2)$ is infinite dimensional.

It is not difficult to see this space in fact has continuum dimension. In the view of Lemma 10.2.2, this means the space $H_{\rm b}^2(F_2)$ has continuum dimension.

10.2.3 Higher rank versus rank one

Recall that Brooks' quasi-homomorphisms (item (i) in Example 10.2.3) has a geometric interpretation. For the construction, the following point is essential: for any two distinct points s, t in the Cayley graph $\operatorname{Cay}(F_2; S)$ (, which is tree) there exists a unique geodesic path from s to t. Epstein–Fujiwara [EpFu] extended Brooks' construction in the case of that G is hyperbolic (, namely, a Cayley graph associated with a finite generating set is a hyperbolic space in the sense of Gromov, see Definition 2.6.19) as follows: a hyperbolic group G acts on its Cayley graph $X = \operatorname{Cay}(G; S)$ (Definition 2.5.9) by isometries from the left, where S is an (arbitrarily) fixed finite symmetric generating set of G. Fix a base point $x_0 \in X$ (for instance, $x_0 = e$). Fix an (oriented) path w on X, and consider the *copies* of w:

$$\{\gamma \cdot w : \gamma \in G\}.$$

For $g \in G$, set $c_w(g) \in \mathbb{Z}_{\geq 0}$ by the following formula:

$$c_w(g) := d(x_0, g \cdot x_0) - \inf_{p:\text{path}} \{ |p| - \max\{\text{number of disjoint copies of } w \text{ on } p \} \}.$$

Here p moves among (oriented) paths from x_0 to $g \cdot x_0$, d denotes the distance on X, and |p| means the length of the path p. Finally, define $\phi_w \colon G \to \mathbb{R}$ as

$$\phi_w(g) := c_w(g) - c_{w^{-1}}(g).$$

Epstein–Fujiwara shown that this ϕ_w is in QH(G), whenever $|w| \geq 2$. It follows from the hyperbolicity of (X, d), and this construction has its origin in (the geometric interpretation) of Brooks' construction of quasi-homomorphisms on F_2 . They also examined the necessary and sufficient condition of a hyperbolic group G such that there exists a path w with $[\phi_w] \neq 0$ in $\widetilde{QH}(G)$. Thus the following theorem has been proven:

Theorem 10.2.5. (Epstein–Fujiwara [EpFu]) Let G be a non-elementary hyperbolic group. Then the space $\widetilde{QH}(G)$ is infinite dimensional. In particular, $H^2_{\rm b}(G)$ is infinite dimensional.

Recall from Definition 7.2.1 a hyperbolic group is *non-elementary* if G is neither finite nor virtually \mathbb{Z} (a group is said to be virtually \mathbb{Z} if it contains \mathbb{Z} with finite index). This condition that hyperbolic group G is non-elementary is equivalent to one that G is not virtually abelian; and to one that G is not amenable. This condition assures existence of such ψ_w (we need some modification to the construction from the original Brooks argument) which is *non*trivial. For example, \mathbb{Z} is a hyperbolic group, but all quasi-homomorphisms on \mathbb{Z} are trivial.

In fact, the same assertion holds true for any amenable group G. To see this, let $E: \ell^{\infty}(G, \mathbb{R}) \to \mathbb{R}$ be a invariant mean. For any $\psi \in G$, define $\delta_g(h) := \psi(gh) - \psi(h)$. Then define

$$\psi' \colon G \to \mathbb{R}; \ g \mapsto E(\delta_g).$$

Then by G-invariance of E, this ψ' is a homomorphism. Also by construction, the difference $\psi - \psi'$ is a bounded map. This means $\widetilde{QH}(G) = 0$ for amenable groups.

Remark 10.2.6. In Theorem 10.2.5, the following properties for a group G is essential to deduce that $\widetilde{QH}(G)$ is infinite dimensional (in particular, that $\widetilde{QH}(G) \neq 0$.)

- (a) The group G acts on a hyperbolic space X isometrically $\alpha \colon G \curvearrowright X$.
- (b) The action α is properly discontinuous, namely, for any $x \in X$ and any r > 0, the set

$$\{g \in G : d(x, g \cdot x) < r\}$$

is finite.

(c) There exists $g \in G$ of infinite order such that g acts on X by a hyperbolic isometry, and for any n > 0 and any $h \in G$, $hg^n h^{-1} \neq g^{-n}$. In addition, G is not virtually \mathbb{Z} .

For details of condition (c), we refer to [BeFu] and [CalFu]. These conditions are explicitly written in the paper [CalFu], and there they show existence of some bound form below for scl.

We note that hyperbolic groups can be seen as discrete analogue of rank 1 groups.

In contrary, Burger–Monod [BuMo1], [BuMo2] (Theorem 7.2.3) shown that totally higher rank lattices have (TT). This in particular implies that for any totally higher rank lattice Γ , $\widetilde{QH}(\Gamma) = 0$ holds. In fact, they shown the following:

Theorem 10.2.7. (Burger-Monod [BuMo1], [BuMo2]) For any higher rank lattice Γ in the sense in Chapter 0, we have

$$\widetilde{QH}(\Gamma) = 0.$$

Recall that for the definition of a higher rank lattice in a semisimple algebraic group G, we task the irreducibility condition on each rank 1 factor, namely we assume for each rank 1 factor G_i , the image of Γ by the projection $G \twoheadrightarrow G_i$ is dense in G_i . A basic example is $SL_2(\mathbb{Z}[\sqrt{2}])$. This group is isomorphic to

$$\Gamma = \{ (g, \tilde{g}) : g \in \mathrm{SL}_2(\mathbb{Z}[\sqrt{2}]) \} \leqslant \mathrm{SL}_2(\mathbb{R}) \times \mathrm{SL}_2(\mathbb{R}) = G,$$

where $g \mapsto \tilde{g}$ is the conjugation $\sqrt{2} \mapsto -\sqrt{2}$. This $\Gamma \leq G$ is an irreducible lattice in a rank 2 (but not totally higher rank) group, and hence $SL_2(\mathbb{Z}[\sqrt{2}])$ is a higher rank lattice in the sense in Chapter 0.

Therefore in terms of QH, there is a "higher rank versus rank 1" phenomena.

10.3 Homogenization of quasi-homomorphism

In this section, we introduce a powerful tool in the study of quasi-homomorphism, which is called *homogenization*.

10.3.1 Homogeneous quasi-homomorphisms

Definition 10.3.1. Let G be a group.

- (i) A quasi-homomorphism ϕ on G is said to be *homogeneous* if ϕ is a homomorphism on each cyclic subgroup of G. Namely, if for any $g \in \Gamma$ and $m \in \mathbb{Z}$, $\phi(g^m) = m \cdot \phi(g)$ holds.
- (ii) We define HQH(G) as the space of all homogeneous homomorphisms on G.
- (*iii*) We define HQH(G) as the following vector space:

$$HQH(G) := HQH(G) / \{\text{homomorphisms } G \to \mathbb{R}\}.$$

Note that any bounded homogeneous quasi-homomorphism is identical to the zero map.

Apparently it seems that homogeneity condition on quasi-homomorphism is strong, but the next proposition shows there is no loss of generality to concentrate on homogeneous cases:

Proposition 10.3.2. Let G be a group. Then for any $\psi \in QH(G)$, there exists a unique element $\phi \in HQH(G)$ such that

$$\|\psi - \phi\|_{\infty} \le D(\psi)$$

Here $D(\psi)$ denotes the defect of ψ . More precisely, ϕ is given by the following formula:

$$\phi(g) = \lim_{n \to \infty} \frac{\psi(g^n)}{n}$$

Proof. Take any element $g \in G$. Let $m \in \mathbb{N}$. Consider $g^{2^m} = g^{2^{m-1}} \cdot g^{2^{m-1}}$ and obtain

$$|\psi(g^{2^m}) - 2\psi(g^{2^{m-1}})| \le D.$$

Here $D = D(\psi)$ is the defect. This inequality is rewritten as

$$\left|\frac{\psi(g^{2^m})}{2^m} - \frac{\psi(g^{2^{m-1}})}{2^{m-1}}\right| \le \frac{D}{2^m}$$

This implies that $(\psi(g^{2^m})/2^m)_m$ is a Cauchy sequence in \mathbb{R} , and thus there exists a limit. We define $\phi(g)$ as this limit. It is obvious that

 $\|\psi - \phi\|_{\infty} \le D,$

and hence $\phi \in QH(G)$. To see $\phi \in HQH(G)$, for any $n \in \mathbb{N}$ we have

$$\begin{aligned} |\phi(g^n) - n\phi(g)| &= \lim_{m \to \infty} \frac{|\psi(g^{n2^m}) - n\psi(g^{2^m})|}{2^m} \\ &\leq \lim_{m \to \infty} \frac{(n-1)D}{2^m} = 0. \end{aligned}$$

Therefore $\phi(g^n) = n\phi(g)$ for any $n \ge 0$. Also observe for any n > 0

$$D(\phi) \ge |\phi(g^n) + \phi(g^{-n})| = n|\phi(g) + \phi(g^{-1})|.$$

Hence for any $g \in G$,

$$\phi(g^{-1}) = -\phi(g).$$

This means $\phi \in HQH(G)$.

The uniqueness of such ϕ in the statement is trivial by homogeneity.

Corollary 10.3.3. For any group G, there is a natural isomorphism

$$\widetilde{QH}(G) \cong \widetilde{HQH}(G)$$

Therefore, from this point, we focus on homogeneous quasi-homomorphisms.

10.3.2 The Bavard duality theorem

Note that any perturbation of a quasi-homomorphism by a bounded map is also a quasi-homomorphism. This implies for quasi-homomorphisms, an exact value on each point has no meaning. One good point for considering *homogeneous* quasihomomorphism is then each exact value enjoys nice properties.

We need one fact about commutators. For the proof, see Subsection 2.2.3, Lemma 2.24 in [Cal2].

Lemma 10.3.4. For any g, h in a group and $n \in \mathbb{N}$, $g^{2n}h^{2n}(gh)^{-2n}$ is a product of n commutators.

Here we state basic properties of homogeneous quasi-homomorphisms.

Lemma 10.3.5. Let G be a group. Then for any $\phi \in HQH(G)$, the following hold:

- (i) Let $H \leq G$ is a subgroup. If ϕ is bounded on H, then $\phi \mid_{H} \equiv 0$.
- (ii) For any commuting pair $g, h \in G$, $\phi(gh) = \phi(g) + \phi(h)$. In particular, if G is abelian, then $\widetilde{HQH}(G) = 0$.
- (iii) The map ϕ is constant on each conjugacy class. Namely, for any $g \in G$ and any $t \in G$,

$$\phi(tgt^{-1}) = \phi(g).$$

(iv) There is an equality

$$D(\phi) = \sup_{g,h\in G} |\phi([g,h])|.$$

Note that item (*ii*) is a special case of the fact that $\widetilde{HQH} = 0$ for amenable groups, which we have seen in the previous subsection.

Proof. Item (i) is clear.

Note that by homogeneity of ϕ for any g, h,

$$\phi(gh) - \phi(g) - \phi(h) = \lim_{n \to \infty} \frac{\phi((gh)^n) - \phi(g^n) - \phi(h^n)}{n}$$

Item (*ii*) then follows because if gh = hg,

$$|\phi((gh)^n) - \phi(g^n) - \phi(h^n)| \le 2D(\phi) + |\phi((gh)^n g^{-n} h^{-n})| = 2D(\phi).$$

For item (*iii*), since $(tgt^{-1})^n = tg^nt^{-1}$,

$$|\phi((tgt^{-1})^n) - \phi(g^n)| \le 2D(\phi) + 2|\phi(t)|$$

for any $n \in \mathbb{N}$. Again by letting $n \to \infty$, we get the conclusion.

10.3. HOMOGENIZATION OF QUASI-HOMOMORPHISM

For item (iv), by item (iii) we have

$$|\phi([g,h])| \le D(\phi) + |\phi(ghg^{-1}) + \phi(h^{-1})| = D(\phi) + |\phi(h) + \phi(h^{-1})| = D(\phi),$$

and hence $D(\phi) \ge \sup_{g,h\in G} |\phi([g,h])|$. To get the inverse inequality, fix $n \ge 1$. By the definition of the defect, there exist $g, h \in G$ such that

$$|\phi(gh) - \phi(g) - \phi(h)| \ge D(\phi) - \frac{1}{n^2}.$$

By Lemma 10.3.4, $g^{2n}h^{2n}(gh)^{-2n}$ is a product of *n* commutators c_1, \ldots, c_n . Therefore, by homogeneity, we have

$$2nD(\phi) - \frac{2}{n} \le |\phi(c_1 \cdots c_n)|$$
$$\le (n-1)D(\phi) + \sum_{i=1}^n |\phi(c_i)|.$$

Since each c_i is commutator, $|\phi(c_i)| \leq D(\phi)$ holds. Therefore, there exists some commutator c_i such that

$$|\phi(c_i)| \ge D(\phi) - \frac{2}{n}$$

This confirms the inequality $D(\phi) \leq \sup_{g,h \in G} |\phi([g,h])|$.

As we mentioned in item (i) in Example 10.1.2, we now proceed in showing the following:

Lemma 10.3.6. Let F_2 be a free group and a, b be free generators. Then

$$\operatorname{scl}([a,b]) = \frac{1}{2}$$

on F_2 .

Proof. First we show that $scl([a, b]) \leq 1/2$. This follows from Lemma 10.3.4. Indeed, by putting g = [a, b] and h = b we have that $[a, b]^{2n}[b^{2n}, a]$ is a product of n commutators. This implies for any $n \in \mathbb{N}$,

$$\operatorname{cl}([a,b]^{2n}) \le n+1.$$

This shows the desired inequality.

Next, we show the inverse inequality: $scl([a, b]) \ge 1/2$. Here we use the following fact:

there exists
$$\phi \in HQH(F_2)$$
 such that $D(\phi) = \phi([a, b]) > 0$

Here for a construction, consider Brooks' quasi-homomorphism for w = [a, b] and take homogenization. We take this ϕ and put $D = D(\phi) > 0$.

Suppose $[a, b]^n$ is written as m_n commutators c_1, \ldots, c_{m_n} . Then by applying ϕ and taking absolute values, we have

$$nD = |\phi(c_1 \cdots c_{m_n})|$$

$$\leq (m_n - 1)D + \sum_{i=1}^{m_n} |\phi(c_i)|$$

$$\leq (2m_n - 1)D.$$

Here we use (the easier direction of) item (iv) of Lemma 10.3.5. Therefore,

$$\frac{m_n}{n} \ge \frac{1}{2} + \frac{1}{2n}.$$

This shows $scl([a, b]) \ge 1/2$, as claimed.

The argument above indicates that there is connection between homogeneous quasi-homomorphisms and stable commutator length. Ch. Bavard [Bav] revealed the following theorem, which is the fundamental theory in the study of scl. Recall in this chapter we assume all groups are discrete and we allow uncountable groups.

Theorem 10.3.7. (The Bavard duality theorem, [Bav]) Let G be a discrete group.

(i) There is the following formula:

for any
$$g \in [G, G]$$
, $\operatorname{scl}(g) = \sup_{\phi \in HQH(G)} \frac{\phi(g)}{2D(\phi)}$.

- (ii) The following two conditions are equivalent:
 - (a) The equality $\widetilde{HQH}(G) = 0$ holds. In other words, the comparison map in degree 2,

$$\Psi_{\rm b}^2 \colon H^2_{\rm b}(G) \to H^2(G)$$

is injective.

(b) The scl (on [G, G]) vanishes identically.

Proof. For item (i), the inequality

$$\operatorname{scl}(g) \ge \sup_{\phi \in HQH(G)} \frac{\phi(g)}{2D(\phi)}$$

can be shown in a similar argument to that in the proof of (the latter half) of Lemma 10.3.6. The proof of the inverse inequality

$$\operatorname{scl}(g) \le \sup_{\phi \in HQH(G)} \frac{\phi(g)}{2D(\phi)}$$

is more involved, and here we will not present it. We refer to Section 2.5 in [Cal2].

Item (ii) is a corollary of item (i). Indeed, thanks to item (i), condition (b) is equivalent to the condition that every $\phi \in HQH(G)$ is zero on [G, G]. Item (iv) of Lemma 10.3.5 implies that this condition is equivalent to one that every $\phi \in HQH(G)$ is a homomorphism. This is the statement of condition (a). \Box

10.4 Question of Abért and Monod

We state a question of Abért and Monod, which is for instance stated in the ICM proceeding paper [Mon2] of Monod. The first solution to this question is given by A. Muranov [Mur], and we will see the statement of the Muranov theorem.

10.4.1 The question

First, recall the definition of bounded generation in Definition 4.2.4.

We see in Theorem 10.2.7 (or Theorem 7.2.3) that specially for $G = SL_3(\mathbb{Z})$, $\widetilde{HQH}(G) = 0$ holds. However this specific result directly follows from the bounded generation of Carter-Keller for $SL_3(\mathbb{Z})$ as follows: recall Cater-Keller's bounded generation (Theorem 4.2.5) states the following:

 $G = SL_3(\mathbb{Z})$ is boundedly generated by elementary matrices.

By commutator relation, every elementary matrix in $SL_3(\mathbb{Z})$ is a single commutator. Thus the statement above implies the following:

the commutator length on [G, G] is bounded, namely, $\sup_{g \in [G,G]} \operatorname{cl}(g) < \infty$.

In this case [G, G] = G, and hence the statement above means the commutator width is infinite. This certainly implies the vanishing of scl and the Bavard duality theorem 10.3.7 leads us to the conclusion.

In the connection to this argument, Abért asked the following question at his website. Monod exhibited this question in his ICM proceedings paper [Mon2], and now this question is called *a question of Abért–Monod*.

Question 10.4.1. (Problem Q [Mon2])

- (i) Does there exist a example of groups G which enjoys both of the following properties?
 - (1) The commutator length on [G, G] is unbounded:

$$\sup_{g\in[G,G]}\mathrm{cl}(g)=\infty$$

(2) The group G has trivial actual (homogeneous) quasi-homomorphism space, namely,

$$\widetilde{HQH}(G) = 0.$$

Equivalently, the stable commutator length vanishes identically on [G, G]:

$$scl \equiv 0.$$

(*ii*) Let K be a (commutative) field of infinite transcendence degree over its subfield. Let $m \ge 3$. Then, is the space

$$\widetilde{HQH}(\mathrm{SL}_m(K[x]))$$

is the zero-space?

The background of question (ii) is the following theorem of Dennis–Vaserstein [DeVa], which is based on the van der Kallen theorem of *un*bounded generation (Theorem 4.3.1).

Proposition 10.4.2. ([vdKa]) Let K a field as in question (ii) in Question 10.4.1. Then for any $m \ge 3$, $G = SL_m(K[x])$ has infinite commutator width:

$$\sup_{g \in G} \operatorname{cl}(g) = \infty.$$

Note that [G, G] = G follows from commutator relation.

Proof. Suppose $\sup_{g \in G} \operatorname{cl}(g) < \infty$ happens for some $G = \operatorname{SL}_m(K[x])$. Then we will show that $G' = \operatorname{SL}_{2m}(K[x])$ is boundedly generated by elementary matrices. This contradicts Theorem 4.3.1. The key observation is the following formulae: for any $V, W, X \in \operatorname{SL}_m(K[x])$,

$$\begin{pmatrix} [V,W] & 0\\ 0 & I_m \end{pmatrix} = \begin{pmatrix} 0 & V\\ -V^{-1} & 0 \end{pmatrix} \begin{pmatrix} 0 & -W^{-1}\\ W & 0 \end{pmatrix} \begin{pmatrix} (WV)^{-1} & 0\\ 0 & WV \end{pmatrix};$$
$$\begin{pmatrix} X & 0\\ 0 & X^{-1} \end{pmatrix} = \begin{pmatrix} I_m & X\\ 0 & I_m \end{pmatrix} \begin{pmatrix} I_m & 0\\ -X^{-1} & I_m \end{pmatrix} \begin{pmatrix} I_m & X - I_m\\ 0 & I_m \end{pmatrix} \begin{pmatrix} I_m & 0\\ I_m & I_m \end{pmatrix} \begin{pmatrix} I_m & -I_m\\ 0 & I_m \end{pmatrix};$$
$$\begin{pmatrix} 0 & X\\ X^{-1} & 0 \end{pmatrix} = \begin{pmatrix} I_m & X\\ 0 & I_m \end{pmatrix} \begin{pmatrix} I_m & 0\\ -X^{-1} & I_m \end{pmatrix} \begin{pmatrix} I_m & 0\\ -X^{-1} & I_m \end{pmatrix} \begin{pmatrix} I_m & X\\ 0 & I_m \end{pmatrix}.$$

These formulae show for each single commutator $[V, W] \in SL_{2m}(K[x])$, the image of it in the left upper corner of $SL_{2m}(K[x])$ is a product of elementary matrices.

Finally, recall the definition of stable range (Definition 4.1.4). Proposition 4.1.5 means that the following: if a commutative ring A has stable range r, then for any $m \ge r+1$, $\operatorname{SL}_m(A)$ is boundedly generated by $\operatorname{SL}_{r+1}(A)$, sitting in the left upper corner, together with the set of elementary matrices. Since $\operatorname{sr}(K[x]) = 1$ (because this ring is euclidean), this result in particular implies that G' is boundedly generated by G (in the left upper corner) and elementary matrices. Therefore, G' must be boundedly generated by elementary matrices. This leads us to a contradiction, and ends our proof.

10.4.2 A theorem of Muranov

The answer to question (i) of the Abért–Monod question is positive. The first example is brought by A. Muranov in his paper [Mur].

Theorem 10.4.3. (Muranov [Mur]) There exists a simple group G generated by 2 elements which satisfies

$$\sup_{g \in G} \operatorname{cl}(g) = \infty.$$

together with

$$scl \equiv 0$$

The proof of this theorem is involved (in fact Muranov shown much more properties on the group G above), and we refer to his original paper [Mur]. One idea of the construction is to use small cancellation theory on group presentations, and to define presentations recursively. This theory has a background in the theory of hyperbolic groups, and the group G above is a direct limit of a sequence of hyperbolic groups (note that a hyperbolic group itself cannot have the properties in the theorem by Theorem 10.2.5).

Note that in finite group theory, it is known that any simple finite group has commutator width 1. This is a deep theorem, and called the Ore conjecture. We note that in contrast, Caprace–Fujiwara [CapFu] constructed nontrivial quasi-homomorphisms on some class of Kac–Moody groups which are simple. By the Bavard duality theorem (Theorem 10.3.7), this provides with examples of (infinite) simple groups on which scl does *not* vanish identically.

10.5 Quasi-homomorphisms on elementary groups

In this section, we study quasi-homomorphisms on the elementary group over certain (commutative) rings. Recall from Chapter 0 we assume all rings are associative and unital. Also recall in this chapter groups are assumed to be discrete, and we allow uncountable discrete groups. We show Theorem E, which is the main result of the paper [Mim2] of the author. In particular, this results answers the second question (question (ii) of Question 10.4.1) of Abért and Monod, and thus provide with the second class of examples in of question (i). This class has completely different background from one appears in Muranov's theorem.

10.5.1 Precise statement of Theorem D

In the paper [Mim2] of the author, we investigated quasi-homomorphisms on elementary groups over certain rings. Here we restate Theorem E with full generality: **Theorem 10.5.1.** ([Mim2]) Let A be a commutative ring and $m \ge 6$ be a natural number. Suppose the group $G = E_m(A)$ is boundedly generated by $E_2(A)$ (in the left upper corner) and the set of single commutators in G. Then

$$\widetilde{HQH}(\mathbf{E}_m(A)) = 0$$

holds true.

In particular, the following hold true:

(i) Suppose A satisfies GE_2 -condition in the sense of P. M. Cohn; and $G = SL_m(A)$ is boundedly generated by $SL_2(A)$ (in the left upper corner) and the set of single commutators in $SL_m(A)$. Then

$$\widetilde{HQH}(\mathrm{SL}_m(A)) = 0.$$

(ii) Suppose A is a commutative principal ideal ring. If A satisfies GE₂-condition, then

$$\widetilde{HQH}(\mathrm{SL}_m(A)) = 0.$$

(iii) Suppose A is a euclidean domain. Then

$$\widetilde{HQH}(\mathrm{SL}_m(A)) = 0.$$

Here the GE_2 -condition for a commutative ring A is the following condition:

$$E_2(A) = SL_2(A).$$

We note that the original definition of GE_2 -condition for a ring R is the following condition

$$\operatorname{GE}_2(R) = \operatorname{GL}_2(R),$$

and this makes sense even for noncommutative rings. Here $GE_2(R)$ denotes the multiplicative group in $M_2(R)$ generated by elementary matrices and diagonal invertible matrices. We refer to [Coh] for details and the proof of equivalence.

Item (*iii*) of Theorem 10.5.1 answers question (*ii*) of Abért–Monod (Question 10.4.1), with the assumption that $m \ge 6$. The following corollary follows from Proposition 10.4.2 (and the Bavard duality).

Corollary 10.5.2. ([Mim2]) Let K be a (commutative) field of infinite transcendence degree over its subfield. Then for $m \ge 6$, the group $G = SL_m(K[x])$ satisfies

$$\sup_{g \in G} \operatorname{cl}(g) = \infty.$$

together with

 $\operatorname{scl} \equiv 0.$

Recall from Subsection 4.3.1 for the examples of such K. Main examples are \mathbb{C}, \mathbb{R} and some countable field, for instance,

$$K = \mathbb{Q}(e^{\sqrt{p_1}}, e^{\sqrt{p_2}}, \ldots)$$

where $(p_n)_n$ is a strictly increasing sequence of primes, and the symbol above means the field generated by these elements.

In addition, we note that it is not known whether $\mathrm{SL}_{m\geq 3}(\mathbb{Q}[x])$ is boundedly generated by elementary matrices. Theorem 10.5.1 shows that, at least the stable commutator length vanishes if $m \geq 6$.

We will prove Theorem 10.5.1 in the next subsection, and here we state some needed lemma.

Firstly, we state the following lemma, which follows directly from Lemma 10.3.5:

Lemma 10.5.3. Let G a group. Let $g \in G$ be a element which is conjugate to its inverse. Then for any $\phi \in HQH(G)$, $\phi(g) = 0$.

Let R be a ring and $m \geq 2$. We call an element $g \in E_m(R)$ a unit upper (respectively lower) triangular matrix if all diagonal entries are 1 and all of the entries below (respectively above) the diagonals are 0. We define $U_m R$ (respectively $L_m R$) as the group of all unit upper (respectively lower) triangular matrices of degree m.

Lemma 10.5.4. Let ϕ be a homogeneous quasi-homomorphism on $E_m(R)$. If $n \ge 3$, then the following hold:

- (i) For any elementary matrix $s \in E_m(R)$, $\phi(s) = 0$.
- (ii) For any $h \in (U_m R) \cup (L_m R)$, $\phi(h) = 0$.

Proof. Item (i) follows from Lemma 10.5.3 because for $n \ge 3$, an elementary matrix is conjugate to its inverse. Hence, ϕ is bounded on $U_m R$ and $L_m R$. By the homogeneity of ϕ , we obtain item (ii).

The following observation plays an important role in this paper.

Lemma 10.5.5. Let Γ be a group and $H \leq \Gamma$ be a subgroup. Let ϕ be a homogeneous quasi-homomorphism on Γ . Suppose ϕ vanishes on H. Then for any $h \in H$ and any $g \in N_{\Gamma}(H)$, $\phi(hg) = \phi(gh) = \phi(g)$ holds. Here $N_{\Gamma}(H)$ means the normalizer of H in Γ .

Proof. We will only show $\phi(hg) = \phi(g)$. Let D be the defect of ϕ . By employing the condition $gHg^{-1} \leq H$ repeatedly, we have that for any $n \in \mathbb{N}$, there exists an element $h' \in H$ such that $(hg)^n = h'g^n$. Hence, we obtain that

$$n \cdot |\phi(hg) - \phi(g)| \le D.$$

This ensures the conclusion.

10.5.2 Key proposition and proof

The following proposition is the key to proving Theorem 10.5.1:

Proposition 10.5.6. ([Mim2], [Mim4]) Let R be a (possibly noncommutative) ring. Suppose $g \in E_2(R)$ and $s \in E_2(R)$ satisfy the following conditions:

(i) The matrix $g + g^{-1} \in M_2(R)$ commutes with s.

(ii) The equality $(s - I_2)^2 = 0$ holds.

Then for any homogeneous quasi-homomorphism ϕ on $E_6(R)$, we have

$$\phi(gs) = \phi(sg) = \phi(g) + \phi(s).$$

Here we put $E_2(R)$ in the left upper corner of $E_6(R)$, and view $g, s \in E_2(R)$ as elements in $E_6(R)$ with this identification.

In particular, if a ring A is commutative, the following holds: Let $g \in E_2(A)$. Let s in $E_2(A)$ be an elementary matrix. Then for any homogeneous quasi-homomorphism ϕ on $E_6(A)$, we have

$$\phi(gs) = \phi(sg) = \phi(g) + \phi(s).$$

Here we put $E_2(A)$ in the left upper corner of $E_6(A)$.

The latter assersion follows form the following easy fact: if A is commutative, then for any $g \in E_2(A)$, $g+g^{-1} \in M_2(A)$ is a diagonal matrix with the same diagonal entries, and is a multiple of the identity element.

For the sake of simplicity, we here give a proof of the latter assertion of the proposition. The basic argument in below also works for general cases. Before proceeding into the proof, we see some facts.

Firstly, we utilize the following result of Dennis–Vaserstein (Lemma 18 of [DeVa]), with k = 3.

Lemma 10.5.7. ([DeVa]) Let J be a (possibly noncommutative) ring and $p, q, r \in$ GL₁(J) such that pqr = 1. Let Δ be the diagonal matrix in GL₃(J) with the diagonal part p, q, and r. Then $\Delta \in (L_3J)(U_3J)(L_3J)(U_3J)$.

Let g and s be as in Proposition 10.5.6. We set $J = M_2(A)$ in the lemma above and set p = sg, $q = g^{-1}$ and $r = s^{-1}$. We need the following explicit form:

$$\begin{pmatrix} I_2 & 0 & 0 \\ p^{-1} & I_2 & 0 \\ 0 & q^{-1} & I_2 \end{pmatrix} \begin{pmatrix} p & 0 & 0 \\ 0 & q & 0 \\ 0 & 0 & r \end{pmatrix} = X_1 Y X_2, \text{ where } X_1 = \begin{pmatrix} I_2 & I_2 - p & 0 \\ 0 & I_2 & I_2 - pq \\ 0 & 0 & I_2 \end{pmatrix}^{-1}$$
$$Y = \begin{pmatrix} I_2 & 0 & 0 \\ I_2 & I_2 & 0 \\ 0 & I_2 & I_2 \end{pmatrix}, \text{ and } X_2 = \begin{pmatrix} I_2 & (I_2 - p)q & 0 \\ 0 & I_2 & (I_2 - pq)r \\ 0 & 0 & I_2 \end{pmatrix}.$$

By applying Lemma 10.3.5 and Lemma 10.5.5 (for $H = L_3 J$), we conclude that the evaluation of the left-hand side of the first equality in above by ϕ is equal to

$$\phi(p) + \phi(q) + \phi(r) = \phi(sg) - \phi(g) - \phi(s) = \phi(sg) - \phi(g)$$

Here we see p, q, r, sg, g, and s as elements in $E_2(A)$. It has no problem because each of three diagonal $E_2(A)$ parts in $E_6(A)$ can be conjugated to each other by permutation matrices. Hence, in order to show Proposition 10.5.6, it is enough to show that $\phi(X_1YX_2) = 0$ for any g and s as in Proposition 10.5.6.

Proof. (Proposition 10.5.6, the latter assertion) We may assume that $s \in U_2A$ without loss of generality. Note that X_1YX_2 is conjugate to X_2X_1Y . Computation shows that $X_2X_1Y = ZXY$, where

$$X = \begin{pmatrix} I_2 & -(p-I_2)(q-I_2) & 0\\ 0 & I_2 & 0\\ 0 & 0 & I_2 \end{pmatrix} \text{ and } Z = \begin{pmatrix} I_2 & 0 & -(p-I_2)(q-I_2)(pq-I_2)\\ 0 & I_2 & 0\\ 0 & 0 & I_2 \end{pmatrix}$$

Indeed, the key observation here is that if one regards X_2X_1 as an element in $\operatorname{GL}_3(\operatorname{M}_2(A))$, then the (2, 3)-th entry of X_2X_1 is $0(=0_2)$. This is because $s + s^{-1} = 2I_2$. We set $x = -(p - I_2)(q - I_2) = -(sg - I_2)(g^{-1} - I_2)$ and $z = -(p - I_2)(q - I_2)(pq - I_2) = -(sg - I_2)(g^{-1} - I_2)$ as elements in $\operatorname{M}_2(A)$. By definition, g and s can be written as

$$g = \begin{pmatrix} a & b \\ c & d \end{pmatrix}$$
 $(a, b, c, d \in A)$ and $s = \begin{pmatrix} 1 & f \\ 0 & 1 \end{pmatrix}$ $(f \in A).$

By substituting these matrix forms of g and s for the expressions of x and z, we continue calculations as follows:

$$\begin{aligned} x &= -\left(\begin{array}{cc} a+fc-1 & b+fd \\ c & d-1 \end{array}\right) \left(\begin{array}{cc} d-1 & -b \\ -c & a-1 \end{array}\right) = \left(\begin{array}{c} * & * \\ 0 & * \end{array}\right),\\ z &= \left(\begin{array}{c} * & * \\ 0 & * \end{array}\right) \left(\begin{array}{c} 0 & f \\ 0 & 0 \end{array}\right) = \left(\begin{array}{c} 0 & * \\ 0 & 0 \end{array}\right).\end{aligned}$$

Here each * respectively represents a certain element in A.

Next we define a (non-unital) subring N of $M_2(A)$ by

$$N = \left\{ \left(\begin{array}{cc} 0 & l \\ 0 & 0 \end{array} \right) : l \in A \right\}.$$

Obviously $z \in N$ and the following holds.

Lemma 10.5.8. In the setting above, the following hold:

(i) For any $u, v \in N$, uv = 0.

(ii) For any $u \in N$, $xu \in N$ and $ux \in N$.

We also define the following subset in $M_6(A)$:

$$\Gamma_N = \left\{ \begin{pmatrix} I_2 + * & * & * \\ * & I_2 + * & * \\ * & * & I_2 + * \end{pmatrix} : \text{ each } * \text{ is in } N \right\}.$$

The lemma below is the key observation in this proof.

Lemma 10.5.9. In the setting above, the following hold:

- (i) The set Γ_N is a subgroup of $E_6(A)$.
- (ii) The matrices X and Y normalize Γ_N .
- (iii) Any homogeneous quasi-homomorphism ϕ on $E_6(A)$ is bounded on Γ_N . This means that ϕ vanishes on Γ_N .

Proof. (Lemma 10.5.9) Items (i) and (ii) are straightforward from Lemma 10.5.8. For item (iii), we observe that any element $\gamma \in \Gamma_N$ can be decomposed as

$$\gamma = \begin{pmatrix} I_2 + * & * & * \\ 0 & I_2 + * & * \\ 0 & 0 & I_2 + * \end{pmatrix} \begin{pmatrix} I_2 & 0 & 0 \\ * & I_2 & 0 \\ * & * & I_2 \end{pmatrix}.$$

Here each * is in N. Lemma 10.5.4 ends our proof.

We also need the following simple fact. Let

$$T = \begin{pmatrix} I_2 & 0 & 0\\ 0 & -I_2 & 0\\ 0 & I_2 & I_2 \end{pmatrix} \in \mathcal{E}_6(A).$$

Lemma 10.5.10. In the setting above, $TXT^{-1} = X^{-1}$ and $TYT^{-1} = Y^{-1}$ hold.

Now we have all ingredients to complete our proof of Proposition 10.5.6. The essential point is that Lemma 10.5.5 applies to the case of that $H = \Gamma_N$, h = Z, and g = XY (this follows from Lemma 10.5.9). Thus, we show that

$$\phi(ZXY) = \phi(XY).$$

By Lemma 10.5.10, we also see that XY can be transformed to $X^{-1}Y^{-1}$ by the conjugation by T. Because $X^{-1}Y^{-1}$ is conjugate to $Y^{-1}X^{-1}$, XY is conjugate to its inverse. Finally, by Lemma 10.5.3, we have

$$\phi(ZXY) = \phi(XY) = 0.$$

This shows

$$\phi(sg) = \phi(g),$$

as desired.

182

Proof. (Theorem 10.5.1) The assertion of the first part of the theorem immediately follows from Proposition 10.5.6 because $E_2(A)$ is generated by elementary matrices. For the latter parts, we only need to check bounded generation properties. For item (iii), Proposition 4.1.5 ensures it. For item (ii), we use the following Proposition by M. Newman, which can be found in the proof of Theorem 2 in [New].

Proposition 10.5.11. ([New]) Let A be a (commutative) principal ideal ring and m, l be positive integers with $m \ge 3l$. Then any element in $G = SL_m(A)$ can be expressed as a product of two commutators in G and some element in $SL_{m-l}(A)$, where we regard $SL_{m-l}(A)$ as the subgroup of G in the left upper corner.

This proposition immediately shows that for a (commutative) principal ideal ring A and $m \geq 3$, $SL_m(A)$ is boundedly generated by $SL_2(A)$ (in the left upper corner) and the set of all single commutators (therefore, the case of item (*iii*) also follows from this).

We have accomplished the proof of Proposition 10.5.6.

10.5.3 Difficulty on universal lattices

In the subsection above, the rings of our main concern are euclidean domains. Here we consider the case of universal lattices.

At the moment, the theorem below seems the best result for universal lattices.

Definition 10.5.12. Let R be a ring and $m \ge 2$. Then we define $U_m(R)$ as the subgroup of $GL_m(R)$ generated by all $m \times m$ unipotent matrices.

We warn that $U_m(R)$ here is completely different from U_mR in the last subsection (the latter denotes the group of all unit upper triangular matrices). We note that $U_m(R)$ contains $E_m(R)$ as a (possibly non-proper) subgroup. Also by definition, $U_m(R)$ is normal in $GL_m(R)$.

Theorem 10.5.13. ([Mim4]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ with $k \in \mathbb{N}$. Suppose $m \ge 6$. Then for any $\phi \in HQH(SL_m(A))$, $\phi \equiv 0$ on the subgroup $U_2(A)$, which sits in the left upper corner.

Proof. We apply Proposition 10.5.6 to the case of that $g \in U_2(A)$ and s is a 2×2 unipotent matrix (strictly speaking, $U_2(A)$ is larger than $E_2(A)$, but there is no change on the proof). We obtain that

for any
$$\phi \in HQH(SL_m(A)), \ \phi(gs) = \phi(g)\phi(s).$$

Therefore, it suffices to show that for any $\phi \in HQH(SL_m(A)), \phi(s) = 0$ holds.

We use the fact that the ring A is a unique factorization domain. Therefore any 2×2 unipotent matrix s is of the following form:

$$s = \left(\begin{array}{cc} 1 + auv & -au^2 \\ av^2 & 1 - auv \end{array}\right).$$

Here $a, u, v \in A$. This matrix is not necessarily an element in $E_2(A)$ (recall Remark 4.1.8), but the following is known (Mennicke): for any such matrix, it is a product of 8 elementary matrices in $E_3(A) (= SL_3(A))$. More precisely, the following is a presentation (see p. 281 of [PaWo]):

$$s = E_{1,3}(-u)E_{2,3}(-v)E_{3,1}(-av)E_{3,2}(au)E_{1,3}(u)E_{2,3}(v)E_{2,1}(av)E_{3,2}(-au).$$

Take any $\phi \in \mathrm{SL}_m(A)$ and consider s as an element of $\mathrm{SL}_m(A)$ in the left upper corner. Recall by Lemma 10.5.4 that ϕ vanishes on the set of elementary matrices. Therefore, $|\phi(s^n)|$ is uniformly bounded for $n \in \mathbb{N}$. This means $\phi(s)=0$. This ends the proof.

Since we have verified property $(FF_{\mathcal{C}})/T$ for universal lattices with degree ≥ 4 for several classes (Theorem B, Theorem C), we have the following corollary.

Corollary 10.5.14. ([Mim4]) Let $A = \mathbb{Z}[x_1, \ldots, x_k]$ with $k \in \mathbb{N}$. Then for $m \ge 6$, the pair $\mathrm{SL}_m(A) \ge \mathrm{U}_2(A)$ has relative (FF_B) for $B = \mathcal{L}_p, C_p$ ($p \in (1, \infty)$) and $B = [\mathcal{H}]$. In particular, $\mathrm{SL}_m(A) \ge \mathrm{U}_2(A)$ has relative (TT). Here $\mathrm{U}_2(A)$ sits in the left upper corner.

We warn the following point. In the view of Vaserstein's bounded generation (Theorem 4.3.2), it may seem that Theorem 10.5.13 implies that $HQH(SL_m(A)) = 0$. However, there is a *big gap* in this guess. The point here is the following (for $k \ge 1$): $E_2(A)$ is a very small subgroup in $SL_2(A)$, and for Vaserstein's bounded generation, $SL_2(A)$ is needed. More precise meaning of very small is given by the following theorem of Grunewald–Menniche–Vaserstein [GMV2]:

Theorem 10.5.15. ([GMV2]) Let F_{∞} be the free group of countable (infinite) rank. Then there is a surjection

$$\operatorname{SL}_2(\mathbb{Z}[x])/\operatorname{U}_2(\mathbb{Z}[x]) \twoheadrightarrow F_{\infty}.$$

In fact Krstić–McCool [KrMc] later shown that for any euclidean domain A which is not a field, the group quotient $SL_2(A[x])/U_2(A[x])$ maps into F_{∞} .

Therefore, to deal with quasi-homomorphisms on universal lattices, one needs inventional method. We make the following remark that universal lattices are truly universal object in the study of quasi-homomorphisms on elementary groups (over commutative rings).

Lemma 10.5.16. ([Mim4]) If there exists $m \ge 3$ such that for any $k \ge 1$, $\widetilde{QH}(SL_m(A_k)) = 0$, then for any (not necessarily finitely generated) commutative ring C, $\widetilde{QH}(E_m(C)) = 0$ holds. Here $A_k = \mathbb{Z}[x_1, \ldots, x_k]$.

Proof. Let $G = E_m(C)$ for a given commutative ring C. Then for any g, there exist $l \in \mathbb{N}$ and elementary matrices s_1, \ldots, s_l in G such that $g = s_1 \cdots s_l$. Consider the subring C' of C generated by all $r \in C$ which appears some entry of some matrix s_1, \ldots, s_l . Then C' is finitely generated, and for sufficiently large k, there is a surjection

$$\operatorname{SL}_m(A_k) \twoheadrightarrow \operatorname{E}_m(C') \ni g.$$

It is easy to see that by homomorphism, scl does not increase. Namely, for any group Γ, Λ and any homomorphism $\Psi \colon \Gamma \to \Lambda$, we have for any $\gamma \in [\Gamma, \Gamma]$

$$\operatorname{scl}_{\Gamma}(\gamma) \ge \operatorname{scl}_{\Lambda}(\Psi(\gamma)).$$

Therefore by combining this with the assumption and the Bavard duality theorem, we conclude that $\operatorname{scl}_{\operatorname{E}_m(C')}(g) = 0$. Since $\operatorname{E}_m(C') \leq G$, this implies $\operatorname{scl}_G(g) = 0$. This holds for any $g \in G$, and thus we have

$$\operatorname{scl} \equiv 0 \quad \text{on } G.$$

Again by the Bavard duality, we get the conclusion.

Chapter 11

Applications to group actions

In this chapter, we apply our rigidity results, which have been obtained in this thesis, to group actions. Firstly, we observe the Navas theorem [Nav1], [Nav2] which concern actions on the circle of a discrete group with $(F_{\mathcal{L}_p})$. As a corollary, we obtain Theorem F.

Secondly, we consider homomorphism rigidity with the target of mapping class groups $\operatorname{MCG}(\Sigma_g)$ (of compact closed oriented surfaces); and of outer automorphism groups $\operatorname{Out}(F_n)$ of (finitely generated) free groups. Since $\operatorname{Out}(F_n)$ (for sufficiently large n) contains the mapping class group of a punctured surface and the automorphism groups of a free group, these two objects above are the essential ones. Here firstly we see known fundamental results on $\operatorname{MCG}(\Sigma_g)$ and $\operatorname{Out}(F_n)$, such as element classifications and subgroup classifications. Secondly, we see quasi-homomorphisms on these groups, respectively due to Bestvina–Fujiwara [BeFu] and Bestvina–Feighn [BeFe]. Thirdly, we see theorems of U. Hamenstädt [Ham]; and Bestvina–Bromberg– Fujiwara on quasi-cocycles on $\operatorname{MCG}(\Sigma_g)$; and on $\operatorname{Out}(F_n)$. These results are one key to establishing Theorem G. With these preliminary facts, we proceed into the proof of Theorem G. Property (TT)/T is the counterpart of theorems of Hamenstädt and Bestvina–Bromberg–Fujiwara. We note that our proof of Theorem G with $\operatorname{Out}(F_n)$ target follows the proof in a paper [BrWa] of Bridson–Wade, in which they shown the homomorphism rigidity theorem for higher rank lattices.

Finally, we consider some shortcut in proving Theorem G in the case of universal lattices. That shortcut also works for noncommutative universal lattices, and this is Theorem H. We argue why this argument may not work in the case of symplectic universal lattices.

In this chapter, we state many theorems on $MCG(\Sigma_g)$ or $Out(F_n)$ without providing with proofs. We refer the readers to the original papers. On actions on the circle, there is a survey [Ghy2] by E. Ghys. Concerning Navas' theorem, we also refer to Section 2.9 of [BHV]. For mapping class groups, main references are survey [Iva] of N. V. Ivanov, and a forthcoming book [FaMar] of Farb-Margalit. For automorphism groups of free groups, we refer to an ICM proceedings paper [Vog2] of K. Vogtmann; surveys [Vog1], [BrVo] of Vogtmann, and Bridson–Vogtmann.

11.1 Actions on the circle

Let S^1 denote the unit circle in \mathbb{R}^2 and identify S^1 with $[-\pi, \pi)$. We denote by $\text{Diff}_+(S^1)$ the group of orientation preserving group diffeomorphisms of S^1 .

Definition 11.1.1. Let $\alpha > 0$ be a real number. The group $\text{Diff}_{+}^{1+\alpha}(S^1)$ is defined as the class of all orientation preserving group diffeomorphisms f of S^1 such that f'and $(f^{-1})'$ are Hölder continuous with exponent α . Here a function g on S^1 is said to be *Hölder continuous with exponent* α if

$$\|g\|_{\alpha} := \sup_{\theta_1 \neq \theta_2} \frac{|g(\theta_1) - g(\theta_2)|}{|\theta_1 - \theta_2|^{\alpha}} < \infty$$

holds.

Firstly, we restate Theorem F:

Theorem 11.1.2. Let Γ be a finite index subgroup either of $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 4)$ or of $Sp_{2m}(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \ge 3)$. Then for any $\alpha > 0$, every homomorphism

$$\Gamma \to \operatorname{Diff}_{+}^{1+\alpha}(\mathrm{S}^1)$$

has finite image.

In particular, the following holds true: Γ be a finite index subgroup either of $E_m(A)$ $(m \ge 4)$ or of $Ep_{2m}(A)$ $(m \ge 3)$ for a commutative finitely generated ring A. Then for any $\alpha > 0$, every homomorphism

$$\Gamma \to \operatorname{Diff}^{1+\alpha}_+(\mathrm{S}^1)$$

has finite image.

This theorem is an immediate corollary of theorem(s) of A. Navas, as we will explain in below.

Navas [Nav1] has shown the following theorem: For any discrete group Γ with property (T), every homomorphism from Γ into $\text{Diff}_{+}^{1+\alpha}(S^1)$ has finite image, for any $\alpha > 1/2$. He has also noted in Appendix of the paper [Nav2] that his theorem can be extended to general L^p cases.

Theorem 11.1.3. (Navas [Nav1], [Nav2]) Let $1 and <math>\Gamma$ be a discrete group with property ($F_{\mathcal{L}^p}$). Then for any $\alpha > 1/p$, every homomorphism

$$\Gamma \to \operatorname{Diff}^{1+\alpha}_+(\mathrm{S}^1)$$

has finite image.

11.1. ACTIONS ON THE CIRCLE

We note that this theorem is one of the motivations of Bader–Furman–Gelander– Monod for their study of $(F_{\mathcal{L}_p})$, see Subsection 1.b of [BFGM].

Here we will give an outline of Navas' theorem. The proof below is an variant of one in Section 2.9 of [BHV], and there an idea of D. Witte is also used.

Proof. (Outlined)

We consider the space $B = L^p(S^1 \times S^1, \mathbb{R})$, where we endow $S^1 \times S^1$ with the Lebesgue measure $d\theta d\phi$. We consider the following isometric representation ρ of $\text{Diff}^{1+\alpha}_+(S^1)$ on B:

$$(\rho(f)K)(\theta,\phi) = \left((f^{-1})'(\theta)(f^{-1})'(\phi) \right)^{1/p} K(f^{-1}(\theta,\phi)).$$

(Note that we will consider the image of the discrete group Γ into $\text{Diff}_{+}^{1+\alpha}(S^1)$ (or slightly modified one) and the restriction of ρ on this group. Therefore we do not care on topology on $\text{Diff}_{+}^{1+\alpha}(S^1)$.) Consider the following kernel F on S^1 :

$$F(\theta, \phi) = \frac{1}{(2\tan((\theta - \phi)/2))^{2/p}}$$

Set a formal coboundary of $\text{Diff}^{1+\alpha}_+(S^1)$,

$$c(f) := F - \rho(f)F.$$

More precisely,

$$c(f^{-1})(\theta_1, \theta_2) = \frac{1}{\left|2\sin\left((\theta_1 - \theta_2)/2\right)\right|^{2/p}} - \frac{\left[f'(\theta_1)f'(\theta_2)\right]^{1/p}}{\left|2\sin\left((f(\theta_1) - f(\theta_2))/2\right)\right|^{2/p}}$$

This map is called the *Liouville* L^{p} -cocycle. The point here is the following:

although F does not belong to $L^p(S^1 \times S^1)$, this formal coboundary belongs to $L^p(S^1 \times S^1)$ for each $f \in \text{Diff}^{1+\alpha}_+(S^1)$.

We omit the proof of this fact: see [Nav1] or Lemma 2.9.2 in [BHV] in the case of p = 2. We note that for the proof of the latter, we need the assumption $\alpha > p^{-1}$. This *formal* coboundary (this is said to be formal because $F \notin B$) becomes an ρ -cocycle.

Now consider a homomorphism $\Gamma \to \text{Diff}^{1+\alpha}_+(S^1)$. Take a triple cover of S^1 , and take a corresponding cover of Γ . We name this group G, and with identify with its image in $\text{Diff}^{1+\alpha}_+(S^1)$. Since G is a extension of Γ by a finite group $\mathbb{Z}/3\mathbb{Z}$, G also has $(F_{\mathcal{L}_p})$. Consider the restriction of ρ and c on G, and rewrite them as ρ and c. Here we employ property $(F_{\mathcal{L}_p})$ for G. Then it indicates that this c is indeed a ρ coboundary. Namely, there exists a kernel ξ on S^1 which belongs to B such that

for all
$$g \in G$$
, $c(g) = \xi - \rho(g)\xi$.

This means

for all
$$g \in G$$
, $F - \xi = \rho(g)(F - \xi)$.

Therefore the (formal) vector $F - \xi$ is $\rho(G)$ -invariant.

We define a (positive) measure μ on $(S^1 \times S^1) \setminus \Delta$ by the following formula:

$$d\mu(\theta,\phi) := (F-\xi)^2(\theta,\phi)d\theta d\phi$$

Here Δ is the diagonal part (recall that F diverges on Δ). By the construction of μ , the following properties can be seen:

- 1. The measure μ is $\rho(G)$ -invariant.
- 2. For any pairwise distinct and cyclically ordered points a, b, c, d in S^1 ,

$$\mu([a,b] \times [c,d]) = \mu([c,d] \times [a,b]).$$

3. For any pairwise distinct and cyclically ordered points a, b, c in S^1 ,

$$\mu([a,a] \times [b,c]) = 0.$$

4. For any pairwise distinct and cyclically ordered points a, b, c, d in S^1 ,

$$\mu([a,b) \times (b,c]) = \infty.$$

Then it can be also shown that these four properties imply the following:

any $g \in G \setminus \{e\}$ does not fix three points of S^1

(see Proposition 2.9.6 in [BHV]). Then in terms of Γ (we also identify Γ with its image in Diff^{1+ α}(S¹)), this exactly means the following:

any $\gamma \in \Gamma \setminus \{e\}$ does not fix any point of S¹.

This means the action of Γ on S^1 is free.

However, it is a well-known theorem that any such group (even inside Homeo₊(S¹)) must be abelian. The proof uses ordering on a group. See Corollary 2.9.10 in [BHV]. Since Γ (recall that we identify with the image) has ($F_{\mathcal{L}_p}$), this implies (the image) Γ is finite, as desired.

For the proof of Theorem 11.1.2, we just need to recall that $(F_{\mathcal{L}_p})$ passes to *p*-integrable lattices, in particular to finite index subgroups.

Remark 11.1.4. (i) In the case of higher rank lattices of \mathbb{Q} -rank ≥ 2 , there is a simple argument by Witte [Wit] focusing on ordering on a group, and it shows that in fact every homomorphism from these into Homeo₊(S¹) has finite image (the author acknowledges A. Furman for this reference). However, in the process of deducing finiteness of the image, we need to appeal to the Margulis finiteness theorem. More precisely, in argument on orders, firstly we show that this homomorphism cannot be injective. This means the kernel of this homomorphism cannot be finite. Then we appeal to the Margulis finiteness theorem, and ends the proof.

However, as we have seen in Lemma 4.1.12, universal lattices and symplectic universal lattices *fail* to have the Margulis finiteness property. Therefore arguments [Wit] does not directly apply to these cases (it is shown in the argument in [Wit] that a homomorphism into $Homeo_+(S^1)$ is never injective).

(*ii*) The argument in Theorem 11.1.3 does not give any information for the case of $\alpha = 0$. Navas pointed out in [Nav3] that a property valid for $\text{Diff}_{+}^{1+\alpha}(S^1)$ for all $\alpha > 0$ is *not* necessarily valid for $\text{Diff}_{+}^1(S^1)$. His example is the following: for any $\alpha > 0$, $\text{Diff}_{+}^{1+\alpha}(S^1)$ does not contain a finitely generated group of intermediate growth, but $\text{Diff}_{+}^1(S^1)$ does contain.

Nevertheless, Burger–Monod [BuMo1], [BuMo2] have succeeded in establishing finiteness results for homomorphisms from higher rank lattices, which is based on the study of Ghys [Ghy1] on the euler class $e \in H^2(\text{Homeo}_+(S^1), \mathbb{Z})$. Burger–Monod shown that if a discrete group Γ satisfies $H^2_b(\Gamma) = 0$ (this symbol denotes the second bounded cohomology with trivial real coefficient) and has finite abelianization, then homomorphism finiteness from Γ into Diff¹₊(S¹) follows. Also in this point of view, studies of quasi-homomorphisms on universal lattices (and on symplectic universal lattices) seems important.

(iii) As we have seen in item (ii) of Remark 4.1.11, universal lattices and symplectic universal lattices are linear groups (Definition 2.7.4). A lemma of A. Selberg states every linear group contains a finite index subgroup which is torsion-free. In study of homomorphism rigidity with target of circle diffeomorphism group, it is much easier to obtain finiteness results if the source group has torsions. Therefore the essential case in statement of Theorem 11.1.2 is the case where we take these torsion-free finite index subgroups.

11.2 Basics on $MCG(\Sigma)$ and $Out(F_n)$

11.2.1 Definitions

Recall from Chapter 0 that the symbol Σ_g denotes a compact connected oriented surface with g genus, and the symbol $\Sigma_{g,l}$ denotes a compact connected oriented surface with g genus and l punctures. Hence Σ_g is equal to $\Sigma_{g,0}$

Definition 11.2.1. Let $g, l \ge 0$ and $n \ge 2$ (finite numbers).

- (i) Let $\Sigma = \Sigma_{q,l}$.
 - (1) The mapping class group of Σ is defined as

 $MCG(\Sigma) := Homeo_+(\Sigma)/isotopy.$

Here a mapping class is allowed to permute punctures, and isotopies can rotate a neighborhood of a puncture.

(2) The surface $\Sigma = \Sigma_{a,l}$ is said to be *non-exceptional* if

$$3g+l \ge 5.$$

Otherwise, it is said to be *exceptional*.

(ii) Let F_n denotes the free group of rank n. The symbol $\operatorname{Aut}(F_n)$ denotes the *auto-morphism group* of F_n . The symbol $\operatorname{Out}(F_n)$ denotes the *outer automorphism group* of F_n , namely,

$$\operatorname{Out}(F_n) := \operatorname{Aut}(F_n) / \operatorname{Inn}(F_n).$$

Here $\operatorname{Inn}(F_n)$ denotes the normal subgroups of $\operatorname{Aut}(F_n)$ consisting all conjugations.

Here are a few words on the terminology "punctures." There is another terminology "boundary component", and these two are distinguished. In the above definition, if we replace punctures with boundary components, then each mapping class must preserve the individual boundary components pointwise, and isotopies must fix each boundary components pointwise. If surface Σ has at least one boundary component, then it is known that MCG(Σ) has no torsion. This case homomorphism rigidity from a finite index group of (symplectic) universal lattices is much easily deduced.

It is a classical fact that $MCG(\Sigma)$ is a finitely generated group, and in fact it is finitely presented (see for instance Chapter 6 of [FaMar]). Also, Nielsen has given finite presentations for $Out(F_n)$ and $Aut(F_n)$. We note that there is an injection

$$\operatorname{Aut}(F_n) \hookrightarrow \operatorname{Out}(F_{n+1}).$$

If $\Sigma = \Sigma_{g,l}$ is exceptional, then it is not difficult to see that $MCG(\Sigma_{g,l})$ is virtually free, which means it contains a free group (possibly $\{e\}$ or \mathbb{Z}) as a finite index subgroup. For instance, $MCG(\Sigma_{0,0}) = \{e\}$; and $MCG(\Sigma_{1,0}) \cong SL_2(\mathbb{Z})$, whose isomorphism is realized by a action on homology class $H_1(\Sigma_{1,0},\mathbb{Z}) \cong \mathbb{Z}^2$. Recall from the proof of Corollary 2.5.11 that $SL_2(\mathbb{Z})$ contains a copy of F_2 with index 12. Virtually free groups are more or less well-understood, and hence we are interested in non-exceptional surfaces case.

Consider a non-exceptional surface $\Sigma = \Sigma_{q,l}$. There is a correspondence:

$$\left\{\begin{array}{l} \text{Free homotopy classes of} \\ (\text{unbased}) \text{ maps } \Sigma \to \Sigma \end{array}\right\} \longleftrightarrow \left\{\begin{array}{l} \text{Conjugacy classes of homo-} \\ \text{morphisms } \pi_1(\Sigma) \to \pi_1(\Sigma) \end{array}\right\}.$$

For details, see Proposition 1 B.9 of a book [Hat] of A. Hatcher (here observe that the universal cover of Σ is contractible). With this correspondence, there is a natural homomorphism

$$MCG(\Sigma) \rightarrow Out(\pi_1(\Sigma)),$$

which is always *injective*. The Dehn–Nielson–Bear theorem states the following:

Theorem 11.2.2. (The Dehn-Nielsen-Bear theorem) For $g \ge 1$, the homomorphism

$$\mathrm{MCG}^{\pm}(\Sigma_g) \to \mathrm{Out}(\pi_1(\Sigma_g))$$

is an isomorphism. Here $MCG^{\pm}(\Sigma_g)$ denotes the extended mapping class group, which is defined by allowing orientation-reversing mapping homeomorphisms in the definition of mapping class groups.

In particular, $MCG(\Sigma_g)$ is a subgroup of index 2 in the outer automorphism of the surface group of genus g,

$$\pi_1(\Sigma_g) = \langle a_1, \dots, a_g; b_1, \dots b_g \mid [a_1, b_1] \cdots [a_g, b_g] = e \rangle.$$

If $l \geq 1$, then $\pi_1(\Sigma_{g,l}) = F_{2g+l-1}$. Thus we have an injection:

$$MCG(\Sigma_{g,l}) \hookrightarrow Out(F_{2g+l-1}) \qquad (l \ge 1)$$

11.2.2 Linear representations

Consider a closed surface Σ_g . On a pair of two homology classes of Σ_g , there is a concept of the *algebraic intersection number*. Namely, for $\alpha, \beta \in H_1(\Sigma_g; \mathbb{Z}), \tilde{i}(\alpha, \beta) \in \mathbb{Z}$ is defined as the sum of the indices of the intersection point of α and β , where an intersection point is of index +1 if the orientation of the intersection agrees

with the orientation of S, and is -1 otherwise. Also, there exists an ordered basis $\{[\alpha_1], \ldots, [\alpha_g]; [\beta_1], \ldots, [\beta_g]\}$ of $H^1(\Sigma_g; \mathbb{Z})$ which realizes the intersection number as the symplectic form with respect to it. Each element f of $MCG(\Sigma_g)$ induces a homomorphism $f_*: H^1(\Sigma_g; \mathbb{Z}) \to H^1(\Sigma_g; \mathbb{Z})$, and f_* leaves the intersection number (as a bilinear form) invariant. Thus we obtain a homomorphism

$$\operatorname{MCG}(\Sigma_g) \twoheadrightarrow \operatorname{Sp}_{2g}(\mathbb{Z}),$$

which is known to be surjective, as we have written as \rightarrow .

For $\operatorname{Out}(F_n)$, by considering its action on the abelianization of $F_n \cong \mathbb{Z}^n$, we obtain a surjection,

$$\operatorname{Out}(F_n) \twoheadrightarrow \operatorname{GL}_n(\mathbb{Z}).$$

These linear representations have enormous kernels. It is an important open problem to determine whether $MCG(\Sigma_g)$ $(g \ge 2)$ and $Out(F_n)$ $(n \ge 2)$ are linear groups.

Definition 11.2.3. For $g \ge 1$, the normal subgroup \mathcal{T}_g of $MCG(\Sigma_g)$ is defined as the kernel of

$$MCG(\Sigma_g) \twoheadrightarrow Sp_{2q}(\mathbb{Z}).$$

It is called the *Torelli group*.

For $n \geq 2$, the normal subgroup $\overline{\mathrm{IA}}_n$ of $\mathrm{Out}(F_n)$ is defined as the kernel of

$$\operatorname{Out}(F_n) \twoheadrightarrow \operatorname{GL}_n(\mathbb{Z}).$$

These groups are of high interest in the theory of $MCG(\Sigma_g)$ and $Out(F_n)$, but still their structures are mysterious (the symbol IA stands for "identity on the abelianization").

It is known that these groups are torsion-free. Also, it is known that $\operatorname{Sp}_{2g}(\mathbb{Z})$ and $\operatorname{GL}_n(\mathbb{Z})$ respectively have finite index torsion-free subgroups. Indeed, we are able to appeal to Selberg's lemma, but there are explicit constructions. For $m \geq 2$, let $\operatorname{SL}_m(\mathbb{Z})[3]$ be the congruence subgroup of $\operatorname{SL}_m(\mathbb{Z})$ associated with the ideal $(3) \leq \mathbb{Z}$. Namely, $\operatorname{SL}_m(\mathbb{Z})[3]$ is the kernel of the surjection

$$\operatorname{SL}_m(\mathbb{Z}) \twoheadrightarrow \operatorname{SL}_m(\mathbb{Z}/3\mathbb{Z}),$$

which sends $\mathbb{Z} \ni a$ to $a \mod 3$. The group $\mathrm{SL}_m(\mathbb{Z})[3]$ is a finite index subgroup of $\mathrm{SL}_m(\mathbb{Z})$, and it is not so difficult to see it is torsion-free (for instance, see Proposition 8.8 in [FaMar]). Thus we obtain the following result:

Theorem 11.2.4. For $g \ge 1$ and $m \ge 2$, the groups $MCG(\Sigma_g)$ and $Out(F_n)$ have finite index subgroups which are torsion-free.

This result shall be used for the proof of Theorem H, which is the shortcut of Theorem G. For the proof of Theorem G, we need the following theorem of Bass–Lubotzky [BaLu] (another proof is given by Bridson–Wade [BrWa]) on \overline{IA}_n .

Theorem 11.2.5. (Bass-Lubotzky [BaLu]) Any nontrivial subgroup of \overline{IA}_n maps onto \mathbb{Z} .

11.2.3 Element classification

Here we state the element classification theorem for $MCG(\Sigma)$, which is so-called *the Nielsen-Thurston Classification theorem* (for instance, see Chapter 13 in [FaMar]). To state this theorem, we need some terminologies on measured foliation on Σ , which is invented by W. Thurston. We briefly see them. For details on this topic, we refer to a book [Bir] of J. S. Birman, Chapter 14 of [FaMar], and a paper of McCarthy-Papadopoulos [McPa]. We call a *c* curve on a surface a *simple closed curve* if *c* has no self-intersecting point and is closed.

Let $\Sigma = \Sigma_{g,l}$. Let $\mathcal{MF}(\Sigma)$ be the space of equivalence classes of measured foliation (here the equivalence is generated by isotopy and by so-called Whitehead moves). Here we exclude the zero-foliation. Let S be the set of isotopy classes of simple closed curves in Σ . Then we can regard $\mathcal{MF}(\Sigma)$ as a subset of $\mathbb{R}^{\mathcal{S}}_{\geq 0} \setminus \{0\}$. We define $\mathcal{PMF}(\Sigma)$ as the image of $\mathcal{MF}(\Sigma)$ by the projection $\mathbb{R}^{\mathcal{S}}_{\geq 0} \setminus \{0\}$ to the projective space $P(\mathbb{R}^{\mathcal{S}})$. It is known that $\mathcal{PMF}(\Sigma)$ is endowed with a natural topology with respect to which it is homeomorphic to a sphere of dimension 6g+2l-7 (in particular, $\mathcal{PMF}(\Sigma)$ is compact).

Definition 11.2.6. Let $\Sigma = \Sigma_{g,l}$ be a surface.

(i) An element $f \in MCG(\Sigma)$ is said to be *pseudo-Anosov* if the following condition holds: there exists a pair of transverse measured foliations $(L_+, \mu_+), (L_-, \mu_-)$ on Σ and a real number $\lambda > 1$ and a representative ϕ of f such that

$$\phi \cdot (L_+, \mu_+) = (L_+, \lambda \mu_+)$$
 and $\phi \cdot (L_-, \mu_-) = (L_-, \lambda^{-1} \mu_-).$

Therefore, a pseudo-Anosov element f fixes some point in $\mathcal{PMF}(\Sigma)$.

(*ii*) Two pseudo-Anosov elements $f_1, f_2 \in MCG(\Sigma)$ are said to be *independent* if their fixed point sets in \mathcal{PMF} are disjoint.

In the above definition, this λ is uniquely determined for a pseudo-Anosov element f, and is called the *dilatation*

Here is the classification theorem. A simple closed curve c on a surface Σ (possibly with punctures) is said to be *essential* if c is neither homotopic to one point nor homotopic to a boundary component.

Theorem 11.2.7. (the Nielson–Thurston classification theorem) Let $\Sigma = \Sigma_{g,l}$ be a surface. Then each $f \in MCG(\Sigma)$, either of the following three holds true:

(i) the element f is a torsion, namely, there exists $n \ge 1$ such that $f^n = e$;

- (ii) the element f is reducible. This means, f fixes a collection of isotopy classes of essential simple closed curves that are pairwise disjoint;
- (iii) the element f is pseudo-Anosov.

Moreover, a pseudo-Anosov element is neither a torsion nor reducible.

Example 11.2.8. Consider $\Sigma = \Sigma_1$ and $\Lambda = MCG(\Sigma) \cong SL_2(\mathbb{Z})$. With the identification $\Lambda = SL_2(\mathbb{Z})$, the classification above corresponds as follows:

$MCG(\Sigma_1)$	\longleftrightarrow	$\mathrm{SL}_2(\mathbb{Z})$
f is a torsion	\longleftrightarrow	f is a torsion ;
f is reducible		f is unipotent,
	\longleftrightarrow	$f \sim \left(\begin{array}{cc} 1 & * \\ 0 & 1 \end{array} ight);$
f is pseudo-Anosov	\longleftrightarrow	f is semisimple, and for some $\lambda > 1$, $f \sim \begin{pmatrix} \lambda & 0 \\ 0 & \lambda^{-1} \end{pmatrix}$.

A basic example of reducibe element is the *Dehn twist* along an (essential) simple closed curve c. Roughly speaking, the Dehn twist along c is a map which twists once along c on the neighborhood of c:

Definition 11.2.9. Let Σ be a surface. Let *c* be an essential simple closed curve on Σ .

(i) Let N be a regular neighborhood of c, and ϕ be an orientation preserving homomorphism from $A = S^1 \times [0, 1]$ to N. A Dehn twist along c is the following map $\Sigma \to \Sigma$,

$$T_c(x) = \begin{cases} \phi \circ T \circ \phi^{-1}(x) & \text{if } x \in N; \\ x & \text{if } x \in \Sigma \setminus N. \end{cases}$$

Here $T: A \to A$ is the following map:

$$T(\theta, t) = (\theta + 2\pi t, t).$$

(ii) An element $f \in MCG(\Sigma)$ is called the *Dehn twist* along c if there exists a representative which is a Dehn twist in the sense of item (i) (for some N and ϕ). We again use the symbol T_c for the Dehn twist along c, as an element in $MCG(\Sigma)$.

As this definition implicitly says, Dehn twists along c are indeed unique up to isotopy.

Next we proceed to the case of $Out(F_n)$. For $Out(F_n)$, there is an analogous concept to that of pseudo-Anosov elements for mapping class groups. This is called that of *fully irreducible elements*.

Definition 11.2.10. Let $n \geq 2$.

(i) A subgroup $L \leq F_n$ is called a *free factor* of F_n if there exists $L' \leq F_n$ such that

$$F_n = L \star L'$$

holds. Here the symbol $L \star L'$ denotes the free product of L and L'.

(*ii*) An element f in $Out(F_n)$ is said to be *fully irreducible* if *no* non-zero power of f fixes the set of conjugacy classes of any free factor of F_n .

There exists an analogue of \mathcal{PMF} for $\operatorname{Out}(F_n)$ in terms of which we can define independence of two fully irreducible elements. We briefly state this. Let $\mathcal{T} = \mathcal{T}_n$ be the space of free simplicial trees endowed with cocompact left F_n -action, which does not have vertices of valence 1 or 2. For a conjugacy class γ in F_n and $T \in \mathcal{T}$, define $\langle T, \gamma \rangle$ as the translation length of γ in T. Namely, take a representative g of γ and define

$$\langle T, \gamma \rangle := \inf_{t \in T} d(t, gt).$$

This definition is independent of the choice of g. Then the group $Out(F_n)$ acts by right by "changing markings", this means for $f \in Out(F_n)$;

$$T \mapsto T \cdot f$$
 such that $\langle T \cdot f, \gamma \rangle = \langle T, f(\gamma) \rangle$.

The group $\mathbb{R}_{>0}$ acts on \mathcal{T} by scaling and these two actions commute. The *Outer* space of Culler-Vogtmann [CuVo1] is the projectivized space

$$\mathcal{PT} := \mathcal{T} / \mathbb{R}_{>0}.$$

We define $\overline{\mathcal{T}}$ as the closure of \mathcal{T} in the space of trees with minimal left F_n action. Both $\operatorname{Out}(F_n)$ and $\mathbb{R}_{>0}$ continues to act on $\overline{\mathcal{T}}$, and we define

$$\overline{\mathcal{PT}} := \overline{\mathcal{T}} / \mathbb{R}_{>0}.$$

This is a compact space and called Culler–Morgan's equivalent compactification of the outer space [CuMo]. For details on outer spaces, we refer to the original paper [CuVo1], and surveys [Vog1], [Vog2], [BrVo]. The following theorem due to Levitt– Lustig [LeLu] shows in the view of actions on $\overline{\mathcal{PT}}$, fully irreducible elements can be seen as an analogue of pseudo-Anosov elements in mapping class groups.

Theorem 11.2.11. (Levitt–Lustig [LeLu]) Every fully irreducible element $f \in \text{Out}(F_n)$ acts on $\overline{\mathcal{PT}}$ with exactly two fixed point T_+, T_- with north-south dynamics.

Definition 11.2.12. We say two fully irreducible elements in $Out(F_n)$ are *independent* if their fixed point sets in $\overline{\mathcal{PT}}$ are disjoint.

11.2.4 Subgroup classification

For $MCG(\Sigma)$, the following subgroup classification theorem, due to McCarthy– Papadopoulos [McPa], plays a fundamental role:

Theorem 11.2.13. (*McCarthy–Papadopoulos* [McPa]) Let $\Sigma = \Sigma_{g,l}$ be a surface. Then each $\Lambda \leq MCG(\Sigma)$ satisfies either of the following:

- (i) the group Λ is finite;
- (ii) the group Λ is reducible. That means, there exists a collection \mathfrak{C} of essential simple closed pairwise disjoint curves such that every element of Λ fixes \mathfrak{C} ;
- (iii) the group Λ has a pseudo-Anosov element f, and there is no pseudo-Anosov element in Λ which is independent to f. In this case, Λ is virtually \mathbb{Z} ;
- (iv) the group Λ contains two independent pseudo-Anosov elements.

Recall that a group is said to be virtually \mathbb{Z} if it contains \mathbb{Z} as a finite index subgroup.

For the case of $Out(F_n)$, analogous results have been desired, and finally, Handel-Mosher [HaMo] have successfully obtained the theorem in below.

Definition 11.2.14. (Handel–Mosher [HaMo]) Let $n \ge 2$.

- (i) A subgroup $\Lambda \leq \operatorname{Out}(F_n)$ is said to be *reducible* if there exists a free factor $L \leq F_n$ with $L \neq \{e\}$ such that each element of Λ preserves the set of conjugacy classes in L.
- (*ii*) A subgroup $\Lambda \leq \operatorname{Out}(F_n)$ is said to be *fully irreducible* if no subgroups of Λ of finite index is reducible.

Theorem 11.2.15. (Handel–Mosher [HaMo]) Let $n \ge 2$. Then each $\Lambda \leq MCG(\Sigma)$ satisfies either of the following:

- (i) the group Λ is not fully irreducible;
- (ii) the group Λ has a fully irreducible element f, and there is no fully irreducible element in Λ which is independent to f. In this case, Λ is virtually \mathbb{Z} ;
- (iii) the group Λ contains two independent fully irreducible elements.

11.2.5 Curve graph for MCG

For a surface $\Sigma = \Sigma_{g,l}$, we can define an associated natural complex, which is called the *complex of curves*. In the study of quasi-cocycles on the mapping class group, 1-sketlon of this complex plays an important role. This graph is called the *curve* graph.

Definition 11.2.16. Let Σ be a surface. Then the *curve graph* associated with Σ , which is written as $C(\Sigma)$, is defines as follows:

- (1) The vertex set V is the set of all homotopy classes of essential simple closed curves on Σ .
- (2) For each distinct pair $[a], [b] \in V$, we connect [a] and [b] by an edge if there exist representatives a of [a] and b of [b] (essential simple closed curves) such that

$$a \cap b = \emptyset$$

holds

The following is a fundamental fact on curve graphs. For instance, see chapter 3 of [Iva].

Theorem 11.2.17. Suppose $\Sigma = \Sigma_{g,l}$ is a non-exceptional surface (, namely, $3g + l \geq 5$). Then $C(\Sigma)$ is a connected graph, and the diameter of $C(\Sigma)$ is infinite.

Thanks to this theorem, for a nonexceptional surface Σ , we regard $C(\Sigma)$ as a metric space with the shortest path metric.

An extreme importance of a curve graph lies in the following two deep theorems, respectively due to Masur–Minsky [MaMi] and B. Bowditch [Bow]. Before stating these theorems, we note that $MCG(\Sigma)$ naturally acts on $C(\Sigma)$ by isometries. Also, recall the definition of hyperbolicity of a metric space from Definition 2.6.19.

Definition 11.2.18. (Bowditch [Bow] Let G be a discrete group and (X, d) be a metric space. Let α be an isometric action of G on X. We say the action α is *acylindrical* if the following condition is satisfied: for any R > 0, there exists $D \ge 0$ and $L \ge 0$ such that:

for any
$$x, y \in X$$
 with $d(x, y) \ge L$,
 $|\{g \in G : d(x, gx) \le R \text{ and } d(y, gy) \le R\}| \le D.$

For example, for any finitely generated group G, the natural isometric action of G on a Cayley graph Cay(G; S) (where S is any finite generating subset) is acylindrical.

Now we state the celebrated theorems of [MaMi] and [Bow]:

Theorem 11.2.19. (*Masur–Minsky* [MaMi]) For a non-exceptional surface Σ , $C(\Sigma)$ is a hyperbolic metric space.

Theorem 11.2.20. (Bowditch [Bow]) For a non-exceptional surface Σ , the natural isometric action of MCG(Σ) on $C(\Sigma)$ is acylindrical.

Remark 11.2.21. One of the main difficulities to deal with a curve graph is that it is *not* locally finite. Namely, the valence of some vertex is infinite. Because of this, much care is needed in studies of mapping class groups, even though the two theorems above are valid.

11.2.6 Homomorphism rigidity from higher rank lattices

There was a question whether $MCG(\Sigma)$ for a nonexceptional surface can be isomorphic to a lattice in a semi-simple algebraic group, and it was answered in negative (see Chapter 9 of [Iva]). Later, Farb-Masur [FaMas] shown that in fact there is homomorphism rigidity from higher rank lattices to mapping class groups. Recall our terminology of "higher rank lattice" from Chapter 0.

Theorem 11.2.22. (Farb-Masur [FaMas]) Let Γ be a higher rank lattice. Then for any surface, every homomorphism

$$\Gamma \to \mathrm{MCG}(\Sigma)$$

has finite image.

Their proof uses the study of the Poisson boundary of $MCG(\Sigma)$ of Kaimanovich– Masur [KaMa], and the theorem above is called the *Farb–Kaimanovich–Masur superrigidity*.

It is natural to expect an analogous result holds with $Out(F_n)$ target. Recently, Bridson–Wade [BrWa] have succeeded in establishing it:

Theorem 11.2.23. (Bridson–Wade [BrWa]) Let Γ be a higher rank lattice. Then for any $n \geq 2$, every homomorphism

$$\Gamma \to \operatorname{Out}(F_n)$$

has finite image.

We note that, as we mentioned in Chapter 1, in proofs of both theorems, the Margulis finiteness theorem plays a key role. However it is *not* valid for (symplectic) universal lattices (Lemma 4.1.12). Also, we recall from Chapter 1 another difficulty, which we will explain in below. Before stating it, we note that Bridson–Wade in fact defined and shown the following:

Definition 11.2.24. ([BrWa]) A discrete group G is said to be \mathbb{Z} -averse if G has no finite index subgroup G_0 that contains a normal subgroup $N \leq G_0$ which maps onto \mathbb{Z} .

Theorem 11.2.25. (Bridson–Wade [BrWa]) Let G be a discrete group. Suppose G is \mathbb{Z} -averse. Then for any surface $\Sigma = \Sigma_{g,l}$ and any $n \ge 2$, every homomorphism

$$\phi \colon \Gamma \to \mathrm{MCG}(\Sigma)$$

and every homomorphism

$$\psi\colon\Gamma\to\operatorname{Out}(F_n)$$

have finite image.

However, we observe the following:

Lemma 11.2.26. Let $A_k = \mathbb{Z}[x_1, \ldots, x_k]$. If $k \ge 1$, then universal lattices $SL_m(A_k)$ $(m \ge 3)$ and symplectic universal lattices $Sp_{2m}(A_k)$ $(m \ge 2)$ are not \mathbb{Z} -averse.

Proof. We will only show that $SL_m(\mathbb{Z}[x])$ is not \mathbb{Z} -averse. Consider the congruence subgroup H associated with the ideal $(x) \leq \mathbb{Z}[x]$, namely, H is the kernel of a map

 $\operatorname{SL}_m(\mathbb{Z}[x]) \twoheadrightarrow \operatorname{SL}_m(\mathbb{Z})$, which sends x to 0.

Define a map on H by the following formula:

$$\sigma \colon H \to \mathbb{Z}; h \mapsto (h' \mid_{x=0})_{1,1}.$$

Here h' is the derivative matrix by the variable x, and $(\cdot)_{1,1}$ means the (1,1)-th entry.

We show that this map is a group homomorphism. Indeed, let $h, l \in H$. Then by the definition of the group $H, h \mid_{x=0} = l \mid_{x=0} = I_m$, Therefore we have

$$(hl)'|_{x=0} = (h'l)|_{x=0} + (hl')|_{x=0} = h'|_{x=0} + l'|_{x=0}$$

This shows our assertion. It is clear that there exists $h \in H$ with $\sigma(h) \neq 0$. \Box

11.3 Quasi-homomorphisms on $MCG(\Sigma)$ and $Out(F_n)$

In this section, we see quasi-homomorphisms on $MCG(\Sigma)$ and $Out(F_n)$. For the former part, Bestvina–Fujiwara [BeFu] shown that $\widetilde{QH}(MCG(\Sigma)) \neq 0$ (in fact, the left hand side is infinite dimensional) by employing theorems on curve graphs. The method they use has its origin in [EpFu], which had extended the construction of Brooks' quasi-homomorphisms (see Subsection 10.2.2). For the latter part, some analogue of a curve graph for $Out(F_n)$ is needed. This is done by Bestvina–Feighn [BeFe]. Although, this construction is not as powerful as that of curve graph, it is sufficient to obtain nonvanishing (and infinite dimensionality) of $\widetilde{QH}(Out(F_n))$. In both cases, also subgroup cases are considered.

11.3.1 A theorem of Bestvina–Fujiwara

Recall from Remark 10.2.6 that (Epstein–Fujiwara [EpFu] and) Calegari–Fujiwara [CalFu] observed the following properties for a group G is essential to deduce that $\widetilde{QH}(G) \neq 0$ (in fact, that $\widetilde{QH}(G)$ is infinite dimensional).

- (a) The group G acts on a hyperbolic space X isometrically $\alpha \colon G \curvearrowright X$.
- (b) The action α is properly discontinuous, namely, for any $x \in X$ and any r > 0, the set

$$\{g \in G : d(x, g \cdot x) < r\}$$

is finite.

(c) There exists $g \in G$ of infinite order such that g acts on X by a hyperbolic isometry, and for any n > 0 and any $h \in H$, $hg^n h^{-1} \neq g^{-n}$. In addition, G is not virtually \mathbb{Z} .

In [BeFu], Bestvina–Fujiwara extended this machinery, and observed the following. Instead of stating this, we shall state some improved version, which is implicitly in [CalFu].

Theorem 11.3.1. (Bestvina–Fujiwara [BeFu], Calegari–Fujiwara [CalFu]) Let G be a group. Suppose there exists a hyperbolic metric space (X, d) such that following hold true:

- (a) The group G acts on a hyperbolic space X isometrically $\alpha : G \curvearrowleft X$.
- (b) The action α is acylindrical, in the sense in Definition 11.2.18.
- (c) There exists $g \in G$ of infinite order such that g acts on X by a hyperbolic isometry, and for any n > 0 and any $h \in H$, $hg^n h^{-1} \neq g^{-n}$. In addition, G is not virtually \mathbb{Z} .

Then we have

$$\widetilde{QH}(G) = 0.$$

Moreover, this space is infinite dimensional.

More precisely, firstly Bestvina–Fujiwara shown that if a group G acts on a hyperbolic space X isometrically which satisfies the WPD condition (weak proper continuity), then $\widetilde{QH}(G)$ is infinite dimensional. This condition is weaker than condition (b') (the condition of the action being acylindrical) together with condition (c). For details of the WPD condition, we refer to Section 3 of [BeFu]. Later Calegari–Fujiwara essentially shown that if the action is moreover acylindrical (and satisfies condition (c) above), then there exists a uniform positive lower bound for scl of certain elements of [G, G] (recall Theorem 10.3.7: the Bavard duality theorem).

By combining this theorem together with Theorem 11.2.15, Theorem 11.2.20 and Theorem 11.2.13, Bestvina–Fujiwara obtained the following result on $\widetilde{QH}(\Lambda)$ for each subgroup of mapping class groups. Recall that a group is said to be virtually abelian if it contains an abelian subgroup with finite index.

Theorem 11.3.2. (Bestvina–Fujiwara [BeFu]) Let Σ be a surface and Λ be a subgroup of MCG(Σ). Then $\widetilde{QH}(\Lambda) \neq 0$ if and only if Λ is not virtually abelian. Moreover, in this case, this space is infinite dimensional.

Remark 11.3.3. By employing this theorem, Bestvina–Fujiwara gave another proof of Theorem 11.2.22 without appealing to the Margulis finiteness theorem. The following is the outline of their argument: let $\phi: \Gamma \rightarrow \Lambda \leq \text{MCG}(\Sigma)$ be a homomorphism, where Γ is a higher rank lattice. The Burger–Monod theorem (Theorem 10.2.7) tells that

$$\widetilde{QH}(\Gamma) = 0$$

Therefore by pulling back argument, we have

$$\widetilde{QH}(\Lambda) = 0.$$

Then Theorem 11.3.2 applies and Λ must be virtually abelian. However, any higher rank lattice Γ (and hence its finite index subgroup) is known to have finite abelianization (if Γ is totally higher rank lattice in the sense in Chapter 0, then it has (T) and this fact follows). Therefore, Λ is finite, as desired.

In the same argument, Theorem G will follow if universal lattices and symplectic universal lattices satisfy $\widetilde{QH} = 0$. However, as we discussed in Subsection 10.5.3, it will be very standing to determine whether $\widetilde{QH} = 0$ holds in these cases.

11.3.2 An $Out(F_n)$ complex of Bestvina–Feighn

In the proof of Theorem 11.3.2, a curve graph plays a key role. In [BeFe], Bestvina– Feighn provided with some analogue for $Out(F_n)$. Recall the definition of $\overline{\mathcal{PT}}$ and translation length from Subsection 11.2.3.

Theorem 11.3.4. (Bestvina–Feighn [BeFe]) Let $n \ge 2$. For any finite collection f_1, \ldots, f_k of fully irreducible elements of $Out(F_n)$, there exists a connected hyperbolic graph X equipped withan isometric action of $Out(F_n)$ such that the following hold:

- (i) The stabilizer in $Out(F_n)$ of a simplicial tree in $\overline{\mathcal{PT}}$ has bounded orbits.
- (ii) The stabilizer in $Out(F_n)$ of a proper free factor $L < F_n$ has bounded orbits
- (iii) The elements $f_1, \ldots f_k$ have nonzero translation lengths.

(iv) The elements $f_1, \ldots f_k$ satisfy the weak proper discontinuity condition: that means, for every $1 \le i \le k$, every $x \in X$ and every R > 0, there exists N > 0such that the following holds:

$$\{g \in \operatorname{Out}(F_n) : d(x, gx) \leq R \text{ and } d(f_i^N x, gf_i^N x) \leq R\} < \infty$$

They observed that this construction is enough to deduce nonvanishing of $QH(Out(F_n))$. Thus they obtained the following theorem. For details, see Corollary 4.30 in [BeFe] and Section 3 of [BeFu].

Theorem 11.3.5. (Bestvina–Feighn [BeFe]) Let $n \ge 2$ be a surface and Λ be a subgroup of $Out(F_n)$. If Λ contains two independent fully irreducible elements, then we have

$$\widetilde{QH}(\Lambda) \neq 0.$$

Moreover, in this case, this space is infinite dimensional.

Bestvina–Feighn stated that this construction of the graph is not ideal (compared with that of the curve graph for a mapping class group), because the graph X depends on the finite collection f_1, \ldots, f_k .

11.4 Quasi-cocycles on $MCG(\Sigma)$ and $Out(F_n)$

In the view of application to homomorphism rigidity from (symplectic) universal lattices to $MCG(\Sigma)$ or $Out(F_n)$, nonvanishing of actual quasi-homomorphism spaces for (certain) $\Lambda \leq MCG(\Sigma)$, $Out(F_n)$ seems insufficient. This is because vanishing of this space is not verified at the present for (symplectic) universal lattices Γ (Remark 11.3.3). However vanishing of $\widetilde{QH}(\Gamma; \pi, \mathfrak{H})$ is established for any unitary Γ representation (π, \mathfrak{H}) with $\pi \not\supseteq 1_{\Gamma}$. This is a corollary of Theorem B and Theorem D.

Thus results on $QH(\Lambda; \pi, \mathfrak{H})$ for certain $\Lambda \leq \text{MCG}(\Sigma)$ or $\Lambda \leq \text{Out}(F_n)$ is ideal for the proof of Theorem G. Here (π, \mathfrak{H}) is a unitary Λ -representation with $\pi \not\supseteq 1_{\Gamma}$. Such example is given by U. Hamenstädt [Ham], for the case of $\Lambda \leq \text{MCG}(\Sigma)$. She consider the actual quasi-cocycle space with *left regular representation* coefficient. In this section, we see her theorem, which is a key to completing the proof of Theorem G.

For the case of $\Lambda \leq \operatorname{Out}(F_n)$, a work of Bestvina–Bromberg–Fujiwara in progress studies isometric actions of a group on *quasi-trees*. By means of this they have obtained nonvanishing result in the setting above.

11.4.1 A theorem of Hamenstädt

In [Ham], Hamenstädt examined the space $QH(\Lambda; \lambda, \ell^2(\Lambda))$ (recall Definition 7.1.6) for groups acting "nicely" on a hyperbolic space. Here λ is the left regular representation of λ (in fact, she considered any ℓ^p setting). The following is the main theorem of [Ham].

Theorem 11.4.1. (Hamenstädt [Ham]) Let Λ be a discrete group. Suppose Λ admits an isometric action on a hyperbolic space which is non-elementary and weakly acylindrical. Then

$$\widetilde{QH}(\Lambda;\lambda,\ell^2(\Lambda))\neq 0,$$

where λ is the left regular representation of λ . Moreover, in this case, $QH(\Lambda; \lambda, \ell^2(\Lambda))$ is infinite dimensional.

The same result holds if $\ell^2(\Lambda)$ is replaced with $\ell^p(\Lambda)$, for any $p \in (1, \infty)$.

We do not explain the definition of non-elementary actions or that of weakly acylindrical actions. However, as the name suggests, all acylindrical (isometric) actions are weakly acylindrical. The proof of this theorem is involved, and employs dynamical properties. We refer to the original paper [Ham] for details. By combining this theorem with Theorem 11.2.15 and Theorem 11.2.20, Hamenstädt obtained the following result.

Theorem 11.4.2. (Hamenstädt [Ham]) Let $\Sigma = \Sigma_{g,l}$ be a surface. Let $\Lambda \leq MCG(\Sigma)$. Suppose Λ contains two independent pseudo-Anosov elements. Then

$$\widetilde{QH}(\Lambda;\lambda,\ell^2(\Lambda))\neq 0,$$

where λ is the left regular representation of λ . Moreover, in this case, $QH(\Lambda; \lambda, \ell^2(\Lambda))$ is infinite dimensional.

The same result holds if $\ell^2(\Lambda)$ is replaced with $\ell^p(\Lambda)$, for any $p \in (1, \infty)$.

We note that roughly speaking, $QH(G; \lambda_G, \ell^2(G))$ easily vanishes. For instance, it is known that if G is a product of infinite groups, then the space above always vanishes (see [MoSh1], [MoSh2]). Therefore, it is not possible to expect the same statement as in Theorem 11.3.2 for $\widetilde{QH}(\Lambda; \lambda, \ell^2(\Lambda))$.

11.4.2 Another approach – actions on quasi-trees

Definition 11.4.3. Let (X, d_X) , (Y, d_Y) be metric spaces.

- (i) A (not necessarily continuous) function $f: X \to Y$ is called a *quasi-isometry* if there exists constants $A \ge 1$, $B \ge 0$ such that the following two conditions hold:
 - (1) For any $x_1, x_2 \in X$,

$$\frac{1}{A}d_Y(f(x_1), f(x_2)) - A \le d_X(x_1, x_2) \le Ad_y(f(x_1), f(x_2)) + A.$$

(2) For any $y \in Y$, there exists $x \in X$ such that

$$d_Y(y, f(x)) \le B$$

- (*ii*) The metric space (X, d_X) is said to be *quasi-isometric to* (Y, d_Y) if there exists a quasi-isometry $f: X \to Y$.
- (*iii*) A quasi-tree is a metric space (X, d) which is quasi-isometric to a tree.

In [Man], J. F. Manning considered a strengthening of Serre's property (FA) (see Definition 2.5.13) and defined property (QFA) as follows: a group G is said to have property (QFA) if any isometric action on any quasi-tree has finite orbits. He shown that $SL_{m>3}(\mathbb{Z})$ enjoys this property.

In a work [BBF] in progress, Bestvina–Bromberg–Fujiwara have been studying in the opposite direction. They shown that the following groups admits an unbounded action on a quasi-trees by isometries, which also satisfies certain additional condition: any non-elementary hyperbolic group; any subgroups of $MCG(\Sigma)$ containing two independent pseudo-Anosov elements; any subgroups of $Out(F_n)$ containing two independent fully irreducible elements (compare with the fact that $Out(F_n)$ $(n \ge 3)$ has (FA), [Bog], [CuVo2]). By this result, they in particular show the following theorem:

Theorem 11.4.4. (Bestvina-Bromberg-Fujiwara [BBF]) Let $n \geq 2$. Let $\Lambda \leq Out(F_n)$. Suppose Λ contains two independent fully irreducible elements. Then

$$\widetilde{QH}(\Lambda;\lambda,\ell^2(\Lambda))\neq 0,$$

where λ is the left regular representation of λ . Moreover, in this case, $QH(\Lambda; \lambda, \ell^2(\Lambda))$ is infinite dimensional.

This result is also a key to ending the proof of Theorem G.

11.5 Homomorphism rigidity into $MCG(\Sigma)$ and $Out(F_n)$

In this section, we prove Theorem G, with the aid of all preliminary facts above in this chapter. We restate our goal:

Theorem 11.5.1. ([Mim4]) Let Γ be a finite index subgroup either of $SL_m(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \geq 3)$ or of $Sp_{2m}(\mathbb{Z}[x_1, \ldots, x_k])$ $(m \geq 2)$. Then for any $g \geq 0$ and $n \geq 2$, every homomorphism

$$\Phi\colon\Gamma\to\mathrm{MCG}(\Sigma_g)$$

206

and every homomorphism

$$\Psi\colon\Gamma\to\operatorname{Out}(F_n)$$

207

have finite image. In particular, every homomorphism

$$\Gamma \to \mathrm{MCG}(\Sigma_{g,l}) \quad (l \ge 1)$$

and

$$\Gamma \to \operatorname{Aut}(F_n)$$

also have finite image.

Instead of establishing this theorem, we will prove the following theorem:

Theorem 11.5.2. ([Mim4]) Let Γ be a group with (TT)/T. Then all of the conclusions in Theorem 11.5.1 hold true.

Proof. (Theorem 11.5.2) Firstly, recall that (TT)/T (= $(FF_{\mathcal{H}})/T$) implies (T) (see Lemma 8.1.2). Also recall that (TT)/T passes to group quotients and to finite index subgroups (see Proposition 8.1.4). In particular, in the statement of the theorem, $\Phi(\Gamma)$ and $\Psi(\Gamma)$ has (TT)/T.

Secondly, the second half of the four assertions follow from the first half. This follows from that $MCG(\Sigma_{g,l})$ $(l \ge 1)$ and $Aut(F_n)$ are subgroups in $Out(F_{n'})$ for sufficiently large n' (see Subsection11.2.1). Therefore we deal with the first two statements.

Case1. with target of mapping class groups

Let $\Phi: \Gamma \to MCG(\Sigma_g)$ be a homomorphism. Firstly, we note that for exceptional cases the conclusion holds, because then $MCG(\Sigma_{g,l})$ is virtually free and has the Haagerup property (see Section 2.5). Therefore, hereafter, we assume every surface which appears in this proof is non-exectional.

Let $\Lambda \leq MCG(S)$ be the image of Φ . We employ Theorem 11.2.13, the subgroup classification result of [McPa]. For convenience we restate here: Λ is either of the following forms:

- (i) the group Λ is finite;
- (*ii*) the group Λ is reducible: there exists a collection \mathfrak{C} of essential simple closed pairwise disjoint curves such that every element of Λ fixes \mathfrak{C} ;
- (*iii*) the group Λ is virtually \mathbb{Z} ;
- (iv) the group Λ contains two independent pseudo-Anosov elements.

Then we appeal to Theorem 11.4.2 of Hamenstädt, and exclude option (iv). This part is the key in this proof. More precisely, Λ has property (TT)/T. However, if option (iv) occurs, then by Theorem 11.4.2 of Hamenstädt,

$$\widetilde{QH}(\Lambda;\lambda,\ell^2(\Lambda)) \neq 0.$$

Since option (iv) (or the condition above already) implies Λ is infinite. Therefore $\lambda \not\supseteq 1_{\Lambda}$. This contradicts (TT)/T for Λ .

Property (T) excludes option (*iii*). Therefore for the proof, it suffices to show that $\Phi(\Gamma)$ must be virtually abelian in the case of option (*ii*). If it is shown, then property (T) forces Λ to be finite.

Suppose option (*ii*) occurs. Take a maximal curve system \mathfrak{C} preserved by Λ . Cut Σ open along \mathfrak{C} and replace each boundary circle of the resulting bordered surface with a puncture. Then we get a possibly disconnected surface Σ' . Let $\Sigma'_1, \ldots, \Sigma'_n$ be connected components of Σ' . Then there is a homomorphism:

$$\Lambda \twoheadrightarrow \Lambda' \leqslant (\mathrm{MCG}(\Sigma'_1) \times \cdots \times \mathrm{MCG}(\Sigma'_n)) \rtimes \mathfrak{S}$$
$$\longrightarrow \mathfrak{S}.$$

Here \mathfrak{S} is a subgroup of \mathfrak{S}_n , the symmetric group of degree n, and \mathfrak{S} acts by permutations on mapping class groups of homeomorphic surfaces among $\Sigma'_1, \ldots, \Sigma'_n$. And the second homomorphism is the projection to \mathfrak{S} . It is known the kernel of the map $\Lambda \to \Lambda'$ is a free abelian group, generated by multiple Dehn twists associated to the curves of the curve system \mathfrak{C} (for instance, see Chapter 4 of [Iva]). Take the kernel $\Lambda'_0 \leq \Lambda'$ of the map $\Lambda' \to \mathfrak{S}$. Then there is a natural map from Λ'_0 to each $\mathrm{MCG}(\Sigma'_i)$, which takes the *i*-th component.

Note that by definition Λ'_0 is a finite index subgroup of Λ' . Therefore again Theorem 11.4.2 tells us that each image of Λ'_0 inside $MCG(\Sigma'_i)$ is either finite or reducible. However it cannot be reducible. Indeed, if an image of Λ'_0 fixes a curve system, then by translation by \mathfrak{S} we have a curve system on $\Sigma'_1 \cup \cdots \cup \Sigma'_n$ which is preserved by Λ' . This contradicts the maximality of the curve system \mathfrak{C} , which we took at the beginning. Therefore Λ'_0 must be finite, and thus Λ is virtually abelian. This ends our proof.

Case2. with target of outer automorphism groups

This part is based on the argument in the work of Bridson–Wade [BrWa], who have utilized study of $\overline{\text{IA}}_n$ to show homomorphism rigidity result. (Recall from Definition 11.2.3 that $\overline{\text{IA}}_n$ denotes the kernel of the map $\text{Out}(F_n) \to \text{GL}_n(\mathbb{Z})$ induced by the abelianization of F_n . First we note that the conclusion holds for n = 2 because then $\text{Out}(F_n)$ has the Haagerup property.

Let $\Lambda \leq \operatorname{Out}(F_n)$ be the image of Γ by Ψ . Firstly, we appeal to Theorem 11.2.15, the classification of subgroups in $\operatorname{Out}(F_n)$ by Handel–Mosher [HaMo]: a subgroup $\Lambda < \operatorname{Out}(F_n)$ is either of the following forms:

- (i) the group Λ is not fully irreducible: there exists a finite index subgroup which preserves each conjugacy class of some proper free factor of F_n ;
- (*ii*) the group Λ s virtually \mathbb{Z} ;
- (*iii*) the group Λ contains two independent fully irreducible elements.

Secondly, we appeal to Theorem 11.4.4 of Bestvina–Bromberg–Fujiwara and exclude option (*ii*). This is done in a similar manner to one in Case 1. Property (T) excludes option (*i*). We shall show that a finite index subgroup $H \leq \Lambda$ as in option (2) must be finite, by induction on n.

If n = 2, then we have already seen it. Suppose the assertion holds true for every natural number $\langle n$, and we will verify the case of n. We take $H \leq \Lambda$ and a free factor L as in option (*ii*). Then for any $[f] \in H$, one can choose an element $g_f \in F_n$ such that $f(L) = g_f^{-1}Lg_f$ holds. Then the map $L \ni h \mapsto g_f h g_f^{-1} \in L$ is an element of Aut(L), and the image in Out(L) is uniquely determined by $[f] \in H$. This induces a well-defined group homomorphism $H \to \text{Out}(L)$. Likewise, the action on $F_n/\langle \langle L \rangle \rangle$ induces a homomorphism $H \to \text{Out}(L')$. Because the ranks of L and L' are strictly less than n, the assumption of induction applies. Hence both images are finite. By taking abelianization of L * L', with respect to the union of basis for L and L' the action of H is of the following form:

$$\left(\begin{array}{cc}G&0*&G'\end{array}\right),$$

where both G and G' are finite. Hence the image of the homomorphism $H \to GL_n(\mathbb{Z})$ is virtually abelian. Since H is a finite index subgroup of Λ , H in particular has property (T). By combining these two, we have the image is in fact finite. Therefore, the kernel H' of the map above is a finite index subgroup of H.

Note that H' is in \overline{IA}_n . Finally we appeal to Theorem 11.2.5 of Bass–Lubotzky and Bridson–Wade that every nontrivial subgroup in \overline{IA}_n surjects onto \mathbb{Z} . On the other hand, H' is a finite index subgroup of H and hence has property (T). Therefore H' must be trivial, and thus Λ is finite. This ends our proof.

In the last part of this section, we make a remark. It is not known whether property (T) itself does imply homomorphism rigidity above. However, for instance, homomorphism rigidity for $\text{Sp}_{m,1}$ into $\text{MCG}(\Sigma)$ is obtained by S.-K. Yeung [Yeu]. Also, J. E. Andersen [Ande] has announced that mapping class groups do *not* have (T).

Proof. (Theorem 11.5.1) Theorem 11.5.2, together with Theorem B and Theorem D completes the proof. \Box

11.6 A shortcut for noncommutative universal lattice cases

Here we see one shortcut of Theorem 11.5.1, for universal lattices. The key is study on distorted elements. We prove Theorem H. Finally, we explain why this argument may not work in symplectic lattices case.

11.6.1 Distorted elements in a finitely generated group

Definition 11.6.1. Let G be a finitely generated group. An element $g \in G$ is called a *distorted element* if

$$\lim_{n \to \infty} \frac{l_S(g^n)}{n} = 0.$$

Here S is a finite generating set (the choice of such S does not affect the definition above), and l_S denotes the word length with respect to S. The element g is said to be *undistorted* otherwise.

Remark 11.6.2. In the literatures, it is more common to ask a distorted element *not* to be a torsion. In this thesis, however, we allow the case of g being a torsion because it fits our purpose.

An obvious example of distorted elements (in the sense in this thesis, see the remark above) is a torsion. The following theorems resprectively Farb–Lubotzky–Minsky [FLM] (and L. Mosher [Mos]); and E. Alibegović [Ali] state these are only examples in $MCG(\Sigma)$; $Out(F_n)$.

Theorem 11.6.3. ([FLM], [Mos]; [Ali]) Let Σ be a surface and $n \geq 2$. Then any element respectively in MCG(Σ); and in Out(F_n) which is not a torsion is undistorted.

This is a key to the shortcut as we are mentioning.

11.6.2 The proof

We restate Theorem H:

Theorem 11.6.4. ([Mim4]) Let Γ be a finite index subgroup either of $E_m(\mathbb{Z}\langle x_1, \ldots, x_k \rangle)$ $(m \geq 3)$. Then for any $g \geq 0$ and $n \geq 2$, every homomorphism

$$\Phi\colon\Gamma\to\mathrm{MCG}(\Sigma_g)$$

and every homomorphism

$$\Psi \colon \Gamma \to \operatorname{Out}(F_n)$$

have finite image.

11.6. PROOF OF THEOREM H–A SHORTCUT

In particular, every homomorphism

$$\Gamma \to \mathrm{MCG}(\Sigma_{g,l}) \quad (l \ge 1)$$

and

$$\Gamma \to \operatorname{Aut}(F_n)$$

also have finite image.

The shortcut has its origin in the proof of Theorem 11.2.22 for the case of that the higher rank lattice is non-cocompact, which was shown by Ivanov before the general case of [FaMas]. We explain (some variant of) the proof shortly. Suppose it is known by a theorem of Lubotzky–Mozes–Raghunathan [LMR] that any such lattice has a distorted element of infinite order. Then by Theorem 11.6.3, this implies the homomorphism has infinite kernel. Finally, the Margulis finiteness theorem ends the proof. We also note that with $Out(F_n)$ target, Bridson–Farb [BrFa] observed a similar result before the general result Theorem 11.2.23.

In our case (universal lattices), this group does not have the Margulis finiteness property. However, we can make a way along this argument. That is the shortcut.

Proof. (Theorem 11.6.4) We only consider the first case because the argument below works without any change. Set $R = \mathbb{Z}\langle x_1, \ldots, x_k \rangle$. Set $H = \text{MCG}(\Sigma)$. By Theorem 11.2.4, there exists a finite index subgroup of H which is torsion-free. We choose one and name it H_0 .

Let $\Psi: \Gamma \to H$ be a homomorphism. Then $\Phi(\Gamma) \cap H_0$ is a finite index subgroup in $\Phi(\Gamma)$ and hence $\Gamma_0 := \Phi^{-1}(\Psi(\Gamma) \cap H_0)$ is a finite index subgroup of Γ . For distinct pair (i, j) with $1 \leq i, j \leq m$, define a subset of R as $R_{i,j}^0 := \{r \in R : E_{i,j}(r) \in \Gamma_0\}$. Since for fixed $i, j \ \{E_{i,j}(r) : r \in R\} \cong R$ as additive groups, $R_{i,j}^0$ is a finite index subgroup in the additive group R.

Note that if $E_{i,j}(r) \in \Gamma$, then it is a distorted element in Γ . This follows form the commutator relation:

$$[E_{i,j}(r), E_{j,l}(s)] = E_{i,l}(rs) \quad (i \neq j, j \neq l, l \neq i; r, s \in R).$$

By the construction of H_0 , we conclude that for any (i, j) and any $r \in R_{i,j}^0$, $\Phi(E_{i,j}(r)) = e_H$. Indeed, the relation above implies each $\Phi(E_{i,j}(r))$ $(r \in R_{i,j}^0)$ is distorted, and Theorem 11.6.3 shows that only distorted element in H_0 is e_H . We next claim that $R^0 := \bigcap_{i,j} R_{i,j}^0$ is a subring of R. Indeed, the commutator relation above implies is R is closed under multiplication. Therefore R^0 is a subring of R of finite index. We use the following theorem of J. Lewin (Lemma 1 in [Lew]): for any finitely generated (possibly noncommutative) ring Q, any finite index subring Q_0 of Q contains a (two sided) ideal J of Q which is a finite index subring of Q_0 . Thus we have a ideal \mathcal{I} of R of finite index, which is included in in R_0 . Finally, we employ the following folklore result (the proof can be found in Lemma 17 of [KaSa]): for a finite ring S the Steinberg group $\operatorname{St}_{m\geq 3}(S)$ is finite. Here recall Definition 9.2.4 for the Steinberg groups. Now we are done because Φ factors through a subgroup of $\operatorname{St}_m(R/\mathcal{I})$, and this group itself is finite.

11.6.3 Difficulty for symplectic universal lattices

In fact, the finiteness of the Steinberg group over a finite ring is extended to general cases (, more precisely, twisted Steinberg groups, see Subsection 9.2.2). For the proof of this fact, we refer to Proposition 4.5 of a paper [Rap] of I. A. Rapinchuk. This implies that part can be extended to the case of symplectic universal lattices.

However, as we have seen in Subsection 9.1.1, the commutator relation among elementary symplectic groups are much complicated. The gap lies in the point that we have deduced R_0 in the proof above is multiplication closed.

Appendix I

Relative Kazhdan constant for uniformly bounded representations

In this appendix, we shall prove Proposition I. For convenience, we restate it here.

Proposition I.0.5. Let $A_k = \mathbb{Z}[x_1, \ldots, x_k]$ and set $G = \mathbb{E}_2(A_k) \ltimes A_k^2$ and $N = A_k^2 \leq G$. Set S be the set of all unit elementary matrices in $G \subset SL_3(A_k)$. Then there is an inequality

$$\overline{\mathcal{K}}(G, N; S; M) > (15k + 100)^{-1} M^{-6}.$$

In the case of k = 0, one has $\overline{\mathcal{K}}(\mathrm{SL}_2(\mathbb{Z}) \ltimes \mathbb{Z}^2, \mathbb{Z}^2; F; M) > (21M^6)^{-1}$. Here the symbol $\overline{\mathcal{K}}(G, N; S; M)$ denotes the relative Kazhdan constant for uniformly bounded representations, which is defined in Definition 3.5.2.

For the proof, we firstly observe the following. Let Γ be a group and $M \geq 1$. Let $(\rho, \mathfrak{H}) \in \mathcal{A}_M$, namely, ρ is a uniformly bounded Γ -representation on a Hilbert space with $|\rho| \leq M$. For this given ρ , we define the norm $\|\cdot\|_{\rho}$ on \mathfrak{H} as the dual norm of the following norm $\|\cdot\|_{\rho*}$:

for
$$\phi \in \mathfrak{H}^*$$
, $\|\phi\|_{\rho*} := \sup_{g \in \Gamma} \|\rho^{\dagger}(g)\phi\|_{\mathfrak{H}^*}$.

Here ρ^{\dagger} denotes the contragredient representation of ρ on \mathfrak{H} (see Definition 3.1.6). This norm $\|\cdot\|_{\rho}$ then satisfies the following three properties:

(1) $\|\cdot\|_{\rho}$ is compatible with $\|\cdot\|_{\mathfrak{H}}$ with the norm ratio $\leq M$. More precisely,

$$M^{-1} \| \cdot \|_{\mathfrak{H}} \le \| \cdot \|_{\rho} \le \| \cdot \|_{\mathfrak{H}}.$$

- (2) ρ is isometric with respect to $\|\cdot\|_{\rho}$.
- (3) $(\mathfrak{H}, \|\cdot\|_{\rho})$ is us.

For item (3) in fact we have some estimate of modulus of smoothness. Indeed, thanks to Lemma 3.1.3, we have the following:

for any $\tau > 0$, the inequality $r_{\|\cdot\|_{\rho}}(\tau) \leq \sqrt{1 + M^2 \tau^2} - 1 \leq M^2 \tau^2/2$ holds.

Proof. (Proposition I.0.5) Let $\epsilon > 0$. We stick to the notation in the proof of Theorem 6.1.1. Suppose that there exists a *G*-representation $(\rho, \mathfrak{H}) \in \mathcal{A}_M$ such that the following holds: ρ admits a non-zero vector ξ in $\mathfrak{H}'_{\rho(N)}$ which satisfies $\sup_{s \in S} \|\xi - \rho(s)\xi\|_{\mathfrak{H}} \leq \epsilon \|\xi\|_{\mathfrak{H}}$. We may assume that $\mathfrak{H}^{\rho(N)} = 0$. For this (ρ, \mathfrak{H}) , we take the us norm $\|\cdot\|_{\rho}$ defined in the paragraph above. Thus by applying Lemma 3.1.5 and (precise estimate of) item (3) above, we can assume that there exists $\xi \in \mathfrak{H}$ with $\|\xi\|_{\rho} = 1$ such that the following two inequalities hold:

- (a) $\sup_{s \in S} \|\xi \rho(s)\xi\|_{\rho} \le M\epsilon.$
- (b) $\sup_{s \in S} \|\xi^* \rho^{\dagger}(s)\xi^*\|_{\rho^*} \le 4M^3\epsilon.$

Here $\xi \mapsto \xi^*$ is a duality mapping.

Thanks to Dixmier's unitarization (Proposition 6.1.2), we have an invertible operator $T \in \mathbb{B}(\mathfrak{H})$ with

$$||T||_{\mathbb{B}(\mathfrak{H})}||T^{-1}||_{\mathbb{B}(\mathfrak{H})} \le M^2$$

such that $\pi := \operatorname{Ad}(T) \circ \rho \mid_N$ is unitary. Let \hat{N} denote the unitary dual of N. By general theory of Fourier analysis, one obtains a standard unital *-hom $\sigma : \operatorname{C}(\hat{N}) \to \mathbb{B}(\mathfrak{H})$ from the unitary operators $\pi(N)$. Indeed, for $i \in \{1, 2\}$, let $z_i \in \operatorname{C}(\mathbb{T}^2)$ be the map $t \mapsto e^{2\pi\sqrt{-1}t_i}$ (t_i is the *i*-th component of $t \in \mathbb{T}^2$) and let $l_i \in S_0$ be $E_{i,3}(1)$. We define σ by setting $\sigma(z_i) = T\rho(l_i)T^{-1}$ for each *i*.

Then from Riesz–Markov–Kakutani theorem, from this σ one obtains the complexedvalued regular Borel measure ν on \hat{N} satisfying the following: for any $f \in C(\hat{N})$, $\int_{\hat{N}} f \, d\nu = \langle T^{-1}\sigma(f)T\xi, \xi^* \rangle$ (we note that T = I in our proof of Theorem 6.1.1). We take the Jordan decomposition of $\text{Re}\nu = \nu_+ - \nu_-$. Here $\nu_+ \perp \nu_-$ (this means they are singular to each other) and both of them are positive regular Borel measure. Then the inequality $\nu_+(\hat{N}) \geq 1$ holds.

For the proof of the proposition, first we discuss the case of that k = 0. We take the following well-known decomposition of $\hat{N} = \mathbb{T}^2 \cong \left\{ \begin{pmatrix} t_1 \\ t_2 \end{pmatrix} : t_1, t_2 \in \left[-\frac{1}{2}, \frac{1}{2}\right] \right\}$:

$$\{0\}, \ D_0 = \{|t_1| \ge 1/4 \text{ or } |t_2| \ge 1/4\}, \\ D_1 = \{|t_2| \le |t_1| < 1/4 \text{ and } t_1 t_2 > 0\}, \ D_2 = \{|t_1| < |t_2| < 1/4 \text{ and } t_1 t_2 \ge 0\}, \\ D_3 = \{|t_1| \le |t_2| < 1/4 \text{ and } t_1 t_2 < 0\}, \ D_4 = \{|t_2| < |t_1| < 1/4 \text{ and } t_1 t_2 \le 0\}.$$

We consider the natural $\operatorname{SL}_2(\mathbb{Z})$ -action on \mathbb{T}^2 defined as follows: for any $h \in \operatorname{SL}_2(\mathbb{Z})$, the action map \hat{h} of $h : t \mapsto \hat{h}t$ is the left multiplication of the matrix $\hat{h} = {}^t h^{-1}$. Then one can check the following equality: for any $g \in \operatorname{SL}(2,\mathbb{Z})$ and any $f \in \operatorname{C}(\mathbb{T}^2)$, $\sigma(\hat{h}f) = T\rho(h)T^{-1}\sigma(f)T\rho(h^{-1})T^{-1}$. With some calculation, one can also obtain the following two estimations:

- (i) The inequality $\nu_+(D_0) \leq 4M^7 \epsilon^2$ holds.
- (*ii*) For any Borel subset $Z \subset \mathbb{T}^2$ and any $h \in S_0 \subseteq \mathrm{SL}_2(\mathbb{Z})$, the inequality $|\nu_+(\hat{h}Z) \nu_+(Z)| \leq 5M^6 \epsilon$ holds.

Indeed, for instance, inequality (i) follows from the argument below. For i = 1, 2, set $D_0^i = \text{supp}\nu_+ \cap \{|t_1| \ge 1/4\} \subset D_0$. By approximating (pointwisely) $\chi_{D_0^i}$ by continuous functions and obtaining an associated projection $P \in \mathbb{B}(\mathfrak{H})$, one can make estimate as follows: for each $i \in \{1, 2\}$,

$$\begin{aligned} 2\nu_{+}(D_{0}^{i}) &\leq \left| \int_{D_{0}^{i}} |1 - z_{i}|^{2} d\nu \right| = \left| \int_{\mathbb{T}^{2}} \overline{(1 - z_{i})} \chi_{D_{0}^{i}}(1 - z_{i}) d\nu \right| \\ &= \left| \left\langle T^{-1} \sigma \left(1 - z_{i} \right)^{*} P \sigma \left(1 - z_{i} \right) T\xi, \xi^{*} \right\rangle \right| \\ &= \left| \left\langle T^{-1} \left(I - T \rho(l_{i}) T^{-1} \right)^{*} P \left(I - T \rho(l_{i}) T^{-1} \right) T\xi, \xi^{*} \right\rangle \right| \\ &= \left| \left\langle T^{-1} \left(I - T \rho(l_{i}^{-1}) T^{-1} \right) PT \left(I - \rho(l_{i}) \right) \xi, \xi^{*} \right\rangle \right| \\ &\left(\text{Recall Ad}(T) \circ \rho \mid_{N} \text{ is unitary} \right), \\ &= \left| \left\langle T^{-1} PT \left(\xi - \rho(l_{i}) \xi \right), \xi^{*} - \rho^{\dagger}(l_{i}) \xi^{*} \right\rangle \right| \\ &\leq \left\| T^{-1} PT \left(\xi - \rho(l_{i}) \xi \right) \right\|_{\rho} \left\| \xi^{*} - \rho^{\dagger}(l_{i}) (\xi^{*}) \right\|_{\rho_{*}} \\ &\leq M \left\| T^{-1} PT \right\|_{\mathbb{B}(\mathfrak{H})} \left\| \xi - \rho(l_{i}) \xi \right\|_{\rho} \left\| \xi^{*} - \rho^{\dagger}(l_{i}) \xi^{*} \right\|_{\rho_{*}} \\ &\leq 4M^{7} \epsilon^{2} \quad \text{(by item (a) and item (b)).} \end{aligned}$$

In above, we remark that for $V \in \mathbb{B}(\mathfrak{H})$, V^* means the adjoint operator of V. Also, by item (1) in appendix, we note that

$$\|V\eta\|_{\rho} \le M \|V\|_{\mathbb{B}(\mathfrak{H})} \|\eta\|_{\rho},$$

holds in general.

Thanks to these two estimations, one can verify $\nu_+(D_i) < 5M^6\epsilon + 4M^7\epsilon^2$ for $1 \leq i \leq 4$ (use for instance, $\widehat{h_{1,-}}(D_1 \cup D_2) \subset D_2 \cup D_0$, where $h_{1,-} = E_{1,2}(-1)$). Hence the inequality $\nu_+(\mathbb{T}^2 \setminus \{0\}) \leq 20M^6\epsilon + 20M^7\epsilon^2$ holds. If $\epsilon \leq (21M^6)^{-1}$, then there must exist a non-zero $\rho(N)$ -invariant vector. It is a contradiction.

For the general case, let us recall Kassabov's argument in [Kas1]. We identify \widehat{A}_k with the set of all formal power series of variables x_l^{-1} $(1 \le l \le k)$ over $\widehat{\mathbb{Z}} \cong \mathbb{T}$. Here the pairing is defined by

$$\langle ax_1^{i_1}\cdots x_k^{i_k}|\phi x_1^{-j_1}\cdots x_k^{-j_k}\rangle = \phi(a)\delta_{i_1,j_1}\cdots \delta_{i_k,j_k}.$$

We define the valuation v on $\widehat{A_k}$ as the minimum of the total degrees of all terms. Here we naturally define $v(0) = +\infty$. We decompose $\widehat{N} \setminus \{0\} = \widehat{A_k}^2 \setminus \{0\}$ as follows:

$$A = \{(\chi_1, \chi_2) : v(\chi_1) > v(\chi_2) > 0\}, \ B = \{(\chi_1, \chi_2) : v(\chi_1) = v(\chi_2) > 0\},\ C = \{(\chi_1, \chi_2) : v(\chi_2) > v(\chi_1) > 0\}, \ D = \{(\chi_1, \chi_2) : v(\chi_1)v(\chi_2) = 0\}.$$

Then from an argument similar to one in [Kas1], we have the following inequalities:

$$\nu_+(A) \le \nu_+(D) + 5(k+1)M^6\epsilon,$$

 $\nu_+(B) \le \nu_+(D) + 5kM^6\epsilon,$
and $\nu_+(C) \le \nu_+(D) + 5(k+1)M^6\epsilon.$

We naturally define the restriction map res: $\hat{N} \to \hat{\mathbb{Z}}^2$ and obtain that $\nu_+(D) = \nu_+(\hat{N} \setminus \text{res}^{-1}\{0\}) \leq 20M^6\epsilon + 20M^7\epsilon^2$. Finally, by combining these inequalities we conclude that

$$1 \le \nu_+(\hat{N}) = \nu_+(\hat{N} \setminus \{0\}) \le (15k + 90)M^6\epsilon + 80M^7\epsilon^2.$$

Here the middle equality in above follows from the assumption that $\mathfrak{H}^{\rho(N)} = 0$. Hence in particular ϵ must be more than $(15k + 100)^{-1}M^{-6}$.

Bibliography

- [Ak] A.G. Aksoy, and M.A. Khamsi, Nonstandard Methods in Fixed Point Theory. Springer-Verlag, New York, 1990
- [Ali] E. Alibegović, Translation lengths in $Out(F_n)$. Geom. Ded. 92, 87–93 (2002)
- [Ande] J. E. Andersen, Mapping class groups do not have Kazhdan's Property (T). Preprint, arXiv:0706.2184
- [Ando] T. Ando, Comparison of norms ||f(A) f(B)|| and ||f(|A B|). Math. Z. 197, no.3, 403–409 (1988)
- [Ara] J. Arazy, The isometries of C_p . Israel J. Math. 22, 247–256 (1975)
- [ArDe] G. N. Arzhantseva and T. Delzant, Examples of random groups. J. Topology, to appear, 2008
- [BFGM] U. Bader, A. Furman, T. Gelander and N. Monod, Property (T) and rigidity for actions on Banach spaces. Acta Math. 198, no.1, 57–105 (2007)
- [BaSw2] W. Ballmann and J. Swiatkowski, On L²-cohomology and property (T) for automorphism groups of polyhedral cell complexes. *Geom. Funct. Anal.* 7, 615–645 (1997)
- [BCH] P. Baum, A. connes and N. Higson, Classifying space for proper actions and K-theory of group C*-algebras. In C*-algebras: 1943-1993 (San Antonio, TX, 1993) vol. 167, Contemp. Math., 240–291, Amer. Math. Soc., Providence, RI (1994)
- [Bav] Ch. Bavard, Longueur stable des commutateurs. L'Enseign. Math. 37, nos. 1-2, 109–150 (1991)
- [Bas] H. Bass, K-theory and stable algebra, Inst. Hautes Études Sci. Publ. Math. 22, 5–60 (1964)
- [BaLu] H. Bass and A. Lubotzky, Linear-central filtorations on groups. In The mathematical legacy of Wilhelm Magnus: groups, geometry and special functions (Brooklyn, NY, 1992), volume 169 of Comtemp. Math., 45–98. Amer. Math. Soc., Providence, RI, 1994
- [BHV] B. Bekka, P. de la Harpe, and A. Valette, Kazhdan's property (T). New Mathematical Monographs, Vol. 11, Cambridge University Press, Cambridge, 2008
- [Beh] H. Behr, $SL_3(\mathbb{F}_q[T])$ is not finitely presentable. Proceedings Durham 1977, London Math. Soc. Lecture Notes **36**, 213–224, Cambridge University Press (1979)
- [BeLi] Y. Benyamini and J. Lindenstrauss, Geometric nonlinear functional analysis, Vol. 1, Amer. Math. Sci, Providence, RI, 2000

- [BBF] M. Bestvina, K. Bromberg and K. Fujiwara, forthcoming paper
- [BeFe] M. Bestvina and M. Feighn, A hyperbolic $Out(F_n)$ -complex. Groups Geom. Dyn. 4, no. 1, 31–58 (2010)
- [BeFu] M. Bestvina and K. Fujiwara, Bounded cohomology of subgroups of mapping class groups. Geom. Topol. 6, 69–89 (2002)
- [Bir] J. S. Birman, Braids, links and mapping class groups. Annals of Math. Studies, no.82, Princeton University Press, 1974
- [Bog] O. Bogopolski, Arboreal decomposability of the group of automorphisms of a free group. Algebra i Logicki 26, no.2, 131–149 (1987)
- [BoPa] M. Bourdon and H. Pajot, Cohomologie l_p et espaces de Besov. J. reine angew. Math. 558, 85–108 (2003)
- [Bow] B. Bowditch, Tight geodesics in the curve complex. Invent. Math. 171, 281–300 (2008)
- [BDK] J. Bretagnolle, D. Dacunha-Castelle and J. L. Krivine, Lois stables et espaces L^p. Ann. Inst. H. Poincaré Sect. B 2, 231–259 (1965/1966)
- [BrFa] M. R. Bridson and B. Farb, A remark about actions of lattices on free groups. *Topology Appl.* 110 (1), 21–24 (2001)
- [BrHa] M. R. Bridson and A. Haefliger, Metric spaces of non-positive curvature., Sprinder-Verlag, Berlin, 1999
- [BrVo] M. R. Bridson and K. Vogtmann, Automorphism groups of free groups, surface groups and free abelian groups. In *Problems on Mapping Class Groups*, edited by B. Farb, Proceedings of Symposia in Pure Mathematics 74, American Mathematical Society, Providence, RI, 301–316 (2006)
- [BrWa] M. R. Bridson and R. D. Wade, Actions of higher-rank lattices on free groups. Preprint, arXiv:1004.3222
- [Bro] R. Brooks, Some remarks on bounded cohomology. In *Riemann surfaces and related topics*: Proceedings of the 1978 Stony Brook Conference (State Univ. New York, Stony Brook, N. Y., 1978) (Princeton, N. J.), Ann. of Math. Stud., vol. 97, Princeton Univ. Press, 53–63 (1981)
- [BrOz] N. Brown and N. Ozawa, C^{*}-algebras and Finite-Dimensional Approximations. Graduate Studies in Mathematics, 88, Amer. Math. Sci, Providence, RI, 2008
- [Bur] M. Burger, Kazhdan constants for $SL(3,\mathbb{Z})$. J. reine angew. Math. 413, 36–67 (1991)
- [BuMo1] M. Burger and N. Monod, Bounded cohomology of lattices in higher rank Lie groups. J. Eur. Math. Soc. 1 no. 2, 199–235 (1999)
- [BuMo2] M. Burger and N. Monod, Continuous bounded cohomology and application to rigidity theory. Geom. Funct. Anal. 12(2), 219–280 (2002)
- [BOT] M. Burger, N. Ozwa and A. Thom, On Ulam stability. Preprint, arXiv:1010.0565

- [Cal1] D. Calegari, Stable commutator length is rational in free groups. J. Amer. Math. Soc. 22, no.4, 941–961 (2009)
- [Cal2] D. Calegari, scl. MSJ Memoirs, 20. Tokyo, Mathematical Society of Japan, 2009
- [CalFu] D. Calegari and K. Fujiwara, Stable commutator length in word hyperbolic groups. Groups, Geometry, Dynamics 4, no.1, 59–90 (2010)
- [CapFu] P.-E Caprace and K. Fujiwara, Rank one isometries of buildings and quasi-morphisms of Kac–Moody groups. Geom. Funct. Anal. 19, 1296–1319 (2010)
- [CaKe] D. Carter and G. Keller, Bounded elementary generation of $SL_n(\mathcal{O})$. Amer. J. Math. 105, 673–687 (1983)
- [CMS] D. Cartwright, W. Mlotkowski and T. Steger, Property (T) for A₂ groups. Ann. Inst. Fourier 44, 213–248 (1993)
- [CCJJV] P.-A. Cherix, M. Cowling, P. Jolissaint, P. Julg and A. Valette, Groups with the Haagerup Property. Birkhäuser, Progress in Mathematics 197, 2001
- [CCLTV] R. Cluckers, Y. Cornulier, N. Louvet, R. Tessera and A. Valette, The Howe-Moore property for real and p-adic groups. Preprint, arXiv:1003.1484
- [Coh] P. M. Cohn, On the structure of the GL₂ of a ring. Inst. Hautes Études Sci. Publ. Math. 30, 5–53 (1966)
- [CuMo] M. Culler and J. W. Morgan, Groups acting on R-trees. Proc. London Math. Soc. (3) 55, 571–604 (1987)
- [CuVo1] M. Culler and K. Vogtmann, Moduli of graphs and automorphisms of free groups. Invent. Math. 84, 91–119 (1986)
- [CuVo2] M. Culler and K. Vogtmann, A group theortic criterion for property FA. Proc. Amer. Math. Soc. 124, no.3, 677–683 (1996)
- [dCo1] Y. de Cornulier, Kazhdan and Haagerup Properties in algebraic groups over local fields. J. Lie Theory 16, 67–82 (2006)
- [dCo2] Y. de Cornulier, Strongly Bounded Groups and Infinite Powers of Finite Groups. Comm. Algebra 34(7), 2337–2345 (2006)
- [dCo3] Y. de Cornulier, Relative Kazhdan Property. Ann. Sci. Ecole Norm. Sup. 39(2), 301–333 (2006)
- [CSV] Y. de Cornulier, Y. Stalder and A. Valette, Proper actions of wreath products and generalizations. Preprint, arXiv:0905.3960
- [Del] P. Delorme, 1-cohomologie des représentations unitaires des groupes de Lie semisimples et résolubles. Produits tensoriels continus de représentations. Bull. Soc. Math. France 105, 281–336 (1977)
- [DeVa] R. K. Dennis, L. N. Vaserstein, On a question of M. Newman on the number of commutators. J. Algebra 118, 150–161 (1988)

- [Dix] J. Dixmier, Les moyennes invariantes dans les semi-groups et leurs applications. Acta Sci. Math. Szeged. 12 Leopoldo Fejér et Frerico Riesz LXX anno natis dedicatus, Pars A, 213–227 (1950)
- [DyJa] J. Dymara and T. Januszkiewicz, Cohomology of buildings and of their automorphism groups. Invent. Math. 150(3), 579–627 (2002)
- [EpFu] D. B. A. Epstein and K. Fujiwara, The second bounded cohomology of word-hyperboloc groups. *Topology* 36, no. 6, 1275–1289 (1997)
- [Ero] I. V. Erovenko, $SL_n(F[x])$ is not boundedly generated by elementary matrices: explicit proof. *Electron. J. Linear Algebra* **11**, 162–167 (2004)
- [ErJa] M. Ershov and A. Jaikin-Zapirain, Property (T) for noncommutative universal lattices. Invent. Math. 179(2), 303–347 (2010)
- [EJK] M. Ershov, A. Jaikin-Zapirain and M. Kassabov, Property (T) for groups graded by root systems. Preprint, arXiv:1102.0031
- [FLM] B. Farb, A. Lubotzky and Y. Minsky, Rank one phenomena for mapping class groups. Duke J. of Math. 106, 194–199 (2001)
- [FaMar] B. Farb and D. Margalit, A primer on mapping class groups. Forthcoming book, currently available at: http://www.math.utah.edu/ margalit/primer/
- [FaMas] B. Farb and H. Masur, Superrigidity and mapping class groups. Topology 37, no. 6, 1169–1176 (1998)
- [FaSh] B. Farb and P. Shalen, Lattice actions, 3-manifolds and homology. Topology 39(3) 573–587 (2000)
- [FiPi] T. Figiel and G. Pisier, Séries aléatories dans les espaces uniformément convexes ou uniformément lisses. C. R. Acad. Sci. Paris Sér. A 279, 611–614 (1974)
- [Ghy1] E. Ghys, Groupes d'homéomorphismes du cerclr et cohomologie bornée. In the Lefschetz centennial conference, Part III (Mexico City, 1984), Contemp. Math. 58 III, Amer. Math. Soc., Providence, RI, 81–106 (1987)
- [Ghy2] E. Ghys, Groups acting on the circle. L'Enseign. Math. 47, 329–407 (2001)
- [Gri] R. I. Grigorchuk, Degrees of growth of finitely generated groups and the theory of invariant means. Math. USSR IZV. 25(2), 259–300 (1985)
- [Gro1] M. Gromov, Volume and bounded cohomology. Inst. Hautes Études Sci. Publ. Math. 56, 5–99 (1982)
- [Gro2] M. Gromov, Hyperbolic groups. In Essays in group theory, Math. Sci. Res. Inst. Publ. 8, Springer, New York, 75–263 (1987)
- [Gro3] M. Gromov, Rigid transformation groups. In Géométrie différentielle, variétés complexes, feuilletage riemanniens (Paris, 1986), vol. 33, Hermann Paris, 65–139 (1988)

- [Gro4] M. Gromov, Asymptotic invariants of infinite groups. In Geometric group theory, vol.2 (Sussex, 1991), London Math. Soc. Lecture Note Ser., 182, Cambridge Univ. Press, Cambridge, 1–295 (1993)
- [Gro5] M. Gromov, Spaces and questions. In GAFA 2000 (Tel Aviv, 1999), Geom. Funct. Anal. 2000, Special Volume, Part I, 118–161 (2000)
- [Gro6] M. Gromov, Random walks in random groups. Geom. Funct. Anal. 13, 73–146 (2003)
- [GMV1] F. Grunewald, J. Mennicke and L. Vaserstein, On symplectic groups over polynomial rings. Math. Z. 206, 35–56 (1991)
- [GMV2] F. Grunewald, J. Mennicke and L. Vaserstein, On the groups $SL_2(\mathbb{Z}[x])$ and $SL_2(k[x, y])$. Israel J. Math. 86, 157–193 (1994)
- [GHW] E. Guntner, N. Higson and S. Weinberger, The Novikov Conjecture for Linear Groups. Inst. Hautes Études Sci. Publ. Math. 101, 245–268 (2005)
- [GTY] E. Guntner, R. Tessera and G. Yu, A notion of geometric complexity and its application to topologicl rigidity. Preprint, arXiv:1008.0884
- [Gui] A. Guichardet, Sur la cohomologie des groupes topologiques. II. Bull. Sci. Math. 96, 305–332 (1972)
- [Haa] U. Haagerup, An example of a nonnuclear C*-algebra, which has the metric approximation property. Invent. Math. 50(3), 279–293 (1979)
- [HaPr] U. Haagerup and A. Przybyszewska, Proper metrics on locally compact groups, and proper affine isometric actions on Banach spaces. Preprint, arXiv:0606794
- [HRS] U. Haagerup, H. P. Rosenthal and E. A. Sukochev, Banach Embedding Properties of Non-Commutative Lp-Spaces. Memoirs of AMS, vol. 163 no.776, 2003
- [Had] U. Hadad, Uniform Kazhdan Constant for some families of linear groups. J. Algebra. 318(2), 607–618 (2007)
- [Ham] U. Hamenstädt, Bounded cohomology and isometry groups of hyperbolic spaces. J. Eur. Math. Soc. 10, 315–349 (2008)
- [HaMo] M. Handel and L. Mosher, Subgroup classification in $Out(F_n)$. Prepreint, arXiv:0908.1225
- [Hat] A. Hatcher, Algebraic topology. Cambridge University Press, Cambridge, 2002
- [Hei] S. Heinrich, Ultraproduct in Banach space theory. J. reine angew. Math. 313, 72–104 (1980)
- [HiKa] N. Higson and G. Kasparov, Operator K-theory for groups which act properly and isometrically on Hilbert space. Electron. Res. Announc. Amer. Math. Soc. 3, 131–142 (1997)
- [HLS] N. Higson, V. Lafforgue and G. Skandalis, Counterexamples to the Baum–Connes conjecture. Geom. Funct. Anal. 12, 330–354 (2002)
- [HLW] S. Hoory, N. Linial and A. Wigderson, Expander graphs and their applications. Bull. Amer. Math. Soc. 43, 439–561 (2006)

- [HoMo] R. Howe and C. C. Moore, Asymptotic properties of unitary representations. J. Funct. Anal. 32, 72–99 (1979)
- [Iva] N. V. Ivanov, Mapping class groups. In Handbook of geometric topology, 523–633, North-Holland, Amsterdam, 2002
- [Joh] B.E. Johnson, Cohomology of Banach algebras. Mem. Amer. Math. Soc. 127, 1972
- [Jol] P. Jolissaint, On Property T for pairs of topological groups. L'Enseign. Math. 51, 31–45 (2005)
- [KaMa] A. V. Kaimanovich and H. Masur, The Poisson boundary of the mapping class group. Invent. Math. 125, no. 2, 221–264 (1996)
- [KaYu] G. Kasparov, and G. Yu, The coarse geometric Nivikov conjecture and uniform convexity. Adv. Math. 206, no. 1, 1–56 (2006)
- [Kas1] M. Kassabov, Kazhdan constants for $SL_n(\mathbb{Z})$. Internat. J. Algebra Comput. 15, no.5-6, 971–995 (2005)
- [Kas2] M. Kassabov, Universal lattices and unbounded rank expanders. Invent. Math. 170, no. 2, 297–326 (2007)
- [KaSa] M. Kassabov and M. Sapir, Nonlinearity of matrix groups. J. Topol. Anal. 1, 251–260 (2009)
- [Kaz] D. A. Kazhdan, Connection of the dual space of a group with the structure of its closed subgroups. Funct. Anal. Appl. 1, 63–65 (1967)
- [Kos] B. Kostant, On the existence and irreducibility of certain series of representations. Bull. Amer. Math. Soc. 75, 627–642 (1969)
- [KrMc] S. Krstić and J. McCool, Free quotients of SL₂(R[x]). Proc. Amer. Math. Soc. 125, no.6, 1585–1588 (1997)
- [Lac] H. E. Lacey, The isometric theory of classical Banach spaces. Springer-Verlag, 88, 1974
- [Laf1] V. Lafforgue, Un renforcement de la propriété (T). Duke Math. J. 143, 559–602 (2008)
- [Laf2] V. Lafforgue, Propriété (T) renforcée banachique et transformation de Fourier rapide. J. Topl. Anal. 1(3), 191–206 (2009)
- [LeLu] G. Levitt and M. Lustig, Irreducible automorphisms of F_n have north-south dynamics on compactified outer space. J. Inst. Math. Jussieu 2, 59–72 (2003)
- [Lew] J. Levin, Subrings of finite index in finitely generated rings. J. Algebra 5, 84–88 (1967)
- [LMR] A. Lubotzky, S. Mozes and M.S. Raghunathan, The word and Riemannian metrics on lattices of semi-simple groups. Inst. Hautes Études Sci. Publ. Math. 91, 5–53 (2000)
- [Man] J. F. Manning, Quasi-actions on trees and property (QFA). J. London Math. Soc. (2) 73 (1), 84–108 (2006), with an appendix by N. Monod and B. Rémy
- [Mar1] G. A. Margulis, Explicit construction of concentrators. Problems of Inform. Transm. 10, 325–332 (1975)

- [Mar2] G. A. Margulis, Discrete Subgroups of Semisimple Lie Groups. Berlin, Springer-Verlag, 1991
- [MaMi] H. A. Masur and Y. N. Minsky, Geometry of the complex of curves I: hyperbolicity. Invent. Math. 138 103–149 (1999)
- [McPa] J. McCarthy and A. Papadopoulos, Dynamics on Thurston's sphere of projective measured foliations. *Comment. Math. Helv.* 64 133–166 (1989)
- [McC] J. McCool, A faithful polynomial representation of Out(F₃). Math. Proc. Cambridge Philos. Soc. 106 no.2, 207–213 (1989)
- [Mim1] M. Mimura, Fixed point properties and second bounded cohomology of universal lattices on Banach spaces. J. reine angew. Math., to appear, arXiv:0904.4650
- [Mim2] M. Mimura, On Quasi-homomorphisms and Commutators in the Special Linear Group over a Euclidean Ring. Int. Math. Res. Not. IMRN. 2010, no. 18, 3519–3529 (2010)
- [Mim3] M. Mimura, Fixed point property for universal lattice on Schatten classes. Preprint, arXiv:1010.4532
- [Mim4] M. Mimura, Rigidity of symplectic universal lattices on free groups. Preprint, in preparation
- [Min] I. Mineyev, Straightening and bounded cohomology of hyperbolic groups. Geom. Funct. Anal. 11, 807-839 (2001)
- [MMS] I. Mineyev, N. Monod and Y. Shalom, Ideal bicombings for hyperbolic groups and applications. *Topology* 43, no. 6, 1319–1344 (2004)
- [Mon1] N. Monod, Continuous bounded cohomology of locally compact groups. Springer Lecture notes in Mathematics, 1758, 2001
- [Mon2] N. Monod, An invitation to bounded cohomology. In Proceedings of the International Congress of Mathematicians, Madrid 2006 Vol. II, Eur. Math. Soc., Zürich, 1183–1211 (2006)
- [MoSh1] N. Monod and Y. Shalom, Cocycle superrigidity and bounded cohomology for negatively curved space. J. Diffr. Geom. 67, 395–455 (2004)
- [MoSh2] N. Monod and Y. Shalom, Orbit equivalence rigidity and bounded cohomology. Ann. of Math. (2) 164, no.3, 825–878 (2006)
- [Mos] L. Mosher, Mapping class groups are automatic. Ann. of Math. (2) 142, no.1, 303–384 (1995)
- [Mur] A. Muranov, Finitely generated infinite simple groups of infinite square width and vanishing stable commutator length. J. Topol. Anal. 2, no. 3, 341–384 (2010)
- [Nav1] A. Navas, Actions de groupes de Kazhdan sur le cercle. Ann. Sci. École Norm. Sup. 35, 749–758 (2002)
- [Nav2] A. Navas, Reduction of cocycles and groups of diffeomorphisms of the circle. Bull. Belg. Math. Soc. Simon Steven 13, no. 2, 193–205 (2006)

- [Nav3] A. Navas, Growth of groups and diffeomorphisms of the interval. Geom. Funct. Anal. 18, 988–1028 (2008)
- [Neu] M. Neuhauser, Kazhdan's property T for the symplectic group over a ring. Bull. Belg. Math. Soc. 10, 537–550 (2003)
- [New] M. Newman, Unimodular commutators. Proc. Amer. Math. Soc. 101, 605–609 (1987)
- [Oll] Y. Ollivier, A Januray 2005 invitation to random groups. Ensaios Matemáticos, 10, Sociedade Brasileira de Matemática, rio de Janeiro, 2005
- [OlWi] Y. Ollivier and D. T. Wise, Cubulating random groups at density < 1/6. Trans. Amer. Math. Soc., to appear
- [Oza] N. Ozawa, Quasi-homomorphism rigidity with noncommutative targets. J. reine angew. Math., to appear, arXiv:0911.3975
- [Pan] P. Pansu, Cohomologie L^p : invariance sous quasiisométrie. Preprint, 1995
- [PaWo] H. Park and C. Woodburn, An algorithmic proof of Suslin's stability theorem for polynomial rings. J. Algebra 178, no.1, 277–298 (1995)
- [Pat] A. T. Paterson, Amenability, Amer. Math. Soc., Providence, 1988
- [PiXu] G. Pisier, and Q. Xu, Non-commutative L_p-spaces. In Handbook of the geometry of Banach spaces, Vol. 2, North-Holland, Amsterdam, 1459–1517 (2003)
- [Pus] M. Puschnigg, Finitely summable Fredholom modules over higher rank groups and lattices. Preprint, arXiv:0806.2759
- [Rap] I. A. Rapinchuk, On linear representation of Chevalley groups over commutative rings. Preprint, arXiv:1005.0422
- [Ray] Y. Raynaud, On ultrapowers of non commutative L_p spaces. J. Operator Theory 48, no. 1, 41–68 (2002)
- [Roe] J. Roe, Lectures in Coarse Geometry. University Lecture Series Vol. 31, American Mathematical Society: Providence, Rhode Island, 2003
- [Rol] P. Rolli, Quasi-morphisms on Free Groups. Preprint, arXiv:0911.4234
- [Sch] I. J. Schoenberg, Metric spaces and completely monotone functions. Ann. Math. 39(4), 811–841 (1938)
- [Ser] J.-P. Serre, Trees. (Translated from the French by John Stillwell), Springer-Verlag, 1980
- [Sha1] Y. Shalom, Bounded generation and Kazhdan's property (T). Inst. Hautes Études Sci. Publ. Math. 90, 145–168 (1999)
- [Sha2] Y. Shalom, Rigidity of commensurators and irreducible lattices. Invent. Math. 141(1), 1–54 (2000)
- [Sha3] Y. Shalom, Rigidity, unitary representations of semisimple groups, and fundamental groups of manifolds with rank one transformation group. Ann. Math. (2) 152, 113–182 (2000)

- [Sha4] Y. Shalom, Explicit Kazhdan constants for semisimple and arithmetic groups Ann. Inst. Fourier 50, 833–863 (2000)
- [Sha5] Y. Shalom, The algebraization of Kazhdan's property (T). In Proceedings of the International Congress of Mathematicians, Madrid 2006 Vol. II, Eur. Math. Soc., Zürich, 1283–1310 (2006)
- [She] D. Sherman, Noncommutative L^p-structure encodes exactly Jordan structure. J. Funct. Anal. 221, 150–166 (2005)
- [Sil] L. Silberman, Addendum to "Random walks in random groups". Geom. Funct. Anal. 13, 147–177 (2003)
- [Sus] A. A. Suslin, On the structure of the special linear group over polynomial rings. Math. USSR Izv. 11, 221–238 (1977)
- [Sto] E. Stømer, On the Jordan structure of C*-algebras. Trans. Amer. Math. Soc. 120, 438–447 (1965)
- [Tu] J.-L. Tu, La conjecture de Baum-Connes pour les feuillages moyennables. K theory 17(3), 215–264 (1999)
- [vdKa] W. van der Kallen, SL₃(ℂ[x]) does not have bounded word length. In Algebraic K-theory, Part I, (Oberwolfach, 1980), Lecture Notes in Mathematics, Vol. 966. Berlin: Springer, 357–361 (1982)
- [Vas1] L. N. Vaserstein, The stable range of rings and the dimension of topological spaces. Funkcional. Anal. i Prilozen. 5 no.2, 17–27 (1971)
- [Vas2] L. Vaserstein, Bounded reduction of invertible matrices over polynomial ring by addition operators. Preprint, 2007
- [Vog1] K. Vogtmann, Automorphisms of free groups and outer space. Geom. Ded. 94, 1–31 (2002)
- [Vog2] K. Vogtmann, The cohomology of automorphism groups of free groups. In Proceedings of the International Congress of Mathematicians, Madrid 2006, vol. II, Eur. Math. Soc., Zürich, 1101–1119 (2006)
- [Wan] S. P. Wang, On isolated points in the dual spaces of locally compact groups. Math. Ann. 219, 19–34 (1975)
- [Wat] Y. Watatani, Property (T) of Kazhdan implies Property (FA) of Serre. Math. Japonica 27, 93–103 (1982)
- [Wit] D. Witte, Arithmetic groups of higher Q-rank cannot act on 1-manifolds. Proc. Amer. Math. Soc. 122, 333–340 (1994)
- [Yea] F. J. Yeadon, Isometries of non-commutative L^p spaces. Math. Proc. Camb. Phil. Soc. 90, 41–50 (1981)
- [Yeu] S.-K. Yeung, Representations of semisimple lattices in mapping class groups. Int. Math. Res. Not. 31, 1677–1686 (2003)

- [Yu1] G. Yu, The coarse Baum–Connes conjecture for spaces which admit a uniform embedding into Hilbert space. Invent. Math. 139, 201–240 (2000)
- [Yu2] G. Yu, Hyperbolic groups admit proper affine isometric actions on ℓ^p -spaces. Geom. Funct. Anal. 15, 1144–1151 (2005)
- [Zim] R. J. Zimmer, Ergodic theory and semisimple groups, Birkhäuser Verlag, Basel, 1984
- [Zuk] A. Żuk, Property (T) and Kazhdan constant for discerete groups. Geom. Funct. Anal. 13, 643–670 (2003)